ERRATA and ADDENDUM

pg. 2, para. 2: 'climate affecting' for 'climate effecting'

pg. 4, para. 1: 'If they consume energy efficiently...' replaces 'If they consume energy efficiency...'

pg. 5, para. 1: 'The study data is from average Australian operations...' replaces 'The study data uses average Australian operations...'

pg. 8, footnote: Mebratu, D. (1998), 'Sustainability and Sustainable Development: Historical and Conceptual Review', Environmental Impact Assessment Review, Volume 18, pp. 493-520.

pg. 13, para. 2: 'would be affected' for 'would be effected'

pg. 17, para. 2: '...discrete sources of information...' replaces '...discreet sources of information...'

pg. 25, para. 3: '...sustainability concept models has its own goals...' replaces '...sustainability concept models has their own goals...'

pg. 29, para. 2: 'most likely follow existing' for 'most likely following existing'

pg. 31, para. 4: Added to sentence 'For example, coal fuelled...': Footnote: 'Other factors, primarily environmental legislation, also hastened this fuel transition from coal to oil.'

pg. 39, figure: replacement of figure with Figure A.

pg. 43, Section 2.4.4;shows the impacts previous sustainability assessments, reviewed in Section 2.4.3, have highlighted as important' replaces ' shows the impacts determined to be important in previous sustainability, environmental, economic and social assessments from the review in Section 2.4.3' and 'Many impacts measured by indicators...' replaces 'Most indicators...'. The following replaces the paragraph beginning 'Many of the indicators...':

'Tables 2.3.5 and 2.3.6 include many more indicators than Table 2.4.1. Many of these additional indicators measure the sustainability of an industry's corporate philosophy (e.g. the Human Rights and Society categories of indicators in Table 2.3.5 and the Voluntary actions, Human-capital, Ethics and Welfare categories of indicators in Table 2.3.6). For a proposed system, the choice of technology is unlikely to affect business philosophy. This may not be valid if the choice results in the use of materials sourced from companies with radically different business philosophies (for example, when deciding between purchasing from companies with very different worker conditions). Moreover, many of the remaining additional indicators measure differences in product quality (i.e. the Product Responsibility category of indicators in Table 2.3.5 and Environmental Efficiency category of indicators in Table 2.3.6). Electricity as a product does not vary with its source technology. Thus, its qualities (i.e. endurance, strength, durability, etc.), and indicators measuring quality differences, also do not vary.'

pg. 45, para. 3: 'using environmental, economic and social indicators and adopting a full "life-cycle" perspective' replaces 'using sustainability indicators and a "life-cycle" perspective'

pg. 49, para. 4: 'substance flow analysis' replaces 'substance flow'

pg. 51: 'Life Cycle Assessment' replaces 'Life Cycle Analysis' in title of Section 3.2.2.3. Both terms have been widely accepted for LCA.

pg. 52, para. 2: 'The LCA method is largely scientific' replaces 'The LCA method is scientific' and 'In contrast, results using EIA are influenced by local legislative requi: ments.' replaces ', without the EIA method's limiting local legislative requirements."

pg. 56, para. 2: 'Documentation of these criteria' replaces 'Documentation of this criteria'

pg. 62, para. 1: 'The choice of CF factors, from these sets, introduces...' replaces 'The choice of CF factors introduces...?

CONTINUED IN ENDPAGES



THESIS ACCEPTED IN SATISFACTION OF THE **REQUIREMENTS FOR THE DEGREE OF** DOCTOR OF PHILOSOPHY

Sec. Research Graduate School Committee Under the Copyright Act 1968, this thesis must be used only under the normal conditions of scholarly fair dealing for the purposes of research, criticism or review. In particular no results or conclusions should be extracted from it, nor should it be copied or closely paraphrased in whole or in part without the written consent of the author. Proper written acknowledgement should be made for any assistance obtained from this thesis.

H24/3690

MONASH UNIVERSITY ON..... 13 January 2004

SUSTAINABILITY OF ELECTRICITY GENERATION USING AUSTRALIAN FOSSIL FUELS

by

John R. May B.E.

A Thesis Submitted to

MONASH UNIVERSITY

For the degree of

Doctor of Philosophy

in the

Department of Chemical Engineering

under the supervision of

Dr. David J. Brennan

Department of Chemical Engineering Monash University PO Box 36 MONASH UNIVERSITY VICTORIA 3800 AUSTRALIA

09 September 2003

Table of Contents
List of Tables
List of Figures
Abstract
Candidate's Certificate
Acknowledgements
1101210
1.1 FUSSIL FUELS, ELECTRICITY GENER.
1.2 AIM AND SUOPE
2.1 OVERVIEW OF SUSTAINABILITY
2.2 DEFINING SUSTAINABILITY
2.3 MEASURING SUSTAINABILITY
2.4 ELECTRICITY GENERATION AND SUS
2.5 SUMMARY OF SUSTAINABILITY
3. METHODS FOR PRODUCING SUST
3.1 OVERVIEW OF METHODS FOR PRODU
3.2 METHODS FOR PRODUCING ENVIRON
3.3 METHODS FOR PRODUCING ECONOM
3.4 METHODS FOR PRODUCING SOCIAL
3.5 PRODUCING INDICATORS OF SUSTAIN
3.6 SUMMARY OF METHODS FOR PRODU
4. ELECTRICITY GENERATION SYST
4.1 OVERVIEW OF ELECTRICITY GENERA
4.2 METHODOLOGY FOR DATA COLLEC
4.3 ELECTRICITY GENERATION SYSTEM:
4.4 APPLICATION OF UNCERTAINTY AND
4.5 SUMMARY OF ELECTRICITY GENERA
5. RESULTS OF THE SUSTAINABILIT
5.1 OVERVIEW OF RESULTS OF THE SUS
5.2 ENVIRONMENTAL INDICATOR RESUL
5.3 ECONOMIC INDICATOR RESULTS
5.4 SOCIAL INDICATOR RESULTS
5.5 SUSTAINABILITY INDICATOR RESULT
5.6 SUMMARY OF RESULTS OF THE SUST
6. IMPLICATIONS OF THE SUSTAINA
6.1 OVERVIEW OF THE IMPLICATIONS OF
6.2 SUSTAINABILITY INDICATORS OF EL
6.3 THE SUSTAINABILITY ASSESSMENT
6.4 POTENTIAL APPLICATIONS OF THE S
7. CONCLUSIONS AND RECOMMEN
7.1 OVERVIEW OF THE RESEARCH FIND
7.2 LIMITATIONS OF THE RESEARCH
7.3 RECOMMENDATIONS FOR FURTHER 1
APPENDICES
APPENDIX 1. DATA SOURCES FOR THE SUS
APPENDIX 2. EXAMPLES OF SUSTAINABILI
APPENDIX 3: CONTRIBUTIONS AND NORM
NOMENCLATURE
GLOSSARY
REFERENCES
MAIN TEXT
APPENDICES
a wala barketeelee aanta a

TABLE OF CONTENTS

	i
iv	1
vii	i
ix	
	i
xii	i
1	
ΑΤΙΩΝΙ ΑΝΙΟ ΣΙΙΣΤΑΙΝΑΡΗ ΙΤΥ 1	
۳۲	†
JJ	,
<i>1</i> ,,,,,,,, .	
	i
)
TAINABILITY	
AINABILITY INDICATORS	
JCING SUSTAINABILITY INDICATORS	1
MENTAL INDICATORS	i
IIC INDICATORS	ł
NDICATORS	
IABILITY WITH A 'LIFE-CYCLE' PERSPECTIVE	
ICING SUSTAINABILITY INDICATORS	,
rems and data99)
ATION SYSTEMS AND DATA	ł
TION	I
S AND DATA	í
SENSITIVITY ANALYSES	í
TION SYSTEMS AND DATA	ł
TY ASSESSMENT 161	
TAINABILITY ASSESSMENT. 161	
TS 161	
175	
179)
те 181	
	Ì
AINADILITT ASSESSMENT	r
АЛІБІТТ АЗЗЕЗЗИЕНТ	
- THE SUSTAINABILITI ASSESSMENT	
ECTRICITY GENERATION	•
207	
USTAINABILITY ASSESSMENT METHOD	+
DATIONS	
NGS	,
	-
RESEARCH)
	,
TAINABILITY ASSESSMENT225	į
TY INDICATOR PRODUCTION STEPS	ŀ
ALISED SUSTAINABILITY INDICATORS	ľ
	ł
	ŝ
	í
	5
	•

LIST OF TABLES

a a mentioned blacking proposed national indicators for the social sphere of sustainability
Table 2.3.1: The United Nations proposed national indicators for the sector opticity optication of the sector opticity of the sector optication opticat
Table 2.3.2: The United Nations proposed national indicators for the environmental sphere of
sustainability (Division for Sustainable Development (2003))
Table 2.3.3: The United Nations proposed national indicators for the economic sphere of sustainability
(Division for Sustainable Development (2003)).
Table 2.3.4: The United Nations proposed national sustainability performance indicators for the
institutional sphere of sustainability (Division for Sustainable Development (2003)).
Table 2.3.5: Proposed default set of indicators for assessing the sustainability of organisations (crocking)
Reporting initiative (2002)).
Table 2.3.6: Proposed default set of indicators for assessing the sustainability of organization (1.1.27
Table 2.3.7: Proposed goals of sustainability (goals presented for each sphere of the three-sphere
sustainability model) (Barbier (1987))
Table 2.4.1: Important indicators of sustainability, selected from a review of assessments of the impacts
of electricity generation systems (naming conventions from Azapagic and Perdan (2000) and
IChemE (2002b))
Table 3.2.1: Impact categories recommended in the ISO standards (ISO 14042, Udo de Haes et al.
(1999)). Bold categories are those recommended for use in all studies by CML (Ountee et al.
(2001)). See 'Glossary' for more information about bold categories
Table 3.2.2: Methods for characterising impacts recommended by the Civic guide (Cantoo of an (Civic))
Table 3.2.3: Matrix of indicator scores ('nedigree matrix') for rating the quality performance of data
(Weidema and Wesnæs (1996))
Table 3.2.4. Table to obtain the shape parameters from 1 for use in the data quality indicator method of
75 Kennedy et al. (1996)
Table 3.2.5: Table to obtain the additional uncertainty from I for the geographical correlation indicator for
use in the data quality indicator method of Meier (Berg et al. (1999))
Table 3.6.1: List of equations necessary to estimate each of the indicators of sustainability. The indicators
are from Table 2.5.1 (Chapter 2). Column A contains the appreviations for each impact
Table 4.3.1: Material requirements for the mining of 1 Mt of coal from an average black coal mine 108
Table 4.3.2: Material emissions from the mining of 1 wh of coal mining
Table 4.3.3: Correlation parameters for the operating costs of black coal mining
Table 4.3.5: Employment, LTI and fatalities data for an average black coal mine
Table 4.3.6: Transport distances and vehicle capacities for domestically consumed black coal
Table 4.3.7; Transport distances and vehicle capacities for black coal exported to Japan (Ass. = assumed
value, Diff. = by difference)
Table 4.3.8: Material requirements for the transport of 1 Mt of coal from an average black coal mile to all
average power station.
Table 4.3.9: Material emissions from the transport of 1 Mit of coal from an average black coal minor of 1/2
average black coal power station
Table 4.3.10: Correlation parameters for the operating costs of black coal mining
Table 4.3.17: Employment LTI and fatalities data for average black coal transport
Table 4.3.13: Material requirements for the generation of 1 net GWh of electricity from an average black
coal power station.
Table 4.3.14: Material emissions from the generation of 1 net GWh of electricity in an average black coal
power station
Table 4.3.15: Correlation parameters for the capital costs of black coal generation.
Table 4.3.16: Correlation parameters for the operating costs of black coal generation.
Table 4.3.17: Employment, L11 and ratannes data for average black coal generation, manual the Table 4.3.18. Metabolic requirements and emissions for the transmission of 1 GWh of electricity from an
Table 4.3.18: Material requirements and emissions for the transmission of the station 121
Table 4.3.19: Correlation parameters for the capital and operating costs of electricity transmission
Table 4.3.20: Employment, LTI and fatalities data for average black coal transmission
Table 4.3.21: Material requirements for the mining of 1 Mt of brown coal from an average mine
Table 4.3.22: Correlation parameters for the capital and operating costs of brown coal mining

Table 4.3.23: Material emissions from the min
Table 4.3.24: Employment, LTI and fatalities
Table 4.3.25: Material requirements for the
power station.
Table 4.3.26: Material emissions from the tran
station.
Table 4.3.27: Correlation parameters for the ca
Table 4.3.28: Material requirements for the g
power station.
Table 4.3.29: Material emissions from the ge
power station.
Table 4.3.30: Correlation parameters for the ca
Table 4.3.31: Employment, L11 and fatalities (
1 able 4.5.52: Material requirements and emis
Table 4.2.22: Metarial amissions from the and
rable 4.5.55. Waterial emissions from the pro
Table 4.3.34: Motorial requirements for the
average natural gas mining subsystem
Table 4.3.35: Correlation parameters for the or
Table 4.3.36: Correlation parameters for the or
Table 4.3.37: Employment 1.TI and fatalities
Table 4.3.38: Material requirements for the t
natural gas mining subsystem to power s
Table 4.3.39: Material emissions from the ti
natural gas mining subsystem to power s
Table 4.3.40: Employment, LTI and fatalities
Table 4.3.41: Material requirements for the p
gas power station
Table 4.3.42: Material emissions from the pro-
Table 4.2.42. Completion approximation for the se
Table 4.3.45: Correlation parameters for the ca
Table 4.3.44. Correlation parameters for the op-
Table 4.3.45: Material requirements and emis
average natural gas power station
Table 4.3.47: Conversion factors to convert the
subsystem to system. Note that the natu
allocation factors. Units: subsystem out
Table 4.3.48: CF_{RD} factors with the world as r
factors with Australia as reference region
Table 4.3.49: Australian normalisation factors
the environmental impacts due to all sou
Table 4.3.50: Factors for the mining subsystem
flows data in Section 4.3.4 from subsys
electricity delivered.
Table 4.3.51: Numbers of subsystems necessa
of greater than one implies that that sub-
will be necessary at the end of its life
Table 4.5.52: Factors to estimate sales revenue $A_{1,2,3,2}$
added (VA) (5 (AUD 2000), per t 10
Table 4.3.53: Labour factors (L) for actime
-1 auto 4.5.55. Labour Tactors (L) for estimation operating and maintenance costs ($O \ R \ \lambda$
Table 4.3.54. Australian normalization factor
economic impacts due to all sources with
Table 4.3.55: Employment multipliers (FM)
indicator from direct employment (E

v

ning of 1 Mt of brown coal from an average mine	125
data for an average brown coal mme	125
transport of 1 Mt of brown coal from an average min	e to 126
asport of 1 Mt of brown coal from an average mine to po	wer
	127
apital and operating costs of brown coal transport	127
eneration of 1 GWh of electricity in an overage brown	coal
	130
neration of 1 GWh of electricity in an average brown	coal 131
apital costs of brown coal generation	131
	100
data for average brown coal electricity generation	132
ssions for the transmission of 1 GWh of electricity from	n an 132
dustion of 1 lin of minad not aloun and dusts in an and	
duction of t kg of mixed perforeant products in an ave	rage 135
production of 1 kg of mixed netroleum products from	1 20
procession of 1 kg of united performin products from	10-
	135
apital costs of natural gas mining	136
	104
perating costs of natural gas mining	130
data for average natural gas and LNG mining	136
station	rage 138
rememort of 1 kg of patural gas or 1 NG from an ave	
Tansport of T kg of natural gas of Elvo nom an ave	ayç
station	138
data for average natural gas and LNG transport.	140
roduction of 1 GWh of electricity from an average nat	ural
	142
aduction of I GWh of electricity in an average natural	000
oduction of FOWN of electricity in all average natural	gas
	143
apital costs of natural gas generation	143
	145
perating costs of natural gas generation.	143
data for average natural gas electricity generation	144
scione for the transmission of 1 GWh of electricity from	
	145
e basis of material flows data in Sections 4.3.2 to 4.3.4 f	tom
Becault for all and the set of Cit	1
aral gas mining values are the product of the conversion	and
put basis per GWh of electricity delivered	147
eference region (Guinée et al. (2001)) and newly develo	med
n (Data Ref. B1)	149
for some environmental impacts. Factors are indicator	's of
rces within Australia.	150
n of NG and LNG systems to convert the basis of econo	mic
tem to system. Units: subsystem output basis per GW	h af
sen to system. Onits, subsystem output basis per Gw	150
$r_{\rm r}$ during the operating life of a system (30 years) A y	مرراه
ay during the operating the of a system (50 years). A vi	aiuç
system's life is less than 30 years and a replacement sys	tem
(0) (1) (1)	1.74
ue (5) values from subsystem outputs, for estimating v	alue
r mining and transport and per MWh for generation	and
Ç	152
	122
ting value added (VA). L expressed as a percentage	e of
<u>А</u>	153
	100
ors for economic impacts. Factors are indicators of	thę
hin Australia	153
) to estimate the value of the indirect amplement	(EI)
To estimate the value of the indirect employment	(EI)
D) (ABS (2001b)). Gen. and Trans. = Generation	and
	154
re for coold imposts - Paston and indicaton of the	1.1
rs for social impacis. Pactors are indicators of the so	ciai

impacts due to all sources within Australia	LIST
when using the economic allocation method. Units: subsystem output basis per GWh of electricity delivered	Figure 2.2.1: A representation of sustainabil (modified from Briassoulis (2001)
Table 4.4.2: Minimum, average and maximum electrical generation efficiencies of generation subsystems. Also shown are multipliers (upstream factors) to estimate the additional impacts for minimum and maximum efficiencies (multiply each conversion factor of Table 4.3.47 by the upstream factor) 158	Figure 2.2.2: Three-sphere representation of the arrows represent optimisation towar
 Table 5.2.1: Environmental indicators for electricity generation from Australian fossil fuels (CC = climate change (kg CO_{2-eq}), AD = acidification (kg SO_{2-eq}), EU = eutrophication (kg PO₄³⁻_{eq}), PS = photochemical smog (kg C₂H_{4-o}), HT and ET = human and eco-toxicity (kg 1,4-dichlorobenzene), PM and SW = particulates and solid waste emissions (kg), WD = water consumption (t), RD A and RD W = Australian and world resource depletion (kg Sb_{-c}), EN and EX = energy consumption and every destruction (G1) all per MWh of electricity delivered) 	Figure 2.2.3: Four-sphere representation of the corresponding branches of economic and Empowerment and community building and reproduction
Table 5.2.2: Estimates of the uncertainty of some environmental indicators from the use of Monte-Carlo simulation (SD = standard deviation).	Figure 2.3.2: An example of a plot to examine Figure 2.4.1: Comparative full-fuel cycle, gr
Table 5.2.3: Simulations using Monte-Carlo simulation to indicate the influence of individual data uncertainty on indicator uncertainty (Triangular distribution, SD = standard deviation), A from Table 5.2.2.	systems (Rogner and Khan (1998)). The Figure 2.4.2: Comparative full-fuel cycle, gr systems (Macdonald <i>et al.</i> (1996)). ST
Table 5.2.4: Qualitative uncertainty scores for the data of the ST-BIC and ST-BIC systems calculated using the Wrisberg method (Min. = Mining, Tsp. = Transport, Gen. = Generation and Trm. =	turbine, OCGT = open cycle gas turbine explanations). Figure 3.2.1: Framework of the LCA method a
Talishission,	Figure 3.2.2: Four common probability distri
Table 5.2.6: Uncertainty in the CC indicator obtained using Monte-Carlo simulation, and the methods of Kennedy at al. and Meier (SD = Standard Deviation).	average, c = maximum) Figure 3.2.3: Box plots of system quality scor
Table 5.2.7: Uncertainty in the difference between indicators for the brown and black coal systems obtained using Monte-Carlo simulation (per MWh)	Figure 3.2.4: An example of a comparison
Table 5.2.8: Changes in environmental indicators when varying electrical generation efficiency as part of a sensitivity analysis (per MWh). Efficiencies appear in Table 4.4.2. EIV is ratio of the average of each indicator's percentage range and the value at the average efficiency.	Figure 3.3.1: Comparison of the discourse annualised costs (AC) methods,
Table 5.3.1: Economic indicators for electricity generation from Australian fossil fuels (I = capital cost (\$ billion (AUD 2000)), I _R , AC, VA and CVA = relative capital cost, annualised cost, value added and	Figure 4.2.1: Diagram of an electricity gener boundaries. Also shown is the simpli impacts only)
capital inclusive value added (\$ (AUD 2000) per MWh of electricity delivered))	Figure 4.3.1: Steps in the mining of black coal Figure 4.3.2: Simplified representation of the s
fatalities (10 ⁹ x fatalities), all per MWh of electricity delivered)	Figure 4.3.3: Simplified representation of the i technology.
employees for Australian electricity generation systems (LTI, Fatal and ED from Table 5.4.1, Chapter 5)	Figure 4.3.4: Representative constituents of generation in Australia, highlighting the
Table 6.2.2: Changes in the rankings (1 = best, 12 = worst) of different electricity generation systems, using the economic indicators, when the sensitivity analysis changes the assumed system capacity from 500 MW to 2000 MW (1 = ranking improves one place, 0 = no change in ranking, -2 =	Higgins (1991), NGGIC (1998a) and oth Figure 4.3.5: View of the mining face of the Y shows the coal dredgers used to mine bro Figure 4.3.6: Simplified representation of the c
Table 6.3.1: Ratio of RD A to RD W to show the relative probability of Australia requiring to import fossil fuels for each of the electricity generation systems (RD A and RD W from Figure 5.5.1)209	Figure 4.3.7: Simplified representation of technology. Figure 5.2.1: Subsystem contributions to the Australian fossil fuels (A = ST-BIC, B = E = 10000 Pro C = ST NG H = 0000T
	= CCGT-LNG). Tabulated subsystem of Figure 5.2.2: Cumulative distribution function using Monte-Carlo simulation, of the bro

vi

IST OF FIGURES

Figure 2.2.1: A representation of sustainability as a balance between economic supply and demand
(modified from Briassoulis (2001)
Figure 2.2.2: Three-sphere representation of the concept of sustainability (adapted from Barbier (1987)).
The arrows represent optimisation towards sustainability.
Figure 2.2.3: Four-sphere representation of the concept of sustainability (Pezzoli (1997)). Marked are
corresponding branches of economic and political literature. The numbers represent: 1 Holism; 2
Empowerment and community building; 3 Social justice and equity; and 4 Sustainable production
and reproduction
righter 2.3.1: Framework for organisations to develop sustainability indicators (veleva and Ellenbecker
(2001)
Figure 2.4.1: Comparative full fuel quele greenbaues are amissione of collected electricity expension
rigure 2.4.1. Comparative full-fuel cycle, greenhouse gas emissions of selected electricity generation
Systems (Rognet and Kital (1996)). The units g per $Rwn = Rg$ per $N w n$
Figure 2.4.2. Comparative fun-fuel cycle, gleenhouse gas emissions of selected electricity generation systems (Macdonald et al. (1996)). ST \Rightarrow atoms turbing LCCC = integrated combined cycle ac
systems (Macdonaid et al. (1990)). $S1 =$ steam furbine, NCC = integrated combined cycle gas turbine (see Glossop) for
explanations)
Figure 3.2.1: Framework of the LCA method according to the ISO 14040 series of standards (ISO 14040)
1 igure 5/2. (1 Yunio work of the Zer's method decording to the 150 14040 series of station us (150 14040).
Figure 3.2.2: Four common probability distributions from industrial measurements ($a = minimum = b =$
r_{ij} average $c = maximum$).
Figure 3.2.3: Box plots of system quality scores for notional 'life-cycle' systems A, B and C. The basis
for the positioning of the 1 st and 3 rd percentiles is the required confidence
Figure 3.2.4: An example of a comparison between notional 'life-cycle' systems A and B using the
cumulative normalised difference method
Figure 3.3.1: Comparison of the discounted costs profiles from the net present costs (NPC) and
annualised costs (AC) methods
Figure 4.2.1: Diagram of an electricity generation system for consuming fossil fuels, with 'life-cycle'
boundaries. Also shown is the simplified perspective (grey = all impacts, spotted = economic
impacts only)
Figure 4.3.1: Steps in the mining of black coal (adapted from NGGIC (1998b))
Figure 4.3.2: Simplified representation of the steam turbine electricity generation technology
Figure 4.3.3: Simplified representation of the integrated gasification combined cycle electricity generation
technology
Figure 4.3.4: Representative constituents of black and brown coals and natural gas for electricity
generation in Australia, highlighting the high water content of brown coals (from Brockway and
Higgins (1991), NGGIC (1998a) and others)
Figure 4.3.5: View of the mining face of the Yallourn brown coal mine, Victoria (Holmes (1993)). Also
snows the coal dredgers used to mine brown coal
Figure 4.3.0: Simplified representation of the open cycle gas turbine electricity generation technology.141
righted 4.5.7. Simplified representation of the combined cycle gas turbine electricity generation
Figure 5.2.1: Subcystem contributions to the equironmental indicators for electricity commission from
Australian face i fuels (A = ST_RIC R = ICCC_RIC C = ST_RICE D = ICCC_RICE R = ST_RC
$F = 1000$ Br($G = ST_NG_H = 0.000$ NG $I = 0.000$ NG $I = ST_1 NG_K = 0.000$ NG $I = 0.000$ NG $K = 0.000$ NG $I = 0.000$ NG
= CCGTLNG. Tabulated subsystem contributions appear in Section A3.2 (Appendix 3) 168
Figure 5.2.2: Cumulative distribution function of the difference between the CC indicators obtained
using Monte-Carlo simulation of the brown and black coal systems
Figure 5.3.1: Subsystem contributions to the economic indicators for electricity generation from
Australian fossil fuels (A-L as in Figure 5.2.1). Subsystem contributions for I and I _P are identical.
Tabulated subsystem contributions appear in Section A3.2 (Appendix 1)
Figure 5.4.1: Subsystem contributions to the social indicators for electricity generation from Australian
fossil fuels (A-L as in Figure 5.2.1). Tabulated subsystem contributions appear in Section A3.2
(Appendix 3)
Figure 5.5.1: Sustainability indicators for electricity generation from Australian fossil fuels normalised
using Australian data. Numbers at top indicate the magnitudes where scores exceed the scale).
Section A3.3, Appendix 3 contains a table of these indicator scores
Figure 6.3.1: The maximum and minimum positions for the mean of triangular distributions. Positions of

b and c calculated using laws for similar triangles and equal probabilities (area left and right must

The reliable and inexpensive supply of electricity is crucial to the development of any nation's economy and the welfare of its population. The prevalent forms of electricity generation in Australia, which consume Australia's abundant black and brown coals and natural gas, have some negative effects on its environment. Furthermore, exports of Australian black coal and natural gas (as LNG) fuel other nations' electricity generation systems, affecting their economies, social welfare and environment. One Australian fossil fuel, brown coal, has a poor reputation, based on high CO₂ emissions from its combustion. Using the same basis, natural gas has a good reputation. Yet, testing of these reputations with respect to their combined environmental, economic and social performance, or sustainability, has not occurred. Therefore, this thesis develops a method for quantifying the impacts on sustainability caused by generating electricity from Australia's abundant fuels supported by a detailed analysis of environmental impacts with key economic and social impacts.

A critical review of sustainability assessment finds no common or standardised method, although it finds that the use of indicators is essential for an objective. quantitative approach. The selection of life cycle assessment (LCA) for the environmental indicators enables the quantification of environmental impacts and the use of extended boundaries. Examined are several steps in the LCA method that cause difficulties during application, including allocation of impacts to products, analysis of uncertainty and the quantification of the resource depletion impact. Testing of several approaches for analysis of uncertainty establishes that separate qualitative and quantitative information is useful in describing the robustness of environmental indicators to uncertainties in data. The creation of an Australian resource depletion indicator enables the examination of the local implications of the consumption of resources. No previous methods for producing economic and social indicators allowed the use of extended boundaries, and thus modifications proved necessary. The extension of the optional LCA practice of normalisation to these indicators enables the comparison of the importance of each different type of impact using a scientific, valuefree basis: comparative contribution to Australia's impact from all sources.

The assessed electricity generation systems feature: boundaries extended to include fuel mining and transport to the power station and electricity generation and delivery; electricity generation technologies currently used in Australia and some in development;

ABSTRACT

Abstract

CANDIDATE'S CERTIFICATE

This is to certify that this thesis contains no material which has been accepted for the award of any other degree or diploma in any university or other institution. Additionally to the best of my knowledge, this thesis contains no material previously published or written by another person, except where due reference is made in the text of the thesis.

and electricity generation fuelled using exported Australian fuels. Data selected for the sustainability assessment of these systems, is from a critical review of an extensive range of data sources.

No one fuel has universally superior sustainability, or environmental, performance as measured using the indicators. However, the indicators of sustainability show that the natural gas systems, using the combined-cycle gas turbine electricity generation technology, are superior to the others for most of the indicators. Yet, various impediments to increasing consumption of natural gas (including its greater resource scarcity), limit Australia's ability to replace coal systems with these natural gas systems. While brown coal systems are generally inferior to black coal systems for climate change, brown coal systems outperform black coal systems for the majority of indicators. Advanced coal technologies are markedly better than the present conventional systems. However, they require substantial further development to become commercially viable. Systems using black coal have markedly poorer safety performance than do other fuels, due to their fuel mining and transport operations. Systems using the export fuels, LNG and black coal, have substantially greater impacts on sustainability, than do domestic systems with the same electricity generation technology. The normalised indicators show that capital expenditure is of greatest importance when comparing different electricity generation systems.

The indicators of sustainability developed have major implications when considering additions, or modifications, to existing electricity supply networks. The developed sustainability assessment method is adaptable for use in other industry and product systems.



John May

Acknowledgements

Sustainability of Australia's Electricity Generation

ACKNOWLEDGEMENTS

The author would like to acknowledge the following persons and organisations:

- Dr. D. J. Brennan for his valued contributions and supervision;
- M. McIntosh and S. Bhattacharya, of the CRC for Clean Power from Lignite, for their assistance with the simulation and economic assessment of advanced coal electricity generation technologies;
- T. Innocenzi (Hazelwood Power), D. Stickney (Loy Yang Power) and R. French (Yallourn Energy) for their help in obtaining data for brown coal electricity generation systems; and
- The many other companies and individuals who willingly provided data and technical information.

The author gratefully acknowledges the financial and other support received for this research from the Cooperative Research Centre (CRC) for Clean Power from Lignite, which is established and supported under the Australian Government's Cooperative Research Centres program.

1. INTRODUCTION

'There is no way to make brown coal environmentally viable for energy production' BROWN COAL SHOULD BE FINISHED AS A FUEL, GOVERNMENT TOLD - THE AGE (MELBOURNE, AUSTRALIA) 18/6/02

1.1 FOSSIL FUELS, **SUSTAINABILITY**

Energy is necessary for daily survival (WCED (1987)). It provides heat for warmth, cooking, and manufacturing, or mechanical power for work and transport, without which we could not support our way of life (WCED (1987)). In our society, electricity is the most pervasive energy provider, delivered to almost every home and business. Thus, the reliable and inexpensive supply of electricity is crucial to the development of any nation's economy and the welfare of its population.

Australia has an abundance of many fossil fuels. Coal in particular, with 90 billion tonnes, or 9 % of world reserves (as at June 2000) within Australia (BP (2000)). Natural gas resources are also large, with over 44×10^{12} cubic feet, or 0.9 % of world reserves, within Australia (BP (2000)). Moreover, the extraction of these fuels is inexpensive, as it is necessary to use only the most easily recoverable reserves. Consequently, the utilisation of these fuels was favoured when planning Australia's electricity generation network. In 1999, over 90 % of Australian electricity generation used these fuels (Electricity Supply Association of Australia (1999)). Of this amount,

^{*} Electricity consumption once contributed only 4% of energy, in the 1930's (OECD (1985)). By 2000, it had risen to become a major contributor, 36 % of energy in 2000 (IAEA (2000)).

1

ELECTRICITY GENERATION AND

Chapter 1:Introduction

55 % used black coal, 30 % brown coal (or lignite) and 5 % natural gas (Electricity Supply Association of Australia (1999)). Imbalances in the distribution of these resources, coupled with the fragmented development of the Australian electricity supply, have resulted in the use of different fuels in different states. The major black coal resources are in New South Wales and Queensland, and thus over 90 % of electricity generation uses this fuel in these states. Similarly, over 90 % of electricity generation in Victoria uses brown coal (Electricity Supply Association of Australia (1999)). In South Australia and Western Australia, natural gas is more abundant than coal, and thus much of the electricity generation uses natural gas. Therefore, each Australian State has its own inexpensive fossil fuel as an energy source for electricity generation.

The combustion of fossil fuels releases a wide range of substances into the environment. Many of these substances have harmful effects on fauna and flora, and significantly on the earth's temperature and climate. Electricity generation is a proportionally large consumer of fossil fuels within Australia, when compared to most other consumers (ABS (2001a)). Thus, electricity generation is a major contributor to these effects. It accounts for over one third of climate effecting gases ('Greenhouse Gases'), and substantial proportions of the acidifying gases sulphur dioxide and nitrogen dioxide (AGO (1999)). A similar pattern occurs in other countries with similar electricity generation systems, e.g. U.K (see DoE (U.K.) (1997)) and U.S.A. (see EPA (U.S.A.) (2000)). Therefore, the assessment of these negative effects, as well as benefits to our economy and social welfare, is essential when evaluating the net benefit (or harm) of electricity generation using fossil fuels.

The sustainability concept specifically considers impacts on the environment, economy and social welfare. The sustainability concept has many different definitions. The basis for most is the off-quoted statement from the Brundtland report (WCED (1987)): 'To meet the needs of the present without compromising the ability of future generations to meet their own needs'. Disagreements surround proposals of measures of present sustainability, and methods for attaining long-term sustainability. Contentious also is the question of how to recognise the attainment of sustainability. However, all agree that the goals for sustainability are the minimisation of all negative impacts and maximisation of all positive impacts on the key areas of the economy, the environment, and social welfare.

Sustainability of Australia's Electricity Generation

Electricity, when consumed, produces no direct negative economic, environmental or social impacts at its point of use. It may thus be an ideal energy source for a sustainable society (Schaap (2000)). Indeed the supply of electricity has since its introduction, enhanced both economic and social welfare, by redistributing employment and social services, between urban and rural areas, and increasing the general standard of living (IAEA (2000)). The sustainability concept requires that future generations have access to these same benefits. Nevertheless, as is earlier intimated, electricity generation has negative sustainability impacts (IAEA (2000)). Therefore, to maintain electricity as an energy provider for sustainability, these negative sustainability impacts must be minimised or removed.

Electricity generation involves the conversion of energy contained in fossil fuels, plants, animals, atomic fission, sunlight, wind, or water. The conversion from energy source to electricity is unique for each. Consequently, the types and magnitudes of effects on sustainability (sustainability impacts) caused by each will be unique (WCED (1987)). In addition, it is possible to use different technologies for this conversion. The choice of technology also modifies the type and magnitude of the sustainability impacts resulting from the conversion. Thus, by nominating a particular energy source and technology combination, a corresponding range and magnitude of sustainability impacts will occur (WCED (1987)). Therefore, assessment of each possible energy source and technology option is necessary to reduce the negative sustainability impacts of electricity generation for Australia.

Traditionally, analysis of individual impacts have utilised physical boundaries, and thus a traditional assessment of the environmental or economic impacts of electricity generation would consider the electricity generation plant and its environs. Inherent in this method is an assumption that the electricity generation plant is not responsible for any of the impacts caused by other industries and electricity consumers. However, this assumption is false. Each electricity generation plant generally consumes an energy source obtained off site and materials produced in different industries. Similarly, its construction, commissioning, decommissioning and demolition, use materials, energy, and equipment produced in different industries. All of these different industries rely on the electricity generation plant to consume their products, and are thus dependent on its existence. Consequently, the electricity generation plant is responsible for a proportion of the impacts caused by these dependent industries. Electricity consumers can

2

Chapter 1:Introduction

3

Chapter 1:Introduction

consume electricity both efficiently and inefficiently to meet a need (i.e. to heat a room). If they consume electricity inefficiency, a greater amount of electricity will be necessary to meet this need, than if they had consumed electricity efficiently. Generation of this additional electricity will result in greater impacts from the electricity generation plant. If the electricity generation plant has a responsibility to ensure the consumption of its product is efficient, then it is responsible for a proportion of the impacts caused through this inefficiency. The inclusion of these proportions of impacts from dependent industries and product use is known as the 'life cycle' perspective, and is of increasing interest in both environmental and sustainability analysis (Azapagic and Perdan (2000), Cowell et al. (1999)).

The use of brown coal, often branded a 'dirty fuel', for electricity generation, has come under particular scrutiny, far more so than do other Australian fossil fuels. Many quotes similar to that provided at the start of this chapter appear in newspapers and reports. The basis for this branding is the comparatively greater emissions of 'Greenhouse Gases' from power stations consuming this fuel. Meanwhile, for natural gas and black coal, other groups have reported varying findings over their relative 'Greenhouse Gas' emissions from electricity generation. For example, the Australian Gas Association, in the context of greenhouse gas emissions, state (AGA 2002): 'In mainstream electricity generation, power stations fuelled by natural gas are jar cleaner than those burning oil, black coal, or brown coal,' While the Australian Coal Association state (BHP Minerals Technology (2001)): 'On a full life cycle basis, gasbased electricity generation may have similar or even higher greenhouse gas emissions than coal-based generation.' The basis for these claims is only one type of negative effect, greenhouse gas emissions. Therefore, these claims may misrepresent the actual impacts of electricity generation using these fuels. Only a sustainability assessment, including all positive and negative effects, can estimate these actual impacts.

1.2 AIM AND SCOPE

Consequently, the aim of this thesis is to assess the relative impacts on sustainability of electricity generation systems, consuming Australian brown coal, black coal and natural gas fossil fuels, and utilising a 'life cycle' perspective.

The study includes fossil fuels prevalent in Australian electricity generation (black

Sustainability of Australia's Electricity Generation

and brown coals, and natural gas) and exported in large quantities for electricity generation (black coal and LNG). The comparison places greater emphasis on the assessment of environmental impacts, due to the desire to establish whether brown coal really is a 'dirtier fuel' than are black coal and natural gas. The bases for selection of technologies for electricity generation are: for currently available technology, common usage; and for proposed technologies, the authority of source and extent of potential benefits. The study data uses average Australian operations, rather than best case or particular plants, to ensure that the assessment conclusions are applicable throughout Australia.

1.3 OVERVIEW OF THE STUDY

Chapter 2 examines sustainability, discusses options for its measurement and establishes which method has greatest advantages. Then, this chapter critically reviews assessments of major negative and positive effects from electricity generation, to establish which effects must be included in the assessment of sustainability.

Chapter 3 reviews methods for quantification of these important effects. It reviews as a 77006 the strengths and weaknesses of the life cycle assessment (LCA) environmental-method-In particular, it examines and proposes solutions for difficulties in: quantifying the effect of resource consumption; comparing different types of effects, through normalisation and weighting; and estimating the uncertainty of the quantified effect measures due to uncertainties in data, decisions and assumptions. It reviews methods for quantifying economic effects and proposes measures to overcome these method s³ shortcomings. Finally, it proposes some methods for quantifying effects on social welfare.

Chapter 4 details the steps necessary to quantify the sustainability effects of electricity generation using the methods of Chapter 3. It does this by, firstly, describing the types of electricity generation systems prevalent in Australia and equivalent systems consuming fossil fuels exported from Australia. Following each description, it provides the information necessary to use the methods of Chapter 3 on these systems. Finally, it presents other data and calculation steps necessary to quantify the sustainability effects and test the solutions proposed in Chapter 3 for dealing with difficulties in the LCA method.

Chapter 1:Introduction

Chapter 5 presents and examines the quantified effects for the electricity generation systems. Additionally, it presents the results of the tests of the proposed solutions to difficulties in LCA. It includes a unique plot of sustainability effects, using normalisation, which enables comparisons of the importance to Australia of contributions to the effects.

Chapter 6 discusses some implications of the sustainability assessment concerning the relative sustainability performance of the different fuels and types of electricity generation systems. Then, it discusses the advantages and limitations of the method developed for quantifying sustainability effects and the proposed solutions for difficulties when using the LCA method. Then, it discusses the possible use of the method in evaluating new systems, or replacing old systems, in electricity generation networks. Finally, is discusses the method's flexibility for assessing the sustainability of other product systems.

Chapter 7 identifies the conclusions and recommendations of the thesis.

Appendices 1 and 2 contain supporting information for Chapters 4 to 6, including tables of data sources and example calculations. Within this report are many abbreviations and industry-specific terms, which often have conflicting definitions in other literature. Therefore, presented are a list of the nomenclature and acronyms ('Nomenclature') and a glossary of terms ('Glossary'). Lastly, appears a list of references and other relevant literature.

Parts of this thesis, including the quantification of the environmental effects of electricity generation using LCA and the difficulty of estimating uncertainty in LCA, have been published in conference proceedings and journals (May and Brennan (2001, 2002, 2003a, 2003b)).

2. SUSTAINABILITY

'Sustainable development is development that meets the needs of the present without compromising the ability of future generations to meet their own needs... The goals of economic and social development must be defined in terms of sustainability in all countries... Interpretations will vary, but... must flow from a consensus on the basic concept of sustainable development and on a broad strategic framework for achieving it.'

2.1 OVERVIEW OF SUSTAINABILITY

Assessing the relative impacts on sustainability of electricity generation systems requires an understanding of the concept of sustainability. Many different perspectives on the concept of sustainability had developed since the publishing of 'Our Common Future' in 1987 (Board on Sustainable Development (1999)). Each of these perspectives represented a different understanding of what it meant to reach sustainability. These perspectives invited establishment of measures to assess progress towards sustainability. Reviewing these perspectives, and the measurement techniques thus derived, should indicate an appropriate methodology for the assessment of electricity generation systems.

Section 2.2 reviews the many perspectives on the concept of sustainability. Section 2.3, describes the measurement techniques developed from these perspectives, and reviews their use. Section 2.4 expands up the links, intimated in Chapter 1, between

WCED (1987).

6

OUR COMMON FUTURE, WCED (1987), pg 43.

Chapter 2:Sustainability

electricity generation and wider sustainability. Many studies have discussed one type of impact on sustainability for electricity generation systems, i.e. the environment. Thus, Section 2.4 also reviews these studies to indicate which impacts on sustainability are of most importance for electricity generation.

2.2 DEFINING SUSTAINABILITY

2.2.1 OVERVIEW OF DEFINING SUSTAINABILITY

An important outcome of Chapter 1 was the recognition that it is important to consider environmental, economic and social impacts to compare electricity generation systems. Sustainability assessment and sustainability impacts were the terms applied to this type of comparison and the range of impacts. For this reason it is important to review the concept of sustainability to define what it is, and what it is not, so that later discussions of the sustainability of electricity generation systems may be better understood.

Section 2.2.2 reviews the history of the concept of sustainability and the breadth of definitions. Section 2.2.3 then examines the reasons for the great number of sustainability definitions. Section 2.2.4 details the common models of sustainability obtained from these concepts. Section 2.2.5 discusses ways to measure progress towards sustainability using the common models of sustainability.

2.2.2 SUSTAINABILITY DEFINITIONS[†]

The long history of the sustainability concept has its beginnings in eighteenth century economic theory, which predicted a limit to economic growth caused by the availability of good agricultural land and scarcity of natural resources (Mebratu (1998)). More than a century later, in the 1960's and 1970's, the economist Ernest F. Schumacher introduced the idea that planners and policy makers should account for social welfare, resource depletion and environmental destruction when determining appropriate technology (Mebratu (1998)). However, it was not until the 1972 UN Conference on

Sustainability of Australia's Electricity Generation

Human Development in Stockholm that an international forum intimated that economic development may need to account for environmental concerns (Mebratu (1998)). At the same time, a group of eminem scientists and concerned citizens met in Rome ('The Club of Rome') and produced a comprehensive report, which emphasised that environmental emissions would soon exceed the limits of the earth to assimilate (Mebratu (1998)). During the next decade, UN reports steadily entwined the terms economic development and environment, until in 1980 a report used the term 'Sustainable Development' in its subtitle (Mebratu (1998)). Nevertheless, widespread discussion of sustainability was limited until the publishing of the UN-sponsored Brundtland report, WCED 1987 (Mebratu (1998)). Its most commonly quoted passage, which appears at the start of this chapter[‡], has spawned many interpretations of sustainability. There were more than 80 such interpretations in 1994, according to Holmberg $(1994)^{\$}$ (Mebratu (1998)). After the Brundtland report, a number of world conferences highlighted sustainability, including the UN Conference on Environment and Development (UNCED)** in 1992, and the World Summit on Sustainable Development (WSSD) in 2002. Despite this long history, a definitive definition of the concept of sustainability is still lacking.

The definitions presented in the next few paragraphs were important in the formation of the author's views on sustainability. The definitions drawn from the Brundtland report usually emphasise needs, limits on development, inter-generational equity and intra-generational equity (Briassoulis (2001)^{††}). They are concerned with the adverse effect that currently accepted methods of development have on the environment and social welfare (Pezzoli (1997)). One such interpretation, using the Brundtland report definition, and its further refinements, as its basis, consists of (Overcash (2001)):

- Direction of investments;
- Exploitation of resources;
- Orientation of technological development; and

* The 'Brundtland definition', in the literature, usually refers to the statement 'To meet the needs of the present without compromising the ability of future generations to meet their own needs' (WCED (1987)). [§] Mebratu (1998) does not provide details of this source. " 'Rio Conference' and 'Earth Summit' are other commonly used terms for UNCED. ^{††} This paper is a review, and thus other works are the source of this and many other statements.

Chapter 2:Sustainability

[†] In keeping with traditional usage, sustainable development and sustainability are used interchangeably in this discussion.

economic theory. One such definition, defines sustainability as a dynamic equilibrium state lying between demand and supply (Figure 2.2.1) (Briassoulis (2001)). The society demands a desired development pattern for itself, in which the products must fit. Thus, suppliers of products that fail to meet the new pattern, modify their products and/or the technologies that produce them, to meet the new constraints.



Figure 2.2.1: A representation of sustainability as a balance between economic supply and demand (modified from Briassoulis (2001).

Most definitions have sought broad acceptability, rather than a precise, consistently interpretable and useable definition (Ayres (1996), Barbier (1987), Upham (2000)^{§§}). Yet, no consensus as to the appropriate definition of sustainability exists (Ayres (1996), Azapagic and Perdan (2000), Briassoulis (2001), Cowell et al. (1999), Overcash (2001), Pezzoli $(1997)^{***}$). Why is this so?

The word 'sustainable' has many meanings, such as 'keep into being', 'to provide with food and drink, or the necessities of life', and 'to endure without giving way or yielding', which are relevant in different ways to different groups: governments and organisations, the impoverished, and farmers and small business owners (Pezzoli (1997)). Thus, differences between the morality of the proponents of the definitions,

^{§§} For example, The Natural Step Theory excludes limits of concentration and damage thresholds as they may introduce the 'potential for disagreement' (Upham (2000)). * The authors of most other publications encountered also hold this opinion.

Institutional change.

The first point, 'Direction of investments', advocates the distribution of investment into areas to enable the fulfilment of the essential needs, e.g. jobs, food, energy, water and sanitation (i.e. social and economic growth) (WCED (1987)). Expressed in this point, is the Brundtland report's desire for intra-generational equity (worldwide equality) through directing funds into third-world (or developing) countries to eliminate absolute poverty (WCED (1987)). The second point, 'Exploitation of resources', advocates reducing industrial material and energy consumption, while distributing equitably the generated wealth (WCED (1987)). The third point, 'Orientation of technological development', advocates design of new technology to reduce environment impacts of the old, while ensuring that this new technology is chosen over less favourable options, if economic costs are similar (WCED (1987)). The last point, 'Institutional change', suggests the modification of government and legal frameworks to enforce the implementation of the first three points (WCED (1987)).

Another similar definition is that sustainability is 'a pattern of social and structural economic transformations (i.e. 'development') which optimises the economic and other societal benefits in the future' (Goodland and Ledec (1986)). This definition suggests that a sustainable economic activity is one in which social welfare is increased, while resource conservation is maximised, and environmental degradation is minimised. within given societal, economic and technical constraints (Barbier (1987)). Thus, human activity, i.e. development, which always involves both the consumption of resources and the emission of wastes, requires a 'pattern of activity...that is more resource- and waste-minimising than maximising' (Barbier (1987)).

Another definition, describes sustainability using 'sustainable livelihoods', which refers to 'a level of wealth and of stocks and flows of food and cash which provide for physical and social well-being and security against becoming poorer' (Chambers (1986)^{‡‡}). This definition advocates a rise in real income for all, and increased stability of that income. This applies in rural society, for example, where natural events, such as droughts and floods, and the results of bad practices, i.e. salinity from over irrigation, can drastically affect income (Barbier (1987)). The global poor, in both urban and rural settings, are highly sensitive to these and other similar stresses (Barbier (1987)).

The previous three definitions applied a social perspective; others have applied

10



2.2.3 REASONS FOR THE MULTIPLICITY OF SUSTAINABILITY DEFINITIONS

^{‡‡} Excerpt from a discussion on this method in Barbier (1987).

Chapter 2:Sustainability

and the society that nurtured them (Buhrs and Aplin (1999)), generate different views on sustainability. Moreover, the views of the proponents are likely to change as their society changes (Briassoulis (2001)). Sustainability has been described in the literature as an 'amorphous concept', unable to be clearly defined (Overcash (2001)), and 'understood intuitively by all but very difficult to express in concrete, operational terms' (Briassoulis (2001)).

Some 'radical environmentalists' view this lack of definition as allowing for 'perverse' interpretations of the sustainability term, to allow 'business-as-usual (activities) or shallow rhetoric to win votes or placate customers or shareholders' (Ayres (1996), Buhrs and Aplin (1999)). Furthermore, some claim that it hinders application of the concept, leading to frustration and devaluing of the concept (Mebratu (1998)^{†††}. Conversely, some view this lack of definition as ensuring continued effort. As the definition can evolve over time, the 'solving' of one problem will lead to a changed definition of sustainability (Wallace (1995)). This new definition may highlight some other problems for solving, rather than reduce the importance of the sustainability concept or encouraging its neglect (Wallace (1995)^{‡‡‡}).

Consequently, it is likely that sustainability will remain a concept, grouping a number of definitions with similar ideals, rather than having a precise definition (Buhrs and Aplin (1999), Cowell et al. (1999)). Thus, described in the next section are models of the concept of sustainability, which rely less on a fixed model.

2.2.4 SUSTAINABILITY CONCEPT MODELS

Sustainability concept models include a number of definitions with similar ideals to formulate simple models of sustainability. The first such model separates sustainable development in three spheres: the economy, the environment and social welfare. Many proponents suggest three spheres as sufficient to define sustainability (Azapagic and Perdan (2000), Giobal Reporting Initiative (2002), Overcash (2001)). This concept is generally visualised using an intersecting goals diagram (see Figure 2.2.2), first used by Barbier (1987). The environmental system provides the raw materials for the economic system, and accepts wastes from both the economic and social systems (Azapagic and

Sustainability of Australia's Electricity Generation

Perdan (2000)). The economic system provides goods and services to the social system (Azapagic and Perdan (2000)). Traditional development optimises economics alone, while Marxist economics optimises both social and economic system goals (Barbier (1987)). Traditional environmental analysis optimises environmental goals or a combination of environmental and economic goals. Sustainability optimises each of the three system goals.

> Economic System,

While the three intersecting spheres model was prevalent in the literature, other models exist. Once such model shows four intersecting spheres ('political ecology'): a. the environment (and history and power); b. the legal-institutional terrain; c. culture and civil society; and d. the economy and technology (Figure 2.2.3) (Pezzoli (1997)). This model disaggregates the social sphere of the three-sphere model into two spheres: b. and c. This four sphere model is perhaps better at defining sustainability, than the threesphere model, as there are differences between how institutions (sphere b.) and its people (sphere c.) would need to act and how they would be effected by a movement towards sustainability. The UN's Division for Sustainable Development also uses a four-sphere model: social, economic, environmental and institutional (Division for Sustainable Development (2003)).

Chapter 2:Sustainability



Figure 2.2.2: Three-sphere representation of the concept of sustainability (adapted from Barbier (1987)). The arrows represent optimisation towards sustainability.

^{†††} Comments attributed to various sources: for more details see Mebratu (1998), pg. 503.

^{***} Statement obtained from its reproduction in Anderson (2001).

Chapter 2:Sustainability





2.2.5 REACHING SUSTAINABILITY

The interaction of the spheres in the three sphere concept model is complex. With increased development of the concept of sustainability, has come an improved understanding of the interrelations of its spheres (Board on Sustainable Development (1999)). Empirical evidence for the interdependence of the three spheres was provided through a study that showed that US states with lower pollution levels and greater environmental controls had stronger economies and reduced income disparity (mentioned in Guy and Kibert (1998)^{\$\$\$}). The optimisation of one impact is unlikely to give the optimum result for all impacts (Barbier (1987), Overcash (2001)). Consequently, ensuring maximum sustainability benefit may require trade-offs between the spheres (Barbier (1987), Overcash (2001)). For example, sacrifice of some measure of financial benefit may result in a more socially attractive employment strategy rather

Sustainability of Australia's Electricity Generation

than a fully mechanised alternative"***. There is some disagreement about the substitutability of items within different spheres (Ayres (1996)): such as how many jobs (man-made capital) are equivalent to a tree (natural capital). Some economists would allow substitution of such items on a cost basis, while others declare many such items unsubstitutable (Ayres (1996)). Trade-off would require some way of quantifying the contribution of each to sustainability by adjudging the relative importance of that measurement. None of the definitions of the sustainability concept allowed quantification of sustainability, as they sought broad acceptance, rather than sharp definition (Ayres (1996), Bell and Morse (2001)). Others have suggested that it is possible to produce such a definition (Matson and Carasso (1999)). Therefore, to use the three-sphere concept model an alternative method of quantification is necessary.

2.2.6 SUMMARY OF DEFINING SUSTAINABILITY

Sustainability and sustainable development are commonly used terms, in industrial, governmental and other reports, when describing actions taken (Cowell et al. (1999), Pezzoli (1997)). Yet, there remains no consensus on their definition. Therefore, practitioners have developed concept models, which consider only areas of broad agreement, to allow further work in sustainability assessment. The concept model most prevalent has three spheres representing the economy, environment, and society. The use of these concept models requires the ability to quantify the contributions to each sphere, which none of the definitions of sustainability allow. Consequently, alternative methods for measuring sustainability are necessary.

2.3 MEASURING SUSTAINABILITY

2.3.1 OVERVIEW OF MEASURING SUSTAINABILITY

Key to the practical use of the sustainability concept is the ability to quantify sustainability. This will require the development of analytical tools (Barbier (1987)), as has been achieved in other forms of assessment. Different definitions require often

**** See Barbier (1987) and Overcash (2001) for further examples.

^{\$\$\$} Study mentioned: Templet, P.H. (1995), The positive relationship between jobs, environment and the economy: an empirical analysis and review, Spectrum, Spring, 37.

entirely different measures, e.g. quality of life (social measure) and resource scarcity (material measure). Thus, the tools developed from definitions will necessarily vary. This section reviews some of the proposed tools for measuring sustainability and assesses their suitability for sustainability assessment using the three-sphere concept model (see Section 2.2.4).

Section 2.3.2 presents this review, and selects an appropriate method. Section 2.3.3 then examines the selected method in greater detail. Section 2.3.4 discusses the application of the selected method for industrial systems. Often assessment methods use criteria to adjudge progress towards their goal. Section 2.3.5 discusses the use of criteria in sustainability assessment. Another important choice in assessments is that of boundary placement. Section 2.3.6 discusses appropriate boundaries for sustainability assessments.

2.3.2 TOOLS FOR MEASURING SUSTAINABILITY

Despite the wide range of tools for sustainability they share a number of common themes. For example, many tools share common tasks, such as to (from Barbier (1987), Briassoulis (2001) and Overcash (2001)):

- evaluate existing sustainability performance and indicate problem areas; a.
- monitor progress towards, or away from, enhanced sustainability; b.
- compare alternative designs for both improvements and new systems; C.
- estimate the effectiveness of the improvements to sustainability performance; d.
- ensure application of identified improvements; and e.
- f. identify unintended consequences of improvements.

A sustainability tool must be able to determine the degree of optimisation of each sphere of sustainability (Overcash (2001)), i.e. environment, economy and social welfare, and estimate the effect on this optimisation of possible trade-offs between the aspects. As the sustainability concept includes many different types of effects, tool development should utilise people from various backgrounds and disciplines, such as science, engineering, ocial science, and economics (Hammond (2000)). Most importantly, for the tool to obtain wide acceptance it must have scientific and social credence and be both practical and effective (Board on Sustainable Development

(1999)).

Early workets suggested a combined single indicator for sustainability, expressed as an index (i.e. 'basic needs' or 'physical quality of life') (Barbier (1987)). In the building industry these and other indicators (i.e. sustainable community, healthy community, human development, and urban environment) have been proposed and developed in an 'attempt to create the most balanced approach' to the three spheres (Guy and Kibert (1998)). Developing these types of aggregated indicators requires the use of techniques such as brainstorming, focus groups, expert opinion, surveys and other quantitative and qualitative measurements (Guy and Kibert (1998)). They are designed to provide a basic test, for the local community, which hints at the 'big picture' of the sustainability of the development, through the measurement of simple, discreet sources of information (Guy and Kibert (1998)).

One such method, the Sustainable Process Index, attempted to estimate the required surface area of earth required to negate the detrimental effects of the process on sustainability. However, its coverage was limited, considering only environmental impacts and employment (Krotscheck (2000)). The World Wide Fund for Nature was reporting an environmental version of this, the Eco-footprint, in 1992 (World Wide Fund for Nature (2002)). Another, widely promoted example, Eco-Efficiency, was developed by 'business for business' by the World Business Council for Sustainable Development, which includes both environmental and economic measures in its indicator, covering seven impacts: material intensity; energy intensity; toxic dispersion; recyclability; non-renewable resource consumption; product durability; and product service intensity (Azapagic and Perdan (2000), de Simone and Popoff (1997)). Many other combined indicators exist, some for use in industrial systems^{††††}. Some of these were for internal use by specific companies (i.e. ICI and Unilever), and others by industrial or governmental organisations (i.e. AIChE, UNEP, the World Resources Institute, etc.) for wider application. Most of these included mainly environmental factors (Azapagic and Perdan (2000)).

Other tools convert social and environmental impacts into economic impacts (IAEA (2000)). For example, Virtual Eco-Costs '99 proposed the use of the ratio of the last and most expensive cost associated with negating an environmental effect, to the sum of the product's economic costs, taxes, and profits (Vogtländer et al. (2001)), as an

tttt See Azapagic and Perdan (2000) for a more detailed review of these indicators.

16

17

Chapter 2: Sustainability

appropriate method for weighting impacts. Dispute continues about the validity of placing economic values on environmental and social impacts (Verbuggen and Kuik (1991)). For example, the valuation of impacts such as illness, human life, and visual intrusion (among others) present difficulties (IAEA (2000)).

The integration procedure is often an area of dispute, as it is not possible, using current methodologies, to provide a scientific weighting system free of value choices (Briassoulis (2001)). This is due mainly to the morality, cultural, and other differences between societies (see Section 2.2.2). For example, while water and air pollution were considered high priority hazards in most nations, industrialised nations also ranked ozone depletion and climate change highly, while less industrialised nations ranked drought or flood, disease, and the availability of living resources in their place (Board on Sustainable Development (1999)). Consequently, a combined measure of sustainability would be pertinent only for those who concur with the values used in its weighting mechanism. Sustainability is too complex to allow its measurement with a single measure, and thus sets of measures (or indicators) are more common (Bell and Morse (2001), Board on Sustainable Development (1999), Briassoulis (2001), Verbuggen and Kuik (1991)).

2.3.3 INDICATORS FOR MEASURING SUSTAINABILITY

Indicators have a long history of widespread use and acceptance in measuring effects in all type of assessments (Bell and Morse (2000), Board on Sustainable Development (1999), Pezzoli (1997)). Indicators are useful for communication with society and governments, and may point out impediments to sustainable development in the community (Guy and Kibert (1998)). Indicators generally give measures of the existing degree of sustainability (compared to a reference), an indication of the difference between modification (or new) options, or monitor the effects of previous changes (Ayres (1996), Briassoulis (2001)).

They should ideally (Bell and Morse (2001), Braiassoulis (2001), UNCED (1992)):

- 1. be comprehensive in their coverage of the social, economic and environmental impacts;
- 2. integrate impacts using an accurate, reliable and comprehensive method;
- 3. consider effects at all time scales; and
- 4. be developed with stakeholders (i.e. the affected members of the community)

and those who will use them. They should be able to (Beli and Morse (2001), Guy and Kibert (1998)): provide valid, reliable, and relevant results; Α. only require measurable information^{‡‡‡‡}; Β. С. be responsive and adaptive to changes in the indicated system and the

- community; and
- D. be understandable to all.

This desire for understandability should not detract from complete guidance on how to measure the indicators, as has often occurred (Bell and Morse (2001), Veleva and Ellenbecker (2001)).

Most environmental indicators had their birth in the 1980's (Briassoulis (2001)), when many environmental methodologies and sustainability were initially developed. Social impacts are 'emotive and normative issues' and cannot easily be considered in simple indicators or then objectively compared to other impacts (Bell and Morse (2001)). Thus, no standard way for determining social performance exists, and social indicators are less well defined (Azapagic and Perdan (2000)). Thus, indicators tend to (Azapagic and Perdan (2000), Briassoulis (2001), Guy and Kibert (1998), Veleva and Ellenbecker (2001)):

- measure environmental and economic features well; i.
- ii. consider social impacts poorly (if at all);
- iii. the desired impact;
- iv.
- V.
- be still under development; vi.
- vii.
- viii. lack detailed guidance on application; and
- ix.

18

be aggregates of individual indicators rather than all embracing measures of

be indicators of local impacts rather than national or global;

use only quantitative measures, excluding immeasurable impacts;

be chosen from a list of indicators for each individual application;

be used by internal management, rather than for external reporting.

Chapter 2: Sustainability

Some have suggested that the use of these existing indicators has little benefit, as they contain little human value content, while measuring something that is 'highly subjective and ridden with human values and desires' (Bell and Morse (2001), Briassoulis (2001)).

There are a great number of sets of indicators of sustainability (Bell and Morse (2001), Board on Sustainable Development (1999)). Some examples of bodies producing indicator sets include: the United Nations (Division for Sustainable Development (2003)); national (i.e. DETR (U.K.) (2001)) and local governments (see International Sustainability Indicators Network (2003) for examples); professional bodies and industry organisations (i.e. IChemE (2002a)); and academic-industry collaborations (i.e. Global Reporting Initiative (2002)). The United Nations indicator set (shown in Table 2.3.1 to Table 2.3.4), based on the four-sphere model, is an example of an indicator set suited to the assessment of a nation's sustainability. In OECD countries, the data required for these indicators was available (Ayres (1996)). Some countries, notably the U.K., publish a number (120) of sustainability indicators (DETR (U.K.) (2001)).

Indicators must be relevant to the policy options under consideration for them to be of value to the decision maker (Cardwell et al. (1995), International Institute for Sustainable Development (2000), Verbuggen and Kuik (1991)). Thus, while there are many indicators available, not all will be relevant in each situation. The chosen indicators cannot be too few, for this may exclude important impacts. Therefore, the choice of an appropriate set of indicators is an important decision, which will dictate the accuracy and defensibility of the sustainability assessment. A scientific basis for the selection of the appropriate indicators for each situation is lacking (Board on Sustainable Development (1999), International Institute for Sustainable Development (2000)). One attempt^{\$\$\$\$} at providing such a basis suggested (International Institute for Sustainable Development (2000)):

Sustainability of Australia's Electricity Generation

of sustainability (Division for Sustainable Development (2003)).

Theme	Sub-theme	Indicator
Equity	Poverty	Percent of Population Living below Poverty Line
		Gini Index of Income Inequality
		Unemployment Rate
	Gender Equality	Ratio of Average Female Wage to Male Wage
Health	Nutritional Status	Nutritional Status of Children
	Mortality	Mortality Rate Under 5 Years Old
		Life Expectancy at Birth
	Sanitation	Percent of Population with Adequate Sewage
		Disposal Facilities
	Drinking Water	Population with Access to Safe Drinking Water
	Healthcare	Percent of Population with Access to Primary
	Delivery	Health Care Facilities
		Immunization Against Infectious Childhood
		Diseases
		Contraceptive Prevalence Rate
Education	Education Level	Children Reaching Grade 5 of Primary Education
		Adult Secondary Education Achievement Level
	Literacy	Adult Literacy Rate
Housing	Living Conditions	Floor Area per Person
Security	Crime	Number of Recorded Crimes per 100,000
		Population
Population	Population Change	Population Growth Rate
		Population of Urban Formal and Informal
		Settlements

- b. Simplicity;
- verifiably and reproducibly);

- information);
- g. Sensitivity (can it detect small changes); and

2.3.4 INDICATORS FOR INDUSTRY

Companies, influenced by public concern, regulatory pressures and other factors, have

Table 2.3.1: The United Nations proposed national indicators for the social sphere

a. Policy relevance (can policies be formulated from it);

c. Validity (does it measure a sustainability impact accurately, defensibly,

d. Time-series data (can the variance with time be followed);

Data availability (can the data be affordably obtained);

f. Scope (does it aggregate several more narrow indicators without losing

h. Reliability (can the same result be obtained by other workers).

^{****} This may include non-technical measures, such as survey results.

^{\$\$\$\$} The authors acknowledge that this set is incomplete.

Table 2.3.2: The United Nations proposed national indicators for the environmental sphere of sustainability (Division for Sustainable Development (2003)).

Theme	Sub-theme	Indicator	
Atmosphere	Climate Change	Emissions of greenhouse gases	
	Ozone Layer	Emissions of ozone depleting substances	
	Depletion		
	Air Quality	Ambient concentration of air pollutants in urban	
l		areas	
Land	Agriculture	Arable and permanent crop land area	
		Use of fertilizers	
		Use of agricultural pesticides	
	Forests	Forest area as a percent of land area	
Ì		Wood harvesting intensity	
	Desertification	Land affected by desertification	
	Urbanization	Area of urban formal and informal settlements	
Oceans,	Coastal Zone	Algae concentration in coastal waters	
Seas and		Percent of total population living in coastal areas	
Coasts	Fisheries	Annual catch by major species	
Fresh Water	Water Quantity	Annual withdrawal of ground and surface water as a	
		percent of total available water	
	Water Quality	BOD in water bodies	
		Concentration of faecal Coliform in freshwater	
Biodiversity	Ecosystem	Area of selected key ecosystems	
	L	Protected area as a percentage of total area	
L	Species	Abundance of selected key species	

become interested in sustainability assessment. Companies desired broadly accepted measures of sustainability to enable both audits of sustainability status and alignment with the sustainability goals of its partners and regulators (Global Reporting Initiative (2002)). Institutions and regulatory authorities, driven by the expectations of international bodies and their own population, also desire companies to measure sustainability (Global Reporting Initiative (2002), Buhrs and Aplin (1999)). Moreover, financial markets are beginning to desire such information, due to demands for socially and ethically responsible investments (Global Reporting Initiative (2002)). Thus, many companies have developed their own indicators.

Many of the impacts on sustainability, measured by indicators for nations, are inconsequential to the operation of a single company or industry. For example (from Table 2.3.1) crime rates, healthcare delivery, and many others. Thus, different indicator sets are necessary. Figure 2.3.1 shows a suggested scheme for organisations to develop their own indicator sets. In this scheme, companies begin with simple, easily Sustainability of Australia's Electricity Generation

sphere of sustainability (Division for Sustainable Development (2003)).

Theme	Sub-theme	Indicator	
Economic	Economic	GDP per capita	
Structure	Performance	Ratio of capital investment to GDP	
Ę 1	Trade	Balance of trade in goods and services	
	Financial Status	Debt to GNP Ratio	
		Total official development assistance given or	
		received as a percent of GNP	
Consumption	Material	Intensity of material use (mass or volume per	
and	Consumption	GDP)	
Production	Energy Use	Annual energy consumption per capita Share of consumption of renewable energy	
Patterns			
		resources	
ļ		Intensity of energy use	
	Waste	Generation of industrial and municipal solid waste	
	Generation and	Generation of hazardous waste	
	Management	Management of radioactive waste	
		Waste recycling and reuse	
	Transportation	Distance travelled per capita by mode of transport	

Table 2.3.4: The United Nations proposed national sustainability performance indicators for the institutional sphere of sustainability (Division for Sustainable Development (2003)).

Theme	Sub-theme	Indicator
Institutional	Strategic	National sustainable development strategy
Framework	Implementation of SD	
	International	Implementation of ratified global agreements
		Number of internet subscribers per 1000
		inhabitants
1	Communication	Main telephone lines per 1000 inhabitants
	Infrastructure	
Institutional	Science and	Expenditure on research and development as
Capacity	Technology	a percent of GDP
	Disaster Preparedness	Economic and human loss due to natural
	and Response	disasters

implemented compliance indicators, before moving on to more complex indicators and life-cycle impacts (Veleva and Ellenbecker (2001)). Industrial indicators have additional requirements to those of general indicators of sustainability. Industrial systems have many operating parameters that may be changed. Indicators must be measurable, at specific, sufficiently frequent time intervals, to show

22

Table 2.3.3: The United Nations proposed national indicators for the economic

Chapter 2:Sustainability



Figure 2.3.1: Framework for organisations to develop sustainability indicators (Veleva and Ellenbecker (2001)).

the variation in indicator values with changes in operating parameters (Briassoulis (2001)). A company or industry may have operations in many locations. Thus, the indicators must also have well defined, and explained, boundaries, indicating whether the measurement is specific to a particular processing area or facility, or to the entire product life cycle (Veleva and Ellenbecker (2001)).

It is unlikely that any one set of indicators could be applicable to all organisations, as each production facility would have its own key impacts (Veleva and Ellenbecker (2001)). Thus, development of an all-encompassing, sustainability indicator list is unlikely. Some have proposed a default set that practitioners either, choose relevant indicators from within it (Azapagic and Perdan (2000)), or add missing indicators to a basic set (Veleva and Ellenbecker (2001)). This may increase the likelihood of the standardisation of an indicator set and method for sustainability assessment (Azapagic and Perdan (2000)). Revision of the standardised group of indicators could occur periodically as the concept, scientific knowledge, technology (Briassoulis (2001)) and definition of sustainability evolve. A standard set of indicators has yet to be accepted (Azapagic and Perdan (2000), Board on Sustainable Development (1999)).

One attempt at a default set of indicators, by the Global Reporting Initiative (Global Reporting Initiative (2002)), developed a detailed scheme with the interaction of a number of academics and some large multinational corporations (e.g. BASF, Bayer, Gaz de France, Rio Tinto, Royal Dutch/Shell, Texaco, TotalFinaElf, Transalta, and TXU Europe). Another attempt, by Azapagic and Perdan (2000), was consequently

Sustainability of Australia's Electricity Generation

adopted (with some modification) by the Institution of Chemical Engineers (UK) (see IChemE (2002a)). Table 2.3.5 and Table 2.3.6 show these two indicators sets.

Despite the development of a great number of indicators and frameworks, there are only a very few published examples of their use in decision-making (by those for whom they are developed) (Bell and Morse (2001)). This may be due to their development by external experts, or executive management (Bell and Morse (2001)). This may orientate them towards symbolic use (publicity), rather than the envisioned practical application (Buhrs and Aplin (1999)). Moreover, evidence has revealed, that many companies fail to locate and reduce the major environmental and social impacts, because standardised methods and measures for these impacts are lacking (Veleva and Ellenbecker (2001))^{*****}. Thus, the borrowing of an existing method for use in this asses nent of the sustainability of electricity generation systems, for instance, is not feasible.

2.3.5 CRITERIA FOR SUSTAINABILITY AND SUSTAINABILITY INDICATORS

Each of the spheres of the sustainability concept models has their own goals, which are 'human-ascribed' (Barbier (1987)), and thus, are value, rather than scientifically based. Table 2.3.7 provides examples of the system goals espoused by Barbier (1987) for his definition of sustainability (see Section 2.2.2). The Brundtland report contains similar goals (WCED (1987)). The goals ascribed, will depend on the relative importance that individuals place on each of the three spheres (Overcash (2001)). It is likely then that these goals will vary as perceptions of what is necessary for sustainability change (Board on Sustainable Development (1999)).

Criteria are goals (targets) for achievement, which are less than the ultimate goal, but represent a short-term goal or requirement. Criteria will usually refer to the reduction or increase of an indicator magnitude over a fixed time. For example, the Kyoto protocol, which required signatories to meet 'greenhouse gas' emissions targets by 2010, is a sustainability criterion for the 'Global Warming' indicator. Criteria may be set internally, by regulators, or social groups (Briassoulis (2001)). Constraints on the setting of appropriate criteria, especially for natural resources, are (Briassoulis (2001),

50 Corporate Sustainability Top newsFthird.cfm?NewsID=13397).

24

As reported in a review of company sustainability reports (SustainAbility (2000), Team Spotlights Reports, GreenBiz, <u>http://www.greenbiz.com/news/</u>

Table 2.3.6: Proposed default set of indicators for assessing the sustainability of

organisations (Azapagic and Perdan

Environmental	Economic
Environmental Impacts	Financial Indicators
Resource use	Value added
Global Warming	Contribution to GDP
Ozone Depletion	Expenditure on environmental
Acidification	protection
Eutrophication	Environmental liabilities
Photochemical Smog	Ethical investments
Human toxicity	Human-capital indicators
Eco-toxicity	Employment contribution
Solid Waste	Staff turnover
Environmental efficiency	Expenditure on health and
Material and energy intensity	safety
Material recyclability	Investment in staff development
Product durability	
Service intensity	
Voluntary actions	
Environmental management system	ms
Environmental improvements abo	ove
the compliance levels	
Assessment of suppliers	

the three-sphere sustainability model) (Barbier (1987)).

Environmental	Economic	Social
Genetic diversity	Satisfying basic needs	Cultural diversity
Resilience	Equity-enhancing	Institutional sustainability
Biological productivity	Increasing useful goods and	Social justice
	services	Participation

- 5. The continual redefinition of sustainability goals;

7. The interdependence of individual targets on the attaining of others.

Analysis of these constraints was uncommon in the development of existing international conventions and agreements for criteria (Board on Sustainable Development (1999)).

Some have suggested that the specification of criteria is necessary for progress to occur, especially as institutions (Ekins and Simon (2001)) and organisations preferably work towards targets. However, each practitioner may suggest different possible

Table 2.3.5: Proposed default set of indicators for assessing the sustainability of	Ì
organisations (Global Reporting Initiative (2002)).	

Sphere	Category A	Aspect
Economic	Direct Economic	Customers
		Suppliers
		Employees
	Impacts	Providers of capital
		Public sector
		Materials
		Energy
		Water
		Biodiversity
Environmental	Environmental	Emissions, effluents, and waste
Environmental		Suppliers
		Products and services
		Compliance
		Transport
		Overall
		Employment
	Labour Drastian	Labour/management relations
	Labour Practices	Health and safety
	and Decent work	Training and education
		Diversity and opportunity
		Strategy and management
		Non-discrimination
		Freedom of association and collective bargaining
	Uumon Dighte	Child labour
	Human Kights	Forced and compulsory labour
Social		Disciplinary practices
		Security practices
		Indigenous rights
	Society	Community
		Bribery and corruption
		Political contributions
		Competition and pricing
		Customer health and safety
	Product Responsibility	Products and services
		Advertising
		Respect for privacy

(George (2001)):

- 1. Scientific uncertainty;
- 2. Limited information;
- 3. Unknown effects from interactions between the system and its surroundings;

Э	ſ	١	
	÷		

n (2000)).				
nic al Indicators	Social Ethics Indicators			
ded	Preservation of cultural values			
tion to GDP	-stakeholder inclusion			
ure on environmental n	-involvement in community projects			
nental liabilities	International standards of			
ivestments	conduct			

regimes

-business dealings

-collaboration with corrupt

-child labour

Intergenerational equity Welfare indicators Income distribution Work satisfaction

Satisfaction of social needs

-fair prices

Table 2.3.7: Proposed goals of sustainability (goals presented for each sphere of

4. Value choices in weighting of impacts and qualitative indicators;

6. The clear specification of spatial boundaries and temporal horizons; and

Chapter 2:Sustainability

criteria for the indicators, as each may consider different impacts on sustainability to be most important (Ekins and Simon (2001)). For example, if one practitioner considers human health most important, then emissions of substances which have worst human health effects is paramount. Thus, this practitioner's criterion might be the reduction or elimination of these emissions. If another practitioner believes a different impact to be most important, then their chosen criteria will reflect this belief, and is unlikely to be the same as the criterion based on human health. Thus, the setting of criteria leads not to greater sustainability, but argument over the validity of the target.

Other alternative methods of judging improvement in sustainability include the multi-criteria (or multi-objective) technique, which optimises all indicators simultaneously to select the lowest impact solution (IAEA (2000)). However, the method requires the setting of criteria. Alternatively, a plot of two or three key impacts may show the optimum choice (see Figure 2.3.2). This method enables the selection of a reasonable compromise, even if there is no 'best' option (IAEA (2000)). Golonka and Brennan (1997) discuss some trade-off and other alternative methods for use with environmental and economic impact aggregates.

Without criteria, the multiplicity of views on sustainability may be more easily integrated, and approaches and tools may then adapt to particular views as necessary (Briassoulis (2001), McQuaid (2000)). Thus, discussion has shifted from determining criteria, to considering progress towards the final goal: sustainability (Briassoulis (2001), George (2001)). For example, a modification that leads to a general improvement in sustainability indicators enhances sustainability.





Sustainability of Australia's Electricity Generation

2.3.6 BOUNDARIES FOR MEASUREMENT OF SUSTAINABILITY

Well-defined spatial and temporal boundaries are additional requirements for industrial indicators over regional indicators (see Section 2.3.3). The spatial boundary of a sustainability assessment should ideally be a naturally bounded area, such as a watershed, or an economically bounded area, such as a nation (Briassoulis (2001)), depending on which paradigm is more central to the definition used. Regional areas are suggested as the most appropriate level as they are most likely to have their environmental, economic and social processes integrated (Briassoulis (2001)). By increasing the size of the boundary, it becomes more likely that all impacts on sustainability will be included, but the complexity increases and the study may become unworkable (Briassoulis (2001)). Conversely, a smaller region may have many influences external to the region, and hence out of direct control (Briassoulis (2001)). Indicator sets using a regional boundary are most effective if developed at a local level (i.e. 'Local Agenda 21'), where a more comprehensive coverage of viewpoints is possible (UNCED (1992)). Thus, the resulting indicator would be specific to the local community and have limited applicability on larger scales.

In an organisation, likely spatial dimensions for a study include single facilities, groups of facilities in a similar region or producing similar products (a sector), or a company as a whole. Such a study would most likely following existing business divisional lines. As with indicator sets using regional boundaries, the indicator sets developed for an organisational boundary are likely to be specific to the organisation, and thus have limited applicability on larger scales.

An alternative to these boundary types is the 'life-cycle' perspective. The 'life cycle' perspective boundary follows the life of a product, from raw materials in nature, through successive processing stages, use, and ultimately to disposal. By choosing the life cycle perspective, comparisons both within sectors and between sectors are possible, and the shifting of poor sustainability operations to other locations, beyond the considered administrative boundary^{ttttt}, can be minimised (Ayres (1996)). This 'life

titt For example, consider a product, which has many processing stages, two of which (A and B) occur at a single location. Stage B is 'unsustainable' in its operation, in comparison to stage A. If the boundary is around the fixed site, then moving stage B to another location (outside the boundary) will improve the 'level of sustainability'. However, for a 'life cycle' boundary, the moving of stage B would not enhance

the 'level of sustainability'.

Chapter 2: Sustainability

cycle' perspective is of increasing interest in environmental analysis, and sustainability (Ayres (1996), Azapagic and Perdan (2000), Board on Sustainable Development (1999), Cowell et al. (1999)). The United Nations suggested it was essential for the sound management of chemicals and hazardous wastes (United Nations (2002)). Others have made similar claims for design and engineering (Heaney (1995)^{‡‡‡‡‡}), and sustainability assessment (Azapagic and Perdan (2000)).

2.3.7 SUMMARY OF MEASURING SUSTAINABILITY

Each of the sustainability tools discussed conceivably is applicable to assessments of electricity generation systems. The methods that utilise sets of indicators have important advantages over the alternatives, including the flexibility to be relevant to a wide range of systems and societies and widespread acceptability. Highlighted are some indicator sets for use in industrial situations, but their application is not simple. Indicators of sustainability must be relevant to the needs of the assessment. Thus, to apply the existing indicator sets to electricity generation systems, modifications will be necessary to select the appropriate indicators of sustainability.

2.4 ELECTRICITY GENERATION AND SUSTAINABILITY

2.4.1 OVERVIEW OF ELECTRICITY GENERATION AND SUSTAINABILITY

An important observation of Section 2.3 was that knowledge of the sustainability impacts important to the study subject is necessary to measure sustainability using indicators. Many studies into economic, environmental, or social performance of electricity generation and supply have covered some impacts of sustainability. Each considers different systems and ranges of impacts. Some, for example, include only the economic impacts of an electricity generation plant. Others apply the 'life-cycle' perspective, but include only specific environmental effects, such as climate change. A

Sustainability of Australia's Electricity Generation

review of these studies provides a perspective of the most important impacts to electricity generation. Sustainability indicators to represent these impacts can then be selected from Section 2.3. Therefore, this section establishes which sustainability impacts are of most importance for the study of electricity generation using fossil fuels. Section 2.4.2 discusses the necessity for a sustainability assessment of electricity generation from fossil fuels. Section 2.4.3 contains a review of studies on electricity generation. Section 2.4.4 reports the most important sustainability impacts and the sustainability indicators, which can represent them. It also examines the reasoning for

the selection and rejection of suitable sustainability indicators.

2.4.2 ELECTRICITY AND SUSTAINABILITY

Chapter 1 expressed a number of important links between electricity generation and sustainability. This section expands on those links, and provides a number of views on strategies to limit the effects of electricity generation on sustainability.

Planning for electricity generation and provision has been ad hoc, to meet the shortterm goals of institutions and organisations (WCED (1987)). Yet, energy development is too important for survival to continue in this manner (WCED (1987)). The primary objective of traditional electricity system development was to produce the required electricity at the lowest cost (IAEA (2000)). Immediate needs were paramount, including customer service, risk, financial and technological flexibility, and competition (IAEA (2000)). These policies use energy market prices or forecasts of energy supply limitations for their decision-making tools (Anderson (2001), IAEA (2000)). Variances in the availability and price of fuels often lead to profound changes in energy source usage (Hammond (2000)). For example, coal fuelled the economic recovery of Europe in the 1950's, but its high price soon led to its replacement with cheaper oil fuels (Hammond (2000)). Thus, primarily the bases for choices between options were economic, such as highest net present value or ranking from a cost/benefit analysis (IAEA (2000), Matson and Carasso (1999)). These methodologies would thus exclude any costs (and benefits) which cannot be expressed in monetary terms, and will discount those, which can be so expressed, but occur at times far removed from the initial investment (Matson and Carasso (1999)). Environmental costs did enter into earlier energy strategy decisions, but only in the costs of technology required to meet pollution standards (IAEA (2000), Matson and Carasso (1999)). Detailed environmental

30

Chapter 2:Sustainability

^{####} From Hatch, H. J. (1994), Accepting the Challenge of Sustainable Development, in American Association of Engineering Societies (AAES), The Role of Engineering in Sustainable Development, AAES, Washington D.C. (U.S.A.).

Chapter 2:Sustainability

assessment, as a separate Environmental Impact Assessment (or Statement), aimed to show compliance with regulatory limitations (IAEA (2000)). Sustainable electricity production criteria should include both the traditional requirements of sufficient and reliable (i.e. diverse sources, and secure supply chain) supply, and protection of human health and the environment (IAEA (2000), Matson and Carasso (1999)). Thus, the basis for these planning methods is a system of values 'that are increasingly in conflict with a sustainable planetary future' (Matson and Carasso (1999)).

It has been concluded that no mix of current electricity generation can be sustainable (WCED (1987)). Further, there was no one perfect method or energy source for electricity generation (IChemE (2002b). All energy sources have their own, unique economic, environmental and social cost and benefits (WCED (1987)), or sustainability. Some examples are: resource scarcity; smog generated in cities from road traffic; acid rain; and climate change from greenhouse gas emissions (Hammond (2000)). Fossil fuel systems are ultimately unsustainable because the reserves of fossil fuels are exhaustible (Graedel (2002), International Energy Agency (2001), and many others).

Only renewable energy technologies appear to be sustainable (Hammond (2000), Matson and Carasso (1999), WCED (1987)), however they too cause impacts \$\$\$\$\$ (OECD (1985)). Moreover, some predictions of the extent of renewable energy technologies development claim minimal use, to 2020 (IAEA (2000), IEA (2001)), and even to 2050 (Hammond (2000)). Other predictions are less pessimistic (Board on Sustainable Development (1999), Graedel (2002)), one stating that transition to sustainability could occur by 2050, while the other claimed that up to 60 % of energy could be produced sustainably by 2050, and 100 % by 2100 (Graedel (2002)). The criterion for sustainability in these predictions was low 'greenhouse gas' emissions, and thus it includes (fissile) nuclear energy sources (Graedel (2002)). However, nuclear energy sources consume resources, and that resource was similarly restrictive (20-50 years), not allowing nuclear to substantially increase its share of world energy production without major advances in the efficiency of energy conversion (Nakž enovč et al. (1998))******. Nuclear technology is unlikely to be practical in Australia, despite the abundance of fuel (uranium), since public opinion in Australia is firmly against its use.

Sustainability of Australia's Electricity Generation

Optimistic predictions assume that the public support indicated for progress ensures political, social, and economic support for the utilisation of technological advances (Board on Sustainable Development (1999)). By contrast, some predictions are pessimistic because of the extant economic and institutional constraints on renewable technology growth (WCED (1987)). The economic constraints are due mainly to the higher capital costs of renewable energy schemes for the same output of electricity. Under the prevailing methods for determining electricity costs, capital expenditure has greater weight than the on-going fuel costs associated with non-renewable technologies (Anderson (2001), Matson and Carasso (1999)). Institutional constraints are those of government control of energy systems, where piecemeal decisions by the government of the day lead to large centralised conventional (non-renewable) energy schemes (Anderson (2001)). Yet, these same institutions must introduce measures to support the implementation of renewable energy technologies if they are to become prevalent (IEA (2001)).

Pessimistic predictions also point out that some renewable technologies, like wind and wave power, produce electricity at unpredictable levels (IChemE (2002b)). Fossil fuel technology is necessary to ensure stable electricity provision, when renewables supply greater than 10 % of the grid power (IChemE (2002b)). IChemE (2002b) claim that this necessity would reduce the possible environmental impact benefit from the use of renewable technologies rather than fossil fuels by 50 %. Further nore, when such technologies produce no power at all (i.e. on calm days), they require back-up power generators, most likely fossil fuel technologies, which add to costs and environmental effects (IChemE (2002b)). Renewable energy systems will become more important with the development of large-scale electricity storage enabling the smoothing of high and low productivity conditions (Royal Dutch/Shell (2001)).

Cogeneration systems can produce electricity with much lower impacts than standard fossil fuel systems. However, they can find the challenge of competing on the electricity and heat markets difficult (Matthes and Timpe (2000)). The high cost of installation of district heating networks precludes the use of cogeneration over wide areas (IAEA (2000)). Therefore, cogeneration systems have greatest potential for steam

years (IAEA (2000), Nakž enovž et al. (1998)).

see For example, wind generation has impacts such as noise, visual, material consumption in construction and maintenance, and bird destruction. Solar generation has impacts in land-use, and material consumption in construction and maintenance.

The successful development of fast breeder reactor technology could provide energy for up to 8000

Chapter 2:Sustainability

supply to industries, across plant boundaries. Yet, only a minority of present cogeneration systems share steam between multiple industries^{tttttt}, indicating that such arrangements are also difficult.

Consequently, renewable, cogeneration and nuclear technologies cannot provide a significant proportion of Australia's electricity production. Therefore, Electricity development strategies must increase the sustainability of the current system, while still using conventional fossil fuels. Advanced technology coal fuelled systems were recognised as an important technology for such schemes, due to the expected ability of such schemes to reduce environmental impacts, and expected increases in the cost of natural gas (IChemE (2002b), United Nations (2002)). Thus, it is necessary to establish the sustainability impacts of both conventional and advanced electricity generation technologies for fossil fuels.

2.4.3 STUDIES OF ELECTRICITY GENERATION

2.4.3.1 OVERVIEW OF STUDIES OF ELECTRICITY GENERATION

This review of studies of electricity generation includes sustainability (Section 2.4.3.2), environment (Section 2.4.3.3), economic (Section 2.4.3.4), and social (Section 2.4.3.5) assessments.

2.4.3.2 SUSTAINABILITY ASSESSMENTS

The review uncovered only two sustainability assessments of electricity generation systems. Many described their studies as sustainability assessments, but included only environmental, or economic and environmental, impacts. The most significant was the Databases and Methodologies for Comparative Assessment of Different Energy Sources (or DECADES) project. Initiation of the DECADES project, in 1992, was by the collaboration of (Rogner and Khan (1998));

- International Atomic Energy Agency (IAEA);
- European Commission (EC);

- Institute for Applied Systems Analysis (IIASA);
- World Bank (IBRD);
- Development (OECD/NEA);
- Organisation of Petroleum Exporting Countries (OPEC);

 World Meteorological Organisation (WMO). The DECADES project provides background data for the development of sustainable electricity systems by governments and regulators. The DECADES project (IAEA (2000)) utilised a full energy chain (FENCH) technique to determine the sustainability impacts of a number of electricity generation technologies, including black coal and natural gas systems. The FENCH technique ignores the sustainability impacts due to construction and decommissioning of the system, and all materials consumed by the system, apart from the fuel consumed in the electricity generation activity, and is thus a subset of the 'life-cycle' perspective (IAEA (2000)). The methodology was designed for application in electricity generation sector design, and utilises the concept of valuation of environmental and social impacts to generate impact-cost of production aggregates for comparisons of the strategy options. Valuing environmental and social impacts may enable the comparison of impacts on a consistent basis, and may enhance usage and acceptability of environmental and social impacts into evaluations (IAEA (2000)). However (IAEA (2000)):

- a. amenity is not);
- b. countries);
- Ç. are discounted); and

the results for some situations provided non-optimal solutions. d. The validity of this valuation procedure was the subject of much research in both the U.S.A. and Europe (IAEA (2000)). Figure 2.4.1 shows an example of results obtained from this study (before valuation). The study has also produced a database for

United Nations Economic and Social Council for Asia and the Pacific (ESCAP);

• Nuclear Energy Agency of the Organisation for Economic Cooperation and

United Nations Industrial Development Organisation (UNIDO); and

impacts not directly linked to a market proved difficult to cost (i.e. building damage and loss of agricultural production are linked to a market, but visual

costs for the same impact vary with location (i.e. urban-rural, or different

impacts occurring in the future were under-represented (as future impact-costs

35

¹¹¹¹¹¹ For example, the Osborne Cogeneration plant in South Australia shares steam between its electricity generation needs and a soda ash plant (see www.penrice.com.au).

Chapter 2:Sustainability

sustainability impact data for various electricity generation technologies and fuels, the Reference Technology Database. The sustainability impacts included: climate change, acidification, photochemical smog, particulates, radioactive releases, water pollution (thermal and chemical), solid wastes, resource depletion, visual amenity, employment, and community relations (IAEA (2000)).

Another, much less comprehensive study (Leca et al. (2002)), produced for an eastern-European country (i.e. a former planned economy) a comparison of the relative sustainability impacts of investment options for upgrading their electricity generation capacity. The sustainability indicators used were: net present cost, GDP share, investment flexibility, fuel diversity (economic); CO₂, NO_x, and SO₂ emissions (environmental); employment to investment ratio, and technology acceptance ratio (social). The technology acceptance ratio is the sum of ratios of acceptance for each type of technology. Ratios are the proportion of population that do not oppose the use of each technology. The study used the generation plant alone, rather than the 'lifecycle' perspective system. The sustainability of each system was determined using a multi-criteria optimisation procedure (see Section 2.3.4), which considered a set number of given scenarios. The practitioners set the criteria using their values. For example,



Figure 2.4.1: Comparative full-fuel cycle, greenhouse gas emissions of selected electricity generation systems (Rogner and Khan (1998)). The units g per kWh = kg per MWh.

Sustainability of Australia's Electricity Generation

the net present cost (18 %) had highest importance, while technology acceptance (3 %) had lowest importance. It was recognised that in other countries a lack of technology acceptance may negate any other benefits. They developed four scenarios of future requirements for electricity generation capacity, based on a perceived rate of economic growth: fast, moderate, slow, and pessimistic. The scenarios were set for 2025, 25 years after the completion of the analysis, and considered a number of projects under construction, some planned for development, and some proposed for the future. The favoured scenario included the upgrade of current coal plants, and the addition of gasfired combined cycle units.

2.4.3.3 Environmental Assessments

There are great numbers of environmental assessments of electricity generation systems. Consequently, to keep this review manageable, only a few will be included, with similar studies provided as references.

The first major study to cover the complete fuel cycle was the COMPASS project, by the Organisation for Economic Cooperation and Development (OECD) (OECD (1985)). This study identified the environmental impacts due to electricity generation from fossil fuels as: land and water use, air emissions, long range transport and deposition of air pollutants, local climatic and visual effects from cooling towers, solid waste and ash disposal, and noise. It also noted that impacts occurred before and after generation, during the processes of mining or production, refining or processing, transportation to the generator, and transmission to the customer. Also included were other forms of electricity generation (nuclear, geothermal, biomass, solar, hydro and wind). The study did not attempt to quantify any impacts.

In response to the needs of governments to produce country 'greenhouse gas' emissions reports (e.g. AGO (1999), DoE (U.K.) (1998), EPA (U.S.A.) (2000)), some industries have produced their own such reports (e.g. AGA (1997), APPEA (1999), Lowe (1998), SECV (1992)). The categories in these major inventories are either too broad (i.e. APPEA (1999) includes oil and gas), or too narrow (i.e. SECV (1992) includes only generation) to discuss the environmental impacts of electricity generation systems using a particular fuel, with the 'life-cycle' perspective.

The International Energy Agency (IEA) created the Greenhouse Gas R&D Programme (IEAGHG) in 1991 to identify and evaluate technologies for reducing

36

Chapter 2: Sustainability

Chapter 2:Sustainability

emissions of greenhouse gases arising from use of fossil fuels (IEAGHG (2000)). They examined the capture and storage of CO_2 and other atmospheric pollutants. Additionally, they produced a comparison of the greenhouse gas emissions of the generation step, without CO_2 capture, for a coal fired steam turbine (ST), natural gas combined cycle (CCGT), integrated coal gasification combined cycle (IGCC) and carbon dioxide recycle system (IEAGHG (2000)).

In the 1990s many 'greenhouse gas' studies, using the 'life-cycle' perspective, appeared. For example, the OECD and International Energy Agency (IEA) in 1992 held the 'Expert Workshop on Life-Cycle Analysis of Energy Systems' included studies for fossil and nuclear fuels, as well as photovoltaic systems (Baumann and Hill (1992), Eyre and Holland (1992), Yasukawa *et al.* (1992)). Other examples include:

- a. Central Research Institute of the Electric Power Industry (CRIEPI) in Japan (CRIEPI (1994));
- b. Saskatchewan Energy Conservation and Development Authority (SECDA) in Canada (Macdonald et al. (1996));
- c. Paul Scherrer Institut (PSI) in Switzerland (Dones and Gantner (1996)); and
- d. Alberta Department of Energy (Macdonald *et al.* (1996)), included separation of upstream and combustion emissions (see Figure 2.4.2).
- e. Oil Industry Group in Japan (Ogawa and Yoon (1998)).

In 1995, a study by the Energy Information Agency (EIA) in the U.S.A. prepared a report on the monetary valuation, by the various state authorities, of environmental impacts incurred by steam-electric generators (Energy Information Agency (U.S.A.) (1995)). This study identified, but did not quantify or aggregate the environmental impacts. Similar studies using valuation include Office of Technology Assessment (1994) and Watkiss and Forster (2000).

In 1999, Vattenfall produced a comparative study with 'life cycle' perspective boundaries (Vattenfall (1999)). It contains comparisons based on emissions of CO_2 , SO_2 and NO_x , consumption of copper, and land use. Generation technologies considered were hydro, nuclear, biofuel^{‡‡‡‡‡‡‡}, combined heat and power (CHP), wind,

Sustainability of Australia's Electricity Generation

Technology

Coal-ST, pre-1970 Coal-ST, post-1970 Coal IGCC Coal, Supercritical Steam ST Gas and Coal co-firing ST Cogeneration

Natural Gas OCGT Natural Gas CCGT Natural Gas ST Hydro Wind Turbine Diesel

Figure 2.4.2: Comparative full-fuel cycle, greenhouse gas emissions of selected electricity generation systems (Macdonald *et al.* (1996)). ST = steam turbine, IGCC = integrated combined cycle gas turbine, OCGT = open cycle gas turbine and CCGT = combined cycle gas turbine (see Glossary for explanations).

solar, oil condensing, natural gas combined-cycle (CCGT), fuel cells, coal fired steam turbine (ST), and open cycle gas turbines (GT). This study determined that the open cycle gas turbines (GT) (used for backup generation in the event of a shortfall in supply) produced the greatest emissions of CO_2 and NO_x per kWh of electricity supplied to customers, while the coal fired ST produced the greatest SO₂ emissions. The study also found that the open cycle gas turbines (GT) and solar power consume the most copper, and the biofuelled CHP and wind power require the greatest land area. They concluded that generation from fossil fuels was far more damaging, based on the considered variables, than their generation mix of hydro and nuclear.

National Renewable Energy Laboratory (NREL) has produced several studies of electricity generation systems with 'life cycle' boundaries (e.g. Spath *et al.* (1999)). This example compared three high rank black coal generation options:

i. the current average (ST particulate filter);



the current average (ST with limestone desulphurisation and baghouse

^{#####} Biofuel refers to unprocessed organic fuels, such as wood, straw and bagasse.

- a plant which meets the U.S.A.'s New Source Performance Standards (NSPS) ii. (same design but more efficient desulphurisation and filters, and low NOx burners); and
- the Low Emission Boiler System (LEBS) (ST with low NO_x burners, catalyst iii. SO_2 and NO_s removal, sulphur (or sulphuric acid) generation, and baghouse filter).

Data collection is extensive, but the Impact Assessment stage is limited to a qualitative discussion, with only greenhouse gas emissions treated quantitatively.

In Australia, studies have occurred of greenhouse gas emissions during the generation step (for example: ACA (1999) and AGO (1999)) and with extended boundaries (for example: BHP Research (1999), AGA (2000), Evans (1995) and Woodside Energy Ltd. (1998a)). The extended boundary studies all include greenhouse gas emissions and most include emissions of other atmospheric pollutants. The study by Evans, through the Cooperative Research Centre: New Technologies for Power Generation from Low-Rank Coal, considered only Victorian systems delivering electricity to Melbourne (Evans (1995)). CSIRO (Australia) completed a study for Woodside Energy Ltd. in 1996 (Woodside Energy Ltd. (1998a)). This study considered the greenhouse gas emissions from electricity generation in Japan of: black coal from the Hunter Valley (NSW), liquefied natural gas (LNG) from the North West Shelf (WA), and oil from the Middle East. The study found that coal produced the highest greenhouse gas emissions per MWh of electricity generated (865 kg CO_{2-e} per MWh), while LNG produced the least (493 kg CO_{2-e} per MWh). These figures are consistent with previous results (see Figure 2.4.1 and Figure 2.4.2). The analysis of coal includes methane emissions from the coal mine, but excludes emissions observed to occur from the decomposition and spontaneous combustion of spoil and washery rejects piles. The analysis of LNG uses operating data from Woodside's North West Shelf LNG plant. LNG transport is by dedicated ocean carrier, while coal transport is by rail to port and ship to Japan. The LNG generation plant is a CCGT, while the coal generator is a supercritical ST, with post combustion SO₂ and NO₃ removal. Approximately 91 % and 77% of the emissions were due to the power plant, for the black coal and LNG systems respectively. In 1998, CSIRO completed a similar study for natural gas and nuclear fuels (Woodside Energy Ltd. (1998a)). In this study, the natural gas fuelled open cycle gas turbine (OCGT) ould emit more greenhouse gases than a black coal system, but a CCGT will emit far less (c.f. Figure 2.4.1 and Figure 2.4.2). The nuclear systems

Sustainability of Australia's Electricity Generation

produced negligible quantities of greenhouse gases.

BHP Research and the Australian Coal Association Research Project (ACARP) completed studies on steel and electricity production with 'life-cycle' boundaries (BHP Research (1999) and BHP Minerals Technology (2001)). They included comparisons of electricity production from 10 different technologies, including black coal fired ST and clean coal technologies, and natural gas open and combined cycles. The indicators included were resource energy, greenhouse gas, NO_x, SO₂ and particulates emissions, and fresh water consumption. The results of this report are similar to those produced by other groups. The greenhouse gas emissions of the open cycle system studied are 20 % lower than that reported in other studies (Macdonald et al. (1996), Vattenfall (1999), Woodside Energy Ltd. (1998a)). The study ignores dust emissions from coal mining, and losses during electricity transmission. Moreover, the methane emission calculation for coal mining was lacking in detail.

In 2000 the Australian Gas Association (AGA) released a study of natural gas with 'life-cycle' boundaries (AGA (2000)), which also contained a comparative study of electricity production using natural gas, and black and brown coal. The comparison shows that for Australia, CCGT systems produce near 530 kg CO_{2-e} per MWh, natural gas GT 600 to 650 kg CO_{2-e} per MWh, black coal fired ST 910 kg CO_{2-e} per MWh, and brown coal fired ST 1246 kg CO_{2-e} per MWh. The study excludes electricity transmission, and uses an unrealistic assumption to estimate emissions from the mining and transport of both black and brown coal. Moreover, the basis for this study was Australian best practice efficiencies, rather than current practice, limiting the applicability of its results.

Much of the discussion on electricity generation sustainability involved 'greenhouse gas' emissions (IChemE (2002b), Van de Vate (1997)). However, included in at least two of the reviewed studies were indicators for climate change, acidification, eutrophication, photochemical smog formation, toxicity and particulates. Some studies also indicated that resource and water consumption, and solid waste generation were important impacts.

2.4.3.4 ECONOMIC ASSESSMENTS

This section examines some important studies, which are representative of, but more comprehensive than, other published studies. Internationally, the Nuclear Energy

Chapter 2: Sustainability

Agency (NEA) and International Energy Agency (IEA) produced comparisons of actual electricity generation capital and operating costs in OECD countries (i.e. western European nations, Japan and the USA), and some non-OECD countries (Nuclear Energy Agency (1998)). It considered electricity generation plant for nuclear, coal and natural gas fuels. It included conventional technologies and some advanced coal technologies. The International Energy Agency periodically updates these estimates and produces other specialised estimates (e.g. IEA (1993)).

In the USA, the Department of Energy produced comprehensive capital cost estimates of electricity generation plant (DOE (U.S.A.) (1999)). This report attempted to cost plant based on the costs of individual components. It, like the NEA study, included conventional and advanced electricity generation plants for coal fuels. For natural gas fuel, it limited its study to combined-cycle gas turbine plants, excluding both steam turbine and open-cycle gas turbine plants. It estimated both capital requirements and electricity production costs. The Department of Energy has also produced a number of other cost estimations for electricity generation (e.g. DOE (U.S.A.) (2000)).

In Australia, the Energy Research and Development Corporation produced a comprehensive study of electricity generation options it considered possible for use in the 1990's (Energy Research and Development Corporation (1992)). This study was the only independent study to include plants for Australian brown coal fuel, as well as black coal and natural gas plants. This study included electricity generation plants using both conventional and advanced technology. It estimated both capital requirements and electricity production costs. During this period, a number of other estimates were produced for Australian state governments, who were in charge of electricity network planning (i.e. Natural Resources and Environment Committee (1987)).

The economic indicators for each of these cost estimations were capital requirements and production costs (e.g. cost of electricity generated). This implies that new electricity systems must not only produce electricity at low long-term costs (production costs), but the capital requirements must be attainable for the proponent, and not place an undue burden upon them (IAEA (2000)).

None of the economic assessments encountered consider the 'life-cycle' perspective, and thus exclude the impacts of fuel production, fuel transport and electricity transmission.

Sustainability of Australia's Electricity Generation

2.4.3.5 SOCIAL ASSESSMENTS

Social assessments have been uncommon for electricity generation systems, although some social impacts are included in Environmental Impact Assessments (e.g. ECNSW (1980)).

The IAEA's sustainability information document reported that the social impacts important to electricity generation are local employment and resource depletion (IAEA) (2000)).

Social assessments of coal mining have highlighted that, along with local employment, health and safety is an important impact of coal mining (e.g. Minerals Council of Australia (2002)). Injuries and fatalities have important social and econoraic impact on communities. As coal mining is within the 'life-cycle' perspective boundary of electricity generation systems that consume coal, indicators of health and safety impacts are necessary.

None of the social assessments encountered consider the 'life-cycle' perspective, and thus exclude the impacts that occur in fuel production, fuel transport and electricity transmission.

2.4.4 INDICATORS FOR ELECTRICITY GENERATION

Table 2.4.1 shows the impacts determined to be important in previous sustainability, environmental, economic and social assessments from the review in Section 2.4.3. The sets of incustrial indicators provided earlier (see Table 2.3.5 and Table 2.3.6) include indicators for each of these impacts. The 'Glossary' contains descriptions of many of these impacts.

The indicators are not entirely separable into the categories given. Most indicators have flow on effects contributing to other categories (e.g. increased wealth generation leads to greater tax revenue, which are used to generate social benefits). The convention adopted categorises resource depletion as an environmental indicator, although resource depletion may lead to both social and environmental impacts. Many of the indicators in Table 2.3.5 and Table 2.3.6 have no corresponding impacts in Table 2.4.1. Existing systems require sustainability assessment to measure the performance of not only the sustainability of their operations, but also their business philosophy (i.e. the Human Rights and Society categories of indicators in Table 2.3.5

Table 2.4.1: Important indicators of sustainability, selected from a review of assessments of the impacts of electricity generation systems (naming conventions from Azapagic and Perdan (2000) and IChemE (2002b)).

Environmental Indicators	Economic Indicators,	Social Indicators
Resource Depletion	Wealth Generation	Employees
Climate Change	Capital Requirements	Health and Safety
Acidification		
Eutrophication		
Photochemical Smog		
Toxicity		
Solid Waste		
Particulates		
Water Consumption		

and the Voluntary actions, Human-capital, Ethics and Welfare categories of indicators in Table 2.3.6). For a proposed system, the choice of technology is unlikely to affect business philosophy. This may not be valid if the choice results in the use of materials sourced from companies with radically different business philosophies (for example, when deciding between purchasing from companies with very different worker conditions). Similarly, if the product qualities (i.e. endurance, strength, durability, etc.) do not vary with the option chosen, then its indicators of product sustainability will also not vary (i.e. the product responsibility category of indicators in Table 2.3.5 and Environmental efficiency category of indicators in Table 2.3.6). Electricity as a product does not vary with its source technology.

2.4.5 SUMMARY OF ELECTRICITY GENERATION AND SUSTAINABILITY

The indicators presented in Table 2.4.1 are sufficient to produce a sustainability assessment of possible electricity generation systems. Examined were previous assessments of sustainability, and individual impacts of sustainability, to select these indicators. Many of these indicators are core indicators in other industrial sets of sustainability indicators.

None of the previous assessments of social and economic impacts on sustainability uses 'life-cycle' perspective boundaries, and thus they underestimate sustainability impacts.

Sustainability of Australia's Electricity Generation

2.5 SUMMARY OF SUSTAINABILITY

Sustainability is a concept, rather than a defined term. Often depictions of this concept have three intersecting spheres, each sphere representing a different element of sustainability (environmental, economic and social). Measuring sustainability requires the use of indicators, as they ensure relevancy to a wide range of systems and societies system, reducing the chance of shifting sustainability impacts outside a fixed boundary.

and allow widespread acceptability. Measuring sustainability should use 'life-cycle' boundaries as they encompass all systems dependent on the existence of the studied The sustainability indicators used for a sustainability assessment must be relevant to the studied system. A review of assessments of various impacts on sustainability enabled the selection of a set of indicators relevant to a sustainability assessment of electricity generation systems. The review highlights the lack of a comparison of sustainability impacts for electricity generation from Australian fossil fuels, especially brown coal. Moreover, previous social and economic assessments have lacked the use of 'life-cycle' boundaries, thus underestimating the social and economic effects of electricity generation systems.

Therefore, no previous study has compared the sustainability impacts of electricity generation systems, using sustainability indicators and a 'life-cycle' perspective, for Australian fossil fuels.

3. METHODS FOR PRODUCING SUSTAINABILITY **INDICATORS**

'Sustainable Development has become something of a holy grail in modern times. Rather like the Yeti and the Loch Ness Monster, there have been many claims of sightings but verification is hard to come by.'

3.1 OVERVIEW **O**F **METHODS** FOR **PRODUCING** SUSTAINABILITY INDICATORS

One of the important outcomes of Chapter 2 was the selection of a set of indicators for sustainability assessments of electricity generation and similar systems. Yet, there has been no mention of how to quantify these indicators. The selection of indicators is the outcome of a review of environmental, economic, social, and sustainability studies of such systems. It is also possible to simultaneously select methods used for quantification of these impacts. Chapter 2 also acknowledges that the 'life-cycle' perspective, where the collection of impacts is from cradle to grave, is essential for sustainability assessment. Only some of the methods for producing environmental indicators used the 'life-cycle' perspective. Either modification of the existing methods or generation of entirely new methods will be necessary to use this perspective. Therefore, this chapter produces a group of methods to enable the sustainability assessment of electricity generation, using indicators with 'life-cycle' coverage.

The first three sections (Sections 3.2 to 3.4) develop methods to produce environmental, economic and social welfare indicators with 'life-cycle' boundaries. These methods have varying levels of detail, depending on their relative maturity.

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

BELL AND MORSE (2001), PG. 292.

47

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

Section 3.5 suggests modifications necessary for the economic and social indicators to ensure the resultant set of indicators is consistent in scope, detail and accuracy.

3.2 METHODS FOR PRODUCING ENVIRONMENTAL INDICATORS

3.2.1 OVERVIEW OF METHODS FOR PRODUCING ENVIRONMENTAL INDICATORS

Chapter 2 identifies resource depletion, climate change, acidification, eutrophication, photochemical smog, toxicity, solid waste, particulates and water consumption as the most important environmental indicators of sustainability for electricity generation. Thus, this section, examines methods for the calculation of these indicators.

Section 3.2.2 examines methods for developing environmental indicators. A detailed discussion of the selected, life cycle assessment (LCA) method appears as Section 3.2.3.

3.2.2 Environmental Assessment Methods

3.2.2.1 OVERVIEW OF ENVIRONMENTAL ASSESSMENT METHODS

Most environmental assessment techniques can be traced to the late sixties (for example, Starr (1969)), as the consequences of the rapid industrial expansion in the Western world of the 1950's and 1960's became apparent. The driver for this expansion was the huge increase in consumption, especially of natural resources, stemming from rapid population growth and greater consumer confidence (Sadar (1996)). The U.S.A. released the first national policy on the environment, the 1969 National Environmental Policy Act, which also contained the concept of Environmental Impact Assessment (EIA) (Sadar (1996)). The oil crisis of the early 1970's quickened the pace of development in environmental analysis techniques, as recognition of oil resource scarcity and economic vulnerability became widespread (Boustead (1995), Sadar (1996)). During this time a number of world modelling exercises (and subsequently meetings) occurred (i.e. 1970 Club of Rome and 1972 UNEP Stockholm Conference), which lead to concern about the survival of society if it remained unchanged (Boustead (1995)). Legislation requiring some form of environmental assessment (usually EIA)

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

for all new projects, which had potential to cause environmental impacts, was introduced in most western (OECD) countries during the 1970's (Sadar (1996), Environment Australia (1996)).

When the 'Brundtland' report (WCED (1987)) popularised the concept of sustainability it shifted the emphasis towards the new approaches to economic development: emphasising carrying capacity and integrated resource management (Sadar (1996)). Such approaches also led to wide use of other environmental assessment techniques, including environmental audits, substance flow and life cycle assessments. Worldwide concern about the impact of development on the environment, especially climate change led to a meeting in Toronto in 1988. At Toronto, most industrialised countries agreed to voluntarily reduce carbon dioxide emissions by 20 % by 2005. In 1992, the UNCED Rio Conference discussed concerns over the sustainability of third world (or developing) countries (Sadar (1996)). The results of these meetings, while encouraging, did not lead to verifiable targets. In the UNFCCC meeting at Kyoto in 1997, many (mostly OECD) countries negotiated targets for reductions in greenhouse gas emissions.

Methods for environmental assessment can be for plants and processes, or products. Plant and process methods consider the effects for a single location or process. The most common method of this type is Environmental Impact Assessment (EIA). This is due, in part, to the statutory requirement for such an assessment as part of the project approval process in many countries (Sadar (1996)). Another important factor is its wide acceptance and standardised methodology. It is the most comprehensive method available and can consider a vast array of environmental stressors and effects. Environmental audits are also common, but to operating, rather than proposed plants (Heijungs (1995)). Environmental audits measure current performance against legislative requirements (current or proposed) or internally set targets. Risk assessments consider the impacts of low probability, high impact effects (i.e. explosions), rather than normal operating conditions (UNEP (1996)).

Product specific methods are not limited to a specific plant or location, rather they consider the stages in the production of a particular product (i.e. the 'life cycle' perspective). Some of the product specific methods used are input-output analysis, substance flow analysis, technology assessment and life cycle analysis (LCA) (Heijungs (1995), UNEP (1996)). The use of the LCA method is becoming increasingly common,

48

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

as it is the only integrated analysis method that is equipped to quantify environmental impacts from 'cradle to grave' (Consoli et al. (1993)). LCA, like EIA, has the advantage of wide acceptance and a standardised methodology (BHP Research (1999)).

3.2.2.2 Environmental Impact Assessment: Advantages and Disadvantages

Environmental Impact Assessments (EIA) are designed to 'evaluate the environmental and related social implications (negative and positive) of carrying out a development project, of any size, before irrevocable decisions are made' (Sadar (1996)). EIA aids the improvement of internal decision making processes and the ability of third parties (such as government regulators) to make informed decisions about future projects (Sadar (1996)). In current practice, EIA has a site-specific scope, identifying the environmental impacts of one economic activity at a specific time (Kniel et al. (1996), UNEP (1996)). This site-specific scope gives EIA its greatest advantage, in that it can consider the entire ecosystem of the site (Sadar (1996)). The boundaries of these systems are natural (i.e. watersheds), not arbitrary or political (Sadar (1996)). Other advantages of this methodology are its ability to determine the effects of the process on each individual in the ecosystem, and to consider the resilience of these individuals and ecosystems (Sadar (1996)). If properly performed EIA can lower project costs (reduced instances of environmental disasters, court cases, and clean-ups), increase the efficiency of natural resource use, and enhance public confidence (through involvement in decision making) (Sadar (1996)). Some have used EIA for sustainability assessment (Sustainability Impact Assessments (SIA)), by including economic and social impacts in the method (George (2001)).

One disadvantage of EIA is the limitations on scope and boundary enforced by extensive legislative control (Kniel et al. (1996), Sadar (1996)). These limits reduce its usefulness and applicability to real situations, which have more than one process step and location. Some modifications to EIA allow the measurement of impacts across multiple locations, such as in strategic environmental assessment, but these methods are not widely used (Sadar (1996)). Another disadvantage, is that the restrictive methods legislated cannot account for all the impacts likely to occur for modern complex technologies (Kniel et al. (1996), Sadar (1996)). Furthermore, such studies do not consider the process technology, concentrating on the inputs and outputs, and therefore do not offer suggestions for improvement (Kniel et al. (1996)). Finally, the success of

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

any EIA depends on the existence of detailed information on the effects of the project on the local environment. Many studies require substantial additional research to overcome a lack of such information (for example, Woodside Energy Ltd. (1998b)).

In Australia, each federal and state government has its own legislation regarding EIA. From 1974^{*}, all projects that may, or cause another party to, affect the environment to a significant extent must submit an EIA to the relevant government environmental department[†] before the granting of approval for construction (Environment Australia (2000)). Thus, it has become a requirement to ensure project approval, rather than an objective tool for investment decisions. Some examples of legislated EIA of electricity related subjects include SECV (1973) and ECNSW (1980).

ASSISSMENT .

Life cycle assessment (LCA) is an objective tool for the determination of environmental impact, to improve the environmental performance of existing systems, or to compare between products, processes or activities (UNEP (1996)). LCA follows the path of the target product or process from its initial beginnings to its final disposal, or 'cradle to grave' (Cowell et al. (1999)). For example, in steel manufacture, the 'cradle' is the mining of iron ore and the 'grave' is the dumping of scrap steel. Between these two end points there are many stages; in general this includes raw material production, processing, manufacturing, transportation and distribution, use, re-use, maintenance, recycling and final disposal (Burgess and Brennan (2001), Consoli et al. (1993), Nash and Stoughton (1994)). Thus, LCA determines the environmental impact with a 'life in LCA cycle' perspective.

The LCA-method's holistic approach/provides its main advantage over other techniques (UNEP (1996), Vattenfall (1999)). Other methods usually concentrate on a particular location or aspect (e.g. manufacture). The 'cradle to grave' approach, considering all stages, highlights the most environmentally important stages and allows optimisation of environmental control efforts. This is especially important given that the end-of-pipe and 'command and control' approaches have been inadequate to

3.2.2.3 LIFE CYCLE ANALYSIS: ADVANTAGES AND DISADVANTAGES

Refers to the Environment Protection (Impact of Proposals) Act 1974 (Commonwealth Government). [†] If the project is wholly within one state then this is the state government environmental department. However if the project spans states then this is the federal government environmental department.

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

maintain or reduce emissions (Mohin (1994), Burgess (1999)).

The LCA method is scientific, uses quantitative system data, and produces quantitative results (UNEP (1996)), allowing the integration of its results with economic analysis techniques (Udo de Haes (1993)). Finally, even though still developing, it is an method internationally standardised process (ISO 14040's), without-the-EIA-method2-s-limiting local-legislative requirements. Therefore, results from this method are internationally recognisable and verifiable.

3.2.2.4 SUMMARY OF ENVIRONMENTAL ASSESSMENT METHODS

For an environmental assessment to truly represent the impacts of a product or process, it must include all the environmental impacts through its entire life cycle. LCA, which assessment has these characteristics, is thus a more appropriate method than EIA for the measuring

environmental impacts-for an-assessment-of-the sustainability of electricity generation systems.

3.2.3 LIFE CYCLE ASSESSMENT

3.2.3.1 OVERVIEW OF LIFE CYCLE ASSESSMENT (HISTORY)

LCA has its origins in the late 1960's, along with most other environmental analysis techniques (see Section 2.2.1, Chapter 2). These early studies were simple, containing quantitative analyses of energy[‡] and solid wastes, but with little detail on environmental effects (Burgess and Brennan (2001), Miettinen and Hämäläinen (1997)). The first of these, the selebrated Coca-Cola packaging study, which claimed that plastic containers were more environmentally friendly than glass, was performed by the Midwest Research Institute in 1969 (Huppes 1996)). Various studies, by the early 1970's, had estimated the energy requirements of steel, pulp and paper and petroleum refining, and provided additional data on raw material and solid waste flows (Fava and Page (1992)). The fading of the energy crisis limited the development of these types of study, and they became infrequent (Fava and Page (1992)). These pioneers in LCA were generally involved with private companies in the U.S.A. (e.g. R. G. Hunt and co-workers (1974)),

Sweden (G. Sundström (1971)), and Switzerland (Basler and Hoffman (1974))[§]. although there was some public work in the U.K. (Burgess (1999), Huppes (1996), Udo de Haes (1993)).

In the 1980s, European governments (including the Dutch and Swiss) became interested in LCA (Huppes (1996)). The Organisation for Economic Cooperation and Development (OECD), through its COMPASS project (1980-88), and the United Nations Environment Programme (UNEP) (1979-86), initiated multi-national studies (Sørensen (1992)). The isolated development of these studies often led to widely different approaches (Huppes (1996)). Thus, the results obtained from studies on the same object were often dissimilar (Burgess and Brennan (2001), Udo de Haes (1993)). Such variations prevented LCA from becoming an established part of environmental analysis (Consoli et al. (1993), Udo de Haes (1993), UNEP (1996)). By the end of the decade it had become apparent that acceptance would come only with a standardized methodology (Huppes (1996)).

The first stage of this process was by the Society of Environmental Toxicology and Chemistry (SETAC) in 1990, with meetings between practitioners in Europe and the U.S.A. (Huppes (1996), Udo de Haes (1993)). Despite fundamental differences in their viewpoints, the meetings lead to the development of a framework, Guidelines for Life-Cycle Assessment: A "Code of Practice" (Huppes (1996)). Superseding this book were the ISO 14040's series of standards, the development of which started barely a year later (Huppes (1996)). The release of the first of these standards, ISO 14040, was in 1997. The standards published are:

- framework;
- definition and inventory analysis;
- assessment;
- interpretation;

§ See Udo de Haes (1993) for references.

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

1. ISO 14040-Environmental management-Life cycle assessment-Principles and

2. ISO 14041-Environmental management-Life cycle assessment-Goal and scope

3. ISO 14042-Environmental management-Life cycle assessment-Life cycle impact

4. ISO 14043-Environmental management-Life cycle assessment-Life cycle

53

[‡] For an explanation of energy analysis, see Sorensen (1979).

- 5. ISO/TR 14047-Environmental management-Life cycle assessment-Examples of application of ISO 14042;
- 6. ISO/TS 14048-Environmental management-Life cycle assessment-Data documentation format; and
- 7. ISO/TR 14049-Environmental management-Life cycle assessment-Examples of application of ISO 14041 to goal and scope definition and inventory analysis.

In 2001, CML published a comprehensive guide (Guinée et al. (2001)) on the practical application of LCA, as defined in the ISO standards.

3.2.3.2 OVERVIEW OF LIFE CYCLE ASSESSMENT (STRUCTURE)

The framework for the LCA method consists of four stages: Goal and Scope Definition, Inventory Analysis (LCI), Impact Assessment (LCIA), and Improvement Assessment. In its early forms, such as that proposed in Consoli et al. (1993), the progression of stages was through the first stage (i.e. after the completion of each stage the first stage was reviewed). Most studies before the publication of ISO 14040, in 1999, followed this framework. In ISO 14040, the LCA method's framework became iterative (see Figure 3.2.1), encouraging review of all stages if proven necessary during the assessment. Moreover, the intention of the fourth stage changed from the narrow assessment of possible improvements to the study subject, to a more broad 'Interpretation' of the LCA results and conclusions.





Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

Some practitioners use simpler frameworks (Fava et al. (1991)). For example, Spath et al. (1999) in their LCA study of coal fuelled electricity generation used only the first two stages. However, the ISO framework is most common. Sections 3.2.3.3 to 3.2.3.6 introduce the four stages in the ISO framework of the LCA method.

3.2.3.3 GOAL AND SCOPE DEFINITION

The first stage, Goal and Scope Definition (see ISO 14040), communicates the goal and scope of the study to readers, to provide an understanding of both the motivation for and limitations of the study. The discussion must maintain a balance between the clarity of the expressed information, and the transparency of its argument (SETAC (1998)). To ensure understanding of the goal it must contain not only the purpose of the LCA, but also the initiator of the study and the intended audience for the study report (ISO 14040). Definition of the study scope enhances understanding of the study subject and its function, and of the reporting procedure and style.

emissions your 4

Also within the bounds of definition is the development of an initial boundary for the system. The pure methodology of LCA (comprehensive LCA)/would follow all materials consumed and emitted from the process to their ultimate environmental sources (i.e. resources in ground) and sinks (air, land, and water) (Tillman et al. (1994)). The nature of the modern economy is complex and interconnected. A single part, for example, may be assembled in one plant, from parts fabricated in several others, and the materials for these parts could come from other plants, and so on. The resulting LCA system, in many cases, would be world-scale (Mann et al. (1996)). Very large systems have so many material streams that the solution process may become difficult and timeconsuming (Mohin (1994), Tillman et al. (1994), Burgess and Brennan (2001)), or even

An important decision is the choice of 'functional unit'. The 'functional unit' is usually a measure of the systems output (e.g. an electricity generation system's functional unit is its output of produced electricity), and is the basis for the study results and comparisons. For example, the 'Greenhouse Gas' emissions of a shoe making system, 'A', might be 12 kg per pair of shoes. If the goal were to compare 'A' with another shoe making system, 'B', then the emissions of 'B' should also be expressed in the same form, i.e. 11 kg per pair of shoes. The use of the same basis allows an understandable and relevant comparison of the two system's)'Greenhouse Gas'

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

infeasible (Udo de Haes (2000)). Much of this system contributes relatively little to the environmental impacts of the study subject, i.e. auxiliary processes not directly influencing the study subject (Mann et al. (1996)), low quantity streams, or low hazard material streams. Thus, in most cases it becomes unnecessary to include all parts of the system (Udo de Haes (2000)). This boundary (the system boundary) excludes the minor contributors, to the environmental impacts of the product, from the important contributors (Consoli et al. (1993), Burgess and Brennan (2001)).

The system boundary is therefore a limitation, involving both value and scientific choices. This process should not remove important inputs or outputs from consideration, as inadequate boundary definition often leads to contradicting results from similar systems (Burgess and Brennan (2001), Clift (1998), Harsch et al. (1996), Keoleian (1993)). Methods for boundary placement are usually not entirely scientific, requiring personal judgement as to the effect of exclusions on the system. A scientific method (e.g. Lee et al. (1995)) would require calculation of the effect that each of the exclusions has on the results (Burgess (1999)). Thus, the individual practitioner must establish criteria for boundary placement (ISO 14040, Burgess and Brennan (2001), Lee et al. (1995)). Documentation of this criteria is necessary, to allow for critical review by an independent expert (ISO 14040).

The level of detail necessary to meet the study's goal and the time and funding available for the study sets the inclusiveness of the system boundary. Common boundaries include:

- 1. single plant or location (subsystem) (i.e. manufacturing, known as a gate-to-gate assessment)
- 2. simple groups of subsystems (i.e. raw material winning to manufacturing, known as cradle-to-gate assessment^{**})

Other simplifications include concentrating on a single environmental impact (Udo de Haes (2000)) or on the inventory analysis stage (Mohin (1994)).

The Goal and Scope Definition only sets an initial system boundary; further information is necessary to establish that these boundaries are sufficiently inclusive. Thus, often modifications occur to this initial boundary placement in the remaining

LCA stages.

The Goal and Scope Definition should also detail the bases for all decisions and assumptions in the remaining stages of the LCA (ISO 14040). Section 3.2.3.4 to 3.2.3.6, contains a discussion of the key decisions and assumptions within these stages.

3.2.3.4 INVENTORY ANALYSIS (LCI)

The second stage, Inventory Analysis (LCI) involves the construction of a process flow chart, collection of data, redefining system boundaries, and processing the data into the form of per functional unit (i.e. per MW) (UNEP (1996)). LCI creates a material and energy balance for the entire life of the study subject. This includes obtaining the materials crossing the system boundary, i.e. materials used in production, all transport and processing stages in production and supply to consumers, use, recycle, and disposal. Numerous rules govern the creation of this mass balance (see ISO 14041), to minimise effects on the objectivity of the result. It is important that operating data, not just design data, be gathered to account for real system operation, where deviations from design can oecur, including low (or high) production rates, planned and unplanned shutdowns, and fugitive emissions. Collection of material and energy data for the mass balance is difficult and time consuming; its progress determined by the accessibility of data of the systems under study, and the resources available for the practitioner. A lack of knowledge or resources often forces the neglect of some materials or material streams, or the use of an assumption in place of real data (Burgess and Brennan (2001), ISO 14040, Lindfors et al. (1995)). Software packages can provide data for many systems (i.e. Simapro and Umberto^{††}), but this data is generally for European systems, is of varying detail, and uses generic data and assumptions, which may not be transparent, and thus introduce other errors (Treloar (2000)).

It is important to define a methodology for data collection, and the desirable level of data quality (ISO 14040). Definitions of data quality use many terms; the ISO standards recommend the following list (ISO 14040):

a. Time-related coverage: age and collection period;

b. Geographical coverage: representative area;

^{††} See Simapro (2001) and Umberto (2003) for more details about this software.

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

One such simple system, for energy systems, is full energy chain (FENCH) analysis, which includes mining, transport and generation subsystems (IAEA (2000)).
- c. Technological coverage: technology mix;
- d. Precision: variance of the data;
- e. Completeness: ratio of reporting sites to total sites;
- f. Representativeness: qualitative assessment of degree to which the data reflects the true population;
- g. Consistency: qualitative assessment of uniformity in methodology use; and
- h. Reproducibility: qualitative assessment of the extent to which information about the methodology and data values allows an independent practitioner to reproduce the results reported.

ISO 14040 recommends collection of all of this information, when the results of an LCA are for public distribution (ISO 14040). The final stage of LCA, Interpretation, may use this data quality information to estimate its influence on the results of the LCA (quality analysis).

Some subsystems of the study subject's system have multiple products, e.g. a petroleum refinery might produce diesel, petrol, aircraft fuel, and any number of other products. Each product is responsible for some portion of that subsystem's inventory. Thus, it is necessary to separate, or allocate, the inventory of products used in the study subject's system from the remainder of the inventory.

Allocation should reflect the physical behaviour of the system (Burgess and Brennan (2001), Consoli *et al.* (1993), ISO 14041). Many parameters fit this description: i.e. economic value, mass, volume, energy or exergy content, surface area, or number of moles (Burgess (1999), Stromburg *et al.* (1997)). The final choice is often subjective or arbitrary. The results of allocation are dependent on this choice (Burgess and Brennan (2001)). Allocation based on economic value has many supporters, since it reflects the social basis for the functioning of all processes (Guinée *et al.* (2001), Stromburg *et al.* (1997)). The ISO standards (ISO 14041) recommend this method when no other method reflects the physical behaviour or other methods are inadequate. For example, the traces is the producing two products of vastly different worth (say gold and copper), with copper present in higher proportions than gold, but gold provides the greatest value and drives production. Thus, allocation based on economic worth more adequately.

represents the situation as plant would not exist if no gold were present. This method

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

requires knowledge of the economic values of all products (Guinée *et al.* (2001)), and perhaps knowledge of their long-term variability. Methods that are more complex have been used, i.e. natural and physical relations of cause and effect (Stromburg *et al.* (1997))^{‡‡}. Other workers have reported that the choice of allocation method affects the outcome of LCA studies (Burgess and Brennan (2001), Guinée *et al.* (2001)). Where more than one allocation method is applicable, testing of both methods is necessary, to ensure that the outcome is robust to this choice (ISO 14041). Allocation may also be required for LCA studies which include reuse or recycling of materials (ISO 14041). Avoidance of allocation is sometimes possible, by dividing the unit process into smaller sub-processes that do not require allocation, or expanding the system to include by-product stages (ISO 14041). However, this may substantially increase data requirements, and in most cases, it is easier to allocate (Stromburg *et al.* (1997)).

Another important function of the LCl is to review the placement of boundaries on the LCA system. Changes to the boundaries may be necessary if there are differences between the expected and true relevance's of material streams. For example, the data might show a stream, initially considered significant, has low impacts (UNEP (1996)). Conversely, data collection might suggest the inclusion of an initially excluded stream. The basis for any changes to the initial system boundary must be a sensitivity analysis, or other proof of insignificance to the results of the study (ISO 14041). Provision of the reasoning 'Jr each inclusion and exclusion will allow for review (ISO 14040).

The main outcome of the LCI will be the production of an inventory of material consumptions and emissions fin kg per functional unit), for the study subject's system.

3.2.3.5 IMPACT ASSESSMENT (LCIA)

The third stage of LCA, Impact Assessment, involves the conversion of the mass balance into scores for a number of defined environmental impacts. LCIA consists of two operations, classification and characterisation (ISO 14040). In addition, other operations are optional, such as normalisation and weighting (ISO 14040).

Classification involves the distribution of the study subject's inventory, prepared in the LCI stage of LCA (see Section 3.2.3.4), into impact categories (ISO 14040). Impact categories are types of environmental impacts, such as climate change and acidification.

** See Burgess and Brennan (2001) for further details on these methods.

58

this

The goal of the study will dictate the choice of impact categories. For example, if the desire is to examine the impact on the environment of a subject's 'Greenhouse Gas' emissions, then the only impact category necessary is climate change.

Impact categories may refer to any part of the environmental impacts mechanism (Udo de Haes *et al.* (1999)). The environmental mechanism is the chain of events (effects) between the process or product and a particular impact. Thus, the choice of the endpoint of the environmental mechanism may include the impact categories: damage to humans, loss of biodiversity and loss of materials (Udo de Haes *et al.* (1999)). The more usual choice is to use midpoints, such as climate change or ozone depletion, as this generally reduces the amount of specific environmental detail required.

Table 3.2.1 shows the impact categories recommended by the ISO standards. An important feature of LCA is that it includes input impacts, such as the 'resource depletion' and 'increase of land competition' impact categories, as well as output related impacts. The CML guide (Guinée *et al.* (2001)) recommends the use of the impact categories highlighted in Table 3.2.1, in all LCA studies. The final choice for environmental impact categories is up to the practitioner.

The definitions of the impact categories have only very recently been standardised (Udo de Haes *et al.* (1999)). Some traditional categories are absent from the list of standard categories (see Table 3.2.1), due to a lack of understanding of how to measure their impact. Some examples of these traditional categories are noise, odour, damage due to accident, and non-toxic human impacts (Udo de Haes *et al.* (1999)). Classification methods for the toxicity categories are still developing (Boustead (1995), Burgess (1999), Curran (1993), Udo de Haes *et al.* (1999)), and do not yet meet the ISO

Table 3.2.1: Impact categories recommended in the ISO standards (ISO 14042, Udo de Haes *et al.* (1999)). Bold categories are those recommended for use in all studies by CML (Guinée *et al.* (2001)). See 'Glossary' for more information about bold categories.

Inputs	Outputs
Extraction of biotic resources	Climate change
Resource depletion	Stratospheric ozone depletion
Land use:	Human toxicity
Increase of land competition	Eco-toxicity
Degradation of life support functions	Photo-oxidant formation
Bio-diversity degradation	Acidification
	Eutrophication

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

requirement that LCIA have a scientific basis (Udo de Haes et al. (1999)).

Many of the impacts described involve the same molecular species. For example, \times nitrogen oxides (NO_x) contribute to the acidification, photochemical smog, eutrophication, and toxicity categories. A material or emission contributing to one impact may be available to all other impacts (known as a serial mechanism), no other impact (known as a parallel mechanism), or modified such that it produces a different impact (indirect mechanism) (Gorokhov *et al.* (2002)). Some materials, or emissions, may need to combine with others to create an impact (combined mechanism) (Gorokhov *et al.* (2002)). Yet, the determination of the exact pathway of each material or emission generated by the systems would involve substantial additional assessor resources, and may prove impossible to achieve. Therefore, practitioners commonly assume that the serial mechanism applies. Consequently, the results from most LCA are high estimates of the impact, rather than necessarily the actual impact in practice.

The LCA method aggregates data by integrating across time and space (Udo de Haes et al. (1999), Huppes (1996)). Integration over time is valid if there is no great change in impacts between any given year and the next, i.e. steady state (Udo de Haes (1996)). Yet, in some systems there are variations in environmental effect over time. For example, the leaching of heavy metals from mining wastes decreases steadily over periods of up to several hundred thousand years (Huppes (1996), Burgess (1999)). Udo de Haes et al. (1999) discuss methods for dealing with such systems. Current best practice involves the use of infinite time, without discounting, with approximation used only if relevant (Udo de Haes et al. (1999)). Spatial aggregation is possible as the impact categories measure midpoint effects, rather than final effects (see Paragraph 3 of this section). The need for spatial differentiation is apparent for impact categories that have only local effects. Some examples are acidification (important over lakes and forests, but has little effect over the sea), photo-oxidant formation (smog precursors have little effect outside urban areas), or human toxicity (effect varies if emitted indoors or outdoors) (Udo de Haes et al. (1999)). For a summary of possible solutions, see Burgess and Brennan (2001).

The characterisation step involves multiplying the classified data by an equivalency factor (CF), which gives an indication of the relative impact of the substances within the classified data. Other workers have used detailed environmental models of the fate and impact of each possible material consumed or emitted to develop these CF factors

(Mohin (1994)). For example, methane (CH₄) has a global warming potential (GWP) of 21 kg CO₂/kg CH₄ (Guinée et al. (2001)), where GWP is measured in kg CO₂, and based on methane effecting global warming for 100 years. The use of different environmental models has led to some categories having multiple sets of CF factors. This choice of CF factors introduces a degree of subjectivity, and therefore the decision process should be considered and transparent (ISO 14040). There has been some debate about the introduction of a scientific method for this choice (Udo de Haes (2000)).

CML recommends methods for characterisation in its Guide (Guinée et al. (2001), Volume 2A, Chapter 4), and provides equivalency factors for each recommended method (Guinée et al. (2001), Volume 2B, Chapter 4, shown here in Table 3.2.2).

Table 3.2.2: Methods for characterising impacts recommended by the CML guide (Guinée et al. (2001)).

Impact Category	Characterisation Method(s)	
Resource depletion	Abiotic depletion potential (ADP)	-
Increase of land competition	un-weighted aggregation of used land $(m^2.y^{-1})$	
Climate change	GWP ₁₀₀ (Houghton <i>et al.</i> , 1994 & 1995) ^{§§}	
Stratospheric ozone depletion	steady state ozone depletion potential (ODP)***	
Human toxicity	steady state human toxicity potential (HTP) ^{†††}	
Eco-toxicity	separate toxicity scores for freshwater, marine and terrestrial phases ^{†††}	
Photo-oxidant formation	high NO _x photochemical oxidant creation potential (POCP) ^{\ddagger}	
Acidification	average European acidification potential (AP) ^{\$\$\$}	
Eutrophication	generic eutrophication potential (EP)	_

Resource depletion is a category that attempts to aggregate the environmental impact due to the extraction of non-living material from the environment. Current definitions propose three subcategories:

- a. deposits, which cannot be replaced (e.g. fossil fuels and mineral ores);
- b. funds, which are replaced naturally over time (e.g. groundwater, sand and clay);

and

and surface water) (Udo de Haes et al. (1999)).

Another difficulty involves the choice of an appropriate indicator for the resource depletion category. It is desirable, for simplicity, to use a single indicator. However, this indicator must be able to account for the differences between non-renewable (deposits) and renewable (funds and flows) materials. This is difficult since the empirical verification of its validity is impossible (Finnveden (1996)). In addition, if the subcategory definitions apply different bases (i.e. present availability, possible future use, or energy potency) then different endpoints might be relevant (Udo de Haes et al. (1999)). In these situations, separate indicators for each basis are more relevant than an overall indicator. For some systems, these simple and complicated methods both produce similar results (Finnveden (1996)).

The most commonly used category indicator for resource depletion, consumption rate, involves the division of consumption by deposit capacity. Yet, is the appropriate ratio reference region local (i.e. coal used to coal in mine), regional (i.e. coal used to coal in region), or global (i.e. coal used to global coal resources)? Practitioners commonly use a global reference region. This introduces a new figure, global reserves, and with it come its own uncertainties. Firstly, the discovered energy resources increase by 0.8 % per year (Müller-Wenk (1998)). Secondly, the figure is often an aggregate of regional figures, compounding the individual errors, or scaled from individual country figures, without considering regional differences (such as in UNEP (1996) and Guinée (1993)). In addition, the figures do not discriminate between the high quality resources, currently utilised (i.e. coal deposits close to the surface, or light, low sulphur oil and gas), and the lower quality resources that may need to replace them (Müller-Wenk (1998)). Thus, a total resources figure overestimates the amount of useful resources. The eventual mount of resources used, the ultimately extractable reserve, is the only relevant measure of resource use (Finnveden (1996), Guinée and Heijungs (1995)). Unfortunately, data for this value is unavailable, because of its dependence on future technologies (Finnveden (1996)). Nevertheless, many LCA studies use a global reference region for resource depletion quantification (Fava et al. (1991), Finnveden (1996), Heijungs et al. (1992), Guinée and Heijungs (1995)). Some other indicators proposed for resource depletion include rareness of resources,

Sustainability of Australia's Electricity Generation. Chapter 3:Methods for Producing Sustainability Indicators

c. flows, which are continuously and completely replaced (e.g. solar energy, wind

^{§§} Model developed by the Intergovernmental Panel on Climate Change (IPCC).

[&]quot; Model developed by the World Metcorological Organisation (WMO)

^{†††} Model developed for USES 2.0 model of RIVM.

^{*}** Model developed by UNECE.

^{§§§} Model developed at IIASA (RAINS10 model).

consumption rate, energy depletion, exergy destruction, mineral concentrations (aggregation with weighting), total material usage (aggregation without weighting), and flow use to total flow (Finnveden (1996), Udo de Haes et al. (1999)).

The CML guide (Guinée et al. (2001)) recommends the method shown in Equation 3.2.1 to produce characterisation factors for resource depletion.

$$CF_{RD,i} = \left[\left(\frac{C_{i,R}}{(R_{i,R})^2} \right) / \left(\frac{C_{Sb,R}}{(R_{Sb,R})^2} \right) \right]$$

Equation 3.2.1

where: $CF_{RD,i}$ = resource depletion equivalency factor for substance i (kg of substance i per kg of Antimony (Sb.)^{****}; $C_{i,R}$ and $C_{Sb,R}$ = consumption of substance i and Sb. in region R (i.e.) the world) for one year (kg per annum); and $R_{i,R}$ and $R_{Sb,R}$ = remaining reserve of substance^{tt††} i and Sb. in region R.

Other commonly used variants include energy and exergy content measures (see Equation 3.2.2).

EDP = Exergy Depletion (MJ per kg) or Energy Content (MJ per kg)

Equation 3.2.2

where: EDP = Exergy or Energy Depletion Potential (MJ per kg).

All of these methods fail to measure the impacts of differences in abundance and social value, which are the bases of the resource depletion category (Finnveden (1996), Guinée and Heijungs (1995)). Thus, debate continues as to the applicability of these methods to measure the impact of resource depletion (Guinée et al. (2001)). Consequently, the validity of the measures as indicators of sustainability is questionable.

Section 2.4 (Chapter 2) indicates that each of the indicators in Table 3.2.2 are important for electricity generation systems, except increase in land competition and stratospheric ozone depletion. For each of these indicators, characterisation involves:

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

$$ICS_{a,c} = m_c x CF_{a,c}$$

of impact category a's reference substance).

Three more impacts are important for electricity generation: solid waste, particulates, and water depletion. Water depletion is not an impact category in the ISO standard (ISO 14042), but is included as a flow in resource depletion. In the dry Australian conditions, there is increased competition for surface water and relatively slow rates of replenishment (Lundie *et al.* (2000)). Thus, the definition of surface water as a flow is questionable. Other substance categorisation difficulties may also occur. For example, groundwater is a fund if removed in sustainable quantities, but over-consumption could produce irreversible depletion. Thus, separation of water depletion from the resource depletion category is desirable (Lundie et al. (2000)). Due to the lack of an accepted characterisation method, aggregation applies no weighting to contributors to these three categories. Thus, all waters (surface, treated, or ground) are equivalent (characterisation factors = 1). Similarly, all particulates (dusts and particulates), and solid wastes (i.e. coal, ash, general waste, etc.) are equivalent for their categories. Quantition of the total impact score involves summing the resultant values, after characterisation, for each impact category (Equation 3.2.4). This total impact score represents the magnitude of each impact produced by the system.

$$ICS_{a} = \sum_{c=1}^{C_{T}} ICS_{a,c} = \sum_{c=1}^{C_{T}} m_{c} x CF_{a,c}$$

where: ICS_a = impact category score for impact category a for the study subject's system (kg of impact category a's reference substance per functional unit); and C_T is the total number of substances.

Equation 3.2.3

where: $ICS_{a,c} = characterised score for impact category a and substance c for the study$ <math>subject's system $\int (kg - of - impact - category - a's - reference substance); m_c = mass of <math>\int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substance); m_c = mass of \int (kg - of - impact - category - a's - reference substa$ substance c emitted or consumed in the study subject's system (kg c per functional unit); CF_{ac} = characterisation factor for impact category a and substance c (kg c per kg

^{*} Antimony is the reference substance for the resource depletion impact category.

^{††††} Includes all mineral sources.

An optional further step involves normalisation of the impact category values (ISO 14040). This process normally involves dividing by a representative value, i.e. the impact category score for the world or Australia (see Equation 3.2.5). This produces an indication of the fractional contribution of the system's environmental impact to the overall contributions from all sources in that region. For example, the emissions of greenhouse gases from a process might be 40 t of CO₂ equivalents per year after characterisation, and 0.1 % of the national yearly emissions after normalisation. In some cases, a local, national or regional representative value is of more value than the global one. This is especially true for localised impacts, such as photo-oxidant formation, acidification, and eutrophication.

$$ICS_a^{R,N} = \frac{ICS_a}{ICS_a^R}$$

Equation 3.2.5

where: $ICS_{a}^{R,N}$ = impact category score a for the study subject's system, normalised for region R (i.e. Australia) (kg of impact category a's reference substance per functional unit per kg of impact category a's reference substance emitted or consumed in region R over one year); ICS_a = impact category score for impact category a for the study $f_{ab} = \int_{a}^{a} \frac{1}{1-a} \int_{a}^{a} \frac{1}{1-a} \frac{$ subject's system (kg of impact category a's reference substance per functional unit); ICS_a^R = impact category score a for region R over one year (kg of impact category a's reference substance emitted or consumed in region R over one year).

There is a lack of consensus on the appropriateness of the normalisation method (Burgess (1999), Ekvall et al. (1997)), and the reference area and data sources it uses (Burgess (1999), Udo de Haes (1996)). However, the CML guide strongly recommends its use in any LCA (Guinée et al. (2001)).

Weighting⁺⁺⁺⁺ produces a single value, often named as an environmental index, from the impact category results. The weighting process multiplies each impact category score by a weighting factor, chosen to indicate that impact categorie; relative

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

importance (see Equation 3.2.6). In most cases, development of weighting factors is by a panel of experts or diverse interested parties, based on their opinion of the importance of these impacts to society. There are many sets of weighting factors available \$\$\$\$, but there is no consensus on the definitive set (ISO 14040, UNEP (1996)). ISO 14040 states 'there is no scientific basis for the determination of an overall environmental index'. For this reason there is no globally acceptable set of weighting factors, nor is their development likely, since they would need to be able to vary with location, time, culture, and changing values (Burgess (1999), Udo de Haes (2000), Udo de Haes et al. (1999)). Moreover, the ISO standards require the presentation of data generated before this step within the report (ISO 14040). Therefore, the CML guide states that weighting is not applicable in most LCA studies (Guinée et al. (2001)).

$OEI = \sum_{i=1}^{l_{T}} WF_i \ x \ ICS_i$

where OEI = overall environmental index; $WF_i = weighting factor for impact category$ i: ICS; = impact category score for impact category i for the study subject's system (kg of impact category a's reference substance per functional unit); and I_T = total number of impact categories.

Alternative approaches for aggregation include the 'multi-criterion' approach, and valuation (see Clift (1998) and Section 2.3.5, Chapter 2).

LCIA is continually developing (ISO 14040, Spath et al. (1999)). The ISO 14042 standard contains the concepts, technical framework, and methodological guidelines and requirements (Udo de Haes (2000)). However, to allow the freedom to match the methodology used with the goal and scope defined, and for changes brought about by technological developments, there is no prescribed method (ISO 14040). Thus, ISO/TR 14047 provides illustrative examples (Udo de Haes (2000)), and the CML guide (Guinée et al. (2001)) provides further guidance on how to apply LCIA.

The main outcome of this stage is a set of impact category results, for use as indicators of environmental sustainability.

\$555 For discussion of various weighting sets, see Burgess (1999).

^{****} Eco-indicator 99 and EPS include operational weighting. The CML guide discourages 'straight forward' use of these methods (Guinée et al. (2001)), and ISO 14042 forbids their use in public studies.

3.2.3.6 INTERPRETATION

The last stage, Interpretation (see ISO 14043) involves combining the outcomes of the other three stages to provide conclusions and recommendations consistent with the goal and scope (ISO 14040). Interpretation may also involve an iterative review of the data analysed in the other stages (ISO 14040). Consequently, it may include comparisons of the relative importance of different impacts and of different systems. Examination of the relative contributions of different subsystems within the study subject's system may also highlight opportunities for the improvement of impacts in individual subsystems. Interpretation may also include: quantitative assessments of completeness (the inclusiveness of the LCA system boundary); sensitivity (the effect of data input and methodological decisions within the LCA), uncertainty (the variability of results due to variability in data used); gravity (determines data with greatest contribution); or consistency (of criteria used for decisions) analyses (ISO 14040). A sensitivity analysis estimates the influence of key decisions within the assessment, such as the choice of allocation and characterisation factor methods, on the environmental indicators, by showing the outcome from each probable choice. Most LCA studies are simplifications, for they contain assumptions and often ignore subsystems for which there is a lack of knowledge, or which are difficult to calculate (Weidema (2000)). Thus, in all LCA studies, an estimate of the uncertainty of its results is necessary (Weidema (2000)). Section 3.2.3.7 examines uncertainty analysis for LCA. The CML guide and ISO 14040 recommend the use of these two analyses when disclosing LCA results to the public (Guinée et al. (2001), ISO 14040).

Under the SETAC system (Consoli et al. (1993)), this last stage was 'Improvement Assessment' and had a different focus. As the name suggests, this was the part of the study designed to suggest possible improvements in the product or process. Improvement assessment reduces the possibility of using LCAs to justify the current system, and highlight possible improvements in all systems (Consoli et al. (1993)). An improvement in one subsystem does not automatically produce an overall improvement, as impacts may increase in other subsystems (Sullivan and Young (1995)). For example, the introduction of a recyclable product reduces the demand for new products (thus lessens upstream impacts), but requires the introduction of a new collection and processing system (thus increases impacts). Thus, testing improvement options may require modifications to the study subject's system boundary. Consequently, the other

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

stages of the LCA method may also require review.

The outcome of this stage of the LCA is a set of conclusions and recommendations about the study subject's impact on the environment, and an analysis of the robustness of the indicators from which they derive.

3.2.3.7 REVIEW OF UNCERTAINTY METHODS

3.2.3.7.1 Introduction

Traditional methods for determining the range of a result (known as error analysis). used in other types of assessment, have proven inadequate for LCA, as they cannot include so many different types of uncertainty (Berg et al. (1999), Guinée et al. (2001)). Thus, substantial effort has occurred to develop a generalised method for LCA.

The International Organization for Standardization (ISO) 14040s series of standards recommend that practitioners document data quality requirements in the study's Goal and Scope Definition (ISO 14040) (see Section 3.2.3.4, points a.-h.). If there is a need for an accurate and detailed study, ISO 14041 recommends three types of analyses: gravity, uncertainty and sensitivity (ISO 14040, ISO 14041, see Section 3.2.3.6). In addition, the CML guide (Guinée et al. (2001)) recommends the optional use of a 'Pedigree Matrix' and/or one of the proposed uncertainty frameworks (e.g. Berg et al. (1999), or Huijbregts (1998)).

The Society of Environmental Toxicology and Chemistry (SETAC) European working group on 'Data Availability and Data Quality' (Huijbregts et al. (2001)) provide both a framework for uncertainty, and possible methods to calculate data uncertainty. The framework classifies data uncertainty into 'data inaccuracy' and 'lack of specific data' (i.e. data gaps and unrepresentative data). They recommend using either inputoutput modelling, data from similar products, or mass balances to fill data gaps. They recommend estimation of unrepresentative data uncertainty using uncertainty factors, obtained through further analysis of material inputs and outputs, and account for temporal, geographical and technological differences, using the pedigree matrix of Weidema (Table 3.2.3). Estimation of total uncertainty uses the Monte Carlo simulation method (see Section 3.2.3.7.2). Empirical justification of the uncertainty factors and data ranges is also necessary to ensure the relevance of the conclusions. This is a substantial task for even a simple LCA study.

Table 3.2.3: Matrix of indicator scores ('pedigree matrix') for rating the qualityperformance of data (Weidema and Wesnæs (1996)).

/			Indicator'Score				
Independent of the study in which the data are applied:							
Reliability of the source	based on measurements	verified data partly based on assumptions or non-verified data based on measurements	Non-verified data partly based on assumptions	(e.g. by an industrial expert)	Non-qualified estimate or unknown origin		
Completeness	Representative data from a sufficient sample of sites over an adequate period to even out normal fluctuations	Representative data from a smaller number of sites but for adequate periods	Representative data from an adequate number of sites but for shorter periods	Representative data from a smaller number of sites and shorter periods, or incomplete data from an adequate number of sites and periods	Representativen ess unknown or incomplete data from a smaller number of sites and/or from shorter periods		
Depen	dent on the goal and s	scope of the study:					
Temporal correlation	Less than 3 years of difference to year of study	Less than 6 years of difference to year of study	Less than 10 years of difference to year of study	Less than 15 years of difference to year of study	Age unknown or more than 15 years of difference to year of study		
Geographical correlation	Data from area under study	Average data from larger area in which the area under study is included	Data from area with similar production conditions	Data from area with slightly similar production conditions	Data from an unknown area or with very different production conditions		
Technological correlation	Data from enterprises, processes and materials under study	Data from processes and materials under study but from different enterprises	Data from processes and materials under study but from different technology	Data on related processes and materials but from same technology	Unknown technology or data on related processes or materials but from different technology		

A review of uncertainty analysis methods indicated that there were three types: a quantitative analysis considering only numerical uncertainties in input data (see Section 3.2.3.7.2); a qualitative analysis using indicators to consider types of uncertainty that are difficult to quantify (see Section 3.2.3.7.3); and a quantitative analysis, which attempts to convert qualitative uncertainty, and combine it with quantitative uncertainty (see Section 3.2.3.7.4).

3.2.3.7.2 Numerical Uncertainty

Quantitative analysis methods use input data variability information to obtain a cumulative uncertainty value for an output, using uncertainty propagation. Proposed

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

analytical methods for uncertainty propagation include uncertainty propagation, interval methods, fuzzy logic simulations, and Bayesian statistics (Huijbregts (1998)). More common is the use of stochastic models, such as Monte Carlo (Huijbregts (1998), Kennedy *et al.* (1996)) or its modified form Latin Hypercube (Huijbregts (1998)) simulation, as they are able to include any number of different probability distributions for the individual data points (Huijbregts (1998)). These stochastic methods randomly vary all individual data values, using their individual uncertainty distributions. For example, data point 'A' has a mean at 100, with a distribution such that the probability that it is 100 is 50 %, while the probability that it is 95 is 12 %. Then, for 100 iterations, the stochastic method will use a value of 100 for this data point in approximately 50 iterations, and a value of 95 in approximately 12 iterations. As the calculation involves many such data points, all varying independently from each other, the result calculated will also vary. The collected results from all these iterations form a new profile, which represents the uncertainty of the result

The Monte Carlo method uses actual ranges and probability distributions of the collected data, when available. However, data limitations often necessitate the use of 'rules of thumb' (i.e. Finnveden and Lindfors (1998), Lindfors *et al.* (1995)) to produce likely ranges and a probability profile for each data source. The most common probability profiles are triangular, uniform, normal, and log-normal (see Figure 3.2.2).



Figure 3.2.2: Four common probability distributions from industrial measurements (a = minimum, b = average, c = maximum).

Despite the seeming completeness of such methods, they do suffer a number of limitations. Firstly, they consider only one type of data uncertainty, variability from the true mean, ignoring the data's fitness for purpose, and most importantly, how accurately the system modelled in the LCA matches reality (Berg *et al.* (1999), Rousseaux *et al.* (2001)). Secondly, combining the uncertainties of the different types of data used at different points in the LCA model has proven difficult (Berg *et al.* (1999), Maurice *et*

al. (2000)), e.g. data uncertainty and uncertainty due to allocation between products. Thirdly, the uncertainty of the LCA model itself is unknown (Guinée et al. (2001)), and testing against real systems is impossible, as the outputs have no defined spatial or temporal characteristics (Berg et al. (1999)). Lastly, a final uncertainty value can only be calculated with confidence for a specific impact category (Berg et al. (1999)). While postulation of the relative contributions of specific impacts to total environmental impact is possible, for example by reference to expert opinion, such relativities are necessarily subjective and their uncertainty cannot be quantified on a scientific basis.

3.2.3.7.3 Qualitative Uncertainty

To measure the qualitative uncertainty in an LCA model a score is given for each of the attributes of the system that add to uncertainty; a lower score in most cases represents higher quality. These scores are of three levels (Berg et al. (1999)):

- data level (individual data sources)
 - Example indicators: see Table 3.2.3.
- process level (physical or notional subsections of the system)
 - Example indicators:
 - completeness (all flows considered in process); and
 - applicability of assumptions for allocation and aggregation.
- system level
 - Example indicators:
 - completeness (all processes considered),
 - applicability of assumptions for aggregation and impact category choices.

This is the 'Pedigree Matrix' method (Funtowicz and Ravetz (1990)).

There are many groups of indicators to describe the quality of an LCA system. For example, Weidema and Wesnæs (1996) proposed the indicators shown in Table 3.2.3. Each of these indicators is data level only, while others (i.e. Rousseaux et al. (2001), or Wrisberg (1997)) list indicators at many levels. Some declare that aggregation of these

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

indicators is inappropriate because the scores do not represent an amount of quality (i.e. Weidema and Wesnaes (1996), Weidema (1998), or Coulon et al. (1997)). Thus, they claim that the aggregated scores have no meaning (i.e. a worse score for one system over another does not imply it has worst data quality). Other researchers (i.e. Wrisberg (1997)) and Lindeijer et al. ***** suggest otherwise, claiming the aggregate score is an indicator of data quality; rather than a measure.

In one method, proposed by Wrisberg (1997), aggregation of quality indicators uses equal weights for all environmental flows. Consider for example, the case where an environmental impact result is developed from two material flows, with data quality scores of (3, 2, 4, 1, 1) and (1, 1, 2, 3, 4), for the five indicators respectively, and contributions of 33 % and 67 % of the environmental impact result respectively) The 2.5).

Another method, proposed by Rousseaux et al. (2001), compares the quality performance of each data point to some decided target quality goal score (e.g. 2 out of 5). The percentage of data points that pass this criterion (called 'Acceptability'), determines the performance of a particular process or system. For example, a hypothetical process A has three data points with quality indicator scores of (1, 2, 1, 3, 1, 3)4) for point B, (2, 3, 2, 4, 1) for point C, and (3, 3, 3, 3, 3) for point D. The chosen quality goal score is 2. For the first indicator: point B's (1) and C's (2) score passes, but point D's score (3) fails. Thus, the 'acceptability' of process A's first indicator is 66 %^{†††††}. The ratio of the quality score variance to the average quality score is helpful when attempting to reduce qualitative uncertainty (i.e. indicator scores). Variance here is the sum of each points difference from the mean divided by the number of points. Using the above example, the average of process A's first indicator is 2, its variance is 2/3, and thus its variability ratio is $33\%^{\ddagger\ddagger\ddagger}$. If this ratio is high, then the quality scores have a wide spread, and may include some very 'good' (1) and 'bad' (5) scores. In this case, improving the total quality involves reducing the number of 'bad' scores. When the ratio is low, then the quality scores are in a narrow range and improving the total

72

***** From: Lindeijer, E., Berg, N.W. van den and Huppes G. (1997), Procedure for Data Quality tittt For the remainder of the indicators the 'acceptability's' are 33 %, 66 %, 0 %, and 33 %.

Assessment, Report for Rioned (in Dutch), September (discussed in Berg et al. (1999)). ##### For the remainder of the indicators the 'reliability's' are 8 %, 33 %, 7 %, and 58 %.

data quality score requires changing greater numbers of data points.

3.2.3.7.4 Methods for Combining Numerical and Qualitative Uncertainty

The use of numerical and qualitative uncertainty results, obtained from the methods shown in Sections 3.2.3.7.2 and 3.2.3.7.3, together may indicate uncertainty in impact results. Alternatively, conversion of an individual data point's qualitative uncertainty into a quantitative uncertainty profile (called 'additional uncertainty') allows addition to the data point's numerical uncertainty profile (see Section 3.2.3.7.2). Using a stochastic method (see Section 3.2.3.7.2) generates a total system uncertainty profile from this summed profile. These methods can include the converted process and system level qualitative uncertainties at the appropriate points in the calculation (i.e. to the estimated process and system uncertainties).

Three different methods have been proposed for the conversion of qualitative uncertainty into additional uncertainty: Betz probability functions (Kennedy et al. (1996)); additional estimated uncertainty ranges (twice the coefficient of variation \$\$\$\$\$) for each type of data elements (Meier (1997)); and obtaining estimates of variation due to the problem indicated """ (Weidema and Wesnaes (1996)). The first method (of Kennedy et al. (Kennedy et al. (1996))) integrates a chosen list of indicators through a parameter, x, known as the percent of attainable data quality (Equation 3.2.7). It does not recommend a prescribed set of qualitative indicators, but that they develop from previous experience of using the method. For example, a data point has a value of 100, with quality scores of (2,2,3,2,1) for five indicators. The sum of quality scores is 10, the sum of maximum quality scores is 25 (maximum score for each indicator is 5), and the sum of minimum quality scores is 5 (minimum score for each indicator is 1). Thus x equals (10-5)/(25-5) times 100 = 25 %. Table 3.2.4 provides the data quality indicator (I) (example I is 2). This indicator (I) creates one or more Beta probability distributions, which are modified normal distributions, described using two shape parameters (α and β) and its extents (A and B). Estimating these parameters from I

quality indicator method of Kennedy et al. (1996).

% of Attainable Aggregated Data Shape Parameters Extents (% variation Data Quality (x) Quality Indicator from mean)							
	(I)	a .	β	\mathbf{A}	B		
100	5	5	5	-10	+10		
$87.5 \le x \le 100$	4.5	4	4	-15	+15		
75 ≤ x < 87.5	4	3	3	-20	+20		
$62.5 \le x < 75$	3.5	2	2	-25	+25		
$50 \le x \le 62.5$	3	1	1	-30	+30		
$37.5 \le x \le 50$	2.5	1	1	-35	+35		
$25 \le x < 37.5$	2	1	1	-40	+40		
$12.5 \le x \le 25$	1.5	1	1	-45	+45		
$0 \le x < 12.5$	1	1	1	-50	+50		

40).

\sum Quality Scores – \sum Minimu Σ Maximum Ouality Scores – Σ Min

The second method (Meier (1997)) uses data, process, and system level indicators, including indicators for assumptions, valuation, and impact category considerations. Development of conversion factors for this method relies on previous experience using the method and in other data quality analysis. For example, the conversions in Table 3.2.5 are for the geographical correlation indicator. In this case, calculation of the additional data uncertainty for the chosen indicators uses the additional uncertainties shown in Table 3.2.5, and is characterised by a normal distribution with standard deviation of 1/4 of this range. For the example data above, (2,2,3,2,1), the additional uncertainty will equal the sum of that for each indicator: $\pm(10+10+20+10+5)\%$, or ± 55 %. Using the mean value of our example, 100, then the variation is \pm 55, and the standard deviation is (45 - (-55))/4 or 27.5.

Despite the mathematical nature of these methods, the results are still subjective, reliant on expert judgement to determine the variation caused by each indicator score

Table 3.2.4: Table to obtain the shape parameters from I for use in the data

requires expert knowledge (i.e. Table 3.2.4) (in the example: $\alpha = 1$; $\beta = 1$; A = -0.4times the value of the data point or -40; and B = 0.4 times the value of the data point or

SSSSS The coefficient of variation is equal to the standard deviation divided by the mean value.

For example, an update to the energy consumption data taken from an older process could use the increased efficiency of the new process. In the general case, this requires additional research into each data point, and thus considerable additional resources may be necessary.

Table 3.2.5: Table to obtain the additional uncertainty from 1 for the geographical correlation indicator for use in the data quality indicator method of Meier (Berg et al. (1999)).

Quality Score (1)/	Additional Uncertainty Range (±%)
1	5
2	10
3	20
4	30
5	50

(Weidema (1998)). As they provide a single range for each impact category, they may enhance perception of the credibility of the LCA study results (Huijbregts (1998)). However, as knowledge of the bases of the conversion methods are limited among practitioners and decision makers alike (Huijbregts (1998)), a combined confidence result may unintentionally reduce understanding of the uncertainty.

3.2.3.7.5 Methods for Comparing System Quality Scores

Presentation of quality analysis results is in general by box plot (see Figure 3.2.3) (Coulon et al. (1997)). The basis for the choice of cut-off probabilities for the boxes (by the decision-maker) (Coulon et al. (1997)) is allowable risk, i.e. 95 % (it is true 95 out of every 100 possible cases). These plots may prove helpful, when there is minimal overlap, as in cases A and B. However, when there is substantial overlap, as in cases B and C, it is more difficult to determine which is the better option.



Figure 3.2. Box plots of system quality scores for notional 'life-cycle' systems A, B and C. The basis for the positioning of the 1st and 3rd percentiles is the required confidence.

The usual method of resolving this difficulty is to perform the LCA on the difference between the cases (i.e. Case A minus Case B). If the generated quality box plot is entirely above zero, then Case A is definitely greater than Case B. When the value straddles zero, deciding between the options involves a value judgement. Generally, the basis for this judgement is the positions of mean (or perhaps the median) values, and the qualitative or quantitative reporting of the degree of crossover.

Another proposed method involves the generation of a normalised difference probability distribution (Coulon et al. (1997)), characterised as the ratio (Result (Case A) -Result (Case B)/Result (Case A). If this ratio is positive (i.e. 0.12) then Case A is better than Case B by the magnitude of the ratio (i.e. 12 %). Conversely, if the ratio is negative, then Case B is better than Case A, again by the magnitude of the ratio. It is also proposed (Coulon et al. (1997)) that preparing a graph of the cumulative distribution function of this ratio (see Figure 3.2.4), using a Monte-Carlo simulation, will allow the estimation of the probability of one case being significantly better than the other (say by a 10 % margin). For example, in Figure 3.2.4 there is a 50 % probability that Case B is better than Case A by at least 10 %, while there is only a 20 % chance that Case A is better than Case B by the same margin. This step is essential only if closer inspection of the system cannot separate the alternatives.



Figure 3.2.4: An example of a comparison between notional 'life-cycle' systems A and B using the cumulative normalised difference method.

3.2.3.7.6 Current Practice

Data quality analysis is not current practice in LCA studies (2, 3, 4). A recent survey of LCA studies reported that only 4 out of 30 studies explicitly reported problems of uncertainty, and of these only one produced a quantitative analysis and two produced a

qualitative analysis (Ross et al. (2002)). However, this situation is an improvement from an earlier study of data quality and databases (Vigon and Jensen (1995)), which reported no use of uncertainty analysis. Similarly, out of approximately 30 reports examined for the review in Section 2.4 (Chapter 2) only one report used a quantitative quality analysis technique (confidential report). Another used a simplified quantitative method on a process level (BHP Minerals Technology (2001)), and one gave a qualitative discussion of uncertainty (Audus (1996)). Additionally, a few others gave indicative ranges based on the variability of one data value (i.e. AGA (2000), Rogner and Khan (1998)).

This practice may reflect the already considerable time requirements for many LCAs and a belief that detailed quality analysis does not provide sufficient benefit to decision makers to justify its expense. Others have suggested that the current unsatisfactory state of data quality analysis, and its lack of transparency, (Guinée et al. (2001)) provide little incentive to include it in LCA studies. Thus, research into possible simplifications is important to reduce the need for detailed analysis of each datum (Huijbregts (1998)), and the time required for data quality analysis (Maurice et al. (2000)). Integration of data quality methods into existing LCA software programs could reduce time requirements (Huijbregts et al. (2001)).

3.2.3.8 GENERAL DIFFICULTIES WHEN APPLYING LCA

There are many difficulties encountered in LCA studies, due mainly to its lack of maturity as an environmental analysis tool (Huppes (1996), Udo de Haes (1993), Burgess and Brennan (2001)). This is despite the process of standardisation, which has laid the framework for LCA, without defining the appropriate methodology (ISO 14040). Causes include the lack of universally applicable methods (for all possible studies), and a need for further development overall (Huppes (1996)). The lack of universal methodology will not change, since no method could cover all possible cases (Guinée et al. (2001), Udo de Haes et al. (1999), Burgess and Brennan (2001)). In addition, it would reduce the ability of practitioners to adapt the methodology to suit their system (Guinée et al. (2001), Burgess and Brennan (2001), ISO 14040), or for technological changes (ISO 14040) and thus possibly the applicability of their results for further analysis. As ISO 14040 states, there is no single method for LCA studies; studies should suit the requirements of the user and study.

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

Common difficulties encountered throughout LCA are decisions not amenable to purely scientific analysis. The bases for such decisions are moral values, professional judgement, or legislative requirements. There are two classes of these judgements, assumptions and value choices (ISO 14042). Assumptions are technical choices that could be validated if data were available (i.e. the background levels of a particular substance), whereas value choices are based on political, ethical or ideological principles (i.e. weighting human health against rain forest destruction) (Udo de Haes et al. (1999)). These principles change with time and place, in a similar way to the concept of sustainability (see Section 2.2.3, Chapter 2). Yet, it is not possible to completely separate an LCA from value choices, because the aim of LCA is to support such decisions (Huppes (1996)). Therefore, the documenting of all decisions based on values will allow others to understand the results in context and establish for themselves the validity of the applied value choices.

3.2.4 SUMMARY OF METHODS FOR PRODUCING ENVIRONMENTAL INDICATORS The method presented here, life cycle assessment (LCA), can produce the required indicators of environmental sustainability for a sustainability assessment of electricity generation systems. LCA applies the 'life-cycle' perspective and gives results in the form of indicators. However, the LCA method has a number of limitations requiring further investigation.

3.3 METHODS FOR PRODUCING ECONOMIC INDICATORS

3.3.1 OVERVIEW OF METHODS FOR PRODUCING ECONOMIC INDICATORS

Chapter 2 identifies wealth generation, and capital requirements as the most important economic indicators of sustainability for electricity generation. Thus, this section, examines methods for the calculation of these indicators.

Sections 3.3.2 and 3.3.3 examine the two commonly encountered methods for developing economic indicators: value added (VA) and net present value (NPV) methods. A limitation of VA methods, highlighted in Section 3.3.2, is that they do not account for capital expenditure. Section 3.3.4 examines a simplified method of

combining VA and an estimate of capital expenditure.

3.3.2 VALUE ADDED

Value added (VA) has been suggested as an indicator of economic sustainability. Value added is the difference in value between what is sold and the non-labour, non-capital inputs that are purchased for its production, i.e. goods, materials and services (Azapagic and Perdan (2000), IChemE (2002a), Richards (1993)) (see Equation 3.3.1).

VA = S - P

Equation 3.3.1

where: VA = value added; S = sales revenue; and P = cost of purchased materials, energy and services.

In this definition, P does not include materials in nature (Wood (1978)). Thus, for coal mining, coal in the ground has no cost. P includes changes in stocks of materials (Wood (1978)). Added value indicates the available cash fund from which labour, company profits, shareholder dividends and any capital investment must come (Gilchrist (1971), Riadhi-Belkaoui (1999)), and is indicative of the wealth generated by the efforts of the owners and employees (Riadhi-Belkaoui (1999)).

For a 'life cycle' perspective system, value added, is the sum of the value added by each subsystem within the system boundary (see Equation 3.3.2) (Azapagic and Perdan (2000)).

$$VA = \sum_{l=1}^{L} \sum_{s=1}^{S} \left[S_{s,l} - (R_{s,l} + MES_{s,l}) \right]$$

Equation 3.3.2

where: $S_{s,l}$ = sales from stage s in year l (AUD 1999); $R_{s,l}$ = cost of raw materials used in stage s in year l (AUD 1999); MES_{s,l} = cost of materials, energy and services purchased for stage s in year l (AUD 1999); L = life of main asset (i.e. power station) (years); and S = stages in system.

In the terminology of this report:

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

$$VA = \sum_{l=1}^{L} \sum_{s=1}^{S} \{S_{s,l} - [(O \& M_{s,l} + F_{s,l}) \}$$

where: O & $M_{s,l}$ = operating costs; $F_{s,l}$ = cost of materials produced upstream; and $L_{s,l}$ = costs of labour.

Some suggested that VA would encourage greater cooperation between workers, investors, and government, and distribute the responsibility for economic performance between all parties (Riadhi-Belkaoui (1999)). It may also be utilised to indicate the net output of a company (Riadhi-Belkaoui (1999)). The sum of VA from all sources within a nation is equivalent to its gross domestic product (GDP), which is widely reported. Thus, VA enables calculation of the contribution to GDP (Riadhi-Belkaoui (1999) Azapagic and Perdan (2000), IChemE (2002a)). Moreover, VA is the basis for many taxation schemes (i.e. value added taxation or VAT) and so is readily obtainable (Azapagic and Perdan (2000)).

Some have questioned the ability and accuracy of VA to be measure of economic value. A similar measure to VA, economic value added (EVA), which includes a capital expenditure term, has been widely applied by industries (Azapagic and Perdan (2000)), indicating widespread belief that calculation of VA should include the costs of capital investment. While VA excludes the costs of labour for the study subject, it still includes some labour costs. VA accounts for material, energy and service purchases; the price paid for these purchases includes an allowance for the labour costs of the seller. Further, high VA does not necessarily indicate high economic performance, as judged using other indicators of wealth (Riadhi-Belkaoui (1999), Richards (1998)). Thus, it may lead to poor management decisions, which increase VA, but reduce other indicators of wealth (Riadhi-Belkaoui (1999)).

Others have suggested, instead, that successful capital investment indicates high economic performance (Richards (1998)). Consequently indicators of this success, i.e. profitability, internal rates of return (IRR) and net present values (NPV), were more able to indicate the generation of value (Richards (1998)), and thus progress towards sustainability. VA was one of a number of short cuts to these more rigorous calculations (Richards (1998)). Consequently, the presentation of other economic indicators, with VA, is necessary to adequately indicate economic performance towards

)-L_{sl}]}

sustainability (Azapagic and Perdan (2000)).

While one set of sustainability indicators (IChemE (2002a)) uses VA as the preferred unit of product or service value, another (Global Reporting Initiative (2002)) regards VA as a descriptor of the scale of a company (or industry), rather than an indicator of sustainable development. Both of these sets present other economic indicators, such as sales; cost of goods, materials, and services; cost of labour; research and development; and capital costs. Thus, VA requires the presentation of other economic indicators to adequately indicate economic impacts. Therefore, it may be better to use a different economic measure as an indicator of the wealth generation aspect of economic sustainability.

3.3.3 NET PRESENT VALUE AND INTERNAL RATE OF RETURN METHODS

3.3.3.1 OVERVIEW OF NET PRESENT VALUE AND INTERNAL RATE OF RETURN METHODS

An important conclusion of Section 3.3.2 was that the Value Added indicator is inadequate to completely describe economic sustainability, due to its failure to account for all economic costs. Net present value (NPV) and internal rate of return (IRR) methods do not suffer this disadvantage. Therefore, this section examines these methods to assess their suitability as descriptors of economic performance.

Section 3.3.3.2 defines NPV and IRR. Sections 3.3.3.3 to 3.3.3.5 then present the methods to calculate the cost information necessary for a NPV or IRR calculation. Section 3.3.3.3 discusses the costs necessary to construct the facility (capital costs). Section 3.3.3.4 discusses the costs necessary to operate and maintain the constructed facility (operating costs). Section 3.3.3.5 discusses the combined cost burden due to capital and operating costs.

3.3.3.2] ET PRESENT VALUE AND INTERNAL RATE OF RETURN

Net present value (NPV) is the sum of the annual discounted titte positive (income) and

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

negative (costs) cash flows for the life of a project (Holland *et al.* (1974)). Positive cash flows for most product systems will include sales of the product, but may include sales of other items, such as wastes or equipment. Negative cash flows include: purchases of materials, energy and services; capital investment; maintenance; insurance; taxes; and other expenses. The sustainability assessment of Leca *et al.* (2002) (see Section 2.4.3.2, Chapter 2) uses NPV as an indicator. For the electricity generation systems, positive cash flows consist of electricity revenue alone^{‡‡‡‡‡‡‡}. If the basis of comparisons between systems is the sale of a common amount of electricity, then each system will have identical positive cash flow. Therefore, with such a basis, differences in the NPV of the electricity generation systems will be due to the negative cash flows. Equation 3.3.4 shows how to calculate these negative cash flows, assuming that taxes have no great influence.

$$NPC = \sum_{y=1}^{Y} O \& M_y + I_y$$

where: NPC = net present cost (sum of negative cash flows); O & M_y = operating costs for year y; I = capital investment costs for year y; and Y = total number of years of operation (life).

This NPC includes two types of costs: those attributable to the operation and maintenance of existing plant (O & M) and those attributable to construction of new plant (C_R). Sections 3.3.3.3 (C_R) and 3.3.3.4 (O & M) describe methods for estimating these costs. Then, Section 3.3.3.5, describes methods for the combination of these parts to produce the NPC indicator.

 $PV = FV/(1-i)^n$

combustion).

^{††††††} Discounting the cash flows takes account of the changes in the value of money over time. The general discounting formula is:

3.3.3.3 CAPITAL COSTS

Capital costs (C_R) are the sum of all costs required to produce a working system. C_R thus includes land, planning and construction (fixed capital cost), commissioning and working capital costs. The choice of method for estimation of capital cost depends on the desired accuracy, and resources (time and money) available (Brennan (1990), Holland et al. (1974)). The most basic of estimates involve the use of factors to estimate the variation in plant capital costs with sales revenue or production capacity (Holland et al. (1974)). More advanced estimates require detailed designs of equipment and entire plants which contribute significantly to the overall system cost (Holland et al. (1974)). Detailed estimates require a detailed design of the entire system (Holland et al. (1974)). In this case, the electricity generation systems have 'life cycle' boundaries, which include several complex plants and transport systems. Resources are not available for detailed estimations of these entire systems. Consequently, adopted is the basic estimation approach.

Basic approaches assume that the cost of any system is a function of its size or sales revenue. The most common approach uses production capacity as the indicator of relative cost (Holland et al. (1974), see Equation 3.3.5). For chemical processing plants the exponent (ψ) is often 0.7, but a wide range of exponents is possible.

$$I_2 = I_1 \left(\frac{Q_2}{Q_1}\right)^{\Psi}$$

Equation 3.3.5

where: I_1 and I_2 = capital costs of system 1 and 2 respectively; Q_1 and Q_2 = production capacity of systems 1 and 2 respectively; and ψ = exponent of the estimate.

When comparing many plants, Equation 3.3.5 can be simplified to Equation 3.3.6.

$$I = A(Q)^{\Psi}$$

Equation 3.3.6

where: I = capital costs of similar systems; Q = production capacities of similarsystems; ψ = the exponent for this type of system; and A = correlation constant.

Obtaining exponents for each system requires correlation of capital costs (I) and

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

production capacities (Q) for existing, similar systems, using Equation 3.3.6. Many studies, reports, and other sources report I and Q information.

Collection of capital costs for each subsystem occurs independently, as capital costs for entire systems are unlikely to be available. Thus, each subsystem requires its own correlation and exponent. Estimates of the required production capacity in each subsystem, corresponding to the system production capacity, allow the estimation of each subsystem's contribution to the system capital cost. For example, to produce one MWh of electricity might require the production of 1 tonne of coal from a coal mine. Summing the subsystem contributions provides the system capital cost (see Equation 3.3.7).

$$I = \sum_{s=1}^{s_{\tau}} \left[A_s(Q_s)^{v_s} \right]$$

subsystem s; and S_T = total number of subsystems.

Economic estimates of electricity generation systems commonly report annualised costs (AC), rather than NPC (for examples, see Section 2.4.3.4, Chapter 2). NPC discounts the costs of a project over its life (Figure 3.3.1). AC also discounts project costs, but modifies the discounted amounts so that they are all the same value (Figure 3.3.1). Using the AC method will enable verification against previous estimates. For the AC method annualised capital costs (CR) replaces Iy in Equation 3.3.4. Estimating C_R requires the use of the standard annuity present worth equation (Equation 3.3.8) (from Holland (1974)).



where: C_R = annual capital costs; I = system capital cost estimate; i = cost of capital; $\int l_i n \frac{1}{2} \frac{d}{d}$

84

Equation 3.3.7

where: I = system capital cost estimate; Q_s = production capacity for subsystem s to obtain the system output; ψ_s = exponent for subsystem s; A_s = correlation constant for

and n = life of loan.





Provision of capital is from two sources: equity and debt. Equity are owned funds, such as earnings from previous years or other activities, funds from share issuing, and any other moneys (Holland et al. (1974)). Debt are funds borrowed from others. Each source has an inherently different rate of interest. The interest rate for equity is the rate if invested in other systems, while for debt is the rate required by the lender (Middleton (1977)). The effective interest rate is the sum of the products of interest applying to a capital source and the proportion of capital obtained from that source^{\$\$\$\$\$\$}. This effective interest rate is the cost of capital (i).

Each of the subsystems may have a different life (n), and thus estimation of n is not simple. For example, the economic life of power station might be 30 years, while that of a coal mine might be only 15 years. Thus, for the power station to operate for its economic life, it requires a replacement coal mine after 15 years. The costs associated in construction and operation, and the technologies employed in this second mine, will probably be different to the original (Brigham and Ehrhardt (2002)). Thus, a method for aggregating these discontinuous costs is required.

\$\$\$\$\$\$ For example (adapted from Middleton (1977)):

Source	Proportion	Rate (%)	Weighted Rate (%)
Debt	0.3	6	1.8
Equity	0.7	12	8.4
Total	1.0		10.2

There are two methods to consider projects with unequal lives: the replacement chain (common life), and equivalent annual annuity approach (Brigham and Ehrhardt (2002)). Of these, the replacement chain approach was most commonly used, as it can be calculated using spreadsheet programs (Brigham and Ehrhardt (2002)), and can apply to cases where the costs associated with the replacement projects vary from the initial project (Clark et al. (1989)). This method obtains equal lives by adding replacement projects to the end of the initial project. For example, a one project (A) has a life of 4 years, but project B has a life of just 2 years. To obtain an equivalent life for project B, a replacement project B of 2 years life is added to the end of the original project B. Thus, it can be assumed that two project Bs are necessary for every project A, and the system capital cost will be the sum of A's and the two B's capital costs. Thus, multiplying each subsystem's capital cost (Equation 3.3.7) by a term representing its equivalent life (λ_s) can account for the effects of non-equivalent lives (see Equation 3.3.9).

$I = \sum_{s=1}^{s_{T}} \left[\lambda_{s} A_{s} (Q_{s})^{\psi_{s}} \right]$

where: I, Q_s, ψ_s , A_s and S_T as in Equation 3.3.7; and λ_s = equivalent life of subsystem s.

Now Equation 3.3.8, using I values from Equation 3.3.9, can estimate the annual capital repayment (C_R) for a system with any number of subsystems.

3.3.3.4 OPERATING COSTS

Operating costs (O & M) include all operating costs, except costs related to the repayment of capital, such as interest and load payments. For example, the operating costs of a fossil fuel power station include: fuels, auxiliary fuels, water, chemicals and other materials, operating, administration and management labour, maintenance materials and labour, contracted services (including sewerage and waste disposal) and automobile fuels. These operating costs exclude taxes. Similarly to capital costs, estimating operating (O & M) can use detailed or simple methods. Detailed methods attempt to obtain real or characteristic values of each individual type of cost for each

86

type of system, while simple methods use factors appropriate for each type of industry (Brennan (1990)). As with capital costs, resources are not available to apply the detailed methods, and this assessment uses the factored approach. There are published factored operating cost estimation techniques for some types of subsystems (see Brennan (1990) for some data and sources).

Similarly to capital costs, obtaining exponents for each subsystem requires correlation of O & M costs (O & M) and some production parameter (P) for existing, similar systems, using the simplified basic approach (see Equation 3.3.10). Many studies, reports, and other sources report O & M and P information. This estimate includes both fixed and variable operating costs, as reports of operating costs normally do not present each of these costs contributors separately.

 $O \& M = B(P)^{\omega}$

Equation 3.3.10

where: O & M = operating costs of similar systems; P = production of similar systems; ω = the exponent for this type of system; and B = correlation constant.

Subsystems, within the system boundary, provide a proportion of their materials, energy and services to other subsystems. For example, a coal mine provides a power station's fuel, both of which are subsystems within the 'life cycle' boundary. Thus, the costs of coal provision must count once only, when estimating the O & M costs of the entire system.

$$O \& M = \sum_{s=1}^{S_{T}} O \& M_{s} - F_{other,s}$$

Equation 3.3.11

where: O & M and $O \& M_s$ = operating and maintenance costs for the system and subsystem s respectively; $F_{other,s} = costs$ for provision of materials, energy and services from other subsystems; and S_T = total subsystems.

For the annualised costs (AC) method (see Section 3.3.3.3), annualised operating costs (O & M) replaces O & My in Equation 3.3.4. Annualised operating costs is the real dollar value of O & M for each year during the study subject's operating life. Thus,

the O & M cost collection should collect O & M costs in real dollars. Therefore, O & M_v in Equation 3.3.4 is equal to O & M.

3.3.3.5 ANNUALISED COSTS

 $AC = O \& M + C_{R}$

costs.

METHODS

The methods presented here can produce indicators of generated wealth and capital requirements for a sustainability assessment of electricity generation systems. These methods can overcome the difficulties inherent in VA. However, AC represents NPV only because of the assumption that positive cash flows (i.e. sold electricity) do not vary between electricity generation systems.

3.3.4 ADDING CAPITAL TO VALUE ADDED

One of the stated exclusions of the value added (VA) measure was capital expenditure. One method for adding capital to VA is economic value added (EVA). EVA is the net operating profit (after tax) minus a capital charge. The capital charge is the capital repaid (i.e. amortised capital charge) times the required rate of return applying for that company (i.e. internal rate of return) (Stern et al. (2001)) (see Equation 3.3.13). The amortised capital charge is determined using sinking fund depreciation, and thus annual capital payments are constant (Stern et al. (2001)). Research and development, advertising and promotions, and employee training are included as a capital charge (Stern et al. (2001)).

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

Equation 3.3.4 shows the net present costs (NPC) for each subsystem. For the annualised costs (AC) method, C_R and O & M replace I_v and O & M_v (Equation 3.3.12).

Equation 3.3.12

where: AC = annualised costs; O & M = operating costs; and C_R = annualised capital

3.3.3.6 SUMMARY OF NET PRESENT VALUE AND INTERNAL RATE OF RETURN

 $EVA = \sum_{i=1}^{n} \sum_{s=1}^{S} [NOPAT_{s,i} - (C_{s,i}.CA_{s,i})]$

Equation 3.3.13

where: NOPAT_{s1} = net operating profit before interest but after tax from subsystem s in year I (AUD 1999); $C_{s,l} = cost$ of capital for subsystem s in year I (%); $CA_{s,l} = capital$ expended by subsystem s in year l; n = life of system (years); and $S_T = subsystems$ in system ******.

EVA proponents claim it to estimate the true economic profit and the creation of shareholder value, better than any other existing measure, and be easily measurable and understandable (Azapagic and Perdan (2000)). Yet, it has also been criticised as too simplistic, misrepresenting the true financial situation, and containing subjective assumptions (Azapagic and Perdan (2000)). EVA calculates over a single year period, and thus is dependent on annual capital investment decisions (Azapagic and Perdan (2000)). Thus, reducing capital expenditure in a year, and increasing it in others, enables manipulation of EVA, for example to influence share prices for personal gain. Some assumptions made during the calculation also distort EVA, such as the distribution of research and development capital (Azapagic and Perdan (2000)). Such flaws have led to the development of similar, alternative measures (Azapagic and Perdan (2000)). Some claim that the cause of many of these difficulties is the use of different techniques to estimate the capital charge for EVA (Richards (1998)).

Subtracting the annualised capital repayment (C_R) (Equation 3.3.8) from VA (Equation 3.3.2) produces an analogous indicator to EVA (Equation 3.3.14). This indicator, given the name capital inclusive value added (CVA), will approximate an EVA indicator, but without many of its flaws. For example, the C_R value is an average capital expenditure over many years, and manipulation on a yearly basis is difficult.

 $CVA = VA - C_{R}$

Equation 3.3.14

where: CVA = capital inclusive value added; VA = value added; and $C_R =$ annualised

Sustainability of Australia's Electricity Generation Chapter 3: Methods for Producing Sustainability Indicators

capital costs.

3.3.5 SUMMARY OF METHODS FOR PRODUCT IG ECONOMIC INDICATORS

The methods presented here can produce the required indicators of economic sustainability for a sustainability assessment of electricity generation systems. It examined several methods for producing the wealth generation indicator: value added (VA), annualised costs (AC), and capital value added (CVA); and a capital requirements indicator (I).

The VA method is an incomplete indicator as it neglects the effects of capital costs on wealth generation. The AC and CVA methods include capital costs and thus overcome this difficulty. Thus, a comparison of the VA and AC and CVA indicators may establish if the inclusion of capital costs is essential to indicate wealth generation.

3.4 METHODS FOR PRODUCING SOCIAL INDICATORS

Chapter 2 identified employment, and health and safety as the most important social indicators of sustainability for electricity generation. Thus, this section, examines methods for the calculation of these indicators. Section 3.4.2 examines methods for developing the employment indicator. Section 3.4.3 examines methods for developing the health and safety indicator.

3.4.2 EMPLOYMENT

The total employment due to the activities of one system is difficult to estimate. Direct employment in the system is the sum of the employees actively working in each subsystem (see Equation 3.4.1).

 $ED = \sum_{s=1}^{s_T} ED_s$

3.4.1 OVERVIEW OF METHODS FOR PRODUCING SOCIAL INDICATORS

This equation is from Azapagic and Perdan (2000).

where: ED and ED_s = direct employment in the system and subsystem s respectively; and $S_T =$ subsystems in system.

The actual employment generated by the system will be greater than the direct employment. The total employment generated will include temporary staff, employment associated with purchased materials and services, and employment generated through the spending of direct employees. Estimation of this indirect employment uses employment multipliers, produced from input output tables (ABS (2001)). Generation of these multipliers assumes linear relationships between output and employment, and the outputs of different industries (ABS (1994)). For example, if an industry (A) increases its output (by say 100 %), demand for outputs from other industries (including B) will also increase (by say 50 %). These multipliers assume that output and employment in these industries will increase linearly to match that demand (i.e. B's output and employment would increase by 50 %). Different types of multipliers exist, depending on the amount of additional employment they include. The types of multipliers of most interest are:

- Type 1A multipliers include all additional employment caused by the employment of one person in that industry from other industries whose output is required for that industry.
- Type 1B multipliers include Type 1A employment. It also includes the additional employment required for all industries that provides outputs for the other industries (of Type 1A indicators) and for these industries, and so on. It includes all additional employment in all industries caused by the employment of one person in the subject industry.
- Type 2A multipliers include Type 1B employment and employment caused through the spending of the employees included in the Type 1B multiplier. This indicator includes the original increase of one person.
- Type 2B multipliers are Type 2A multipliers minus the original increase of one person. Thus, 2B represents the industry wide additional increase in employment when employing one person in an industry.

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

Estimation of indirect employment first requires classification of direct employment into industry, using the classifications provided with the multipliers (ABS (2001)). Application of the multiplier appropriate for that industry (EM) converts the classified direct employment into indirect employment. The sum of these indirect employees is the total indirect employment (Equation 3.4.2).

$$EI = \sum_{s=1}^{S_T} \sum_{c=1}^{C_T} EM_c \ x \ ED_{s,c}$$

3.4.3 EMPLOYEE HEALTH AND SAFETY

The most common safety indicators for industries are lost time injuries (LTI) and fatalities. LTI are illnesses and injuries, which cause an absence of at least one day. Many companies report the number of LTI or a LTI frequency (LTIF, LTI per million work hours) as part of their Operational Health and Safety schemes. Equation 3.4.3 shows how to convert from LTIF to LTI.

LTIF = LTI x ED x WP

employee).

indicators.

$$LTI = \sum_{s=1}^{S_T} LTI_s$$

Equation 3.4.2

where: EI = indirect employment for the system, $EM_c =$ employment multiplier for industry classification c, ED_{sc} = direct employment in industry classification c in subsystem s; S_T = subsystems in system, and C_T = total industry classifications.

Equation 3.4.3

where: ED = direct employment; and WP = worker productivity (work hours per direct

Equation 3.4.4 and Equation 3.4.5 show how to produce the system LTI and fatalities

 $Fatal = \sum_{s=1}^{S_T} Fatal_s$

Equation 3.4.5

where: LTI and LTI_s = LTI for the system and subsystem s respectively; Fatal and Fatal_s = fatalities for the system and subsystem s respectively; and S_T = the total number of subsystems.

LTI and fatalities indicators include immediate health impacts, such as illness and diseases that occur during work, but fail to recognise those that occur later in life. The LTI indicator is an imperfect indicator of safety as it neglects injuries that result in absences of less than one day. It is thus possible for a task to cause innumerable small injuries, without contributing the LTI indicator. They also neglect such injuries to non-employees. The toxicity indicator of environmental sustainability will account for some LTIs and fatalities, caused by material emissions, but neglect injuries to non-employees from transport vehicle accidents. Thus, they are low estimates of the ultimate health and safety impact of systems.

3.4.4 SUMMARY OF METHODS FOR PRODUCING SOCIAL INDICATORS

The methods presented here can produce the required indicators of social sustainability for a sustainability assessment of electricity generation systems. Examined are methods for producing employment indicators (ED and El), and health and safety indicators (LTI and Fatal).

3.5 PRODUCING INDICATORS OF SUSTAINABILITY WITH A 'LIFE-CYCLE' PERSPECTIVE

Of the indicator methods, only the LCA method (Section 3.2.3) was developed to have boundaries with a 'life-cycle' perspective. Thus, for the economic and social indicators to have 'life-cycle' boundaries a number of modifications to their methods are

Sustainability of Australia's Electricity Generation_Chapter 3:Methods for Producing Sustainability Indicators

necessary. This section examines whether the components that allow the LCA method to produce indicators with this perspective may apply to the economic and social indicators of sustainability (economic and social).

Allocation distributes the environmental impact causing materials between the individual products of a subsystem (see Section 3.2.3.4). If the studied system has subsystems with multiple products, it will be necessary to allocate all types of impact (environmental, economic and social) between the products. It may be necessary when allocating economic and social impacts to use a different basis than for environmental impacts.

Sensitivity analysis estimates the influence of key decisions, such as the choice of allocation methods, on the indicators, by showing the outcome from each probable choice (see Section 3.2.3.6). Detailed economic assessments often use this type of analysis to test the influence of uncertain parameters in their models, such as future interest rates and inflation. Estimation of the influence of decisions, such as the choice of basis for allocation, on the indicators of sustainability will help to prove their robustness. Therefore, sensitivity analysis will be of benefit for all types of indicators.

Uncertainty analysis estimates the influence of data uncertainties on the indicators (see Sections 3.2.3.6 and 3.2.3.7). Detailed economic assessments sometimes use this type of analysis to test the influence of uncertainties in base data. Uncertainties are present in all types of data, and thus uncertainty assessment is valid for all types of indicators of sustainability. However, there is no standardised method for uncertainty assessment of systems with a 'life-cycle' perspective, and thus application of uncertainty analysis for this case is limited to the environmental indicators.

Normalisation is an optional step in LCA, which divides each of the study subject's indicators by a corresponding indicator for an entire region (i.e. the world or Australia) (see Section 3.2.3.5). This aids in understanding the relative importance and magnitude of the indicators produced by an LCA (Guinée *et al.* (2001)). Normalisation may thus allow a similar understanding of relative importance and magnitude between all types of indicators of sustainability.

95

3.6 SUMMARY **METHODS O**F FOR PRODUCING SUSTAINABILITY INDICATORS

Methods shown here can produce values for each of the indicators listed in Chapter 2. Table 3.6.1 shows the methods selected to obtain values for the indicators listed in Table 2.5.1 (Chapter 2).

While the environmental method (LCA) assesses impacts using the 'life-cycle' perspective, equivalent methods are not available for economic and social indicators. Therefore, the assessment will compare several economic and social indicator methods for each indicator. The use of some of elements of the LCA method may aid the adaptation of the economic and social indicator methods to the 'life-cycle' perspective. Of these, the allocation procedure allows distribution of impacts from subsystems if they have multiple products. Other elements, such as the uncertainty and sensitivity analyses, allow consistent reporting of the indicators robustness to assumptions and data uncertainties. Lastly, the normalisation procedure enhances understanding of the relative importance and magnitude of the indicators.

Consensus is still lacking in several areas of application of the LCA method. The most glaring examples are the lack of accepted methods for characterising the impact of resource depletion and for uncertainty analysis. Therefore, it would seem that some examination of the effect of these problems on the environmental indicators is necessary to ensure wide acceptance of the conclusions from this study.

Table 3.6.1: List of equations necessary to estimate each of the indicators of sustainability. The indicators are from Table 2.5.1 (Chapter 2). Column A contains the abbreviations for each impact.

Environmental Indicators								
	Name	A*`.	Е	quations				
Recourse	CML method	RD	Equation 3.2.1	Fountion 3.2.4.&				
Depletion	Energy depletion	EN	Equation 3.2.2	Equation 3.2.5				
Depiction	Exergy destruction	EX	Equation 3.2.2	Equation 5.2.5				
Clin	nate Change	CC						
Ac	idification	AD						
Eut	rophication	EU						
Photoc	chemical Smog	PS						
Hun	nan Toxicity	HT	Equation 3.2	2.4 & Equation 3.2.5				
Ec	o-Toxicity	ET						
Sc	olid Waste	SW						
P	articulates	PM	and a company of the second					
Water	Consumption	WD						
	Econo	mie Ind	dicators					
Wealth	Value Added	VA	Eq	uation 3.3.3				
Generation	Annualised Cost	AC	Equation 3.3.12					
Generation	Capital Inclusive VA	CVA	Equation 3.3.14					
Capital	Capital	I	Eq	uation 3.3.9				
Requirements	Annualised Capital	CR	Eq	uation 3.3.8				
	Soci	al Indi	cators					
Employees	Direct	ED	Equ	uation 3.4.1				
Employees	Indirect	EI	Eq	uation 3.4.2				
Health and	Lost Time Injuries	LTI	Equation 3.4	4.3 & Equation 3.4.4				
Safety	Fatalities	F	Eq	uation 3.4.5				

Sustainability of Australia's Electricity Generation Chapter 3:Methods for Producing Sustainability Indicators

97

4. ELECTRICITY GENERATION SYSTEMS AND DATA

'Access to electricity services has proven to enhance economic development and social welfare. For example, electrification of rural areas in developing countries contributes to a better distribution of employment opportunities and a more equitable access to health and education services, as well as improving the overall standard of living.'

INTERNATIONAL ATOMIC ENERGY AGENCY (IAEA) (2000), PG. 9.

4.1 OVERVIEW OF ELECTRICITY GENERATION SYSTEMS AND DATA

The previous two chapters developed a methodology and practical methods for sustainability assessment using a set of indicators. These practical methods require detailed information about the system to which they are applied. The methods for producing environmental indicators require information about material consumption and emissions. The methods for economic indicators require information about the costs of construction and operation. Finally, the methods for social indicators require information about the number of employees and rates of injuries and illnesses. This chapter details this information for Australian electricity generation systems.

Section 3.2.3.4 (Chapter 3) establishes that a methodology for data collection is necessary for a life cycle assessment (LCA) before data collection. LCA is the method proposed for the production of the environmental indicators of sustainability. Section 4.2 provides details of this methodology for data collection and a justification for using a similar methodology for the economic and social indicators. Section 4.3 describes some important, operating electricity generation systems which consume Australian fossil fuel, as well as some proposed alternatives. Additionally, Section 4.3 details the

Chapter 4: Electricity Generation Systems and Data

Sustainability of Australia's Electricity Generation

Chapter 4: Electricity Generation Systems and Data

collected data and operations necessary to produce a whole-of-system data set and sustainability indicators for each electricity generation system. During data collection, it is often necessary to make choices and assumptions in the presence of uncertainty. Similarly, it is often necessary to use data where there is a degree of uncertainty as to its accuracy. Section 4.4, details how to apply the uncertainty and sensitivity analyses to estimate the affect of such uncertainties on the sustainability indicators.

4.2 METHODOLOGY FOR DATA COLLECTION

4.2.1 OVERVIEW OF METHODOLOGY FOR DATA COLLECTION

Utilisation of the methods for obtaining sustainability indicators, described in Chapter 3, requires the collection of various data from the electricity generation systems. Data from these systems may vary with employed technology and operational experience. Furthermore, data collection may employ different techniques, i.e. direct measurement (once only or continuous), simulation, or estimation. Consequently, data from each source will have its own inherent accuracy.

ISO 14041 recommends, for an LCA, documentation of the methodology used for data collection. This will increase other's confidence in the conclusions, by ensuring transparency of data inclusion and rejection decisions and assisting duplication the results of the study (ISO 14041). These aims are equally important to the economic and social indicators. Thus, a defined methodology for data collection should exist before data collection begins.

Section 4.2.2 establishes and explains the methodology for data collection. An important decision in data collection is the scope, or system boundary. Section 4.2.3 discusses appropriate system boundaries for the electricity generation systems.

4.2.2 METHODOLOGY FOR DATA COLLECTION

Two procedures are central to data collection: criteria for inclusion; and accuracy requirements. The included data should be relevant to the scope of the assessment: electricity generation systems consuming Australian black coal, brown coal and natural gas. Consequently, the accuracy of data from systems within the scope is greater than

Sustainability of Australia's Electricity Generation

data from without. Use of data from outside the scope is restricted to occasions where data from within the scope is lacking. Where multiple sources for data exist within the scope, the average is preferred. The reference unit for data will be the major output. For example, in coal mining the major output is the mass-rate of coal mined per annum, whereas in electricity generation it is the rate of electricity production per annum.

The accuracy of data should be marked to allow for assessment of the accuracy of the generated sustainability indicator values. Section 3.2.3.6 and Section 3.2.3.7 (Chapter 3) discusses methods for estimating the sensitivity and accuracy of environmental indicators. These methods require pedigree matrix scores (see Table 3.2.3, Chapter 3) and range data (average, minimum and maximum values). These features are also common to economic and social data. Thus, collection of these features for economic and social data will enable the use of these methods for these indicator types. Balances of mass (for environmental indicators), cashflow (economic) and employment (social) ensure consistency with the laws of conservation.

4.2.3 SYSTEM BOUNDARIES FOR DATA COLLECTION

Chapter 2 acknowledges that sustainability assessment requires the use of the 'lifecycle' perspective. Consequently, a fossil fuel electricity generation system, viewed using this perspective, will include all processes from the extraction of the fossil fuel from the ground to the use of the generated electricity (Figure 4.2.1). The 'life cycle' perspective requires that the system should obtain all the materials it consumes, directly from the environment. Thus, a full 'life-cycle' system would include all of the boxes in Figure 4.2.1.

Data for the box labelled 'Materials and Other Fuel Production', which includes the production systems for materials (such as, parts, chemicals and drinking water) and other fuels (such as, petrol), is not available yet for Australian systems and resources for their collection are insufficient to obtain accurate estimations. Thus, the assessment ignores impacts caused by this box.

Detailed investigations (Spath *et al.* (1999), IAEA (2000), BHP Minerals Technology (2001), Gorokhov *et al.* (2002)) have shown that the contribution from the 'Construction and Commissioning' and 'Decommissioning and Rehabilitation' boxes for electricity generation systems, when averaged over the life of the plant (at least 30

Chapter 4:Electricity Generation Systems and Data

Chapter 4: Electricity Generation Systems and Data Sustainability of Australia's Electricity Generation





years), are minor in comparison with operational impacts. During the short construction period (2-4 years), these impacts may be important. However, the assessment ignores environmental and social impacts caused by this box.

The assessment includes economic impacts from 'Construction and Commissioning' in the capital requirements indicator (I, see Section 3.3.3.3, Chapter 3) and from 'Material and Other Fuel Production' in the wealth generation indicators (AC, VA and NPC, see Sections 3.3.2, 3.3.3.5 and 3.3.4, Chapter 3)[†].

Electricity delivery involves high-voltage transmission over cables, conversion to a lower voltage in a substation, and distribution to consumers over more cables. Once delivered, consumers are free to use the electricity as they like. Voltage reduction, distribution and end use are highly differentiated and location specific. Consequently, they are hard to quantify as single measures. The features of these systems are insensitive to the source of the electricity they transport and consume, and thus to the choice of electricity generation technology. Therefore, the impacts they produce will be

Sustainability of Australia's Electricity Generation

identical for all cases and not influence the relative impacts between different systems. The transmission subsystem thus excludes the 'Distribution and End-Use' box.

Figure 4.2.1 shows this simplified perspective, bounded by the system boundary. This simplified perspective is similar to that of the FENCH technique (see Section 2.4.3.2, Chapter 2).

Within the simplified perspective's system boundary, Figure 4.2.1 shows four subsystems: fuel mining; fuel transport; electricity generation; and electricity transmission. These subsystems represent physical separations, often divisions between company ownership, and are present for each fossil fuel. They are thus convenient subdivisions (subsystems) for each system's description.

4.2.4 SUMMARY OF METHODOLOGY FOR DATA COLLECTION These procedures allow the development of a consistent and accurate data set for use in producing sustainability indicators.

4.3 ELECTRICITY GENERATION SYSTEMS AND DATA

Australia uses many different types of systems for generating electricity from fossil fuels. This section presents a simple description of these systems, and an estimation of their prevalence in Australia. The sustainability assessment method requires information (data) about the impact, on the environment, economy and social welfare, of each of these systems. Rules and limitations are placed on data collection for the reasons given in Section 4.2. This section describes the data collected, indicates its source, and details the manipulations necessary to produce sustainability indicators.

Sections 4.3.2 to 4.3.4 describe and show the collected data for the electricity generation systems: Section 4.3.2 for systems consuming either domestic or export black coal; Section 4.3.3 for systems consuming brown coal; and Section 4.3.4 for systems consuming natural gas or LNG. Information sources comprise public reports of corporations, research organisations, industry organisations, national and state government departments (mainly Australian but including some from other nations,

Chapter 4: Electricity Generation Systems and Data

4.3.1 OVERVIEW OF ELECTRICITY GENERATION SYSTEMS AND DATA

One study found that by including the greenhouse gas emissions of the manufacture of equipment used in the upstream processes, which is necessarily greater for oil and natural gas than for coal, reduced the differences between the climate change impact oil and gas systems (Ogawa and Yoon (1998)).

[†] The included economic impact of consuming materials and other fuels is the purchase price.

Sustainability of Australia's Electricity Generation

Chapter 4: Electricity Generation Systems and Data

notably the U.S.A. and U.K.), and other sources, such as computational process models. Additional data sources include contact with personnel from research, industry, and government organisations. Appendix 1 provides tables of the sources for each data value. A lack of sufficient data in some areas has forced the omission of some data. While Sections 4.3.2 to 4.3.4 report some minor omissions. Section 4.3.5 details omissions relevant to all sections. Each section contains four subsections, which independently detail each subsystems data collection. This data is presented using that subsystem's output as a basis, e.g. coal mining data is presented per mass of coal mined. The sustainability assessment aims to produce indicators of sustainability for the system, and thus needs data with the system's output as a basis, e.g. coal mined per MWh of electricity delivered. Section 4.3.6 presents conversions and other similar data manipulations necessary to produce system indicator scores.

4.3.2 DOMESTIC AND EXPORT BLACK COAL

4.3.2.1 MINING SUBSYSTEM

4.3.2.1.1 Description of the Mining Subsystem

Black coal occurs naturally in many locations within Australia. It normally occurs in sub surface layers. Often multiple layers of coal will occur at the same site, interspersed with other mineral matter. Finding these layers involves geological and mining engineering studies, both airborne and surface, and finally sampling (Blackham (1993)).

Mining is the extraction of black coal from these layers. Mining in Australia uses both underground and open cut methods. In NSW, approximately 60 % of coal is from open-cut mines, with the remainder from underground mines (Lowe (2000)). In Queensland, nearly 90 % of coal is from open cut mines (Department of Mines and Energy (2000)).

There are three types of open cut mines: strip mining, open pits and highwall. Strip mining is the progressive mining of a coal layer, with overburden from above the layer dumped behind the extracted coal (Figure 4.3.1, top). Open pits have multiple benches, do not use draglines, and dump overburden distant from the mining operation (in or outside the mine). Overburden removal is by blasting and removal of spoil piles by draglines (strip) or shovels and dump trucks (open cut) (Aspinall et al. (1993), Kukla et



Figure 4.3.1: Steps in the mining of black coal (adapted from NGGIC (1998b)).

al. (1993)). Coal removal is by drilling and blasting (seams greater than 2 m thick), direct digging or ripping, using shovels, excavators and front end loaders (Kukla et al. (1993)). Coal transport to the mine's stockpile is by haulers or conveying systems. Highwall mining is the horizontal cutting of coal unobtainable by conventional open cut methods. It is not widely used in Australia.

There are two main methods of underground mining: Bord and Pillar, and Longwall (Hedley and McDonald (1993)). Both methods have relatively minor impacts on the surface. Bord and Pillar methods break the coal seam into regular sections. Pillars of coal support the roof material while expanding the mine. Often extraction of these pillars occurs after mining of the section is complete. Longwall mines create very large pillars (longwalls) for extraction. This method often produces coal at a faster rate than Bord and Pillar methods. In both cases, transport of coal is by conveying belt systems, which may include shuttle cars (only for some underground mines) and trucks. Ventilation is highly important in underground mining as most coal seams contain carbon dioxide (CO_2) and methane (CH_4) gases. CO_2 gas can cause suffocation. Methane, or even coal dust, can cause fire or explosion in air.

In all cases, coal is stored, in open stockpiles (for large quantities) or bins (for small quantities (up to 6 kt)), before transportation. This enables smoothing of production rates, which reduces the costs of conveying the coal. To ease its transport, crushing of coal may occur at various points.

In both surface and underground mining, the water table may need lowering (by extracting water from below the mine), to ensure the safety of mining and minimise the chance of flooding. Rehabilitation of surface areas to pre-mine condition is a requirement for all mines. It must restore the productivity, ecological integrity, and economic and aesthetic value of the land (Chadwick et al. (1987)).

Black coal production in Australia has, unlike brown coal, two markets; domestic and export. Australia exports more coal than it consumes domestically (around 60 %), which is in sharp contrast to most other countries (Doherty (1993)). Thus, Australian coal producers must be able meet both domestic and export economic and quality demands. Mineral matter reduces the thermal value of coal as it reacts in high temperatures to form ash. Reduction of mineral matter is often by gravity separation techniques, which separate the heavy ash from the light coal. The heavy ash material is around 15 % of the raw coal for underground mines and 25 % for open cut mines (Joint Coal Board (2000)).

Infrastructure for the 'average mine' in this assessment consists of four sets of mining equipment; two each for underground and surface mining. The exact machinery used will depend on the amount of coal produced from each set. In underground mining, the two methods used are Bord and Pillar and Longwall, and for surface mining the two methods are truck and shovel and dragline. Other equipment, necessary for all methods, includes crushing equipment, stockpile spreading, maintenance and reclaiming equipment, transport loader(s), mine water extraction and conveyance equipment, and service and maintenance areas.

4.3.2.1.2 Material Flows

4.3.2.1.2.1 Assumptions

Data in this section refers to black coal mines in New South Wales and Queensland only, since it is in these states that most of Australia's black coal production occurs. It includes only those mines producing coal for the domestic power production industry. The average coal mine produces 2686 kt of coal per annum, of which 1166 kt is for domestic consumption (the remainder is for export) and 24.7 % of total coal production

Sustainability of Australia's Electricity Generation

is from underground mines (Data Ref. C1), While the data excludes material flows due to construction and final rehabilitation (see Section 4.2.3), it includes rehabilitation during normal operation.

4.3.2.1.2.2 Material Requirements

Table 4.3.1 shows the materials used in an average black coal mine. Water (low quality) includes water extracted from subterranean aquifers, as well as from surface sources. This analysis omits a number of minor inputs, for example, chemicals, maintenance materials and lubricating oils for machinery, due to a lack of data.

Table 4.3.1: Material requirements for the mining of 1 Mt of coal from an average black coal mine.

Material	Giyh	Ref.
Electricity	19.7	C2
	kg	Ref.
Coal	1.00×10^9	Basis
Diesel	$1.50 \ge 10^6$	C3
Petrol	1140	C4
Explosives (Ammonium Nitrate)	$1.50 \ge 10^6$	C5
Limestone	4.01 x 10 ⁶	C6
Overburden	6.50 x 10 ⁹	C7
Water (high quality)	45.6	C8
Water (low quality)	7.97 x 10 ⁸	C9

4.3.2.1.2.3 Material Emissions

Table 4.3.2 shows the emissions from an average black coal mine. Mining of overburden allows access to the black coal. However, as mining this overburden is a moving process, rather than a substantial modification, the solid waste value does not include overburden. The methane and carbon dioxide emissions values include emissions, which naturally occur when coal is exposed to air. Emissions from the oxidation of waste coal are not included due to a lack of verified data. Dust emissions due to blasting, movement and transport are negligible in comparison with that caused by coal transport. Thus, the conservative dust emissions estimate in Section 4.3.2.2 will

106

Chapter 4: Electricity Generation Systems and Data

Table 4.3.2: Material emissions from the mining of 1 Mt of coal from an average

black coal mine.

Material	lg	Ref.
Black Coal	1.00 x 10 ⁹	Basis
Overburden	6.50 x 10 ⁹	C10
Water (low quality)	1.21×10^8	C11
Carbon dioxide	4.71 x 10 ⁶	C12
Methane	3.52 x 10 ⁶	C13
NMVOC	7290	C14
N ₂ O	135	C15
NO _x	6.79 x 10 ⁴	C16
со	2.63×10^4	C17
SO ₂	1.38×10^4	C18
Particulates	2800	C19

include mining emissions.

4.3.2.1.3 Economic Flows[‡]

Table 4.3.3 shows the capital costs, I, of open cuts (Equation 4.3.1) and underground mines (Equation 4.3.2) with yearly production capacity, Q. The average mine, which produces 24.7 % of its coal underground, has a capital cost as shown in Equation 4.3.3 (Data Ref. C1).

Table 4.3.3: Correlation parameters for the capital costs of black coal mining.

 Equation	A	Ψ	Range 1 (\$ million (AUD 2000))	O (Mitna)	Data Rel
Equation 4.3.1	158	0.690	50.4 - 1080	0.3 - 15	B1
Equation 4.3.2	107	0.616	46.6 - 271	1 - 4	Al

 $I = 130(Q)^{0.690} + 90.0(Q)^{0.616}$

Equation 4.3.3

for: Q = 1 to 4 Mtpa and I = \$ 50.4 to 271 million (AUD 2000).

Table 4.3.4 shows the operating costs, O & M, of open cuts (Equation 4.3.4) and underground mines (Equation 4.3.5) with yearly production, P. Thus, the average mine, which produces 24.7 % of its coal underground, has an operating cost as shown in

Sustainability of Australia's Electricity Generation

Equation 4.3.6 (Data Ref. C2). Of these operating costs, some 16 % is due to washing, and thus domestic coal, which is generally unwashed, has an operating cost 84 % of the value determined by the correlation.

Table 4.3.4: Correlation parameters for the operating costs of black coal mining.

Equation	B	(). 	Range O & M (S million (AUD 2000) per annum)	Q (Mtpa)	Data Ref
Equation 4.3.4	24.9	0.966	0.116 - 277	0.00386 - 12.1	B2
Equation 4.3.5	45.1	0.969	0.196 - 215	0.00354 - 5	A2

 $O \& M = 19.0(P)^{0.966} + 34.3(P)^{0.969}$

annum.

4.3.2.1.4 Social Flows

Table 4.3.5 shows the social flows in average black coal mining. LTIF for black coal mines is 26 LTI per 10^6 WH and WH per employee is 2237.

Table 4.3.5: Employment, LTI and fatalities data for an average black coal mine.

Parameter	Value per Mt	Ref.
Employment	85	Cl
LTI	9.7	C2
Fatalities	0.015	<u>C3</u>

4.3.2.2 TRANSPORT SUBSYSTEM

4.3.2.2.1 Description of the Transport Subsystem Transport of coal for use in domestic power stations can be by conveyors, trucks or trains. For long distance transport over land, railways are the most common form of transport. Railway cars have a normal capacity of 75 tonne, and trains of up to 45 cars are becoming common (Horrocks and Gadsden (1993)). Over short distances, transport is by either fleets of trucks or conveyor systems. Trucks most often use dedicated roads, but in some cases, smaller trucks operate on public roads. Conveyor systems rely

Chapter 4: Electricity Generation Systems and Data

Equation 4.3.6

for: P = 0.00386 to 5 Mtpa and O & M = \$ 0.246 to 253 million (AUD 2000) per

^{*} See Sections 3.3.3.3 and 3.3.3.4, Chapter 3 for explanation of I, O & M, Q, P, A, B, ψ and ω .

Chapter 4: Electricity Generation Systems and Data

on long term supply agreements between mines and power stations.

Transport of coal for export is by rail, then bulk sea freighter, and finally rail to the power station stockpile. Trucks or coastal barges transport a small amount to ports in NSW from nearby mines. Port facilities can stockpile up to 2.50 Mt, load up to 10500 tonne/hr, and have annual capacities of up to 28 Mt of coal. Sea transport is by ships of up to 250000 DW'r (dead weight tonnes). Japan, South Korea, and Taiwan consume most of Australia's export black coal, but many other countries consume smaller amounts.

Power stations stockpile coal to ensure the feed to the power station is continuous.

These transport systems have loading stations for each type of vehicle, vehicles, pathways (rails, roads, or conveyor), and transfer and unloading stations.

4.3.2.2.2 Material Flows

4.3.2.2.2.1 Assumptions

The important assumptions in black coal transport are those of distance and the capacity of vehicles. Table 4.3.6 and Table 4.3.7 shows these assumptions for domestic and for export coal respectively. All transport which occurs within the mine boundary is not considered in this analysis, neither is transport associated with the power station.

4.3.2.2.2.2 Material Requirements

Table 4.3.8 shows the materials used in average black coal transport. Diesel consumption in the road and rail transport of coal is dependent on weight, and thus they will consume less fuel when returning to the mine for a new cargo. The calculation assumes that the vehicle only operates between the mine and power station, and always has a full load or is empty. The mining and generation subsystems include water consumption for domestic black coal transport[§]. This analysis omits a number of minor inputs, for example, chemicals, maintenance materials and lubricating oils for machinery, due to a lack of data.

Sustainability of Australia's Electricity Generation

consumed black coal.

Transport	Distance	Ref.	Vehicle	Ref.	% of coal	Ref.
Method	(km) .		Capacity (t)		transported	
Road	10	XI	23 (28)	X2	15.5	X3
Rail	209	X4	5400 (7200)	X5	22,4	X6
Conveyor	5.1	X7	NA		62.1	X8

Table 4.3.7: Transport distances and vehicle capacities for black coal exported to

Japan (Ass. = assumed value, Diff. = by difference).

Transport	Distance	Ref.	Vehiele	Ref.	% of coal	Ref.
Method	(km)		Capacity (1)		transported	
Road	2	Ass.	23 (28)	X2	11.5	Diff.
Rail (Domestic)	209	X9	5400 (7200)	X5 -	88.5	X10
Ocean	8100	X11			100	,
Rail (Foreign)	50	Ass.	5400 (7200)	X5	100	

Table 4.3.8: Material requirements for the transport of 1 Mt of coal from an average black coal mine to an average power station.

Material	Domest GWh	ic ReL	Export GWX	Ref.
Electricity	0.508	Al	5.33	Bl
	kg	Reli	kg	Ref.
Coal	1.00×10^9	Basis	1.48 x 10 ⁹	B2
Diesel	7.27 x 10 ⁵	A2	3.81 x 10 ⁶	B3
Fuel Oil			1.44×10^7	B4
Water (low quality)			2.51×10^8	B5

4.3.2.2.2.3 Material Emissions

Table 4.3.9 shows the material emissions from average black coal transport. Dust emissions are included with particulates from combustion of fossil fuels. Due to a lack of data, it is assumed that 50 % of the black coal lost in transport is lost as dust, and the remainder as solid waste. The wastewater value assumes that no water is lost.

4.3.2.2.3 Economic Flows

Table 4.3.10 shows the capital costs, I, of rail (Equation 4.3.7), road (Equation 4.3.8) and conveyor (Equation 4.3.9) transport, with yearly production capacity, L.Q.

110

Chapter 4: Electricity Generation Systems and Data

Table 4.3.6: Transport distances and vehicle capacities for domestically

[§] Water consuming activities, such as vehicle washing and dust suppression, usually occur at transfer points located at the mine and power station.

Sustainability of Australia's Electricity Generation

Table 4.3.9: Material emissions from the transport of 1 Mt of coal from an

average black coal mine to an average black coal power station.

Material	Domestic		Expor	1
	kg	Ref.	kg	Ref.
Coal	9.98 x 10 ⁸	A3	1.00 x 10 ⁹	Basis
Solid waste	1.10 x 10 ⁵	A4	4.73 x 10 ⁸	B6
Wastewater			2.51 x 10 ⁸	B7
Carbon dioxide	2.27 x 10 ⁶	A5	5.47 x 10 ⁷	B8
Methane	186	A6	4590	B9
NMVOC	3520	A7	5.06×10^4	B10
N₂O	65.3	A8	3290	B11
NO _x	3.28×10^4	A9	9.62×10^5	B12
со	$1.27 \ge 10^4$	A10	1.74 x 10 ⁵	B13
SO_2	6660	A11	8.91 x 10 ⁵	B14
Particulates	$1.12 \ge 10^{5}$	A12	1.17 x 10 ⁶	B15

Equation 4.3.10 (ports) and Equation 4.3.11 (shipping) show the capital costs of sea transport (Data Ref. A4). The combined capital costs are factored estimates using the proportions in Table 4.3.6 and Table 4.3.7, and Equation 4.3.12 for domestic and Equation 4.3.13 for export black coal transport.

Table	4.3.10:	Correlation	parameters fo	r the capital costs of	f black coal transport.

Equation	×Λ	Ψ	Range		Data Ref
		<u>.</u>	I (S million (AUD 2000))	L.Q (km.Mipa)	
Equation 4.3.7	2.52	0.806	181 - 3900	200 to 9019	Al
Equation 4.3.8	1.21	0.680	0.763 - 22.3	0.511 - 73	A2
Equation 4.3.9	1.91	0.889	1.06 - 673	0.511 - 730	A3

 $I = 33.7(Q)^{0.881}$

Equation 4.3.10

for: Q = 0.8 to 22.6 Mtpa and I =\$ 27.7 to 525 million (AUD 2000).

I = 1.46 + 0.0036 (L.Q)

Equation 4.3.11

for: L.Q = 10327 to 21243 km. Mtpa and I = \$ 39.7 to 78.9 million (AUD 2000).

 $l = 0.339 (L_{Road} .Q)^{0.680} + 0.755 (L_{Roil} .Q)^{0.806} + 1.26 (L_{Conv} .Q)^{0.889}$

$$I = 0.276 (L_{Road} .Q)^{0.680} + 2.29($$

+ 44.9(Q)^{0.881} + 2.52(L_{Ra}

Table 4.3.11 shows the operating costs, O & M, of rail (Equation 4.3.14), road (Equation 4.3.15), conveyor (Equation 4.3.16) and sea (Equation 4.3.17) transport, with vearly production, L.O. The combined operating costs are factored estimates; Equation 4.3.18 for domestic and Equation 4.3.19 for export black coal transport.

Equation	B	e m	Range		Data
			O & M (S million (AUD	Ĺ.P	Ref
			2000) per annum)	(km.Mtpa)	, 1997 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 199
Equation 4.3.14	1.64	0.621	44.0 - 2200	200 - 109000	A5
Equation 4.3.15	0.182	0.796	0.107 - 5.54	0.511 - 73	A6
Equation 4.3.16	3.24	0.588	2,19 - 190	0.511 - 1010	A7
Equation 4.3.17	0.155	0.388	5.43 - 739	9580 - 21200	A8

$$O \& M = 0.0412 (L_{Road} .P)^{0.796}$$

$$O \& M = 0.0325 (L_{Road} .P)^{0.796} + 1.64 (L_{Rait 2} .P)^{0.621}$$

4.3.2.2.4 Social Flows

Table 4.3.12 shows the social flows in average black coal transport. LTIF for road transport is 22.6 LTl per 10⁶ WH and WH per employee is 2255. LTIF for rail transport is 17.4 LTI per 10⁶ WH and WH per employee is 1931. LTIF for ship transport is 7.8 LTI per 10⁶ WH and WH per employee is 1949. This estimate excludes port employees and their LTI and fatalities.

Chapter 4: Electricity Generation Systems and Data

Equation 4.3.12

 $(L_{Ruit1}.Q)^{0.806} + 1.46 + 0.0036 (L_{Ocean}.Q)$ $(Q_{ail 2}, Q)^{0.806}$

Equation 4.3.13

Table 4.3.11: Correlation parameters for the operating costs of black coal mining.

+ 0.647 $(L_{Rail}.P)^{0.621}$ + 2.45 $(L_{Conv}.P)^{0.588}$

Equation 4.3.18

 $+1.52(L_{Rail1}.P)^{0.621}+0.155(L_{Ocean}.P)^{0.388}$

 Table 4.3.12: Employment, LTI and fatalities data for average black coal transport.

Parameter	Domestie Value per Mf	c Ref.	Export Value per Mt	Ref
Employment	27.2	Al	105	B1
LTI	1.05	A2	3.85	B2
Fatalities	0.0053	A3	0.0083	B3

4.3.2.3 GENERATION SUBSYSTEM

4.3.2.3.1 Description of the Generation Subsystem

Steam turbine systems are the most widely used technology for electricity generation from black coal (Figure 4.3.2). This technology uses steam to turn the blades of a steam turbine, and a generator converts this rotational energy into electricity.

At steam turbine plants, an amount of black coal is stored. Before consumption of the fuel, pre-heating and milling occurs. Transport of the milled fuel into the boiler is by the air needed to combust the fuel. The boiler uses a particular arrangement of fuel entry points to maximise steam generation, and minimise the amount of nitrogen in the air converted to NO_x . Water pipes form the walls of the boiler. The heat generated by





Sustainability of Australia's Electricity Generation ____ Chapter 4: Electricity Generation Systems and Data

fuel combustion converts the water within the pipes to steam. Further pipes in the top of the boiler, heat the steam to higher temperatures (superheating). Superheating dramatically improves both the amount of heat extracted from the boiler and the amount of electricity generated per unit of fuel consumed. Mineral matter in the coal becomes ash on combustion. Large ash particles collect on the boiler walls and floor. Removal of ash from the floor is either continuous or periodic. Removal of ash from the walls requires the boiler to be not operating. Collection of smaller ash p_{eat} ticles from the flue gas uses bag filters or electrostatic precipitators (Woodruff *et al.* (1998)). Some ash is sold, but most becomes landfill. Combustion converts much of the sulphur and nitrogen in the coal into SO₂ and NO_x. Control of these emissions occurs in other countries (Lavely and Ferguson (1996)), but not in Australia because of the low sulphur and nitrogen levels in Australia coals. After removal of ash particles the gas created by the fuel combustion ('flue gas') passes through a fan (used to ensure a constant flow of flue gas) and out to the environment through a stack.

The water used in the boiler and turbine must be very pure, to reduce corrosion, in the boiler and steam system and is thus heavily treated. The source of feed water for the treatment plant is municipal drinking water supplies ('high quality'). After it has left the turbine, the steam is condensed and recycled to the boiler. Two types of condenser system are prevalent in Australia: once-through and circulating. In the once-through system, water from a large reservoir (usually a lake or ocean) passes through the condenser and returns to the reservoir. In the circulating system, water circulates between the condenser and a cooling tower, where air reduces the temperature of the cooling water resulting in the evaporation of much of the water. Water from an external reservoir (often a river) replaces this evaporated water. These condenser systems use large quantities of lower-quality water.

The electricity generation industry in Australia has grown in a fragmentary manner. Until the 1990's, electricity provision was the responsibility of state governments, who developed their own state electricity supply network. This fragmentary development has ensured that most existing power stations have different designs, even though they all use the steam turbine technology. Consequently, a wide range of electricity generation efficiencies (electrical energy exported from the power station divided by the energy consumed as fuel) occurs. Age and steam conditions have important influences on the electricity generation efficiency of steam plants. Newer plant incorporates

114

design features found to improve older plants. Higher temperature and pressure steam increases the efficiency of steam turbines, as it allows greater energy extraction at the turbine (for example, see Hunt (1991)). The maximum efficiency obtainable from steam turbine systems is in the high 30's, but the average efficiency in Australia is around 34 %.

Some proposed technologies promise to increase efficiencies above this maximum. One of the most promising technologies is the integrated gasification, combined cycle (IGCC) technology (Figure 4.3.3). In this technology, a gasifier converts the solid coal into a combustible gas by partial oxidation (CO, CH₄, H₂, plus other non-reactive gases). Removal of the sulphur and other environmental contaminants in this gas can then occur before combustion. As the volume of this combustible gas is much less than the flue gas, this removal is easier before combustion. A combined cycle system, much the same as that used for natural gas (see Figure 4.3.7), consumes the cleaned gas. These systems consist of a combustor, a gas turbine, a heat recovery steam generator (HRSG), and a steam turbine. The gas turbine uses the heat and momentum of the product gas from the combustor to turn the blades of a turbine. A generator produces electricity from the rotational energy of the turbine. The HRSG and steam turbine operate as a steam turbine system. Some plants in Europe and North America have demonstrated the IGCC technology (i.e. Buggenum (NI.) and Pinon Pine (U.S.A.)).





Sustainability of Australia's Electricity Generation

Many variations exist for the IGCC technology.

4.3.2.3.2 Material Flows

4.3.2.3.2.1 Assumptions

The steam turbine technology data estimates are averages of all New South Wales and Queensland black coal power stations. The basis for the IGCC technology data estimates is a report produced by the proponent (DOE (U.S.A.) (2000)). The bases for this choice are:

- and
- process simulation software.

4.3.2.3.2.2 Material Requirements

Table 4.3.13 shows the materials used in an average power station. The auxiliary fuel is fuel oil for steam turbines, but in practice some plants use natural gas. Coke breeze and natural gas are the auxiliary fuels for the IGCC technology.

4.3.2.3.2.3 Material Emissions

Table 4.3.14 shows the material emissions from an average power station. Some power stations have begun recycling and treating all water and have zero water emissions (Delta electricity (1999)). Consumption of ash for other use is around 25 % of total ash production. Oily waste production equals oil and lubricant consumption.

4,3,2.3.3 Economic Flows

Table 4.3.15 shows the capital costs, I, of steam turbine (Equation 4.3.20) and IGCC technologies (Equation 4.3.21), with production capacity, Q. These costs assume that the IGCC technology is mature. The first IGCC system is likely to have considerably

Chapter 4: Electricity Generation Systems and Data

• the technology offers the greatest improvement in electricity generation efficiency; • the source provides detailed design and operating information about the technology;

 the proponent, the Department on Energy (U.S.A.), is of high standing. The reported performance and emissions for the IGCC are verified using ProSim[©]

Sustainability of Australia's Electricity Generation

Table 4.3.13: Material requirements for the generation of 1 net GWh of electricity from an average black coal power station.

Material	🗢 SI		IGCC		
	Ġ₩ĥ	Ref,	. Gwh	Ref.	
Electricity	0.0567	Al	0.0409	Bl	
	· kg	Ref.	kg	Ref.	
Coal	4.49 x 10 ⁵	A2	2.52 x 10 ⁵	B2	
Natural Gas			7340	B3	
Diesel	9.95	A3	10.6	B4	
Petrol	10.9	A4	11.6	B5	
Fuel Oil	302	A5		, I	
Coke Breeze			50	B6	
Limestone			3.35 x 10 ⁴	B7	
Sulfuric Acid	65	A6	76.8	B8	
Caustic Soda	45	A7	218	B9	
Ammonia	7.5	A8	2.14	B10	
Chlorine	6	A9	1.71	B11	
Hydrazine	0.2	A10	0.057	B12	
Aluminium Sulfate (SG1.3)	1.5	A11	0.427	B13	
Bromide	0.75	A12	0.214	B14	
Ferriclear	0.5	A13	0.142	B15	
Hydrochloric Acid	0.2	A14	0.057	B16	
Antifoam	0.2	AI5	0.057	B17	
Lubricating Oil	1.7	A16	1.7	B18	
Water (high quality)	1.97 x 10 ⁵	A17	4.60 x 10 ⁵	B19	
Water (low quality)	1.78 x 10 ⁶	A18	6.24 x 10 ⁵	B20	

greater costs.

Table 4.3.16 shows the operating costs, O & M, of steam turbine (Equation 4.3.22) and IGCC (Equation 4.3.23) technologies, with yearly production, P.

4.3.2.3.4 Social Flows

Table 4.3.17 shows the social flows in average black coal generation. LTIF for steam turbine technology is 7.6 LTI per 10⁶ WH and WH per employee is 1373. LTIF for IGCC technology is not available, but an estimate is 17.4 LTI per 10⁶ WH. An estimate of WH per employee for IGCC systems is 1931. There are no reports of fatalities in Australian black coal generation facilities. In the absence of an existing system, LTIF rates for IGCC systems are not available. As these systems hold no greater danger in

Table 4.3.14: Material emissions from the generation of 1 net GWh of electricity in

an average black coal power station.

Material	ST		- IGCC	
	GWI	Ref.	GWh	Ref
Electricity	1.06	A19	1.04	B21
	i kg	Ref.	kg	Ref.
Solid Waste		ļ	3.51×10^4	B22
Ash	8.41 x 10 ⁴	A20	2.58×10^4	B23
Evaporation	1.51×10^6	A21	4.71 x 10 ⁵	B24
Wastewater	3.39×10^{5}	A22	5.13 x 10 ⁵	B25
Sewerage	0.0273	A23	9.0257	B26
Saline water	3.15×10^{5}	A24	1.01×10^5	B27
Carbon dioxide	9.83 x 10 ⁵	A25	6.34 x 10 ⁵	B28
Methane	9.85	A26	30.2	B29
NMVOC	18.4	A27	0.721	B30
N ₂ O	8.64	A28	9.15	B31
NO _x	3200	A29	156	B32
со	119	A30	149	B33
SO2	4010	A31	50.7	B34
Particulates	477	A32	7.18	B35

Equation	A	ψ	Ran I (S million (AUD 2000))	ge Q (MW)	Data Ref
Equation +.3.20	12.5	0.663	168 - 2410	50 - 2800	Al
Equation 4.3.21	22.7	0.615	<u>430 - 891</u>	467 - 1090	B1

generation.

Équation	∽в	Ð	Range O & M (5 million (AUD 2000) per annum)	P (GWlipa)	Data Ref
Equation 4.3.22	0.359	0.573	25.7 - 61.8	296 - 8060	A2
Equation 4.3.23	0.619	0.479	19.7 - 29.6	<u>1373 - 9461</u>	<u>B2</u>

operation, the values determined for conventional systems are equally applicable.

generation.

Table 4.3.15: Correlation parameters for the capital costs of black coal generation.

Table 4.3.16: Correlation parameters for the operating costs of black coal

Table 4.3.17: Employment, LTI and fatalities data for average black coal

Sustainability of Australia's Electricity Generation

Chapter 4: Electricity Generation Systems and Data

Рагатет	Domestic Value per GWh	Ref.	Export Value per GWh	' Ref.
Employment	0.027	Al	0.027	B1
LTI	0.0003	A2	0.0003	B2
Fatalities	0		0	

4.3.2.4 TRANSMISSION SUBSYSTEM

4.3.2.4.1 Description of the Transmission Subsystem

Transmission occurs over high voltage cables on large transmission towers. These systems include substations, management facilities, and maintenance facilities and equipment. Kinhill Sterns (1987) details the materials necessary to construct transmission towers.

Electricity is lost during transmission, due to various factors. An important factor is distance; the greater the distance travelled by the electricity, the greater the losses. In general, black coal electricity generation systems are near major mines, and distant from major population centres. The losses attributable to transport over these distances may be significant.

4.3.2.4.2 Material Flows

4.3.2.4.2.1 Assumptions

The use of electricity from all power stations is almost exclusively in major cities, i.e. Melbourne, Sydney and Brisbane, and thus the distance over which electricity transmission takes place is the average distance between the power stations and the major capital in each state (and territory), weighted by the magnitude of electricity sent over this distance. This assessment assumes transmission loss is independent of load, varying only with transmission distance. Transmission loss is set at 0.75 % electricity loss per 100 km (Evans (1995)).

4.3.2.4.2.2 Material Requirements and Emissions

Table 4.3.18 shows the materials used in average transmission. Transmission consumes

Sustainability of Australia's Electricity Generation

sulphur hexafluorine (SF₆), a major greenhouse gas^{**}, in gas-insulated substations, circuit breakers, and other switchgear (EPA (U.S.A.) (2000)). Consumption or emission data for electricity transmission systems could not be obtained^{††}. Electricity loss estimates require yearly generation figures from power stations, estimates of distances between power stations and distribution networks, as well as estimates of interstate transfers of electricity and the transfer distance. These values allow the estimation of the average transmission distance traversed by electricity generated from black coal in Australia, 180 km. This distance, with the transmission loss assumption (see Section 4.3.2.4.2.1), can then estimate the transmission losses.

of electricity from an average black coal power station.

Material	GWh	Ref.	Material	GWh	Ref.
Electricity	1	Basis	Electricity	0.986	Al

4.3.2.4.3 Economic Flows

Table 4.3.19 shows the capital costs, I/L, with transmission capacity, Q (Equation 4.3.24), and operating costs, O & M, with transmission productivity, P (Equation 4.3.25) for electricity transmission.

Table 4.3.19: Correlation parameters for the capital and operating costs of electricity transmission.

Equation	. <u>А</u>	. У .	Range I/L (S million (AUD	Q (MW).	Data Ref
Equation 4.3.24	0.0109	0.636	0.338 - 1.4	250 - 4000	Al
Equation	* _/ B	ĨΦ.,	O & M (S million (AUD 2000) per annum)	P (GWhpa)	Data Ref
Equation 4.3.25	0.00333	0.503	5.63 - 107	2.63 - 923	A2

It has the greatest global warming potential (GWP) of any substance, 23 900 kg carbon dioxide (CO₂) per kg emitted (i.e. 23 900 times the effect of CO₂) (EPA (U.S.A.) (2000)) ^{††} 74 tonnes of SF₆ was released in 1997 from all Australian industries (NGGIC (1999)).

Chapter 4: Electricity Generation Systems and Data

Table 4.3.18: Material requirements and emissions for the transmission of 1 GWh

4.3.2.4.4 Social Flows

Table 4.3.20 shows the social flows in average black coal transmission. LTIF for transmission is 5 LTI per 10⁶ WH and WH per employee is 1373. Data generally refers to joint transmission and distribution companies, as there are no extant transmission only companies in Australia.

Table 4.3.20: Employment, LTI and fatalities data for average black coal transmission.

Parameter	Value per km	Ref.
Employment	0.0527	Al
LTI	0.00036	A2
Fatalities	2×10^{-5}	A3

4.3.3 BROWN COAL

4.3.3.1 MINING SUBSYSTEM

4.3.3.1.1 Description of the Mining Subsystem

The brown coal mines of Victoria, Australia are unique, as they are of shallow depth. high thickness, and low hardness. The coal contains a great deal of water (see Figure 4.3.4), in contrast to the lower moisture, harder black coal. There have been only six mines of commercial interest, of which five remain in operation, and four are for electricity production. Export of brown coal does not occur. Other domestic industries,





Sustainability of Australia's Electricity Generation

such as briquette manufacturing, use a small amount of brown coal.

Mining of coal and overburden (waste material above the coal) is continuous, as electricity generation plants require continuous supply and storage of brown coal is difficult^{‡‡}. A bunker maintains 10-16 hours of supply to ensure consistent feed. Overburden removal constitutes one level or bench, while coal extraction occurs on subsequent, lower benches (see Figure 4.3.5). Both the overburden and coal are generally soft and can be removed using coal dredgers ((Holmes (1993)). All material transport uses conveyors, as rail systems cannot handle the vast quantities of material handled^{§§}. Two of the three large mines employ continuous water removal to ensure mine stability, as they operate at great depths.



(Holmes (1993)). Also shows the coal dredgers used to mine brown coal.

Other infrastructure includes overburden conveyors, overburden stacker (used for spreading overburden in the overburden dump), mine water extraction and conveyance equipment, and service and maintenance facilities.

The great size of brown coal seams (at some points thicker than 100 m) results in the creation of massive holes during mining. As the overburden thickness is generally around 10 m, there is not enough material to refill these holes (Holmes (1993)). Consequently, rehabilitation presents greater difficulties than for black coal (see, for example, Woodward-Clyde (1999)).

** Requires vast storage volume, which is costly to provide. Additionally, it is a safety hazard, as drying brown coal is prone to spontaneous combustion. ^{§§} Loy Yang, the largest of the mines, produces 30.5 Mt of coal annually (Loy Yang Power (1999)).

122

Chapter 4: Electricity Generation Systems and Data

Figure 4.3.5: View of the mining face of the Yallourn brown coal mine, Victoria

Chapter 4: Electricity Generation Systems and Data

4.3.3.1.2 Material Flows

4.3.3.1.2.1 Assumptions

An average brown coal mine has the characteristics of the weighted production (mass) average of the existing three Latrobe Valley mines: Yallourn (18 Mtpa), Morwell (17.6 Mtpa), and Loy Yang (30.5 Mtpa) (Yallourn Energy (1999), Hazelwood Power (1999), Loy Yang Power (1999)).

4.3.3.1.2.2 Material Requirements

Table 4.3.21 shows the materials used in an average brown coal mine. Omissions include a number of minor materials, for example, chemicals and maintenance requirements for all equipment.

Table 4.3.21: Material requirements for the mining of 1 Mt of brown coal from an average mine.

Material	GWb	Refe
Electricity	4.25	Al
	kg	Ref.
Brown Coal	1.00×10^9	Basis
Overburden	3.70×10^8	A2
Diesel	6.67×10^4	A3
Petrol	1140	A4
Lubricating Oil	85.7	A5
Water (low quality)	8.52×10^7	A6

4.3.3,1.2.3 Material Emissions

Table 4.3.23 shows the material emissions from an average brown coal mine. Dust emissions when mining coal with moisture content over 10 % the amount are less than 1 g per Mt of raw coal mined (NPI (1999a)). Thus, as brown coal has moisture contents of 60-66 % (see Figure 4.3.4), dust generation is negligible.

4.3.3.1.3 Economic Flows

Table 4.3.22 shows the capital costs, I, with capacity, Q (Equation 4.3.26), and operating costs, O & M, with production, P (Equation 4.3.27) for brown coal mining.

Sustainability of Australia's Electricity Generation

average mine.

Material .	kg.	Ref.
Brown Coal	1.00×10^9	Basis
Overburden	3.70 x 10 ⁸	A7
Oily Waste	881	A8
Water (low quality)	3.12 x 10 ⁸	A9
Carbon dioxide	2.12×10^{5}	A10
Methane	30.9	A11
NMVOC	325	A12
N ₂ O	6.04	A13
NOx	3020	A14
co	1170	A15
SO ₂	612	A16

coal mining.

Equation	Α	Ψ	Range 1 (S million (AUD 2000))	Q (Mtpa)	Daía Ref
Equation 4.3.26	194_	0.536	134 - 1746	0.5 - 60	Al
Equation	B	0	O & M (S million (A)) 2000) per annum)	P (Mtpa)	Data Ref
Equation 4.3.27	13.7	0.593	9.10 - 155	0.5 - 60	A2

4.3.3.1.4 Social Flows

Table 4.3.24 shows the social flows in average brown coal mining. Reports of LTI and fatalities are for the mining, transport, and generation subsystems together (see Section 4.3.3.3.4). One estimate of mining LTIF is 9.4 LTI per 10⁶ WH and indicates no fatalities for at least 8 years (Natural Resources and Environment (2001)).

Table 4.3.24: Employment, LTI and fatalities data for an average brown coal mine.

Parameter Employment

Table 4.3.23: Material emissions from the mining of 1 Mt of brown coal from an

Table 4.3.22: Correlation parameters for the capital and operating costs of brown

Value per Mtpa	Ref.
11.8	<u>A1</u>

125

Chapter 4: Electricity Generation Systems and Data

Sustainability of Australia's Electricity Generation

Table 4.3.26: Material emissions from average mine to power station.

Material Brown Co Solid wast Particulate

4.3.3.2.3 Economic Flows

Table 4.3.27 shows the capital costs, I, with capacity, L.O (Equation 4.3.28), and operating costs, O & M, with production, L.P (Equation 4.3.29) for brown coal transport.

Table 4.3.27: Correlation parameters for the capital and operating costs of brown coal transport.

Equation	A	Ψ.	Range I (S million (AUD 2000))	{0	Data Ref
				(km.Mtpa)	
Equation 4.3.28	19.1	0.5	58 - 1721	1.75 - 3660	Al
Equation .	В	ω.	O & M (Semillion (AUD)	L.P	Data
			2000) per annum)	(km.Mtpa)	Ref
Equation 4.3.29	1.93	0.421	2.44 - 60.9	1.75 - 3660	A2

4.3.3.2.4 Social Flows

Reports of employment, LTI and fatalities for transport are contained in the combined mining, transport, and generation subsystem estimate (see Section 4.3.3.3.4).

4.3.3.3 GENERATION SUBSYSTEM

4.3.3.3.1 Description of the Generation Subsystem As with black coal, electricity generation from brown coal uses steam turbine systems (see Figure 4.3.2). However, brown coal is a very different fuel to black coal (see Figure 4.3.4), which necessitates considerable modification of the standard designs used for black coal steam turbine systems. Firstly, the softer brown coal fuel requires only low-impact fan milling for size reduction. Secondly, the high moisture level (up to 2 tonnes of water per tonne of dry coal) increases the volume of flue gas generated on

4.3.3.2 TRANSPORT SUBSYSTEM

4.3.3.2.1 Description of the Transport Subsystem

Transport of brown coal involves the use of conveyors. Between the mine and power station is the raw coal bunker (see Section 4.3.3.1). Transport of equipment and personnel uses rubber tyred vehicles. Other infrastructure includes motors, conveyor transport equipment, switching stations, dust minimisation equipment, and service and maintenance facilities.

4.3.3.2.2 Material Flows

4.3.3.2.2.1 assumptions

A brown coal conveyor has energy consumption equivalent to a black coal conveyor. The distance traversed by the conveyor is 4 km. Mass lost between the mine and the power station is 0.0005 % of the mass transferred, of which 1 % is dust and the remainder is solid waste.

4.3.3.2.2.2 Material Requirements

Table 4.3.25 shows the materials used in average brown coal transport. This analysis omits a number of material requirements, for example, water, chemicals and maintenance materials.

Table 4.3.25: Material requirements for the transport of 1 Mt of brown coal from an average mine to power station.

Material	GWh	Ref.
Electricity	0.643	Al
	kg	Ref.
Brown Coal	1.00×10^9	Basis

4.3.3.2.2.3 Material Emissions

Table 4.3.26 shows the material emissions from average brown coal transport.

Chapter 4: Electricity Generation Systems and Data

'no	the	trans	port	of	1	Mt of	brown	coal	from	an	
-----	-----	-------	------	----	---	-------	-------	------	------	----	--

	kg 👘 🔨	Ref.
al	1.00 x 108	A2
te	4950	A3
es	50	A4

127

Sustainability of Australia's Electricity Generation

Chapter 4: Electricity Generation Systems and Data

combustion. Thus, to accommodate this extra moisture the size of the boiler is increased. Thirdly, the vaporisation of this higher moisture consumes a greater proportion of the heat generated from fuel combustion, reducing the amount available to generate steam. Consequently, rather than efficiencies of up to the high 30's, efficiencies can only reach the low 30's, resulting in greater emissions of most gases for the same output of electricity. Contrariwise, brown coal systems emit less NO_x than do black coal systems. Considerable NO_x forms in the boiler due to oxidation of N₂ at high temperatures. The high moisture level of brown coal ensures that the combustion temperature is lower for brown coal than black coal. Lastly, electrostatic precipitation is the preferred method of extracting small ash particles, from the flue gases.

Other differences include the use of water to transport ash to dumping sites, and the use of cooling towers for three plants, and once-through lake cooling for the other plant.

Substantially improving the efficiency of brown coal generation requires the reduction of coal water content before combustion. A promising technology for water removal is Mechanical Thermal Expression (or MTE). The MTE technique utilises heat and pressure in combination to force the moisture from the coal as liquid. This can reduce water contents from 60-66 % to 20-30 % moisture content (or 0.75-1.4 kg water removed per kg dried coal), and double the energy density (energy content per mass) of the brown coal fuel (Strauss *et al.* (2001)^{***}). This increases electricity generation efficiency, as less heat is required for water vaporisation. In combination with the IGCC technology, discussed in Section 4.3.2.2.2 (see Figure 4.3.3), some have predicted electrical efficiencies as high as 48 % (McIntosh (2003)). Reducing the moisture in the fuel also results in less water vapour production. Thus, the flue gas has a reduced volume, and the boiler and flue gas parts can be of smaller size. As boiler cost is one of the largest contributors to capital costs, usis size reduction can substantially decrease the capital costs of a power station.

Sustainability of Australia's Electricity Generation _

4.3.3.3.2 Material Flows

4.3.3.3.2.1 Assumptions

The steam turbine technology data estimates are averages of all Victorian brown coal power stations. The basis for the IGCC technology data estimates is a report produced by the proponent (Department of Energy (U.S.A.) (2000)). Section 4.3.2.3.2.1 explains the bases for the choice of this technology for black coal. The brown coal IGCC uses the same technology to ensure comparability between the two fuels. The brown coal IGCC technology also employs the MTE technology (see Section 4.2.3.3). The reported performance and emissions for the IGCC are based on a ProSim[©] process simulation. The basis for data in this section is GWh of electricity generated, rather than GWh of electricity sent out.

4.3,3.3.2.2 Material Requirements

Table 4.3.28 shows the materials used in an average brown coal power station. Lubricating oils consumption estimated from emissions of oily waste.

4.3.3.3.2.3 Material Emissions

Table 4.3.29 shows the materials emitted from an average brown coal power station. Assumed ash density is 900 kg per m^3 . Simple mass balances validate the carbon dioxide and SO₂ emissions results.

4.3.3.3.3 Economic Flows

Table 4.3.30 shows the capital costs, I, of steam turbine (Equation 4.3.32) and IGCC technologies (Equation 4.3.33), with production capacity, Q. The basis of the brown coal IGCC capital and operating cost estimations are reported costs for black coal IGCC systems, modified by replacing the coal preparation costs of such systems with a predicted cost for the MTE coal drying technology (MTE cost estimate from McIntosh (2003)). These costs assume that the technology is mature. The first IGCC system is likely to have considerably greater costs. As the MTE and IGCC technologies for brown coal have yet to be tested, these cost estimates have high uncertainties.

^{***} Additionally many publications of the CRC for Clean Power from Lignite describe the effects of MTE dewatering on efficiencies with ST and IGCC electricity generation technologies (i.e. Kealy *et al.* (2001)).
Table 4.3.28: Material requirements for the generation of 1 GWh of electricity in

 an average brown coal power station.

Material *	ST		JGCC	
/ 1.	GWh	Ref.	GWh	Ref.
Electricity	0.0801	Al	0.0545	BI
	kg	Ref.	kg v	Ref.
Coal	1.39 x 10 ⁶	A2	7.19 x 10 ⁵	B2
Fuel Oil	177	A3		
Briquettes	871	A4		
Natural Gas	13.9	A5	7320	B3
Diesel	0.441	A6	0.446	B4
Petrol	5.59	A7	5.65	B5
LPG	5.01	A8		
Sulfuric Acid	382	A9	334	B6
Caustic Soda	38.1	A10	947	B7
Brine solution (SG 1.2)	63.5	A11		
Aluminium Sulfate (SG1.3)	15.9	A12	1.86	B8
Eliminox	0.68	A13		
Ammonia	0.113	A14	9.3	B9
Trisodium Phosphate	0.793	A15		{
Polymer	13.4	A16		
Anti Scalant treatment	0.732	A17	0.62	B10
Chlorine			7.44	BII
Hydrazine			0.248	B12
Bromide			0.93	B13
Hydrochloric Acid			0.248	B14
Antifoam			0.248	B15
Lubricating Oil	3.85	A18	1.7	B16
Water (low quality)	2.48×10^6	A19	1.40 x 10 ⁵	B17
Water (high quality)	9.25×10^4	A20	4.24 x 10 ⁵	B18

Equation 4.3.30 and Equation 4.3.31 show the operating costs, O & M, for brown coal steam turbine and IGCC power plants, with annual production, P (GWhpa), and capacity, Q (MW) (Data Refs. A2 and B2).

O & M = \$4.51 (P) + \$31.5 (Q)

Equation 4.3.30

Table 4.3.29: Material emissions fro average brown coal power station.

Material	ST .		IGCC	S. S
	GWh	Ref.	GWh	Ref.
Electricity	1.08	A21	1.05	B19
	' kg	Ref.	kg .	Ref
Ash	2.07 x 10 ⁴	A22	6520	B20
Solid Waste			6520	B21
Lubricating Oil	4.39	A23	1.70	B22
Evaporation	1.37 x 10 ⁶	A24	4.15 x 10 ⁵	B23
Wastewater	9.26 x 10 ⁵	A25	4.70 x 10 ⁵	B24
Sewerage	0.0273	A26	0.0265	B25
Saline water	1.30 x 10 ⁵	A27	2.54×10^4	B26
Carbon dioxide	1.15 x 10 ⁶	A28	7.03 x 10 ⁵	B27
Methane	6.09	A29	301	B28
NMVOC	20.6	A30	2.54	B29
N₂O	16.8	A31	9.75	B30
NOx	1590	A32	165	B31
co	210	A33	196	B32
SO2	1810	A34	1840	B33
Particulates	469	A35	20	B34

Table 4.3.30: Correlation parageneration.

Equation	× 4	Ψ	Range		Data Ref
			1 (S million (AUD 2000))	- Q (MW)	
Equation 4.3.32	9.37	0.786	1900 - 6320	1000 - 4000	A1
Equation 4.3.33	10.5	0.663	517 - 941	357 - 883	B1

 $O \& M = 0.743 (P)^{0.479}$

for: P = 1373 to 3203 GWh per annum and O & M = \$ 19.7 to 29.6 million per annum.

4.3.3.3.4 Social Flows

Table 4.3.31 shows the social flows in average brown coal generation. LTIF for steam turbines is 5.6 LTI per 10⁶ WH. Actual WH per employee information for brown coal generation is not available. The value used, 1987 WH per employee, is for brown coal mining (Natural Resources and Environment (2001)). This estimate combines LTI and

130

for: any P and Q.

Chapter 4: Electricity Generation Systems and Data

om	the generation) of 1	GWh of	electricity	in ar	1
----	----------------	--------	--------	-------------	-------	---

Table 4.3.30: Correlation parameters for the capital costs of brown coal

Equation 4.3.31

Sustainability of Australia's Electricity Generation

fatalities for mining, transport, and generation subsystems, and employment for transport and generation. No fatalities in this subsystem was reported.

Table 4.3.31: Employment, LTI and fatalities data for average brown coal electricity generation.

Parameter 👘	Steam Turbin	e	JGCC	
	Value per GWhpa	Ref.	Value per km	Ref.
Employment	0.018	Al	0.018	B1
LTI	0.0038	A2	0.0038	F2
Fatalities	0		0	

4.3.3.4 TRANSMISSION SUBSYSTEM

4.3.3.4.1 Description of the Transmission Subsystem

Section 4.3.2.3.2 discusses electricity transmission. Like black coal, brown coal electricity generation systems are situated near brown coal mines, and distant from major electricity consumers.

4.3.3.4.2 Material Flows

Table 4.3.32 shows the material consumption and emissions for average transmission. The average transmission distance traversed by electricity generated from brown coal in Australia is 279 km. It is greater than for black coal due to greater exports of electricity from Victoria. See Section 4.3.2.4.2.1 for assumptions.

Table 4.3.32: Material requirements and emissions for the transmission of 1 GWh of electricity from an average brown coal power station.

In 💎	- GWh	Ref.	Out	GWh	Ref.
Electricity	1	Basis	Electricity	0.979	Al

4.3.3.4.3 Economic Flows and Social Flows

See Section 4.3.2.4.3 for capital and operating cost data. See Section 4.3.2.4.4 for employment, LTI and fatalities data.

Sustainability of Australia's Electricity Generation

4.3.4 NATURAL GAS AND LNG

4.3.4.1 MINING SUBSYSTEM

4.3.4.1.1 Description of the Mining Subsystem The mining subsystem^{†††} of the natural gas system is much different to that of the coal systems, because natural gas exists in nature as a gas, rather than a solid. Thus, instead of using the crushing and carrying equipment (used in coal processing), the natural gas mining process must use drilling and pumping equipment. Natural gas also exists with other hydrocarbons, such as liquefied petroleum gas (LPG) sources, condensates and crude oils, which have other uses, and water, and gases like hydrogen sulphide (H_2S) , carbon dioxide (CO_2) and nitrogen (N_2) , which must be removed before it can safely be used. The amounts and types of contaminants will vary with well location.

The first requirement is to find the natural gas. Natural gas commonly collects in particular sedimentary rock formations, such as anticlines (upward folds in rock layers) and faults (fractures where layers are broken) (EPA (U.S.A.) (1999)). It can occur both on land and offshore. Searching for natural gas requires the use of geological methods, such as aerial photographs and MAD (Magnetic Anomaly Detection), rock outcrop studies, gravity surveys, remote sensing, core samples, seismic surveys, high-energy common depth point (CDP) and 'wildcat' (test) drilling (EPA (U.S.A.) (1999), Speight (1993)), and exploratory drilling. Rotary drilling is the standard method, because it is rapid, efficient, economical, and can reach depths of 9150 m (Speight (1993)). The drill bit is rotated to chip off pieces of rock, thus increasing the hole depth (EPA (U.S.A.) (1999)). A drilling fluid enters through many hollow pipes, and returns to the surface through the annular space between the hollow pipes and the rock wall (Speight (1993)). Once the well reaches the required depth, it is lined with cemented material called casing (Katz and Lee (1990)). Perforation of the casing allows the gas to enter the well and completes the well (Katz and Lee (1990)). A production system is then developed, usually consisting of a producing well, wellhead, manifold, phase separation equipment, loading facility or transport terminal, and support systems (MMS (1998)).

The second requirement is to extract the hydrocarbons. Gas usually exits spontaneously, as long as there is adequate pressure within the reservoir. Then it passes

ttt Mining refers to what are commonly known as exploration, development, production, and processing.

Sustainability of Australia's Electricity Generation

Chapter_4: Electricity Generation Systems and Data

through phase separation equipment, which also removes any solid impurities, and separates the liquid and gas for transport. Gas transportation is by pipeline to a processing plant, which is usually on land. Gas processing extracts hydrocarbons such as propane and butane for sale, and removes unwanted components such as water, carbon dioxide (CO₂), hydrogen sulphide (H₂S), mercury, and aromatics (or BETX's). The product after cleaning is natural gas.

4.3.4.1.2 Material Flows

4.3.4.1.2.1 Assumptions

Data in this section refers to natural gas production and processing facilities in all Australian states. The basis for much of the included data is APPEA (1999), which includes 98 % of Australia's oil and gas industry. The material consumption and emissions data includes contributions from Australian exploration and development activities (see Section 4.3.4.1.1). It includes both failed and successful exploration. In 1997 there were 376 producing wells, and a production rate of 50423 kt of hydrocarbon (26.1 % natural gas), and thus each producing well produces 134 kt of hydrocarbon, with 35.1 kt of natural gas (AGSO (1998), APPEA (1999)).

4.3.4.1.2.2 Material Requirements

Table 4.3.34 shows the materials used in average natural gas mining. Natural gas consumption includes natural gas for delivery, natural gas consumed internally for heat and electricity, and natural gas flared or leaked. Omission of chemicals is due to difficulties identifying individual chemical use. Information sources state only the combined use, 600 tpa (UKOOA (1999))^{‡‡‡}. Explosives use in seismic surveys is not reported.

4.3.4.1.2.3 Material Emissions

Table 4.3.33 shows the material emissions from average natural gas mining.

Sustainability of Australia's Electricity Generation

Table 4.3.34: Material requirements for the production of 1 kg of mixed petroleum products from an average natural gas mining subsystem.

Material	kġ	Ref.
Natural Gas	0.476	Al
Diesel	4.58 x 10 ⁻³	A2
Petrol	1.31 x 10 ⁻⁶	A3
Aviation Fuel	3.71 x 10 ⁻⁴	A4
Crude Oil	1.66 x 10 ⁻⁴	A5
Drilling Solids	1.15 x 10 ⁻³	A6
Water (low quality)	0.0392	A7

Table 4.3.33: Material emissions from the production of 1 kg of mixed petroleum

products in an average natural gas mining subsystem.

Material	kg	Ref.
Natural Gas	0.396	A8
Crude Oil	1.65 x 10 ⁻⁴	A9
Drilling Fluid Waste	4.28 x 10 ⁻³	A10
Drill Cuttings	2.34×10^{-3}	A11
Produced Sand	1.34×10^{-4}	A12
Solid Waste	1.30 x 10 ⁻⁴	A13
Water (low quality)	0.0361	A14
Sewerage	5.01 x 10 ⁻⁶	A15
Produced Water	0.647	A16
Carbon dioxide	0.336	A17
Methane	3.56 x 10 ⁻³	A18
NMVOC	1.35 x 10 ⁻³	A19
N ₂ O	1.86 x 10 ⁻⁵	A20
NO _x	2.12 x 10 ⁻³	A21
со	5.94 x 10 ⁻⁴	A22
SO ₂	1.95 x 10 ⁻⁴	A23
Particulates	1.02×10^{-4}	A24
H ₂ S	4.19 x 10 ⁻¹¹	A25

4.3.4.1.3 Economic Flows

Table 4.3.35 shows the capital costs, I, of production (Equation 4.3.34) and processing (Equation 4.3.35), with production capacity, Q. Table 4.3.36 shows the operating costs, O & M, of production (Equation 4.3.36) and processing (Equation 4.3.37), with yearly production, P. The costs of LNG processing are included in the transport subsystem (see Section 4.3.4.2.3). Published cost estimates

134

Chapter 4: Electricity Generation Systems and Data

^{***} EPS (Canada) (1989) estimate that oil and gas operators use over 700 different chemicals.

Sustainability of Australia's Electricity Generation Chapter 4: Electricity Generation Systems and Data

for operating natural gas mining facilities are rare, and thus these operating cost estimates have high uncertainties.

Table 4.3.35: Correlation parameters for the capital costs of natural gas mining.

Equation	Å	Ŵ	Range I (S million (AUD 2000))	Q (Mitpa)	Data Ref
Equation 4.3.34	418	0.757	91.3 - 3200	0.134 - 14.7	Al
Equation 4.3.35	96.3	0.881	7.96 - 2010	0.1 - 31.5	B1

Table 4.3.36: Correlation parameters for the operating costs of natural gas mining.

Equation	В	O,	Range O & M (S million (AUD 2000) per annum)	P (Mtpa)	Data Ref
Equation 4.3.36	13.7	1.22	102 - 9280	5.17 - 209	A2
Equation 4.3.37	86.4	1.00	292 - 486	3.38 - 5.62	B2

4.3.4.1.4 Social Flows

Table 4.3.37 shows the social flows in average natural gas and LNG mining. LTIF for oil and gas production is 5.5 and 11.6 LTI per 10⁶ WH for employees and contractors respectively for 1931 WH per employee. Employees allocated between oil, LNG and natural gas using mass production.

4.3.4.2 TRANSPORT SUBSYSTEM

4.3.4.2.1 Description of the Transport Subsystem

Natural gas transport within Australia is exclusively by pipeline. On long pipelines, intermediate, unmanned compressor stations are required to maintain pressure within the pipeline. While natural gas storage is possible, this assessment assumes no storage,

Table 4.3.37: Employment, LTI and fatalities data for average natural gas and LNG mining.

Fatalities	0.011	A3	0.011	B3
LTI	4.61	A2	4.61	B2
Employment	187	A1	166	B1
	Value per Mtpa	Ref.	Value per Mtpa	Ref.
Parameter	Natural Ga	s -	LNG	

Sustainability of Australia's Electricity Generation

as in use currently are only two small storage facilities: Iona, Victoria (underground) and Dandenong, Victoria (LNG).

Liquefaction enables natural gas transport over long distances, where pipeline transport is infeasible. Natural gas does not liquefy until about -161 °C, and thus requires substantial cooling. This cooling process is highly energy intensive. There is currently only one Australian LNG plant (Karratha, WA). LNG can be stored in large quantities before transport. LNG transport from Australia is by sea. LNG transport vessels consume a part of their cargo to power the vessel, and consume fuel oil when empty and on return journeys. Japan and South Korea consume most Australian LNG, but the U.S.A. has consumed some in the past. On delivery, LNG can be stored in large quantities, and vaporised when needed. Vaporisation is by heat exchange with seawater. When the water temperature is too low, the consumption of some of the vaporised LNG may also be necessary (Carlson (1995)). Final transport to power stations is by transmission pipeline.

4.3.4.2.2 Material Flows

4.3.4.2.2.1 Assumptions

The basis for much of the domestic gas transport, material emissions flows data is AGA (1997), which includes 89 % of Australia's gas transmission by flow volume. The material flows assume natural gas delivery to power stations is through high-pressure transmission lines. The average length of these domestic pipelines is 927 km (from AGSO (1998), NGGIC (1998a)^{§§§}). LNG transport is to Japan only, 6800 km. LNG assumes a short (15 km) pipeline delivers vaporised gas to power stations.

4.3.4.2.2.2 Material Requirements

Table 4.3.38 shows the materials used in average natural gas and LNG transport. Water use in pipeline transport is minimal. Natural gas consumption for LNG production is included in the mining subsection. The estimate assumes no supplemental LPG use in LNG transport (LPG may be used to boost the vaporised LNG's energy content where necessary).

§§§ Weighted average based on mass flow through all major pipelines.

Chapter 4: Electricity Generation Systems and Data

_Chapter 4: Electricity Generation Systems and Data

Table 4.3.38: Material requirements for the transport of 1 kg of natural gas or LNG

from an average natural gas mining subsystem to power station.

Material	Pipelin		LNG	
	GWII	Reit	GWb	Ref.
Electricity	6.19 x 10 ⁻¹⁰	Al	1.13 x 10 ⁻⁸	B1
	lg	Ref.	kg	Ref.
Natural Gas	1	Basis	1	Basis
Fuel Oil			0.0348	B2
Diesel	3.94 x 10 ⁻⁵	A2	6.23 x 10 ⁻⁷	B3
Petrol	1.22 x 10 ⁻⁵	A3	3.85 x 10 ⁻⁷	B4
LPG	1.40 x 10 ⁻⁷	A4	4.43 x 10 ⁻⁹	B5
Water (low quality)			17.7	B6

4.3.4.2.2.3 Material Emissions

Table 4.3.39 shows the material emissions from average natural gas and LNG transport.

Table 4.3.39: Material emissions from the transport of 1 kg of natural gas or LNG from an average natural gas mining subsystem to power station.

Material	Pipelin	6	UNG	
	GWb	Ref	GWh	Ref.
Natural Gas	0.989	A17	0.977	B10
Solid Waste			4.02 x 10 ⁻⁵	B11
Wastewater			17.7	B12
Carbon dioxide	0.0304	A18	0.129	B13
Methane	3.86 x 10 ⁻⁴	A19	4.87 x 10 ⁻⁴	B14
NMVOC	7.84 x 10 ⁻⁵	A20	9.85 x 10 ⁻⁵	B15
N₂O	5.55 x 10 ⁻⁸	A21	4.82 x 10 ⁻⁶	B16
NO _x	1.89 x 10 ⁻⁶	A22	3.46 x 10 ⁻⁴	B17
со	2.03 x 10 ⁻⁵	A23	2.59 x 10 ⁻⁵	B18
SO₂	3.70 x 10 ⁻⁷	A24	2.07 x 10 ⁻⁴	B19
Particulates	9.35 x 10 ⁻⁶	A25	2.82 x 10 ⁻⁵	B20
H ₂ S	4.17 x 10 ⁻¹⁰	A26	5.20 x 10 ⁻¹⁰	B21

4.3.4.2.3 Economic Flows

Equation 4.3.38 and Equation 4.3.39 show the capital cost, I, of natural gas and LNG transport, with annual production capacity, L.Q respectively (Data Ref. A1 and B1).

Sustainability of Australia's Electricity Generation

$$I = 2.15(L.Q)^{0.735}$$

$I = 0.622(Q)^{0.854}$	(LN
+ $[114 + 0.0049(L_0)](Q)$	(Oc
+151(Q)	(Va
$+2.15(L_{\rm P}.Q)^{0.735}$	(Pij

2000).

(Data Ref. A2) and Equation 4.3.41 (Data Ref. B2).

$$O \& M = 1.61(L.P)^{0.62}$$

per annum.

 $O \& M = 15.9(L_0.P)^{0.442} + 1.61(L_P.P)^{0.620}$

for: P=2 to 22 Mtpa, $L_0 = 6800$ km (ocean), L_p (pipeline) = 15 km and O & M = \$763 to 6990 million (AUD 2000) per annum.

4.3.4.2.4 Social Flows

Table 4.3.40 shows the social flows in average natural gas and LNG transport. LTIF for pipeline transport is 3.2 LTI per 10⁶ WH for 1931 WH per employee. LTIF for LNG ship transport is 0.48 LTI per 10⁶ WH for 1949 WH per employee. Employment excludes employees in natural gas regasification facilities.

Equation 4.3.38

for: L.Q = 492 to 50435 km.Mtpa and I = \$ 204 to 6120 million (AUD 2000).

IG production)

cean transport)

aporisation)

peline)

Equation 4.3.39

for: Q = 4.8 to 8.3 Mtpa, $L_0 = 6800$ km, $L_p = 15$ km and I =\$ 6.16 to 10.6 billion (AUD

The operating costs, O & M (in \$ million (2000) per annum), of natural gas and LNG transport, with annual transport task, L.Q (km.Mtpa), are shown in Equation 4.3.40

Equation 4.3.40

for: L.P = 34.5 to 208000 km.Mtpa and O & M = \$ 14.4 to 3170 million (AUD 2000)

Equation 4.3.41

Sustainability of Australia's Electricity Generation Chapter 4: Electricity Generation Systems and Data

Table 4.3.40: Employment, LTI and fatalities data for average natural gas and LNG transport.

Parameter	 Steam Turbin 	e	IGCC .	
	"Value per Mipa	Ref.	Value per Mtpå	Ref.
Employment	0.0477 ****	Al	23.7	BI
LTI	0.186	A2	0.315	B2
Fatalities	0.00706	A3	0.00305	B3

4.3.4.3 GENERATION SUBSYSTEM

4.3.4.3.1 Description of the Generation Subsystem

Unlike black and brown coals, no one electricity generation technology dominates for natural gas. A survey, by the author, of currently used technologies in 2000, showed equal installed capacities of steam turbine and open cycle gas turbine technologies in Australia, and little combined cycle gas turbine technology capacity. Additions since 2000 include a large combined cycle gas turbine plant (486 MW), and a number of open cycle gas turbine projects.

Steam turbine technology is similar for natural gap and black coal (see Figure 4.3.2). However, natural gas plants do not store fuel, require no mills, and as they produce no ash, ash removal is not necessary. Efficiencies for these plants are similar to black coal plants (mid to low 30's). This technology was once standard for natural gas, until the development of high efficiency gas turbines in the 1960's reduced their use.

Gas turbine systems produce electricity without the need for steam generation (Figure 4.3.6). They use the heat and momentum of flue gases from fuel combustion to turn the turbine blades. A rotating compressor provides air for combustion at elevated pressure to the combustor. The compressor typically requires 50 % of the power generated (Smith (1996)). Mixing of the air and fuel occurs in a separate combustion chamber, before the turbine. The flue gas immediately enters the turbine. Release of the flue gases to atmosphere occurs after the turbine. Only minimal cooling is required, and thus water consumption is much lower than for steam turbine systems. Their simple design enables them to start generating electricity quicker than other technologies. Consequently, it is the preferred technology for when the demand for

Sustainability of Australia's Electricity Generation

electricity rapidly increases. These systems have a wide range of efficiencies, 17-37 % in Australia, due to rapid advances in their design.





generation technology.

The combined-cycle gas turbine technology is the combination of an open cycle gas turbine with a steam turbine (Figure 4.3.7). The flue gas from the gas turbine provides the heat source to generate steam in a heat recovery steam generator (HRSG). This technology can generate electricity at efficiencies approaching 60 %.



generation technology.

140

Chapter 4: Electricity Generation Systems and Data

Figure 4.3.6: Simplified representation of the open cycle gas turbine electricity

Combustor

Figure 4.3.7: Simplified representation of the combined cycle gas turbine electricity

i41

^{•••••} employees per km.

Sustainability of Australia's Electricity Generation Chapter 4: Electricity Generation Systems and Data

Consumption of LNG occurs after vaporisation, and thus the same technologies (and range of efficiencies) apply as for natural gas.

4.3.4.3.2 Material Flows

4.3.4.3.2.1 Material Requirements

Table 4.3.41 shows the materials used in average natural gas power stations. Minor mass rate air emissions, such as heavy metals (from auxiliary fuel) and large hydrocarbons are omitted from the analysis.

4.3.4.3.2.2 Material Emissions

Table 4.3.42 shows the material emissions from average natural gas power stations.

4.3.4.3.3 Economic Flows

Table 4.3.43 shows the capital costs, I, of steam turbine (Equation 4.3.42), OCGT

Table 4.3.41: Material requirements	for the	production	of 1	GWh	of	electricity
from an average natural gas power sta	ation.					

Material	ST.		. 0061		CCGT /	
	GWh	Ref.	GWh	Ref.	GWh	Ref.
Electricity	0.0471	A1	0.0293	B1	0.0531	Cl
	kg	Ref	kg	Rela	kg	Ref.
Natural Gas	2.01×10^5	A2	2.74 x 10 ⁵	B2	1.40×10^{5}	C2
Fuel Oil			321	B3	164	C3
Diesel	0.462	A3	0.301	B4	0.112	C4
Petrol	5.85	A4	3.82	B5	1.42	C5
LPG	5.25	A5	3.42	B6	1.28	C6
Sulfuric Acid	49.8	A6	0.122	B7	28.9	C7
Caustic Soda	19	A7	0.0498	B8	11.3	C8
Ammonia	0.606	A8			0.228	C9
Chlorine	6	A9			2.26	C10
Hvdrazine	0.2	A10			0.0752	C11
rous Sulphate	8.68	A11			3.27	C12
Trisodium Phosphate	0.793	A12			0.298	C13
Lubricating Oil	0.688	A13	0.165	B9	0.362	C14
Water (low quality)	$1.32 \ge 10^8$	A14	2		2.37×10^7	C15
Water (high quality)	6.47×10^4	A15	256	B10	1.14 x 10 ⁵	C16

Sustainability of Australia's Electricity Generation

Table 4.3.42: Material emissions from the production of 1 GWh of electricity in an

average natural gas power station.

Material	ST		OCGT		CCGT	
	GWh	Ref.	GWh	Ref.	GWh	Ref.
Electricity	1	A19	1	B11	1	C17
	kg }	Ref.	lg	Ref.	kg -	Ref.
Ash	0.214	A20			0.0164	C18
Oily Wastes	0.688	A21	0.165	B12	0.362	C19
Evaporation	9.66 x 10 ⁵	A22			4.87 x 10 ⁵	C20
Wastewater	1.32×10^8	A23	95.1	B13	2.33 x 10 ⁷	C21
Sewerage	0.0266	A24	0.0186	B14	6.95 x 10 ⁻³	C22
Carbon dioxide	6.15×10^{5}	A25	7.20×10^5	B15	3.80×10^5	C23
Methane	1.28	A26	95.4	B16	15.9	C24
NMVOC	7.83	A27	22.1	B17	8.9	C25
N ₂ O	1.2	A28	1.41	B18	6.97	C26
NO _x	1400	A29	2300	B19	497	C27
со	194	A30	533	B20	23.9	C28
SO ₂	0.038	A31	2.98	B21	3.47	C29
Particulates	15.6	A32	252	B22	22.9	C30

generation.

Equation	A	Ψ	Range		Data Rel
			I (\$ million (AUD 2000))	- Õ (MŴ)	
Equation 4.3.42	4.06	0.803	177 - 1039	110 - 1000	Al
Equation 4.3.43	1.13	0.881	4.7 - 271	5 - 500	B1
Equation 4.3.44	11.3	0.593	140 - 451	70 - 500	C1

production capacity, Q.

Table 4.3.44 shows the operating costs, O & M, of steam turbine (Equation 4.3.45), OCGT (Equation 4.3.46) and CCGT (Equation 4.3.47) technologies, respectively, with

generation.

Equation	B	ω	Range O & M (S million (AUD 2000) per annum)	P (GWhpa)	Data 'Ref
Equation 4.3.45	0.00606	0.917	2.54 - 19.2	723 - 6570	A2
Equation 4.3.46	0.387	0.676	0.31 - 14.7	22 - 6570	B2
Equation 4.3.47	0.00234	0.826	2.64 - 27.4	307 - 5220	C2

Chapter 4: Electricity Generation Systems and Data

Table 4.3.43: Correlation parameters for the capital costs of natural gas

(Equation 4.3.44) and CCGT (Equation 4.3.43) technologies, respectively, with

Table 4.3.44: Correlation parameters for the operating costs of natural gas

Chapter 4: Electricity Generation Systems and Data

yearly production, P.

4.3.4.3.4 Social Flows

Table 4.3.45 shows the social flows in average natural gas generation. No LTI figures are available for natural gas plants. Values assume negligible LTIF of 0.1 LTI per 10⁶ WH and 1373 WH per employee. In addition to the employment listed for steam turbine and OCGT power stations, they require 13 and 6 administration employees respectively.

Table 4.3.45: Employment, LTI and fatalities data for average natural gas electricity generation.

Parameter	Steam Turbi	ne .	OCGT		CCGT	
	per MW	Ref.	per MW	Ref.	per GWhpa	Ref.
Employment	0.044	Al	0.021	B1	0.0095	Cl
Parameter	per GWhpa	Ref.	per GWhpa	Ref.	per GWhpa	Ref.
LTI	1 x 10 ⁻⁶	A2	1 x 10 ⁻⁶	B2	1 x 10 ⁻⁶	C2
Fatalities	0		0		_0	

4.3.4.4 TRANSMISSION SUBSYSTEM

4.3.4.4.1 Description of the Electricity Transmission Subsystem

Section 4.3.2.3.2 discusses electricity transmission. Unlike black coal and brown coal, delivery of natural gas to major electricity consumption centres is common. Thus, natural gas electricity generation systems are often near major electricity consumers, and transmission distances are much shorter than in black and brown coal systems.

4.3.4.4.2 Material Flows

Table 4.3.46 shows the material consumption and emissions for average transmission. The average transmission distance traversed by electricity generated from natural gas in Australia is negligible, perhaps 15 km. See Section 4.3.2.4.2.1 for assumptions.

4.3.4.4.3 Economic Flows and Social Flows

See Section 4.3.2.4.3 for capital and operating cost data. See Section 4.3.2.4.4 for

Sustainability of Australia's Electricity Generation

employment, LTI and fatalities data.

of electricity from an average natural gas power station.

In	GWh .	Ref. &	Out	GWh	Ref.
Electricity	1	Basis	Electricity	0.999	Al

4.3.5 DATA OMISSIONS

The material flows data omits air for fuel use in fuels combustion (effectively neutral in sustainability) and emissions to air and water of heavy metals and other minor substances. Data limitations force the omission of heavy metals in all systems. The relative toxicities (CF factors) of heavy metals are high in comparison to other emissions (see Guinée et al. (2001)). Such materials are common in wastewaters, and are present in black and brown coals and coal ash. Thus, it is probable that implications drawn from the results of the eco- and human toxicity environmental indicators, that exclude contributions from such substances, will be uncertain. Chapter 6 examines these and other difficulties of data collection in greater depth.

The social flows data for employment is probably an underestimation, as it omits some contractor employees. In black coal systems, some of the mines included in the analysis are government owned. Often government owned mines will use in-house labour rather than contractors, and consequently the relative proportion of contract labour may be less in these systems.

4.3.6 CONVERSIONS NECESSARY TO PRODUCE INDICATORS

4.3.6.1 OVERVIEW OF CONVERSIONS NECESSARY TO PRODUCE INDICATORS

The sustainability indicator methods require material, economic and social flow information. The data presented in Sections 4.3.2 to 4.3.4 is not in the form necessary to produce indicators for the systems. The basis of the present data is the output of its own subsystem (i.e. 1 Mt of black coal sent out is the basis of the black coal mining subsystem data), not the system (i.e. 1 GWh of electricity delivered). This section develops factors for conversion of the data collected from their subsystem basis to the system basis.

Chapter 4: Electricity Generation Systems and Data

Table 4.3.46: Material requirements and emissions for the transmission of 1 GWh

2ystar

Sustainability of Australia's Electricity Generation Chapter 4: Electricity Generation Systems and Data

Sections 4.3.6.3, 4.3.6.2 and 4.3.6.4 explain the development of conversion factors for the environmental, economic and social flows.

4.3.6.2 Environmental Factors

4.3.6.2.1 Factors to Produce System Material Flows

The conversion of material flows from subsystem basis to system basis requires three steps. The first step is to allocate material flows between products for subsystems with multiple products (see Section 3.2.3.4, Chapter 3). Two subsystems produce multiple products: black coal mining, producing domestic (BIC) and export (BICE) quality coals; and natural gas mining, producing natural gas (NG), LNG and other hydrocarbons (such as, crude oil, condensate and LPG). BIC and BICE production processes are very similar. The difference is that for BICE systems raw coal treatment includes reduction of ash content. This assessment assumes, due to a lack of data, that the ash reduction process minimally affects environmental impacts. Thus, allocation of material flows between BIC and BICE assumes they have equivalent environmental impacts per mass. Very different amounts of energy are necessary to produce each product of the natural gas mining subsystem. Combustion of gas is the primary source for this energy. An examination of Table 4.3.34 indicates that the gas is the primary material requirement. Table 4.3.33 indicates that its combustion (for energy production and flaring) and venting produce most of the emissions^{††††}. Therefore, it is probable that the products requiring greatest energy will result in greatest environmental impacts. Thus, relative gas consumption is the most appropriate allocation parameter, as it reflects the physical operation of the system (see Section 3.2.3.4, Chapter 3). The allocation factors, based on relative gas consumption, are 0.758 for natural gas and 2.20 for LNG (Data Ref. A1) (see Section A2.2.1.2, Appendix 2, for derivation).

The second step is to produce factors to convert between subsystem and system bases. The product of the material flows and conversion factor of each subsystem represents the necessary material flows for that subsystem to produce 1 GWh of delivered electricity. Section A2.2.1.1 (Appendix 2) shows how to produce conversion

^{††††} APPEA (1999) reports that over 94 % of climate change emissions were due to energy production from gas, flaring and fugitive emissions. Sustainability of Australia's Electricity Generation Chapter 4: Electricity Generation Systems and Data

factors for the ST-BIC system. Table 4.3.47 shows the conversion factors for all systems.

Table 4.3.47: Conversion factors to convert the basis of material flows data in Sections 4.3.2 to 4.3.4 from subsystem to system. Note that the natural gas mining values are the product of the conversion and allocation factors. Units: subsystem output basis per GWh of electricity delivered.

Fuel	Technology		Subsy	stem	
		Mining	Transport	Generation	Transmission
BIC	ST	4.56 x 10 ⁻⁴	$4.56 \ge 10^{-4}$		
	IGCC	2.56 x 10 ⁻⁴	2.56 x 10 ⁻⁴	1.01	1.01
DICE	ST	6.72 x 10 ⁻⁴	4.55 x 10 ⁻⁴	1.01	1.01
DICE	IGCC	3.77 x 10 ⁻⁴	2.56 x 10 ⁻⁴		
DrC	ST	1.42×10^{-3}	1.42 x 10 ⁻³	1.02	1.02
ыс	IGCC	7.34 x 10 ⁻⁴	7.34 x 10 ⁻⁴	1.02	1.02
	ST	1.54×10^5	2.03×10^5		
NG	OCGT	2.10×10^5	2.77 x 10 ⁵		
}	CCGT	1.07 x 10 ⁵	1.42×10^5	1	1
	ST	4.53 x 10 ⁵	2.06×10^5		1
LNG	OCGT	6.16 x 10 ⁵	2.80 x 10 ⁵		
	CCGT	3.15 x 10 ⁵	1.43 x 10 ⁵	İ	<u> </u>

The third step replaces the electricity flow with the material flows associated with its production. Just as the electricity produced will have sustainability impacts, the electricity it consumes will also have sustainability impacts. Therefore, the systems should include the material flows associated with electricity consumption. Unlike other materials, for whom exclusion of material flows is due to data availability, material flows are available for electricity. They are the sum of the converted data for each subsystem. The process involves substituting electricity consumption by a factored amount of these sums (for example, see Section A2.2.1.2, Appendix 2).

This step assumes that the source of the electricity is an identical system elsewhere. For example, the source of electricity used in a brown coal, steam turbine technology system (ST-BrC) is an identical ST-BrC. This assumption is valid for Australian black and brown coal generation, as electricity generation in the regions that they are located, is predominately produced by similar systems. The same process can replace the natural gas consumed in the BIC, BICE, and BrC generation subsystems.

146

Sustainability of Australia's Electricity Generation

Chapter 4: Electricity Generation Systems and Data

This material consumption and emission data collection and manipulation corresponds to the life cycle inventory analysis (LCI) step of life cycle analysis (LCA) (see Section 3.2.4, Chapter 3).

4.3.6.2.2 Factors to Convert Material Flows to Environmental Indicators

The characterisation step of the LCA method involves the conversion of material flow data into environmental indicators, using Equation 3.2.4 (Chapter 3) (for example, see Section A2.2.1.2, Appendix 2). To use this equation, equivalency factors (CF) for each indicator and substance are necessary. CML (Guinée et al. (2001)) provides CF factors for most indicators, i.e. climate change, acidification, eutrophication, photochemical smog, eco-toxicity, human toxicity, resource depletion, and exergy destruction. The LCA computer program Simapro (Simapro (1999)) provides CF factors for energy. CF factors for water consumption, particulates and solid waste are unity¹¹¹¹.

Section 3.2.3.5 (Chapter 3) acknowledges that the resource depletion CF factors are the subject of debate. For example, the choice of the appropriate reference region, R, in CF factor development (see Equation 3.2.1, Chapter 3). The CML factors use the world as reference region. However, the use of other reference regions, such as Australia, may allow a more pertinent description of the impact of resource depletion on Australia. Table 4.3.48 shows both the CML factors and factors developed for this assessment using Australia as the reference region.

4.3.6.2.3 Factors to Normalise Economic Indicators to the Australian Region

Normalisation is an optional, but recommended step for environmental indicators (see Section 3.2.3.5, Chapter 3). Normalisation involves division of system environmental indicators by regional environmental indicators (for example, see Section A2.3, Appendix 2). These regional indicators are readily available for some regions, such as Western Europe, and for the world (see Guinée et al. (2001)). Australia normalisation factors are less widely available. Previous LCA studies have produced normalisation factors for energy consumption, water consumption, climate change, acidification,

Sustainability of Australia's Electricity Generation

2 (A 14)	mustance
E	Black Coal
E	Brown Coal
ľ	latural Gas
C	Dil
C	Combusted Oil
ί	Jncombusted Oil
C	Condensate
ĮĽ	_PG
F	Petrol
E	Diesel
F	uel Oil

G 1 2

photochemical oxidant formation, solid waste generation, and abiotic depletion (Burgess (1999), Golonka and Brennan (1996)). However, these factors are not relevant for the base year of 1999. Thus, the estimation of new normalisation factors for Australia is necessary.

Production of normalisation factors for Australia requires the collection of the material requirements and emissions of all human activities in Australia for one year (see Section A1.9, Appendix 1). The conversion of this data into normalisation factors requires the use of the classification and characterisation steps of LCIA (see Section 3.2.3.5, Chapter 3). The characterisation step (i.e. Equation 3.2.4, Chapter 3) should use the same CF factors for both the system environmental indicators and the normalisation factors.

Table 4.3.49 shows the estimated normalisation factors for some indicators. Lack of comprehensive Australia data precludes the estimation of normalisation factors for the remaining indicators.

4.3.6.3 ECONOMIC FACTORS

4.3.6.3.1 Factors to Produce System Economic Flows To produce factors to convert between subsystem and system bases requires a similar process as for material flows. Unlike material flows, the economic data are correlations

148

Table 4.3.48: CF_{RD} factors with the world as reference region (Guinée et al. (2001)) and newly developed factors with Australia as reference region (Data Ref. b.).

Referen	ce Region
World	Australian
0.0134	5.28 x 1 ₀ -5
6.71 x 10 ⁻³	1.59 x 10 ⁻⁵
0.0245	0.0226
0.0201	0.544
0.0201	0.544
0.0201	0.544
0.0201	0.459
0.0245	0.406
0.0201	0.544
0.0201	0.544
0.0201	0.544

^{####} Assumes the consumption of high and low quality water has equivalent effects on total water consumption, and that the environmental impacts of particulate and solid wastes are independent of their compositions.

Table 4.3.49: Australian normalisation factors for some environmental impacts.Factors are indicators of the environmental impacts due to all sources withinAustralia.

Indicator	Туре	Val	UC-
Resource Depletion	World	4.80 x 10 ⁹	kg Sb per year
	Australian	$4.04 \ge 10^{10}$	kg Sb per year
	Energy Depletion	4.82×10^{12}	MJ per year
	Exergy Destruction	8.78 x 10 ¹²	MJ per year
Climate Change		4.78 x 10 ¹¹	kg CO _{2-c} per year
Eutrophication		3.44×10^8	kg PO_4^{3-} per year
Acidification		3.47 x 10 ⁹	kg SO _{2-e} per year
Photochemical Smog	;	9.88 x 10 ⁸	kg C ₂ H _{4-e} per year
Solid Waste		2.12×10^{10}	kg per year

(see Equations 3.3.6 and 3.3.10, Chapter 3). Thus, conversion to system basis requires knowledge of how independent variable(s) of each subsystem are dependent on the system output (see Section A2.2.2.5, Appendix 2). These factors are the same as the conversion factors for environmental indicators (Table 4.3.47), except for the NG and LNG mining subsystem factors (see Table 4.3.50). Since the economic data is pertinent for natural gas mining alone, allocation between the products of the natural mining stage is unnecessary. For an example of conversion for economic indicators, see Section A2.2.2.1 (Appendix 2).

Table 4.3.50: Factors for the mining subsystem of NG and LNG systems to convert the basis of economic flows data in Section 4.3.4 from subsystem to system. Units: subsystem output basis per GWh of electricity delivered.

Fuel ,	Technology	Mining
	ST	1.10 x 10 ⁻⁴
NG	OCGT	1.50 x 10 ⁻⁴
	CCGT	7.65 x 10 ⁻⁵
	ST	1.62 x 10 ⁻⁴
LNG	OCGT	2.20 x 10 ⁻⁴
ļ	CCGT	1.13 x 10 ⁻⁴

Multiple users share the output of some subsystems. For example, presently the primary use of natural gas (and LNG) from mining and transport subsystems is

Sustainability of Australia's Electricity Generation

domestic heating and cooking (90 % by mass^{§§§§}). In a slightly different case, domestic electricity generation consumes only 43 % of the black coal from mining subsystems, while exports consume the remainder (Data Ref. A1). Thus, the size of some subsystems is greater than strictly necessary for the system. Equation 4.3.48 shows how to estimate the proportion of the costs of the larger subsystem due to the electricity generation system alone (for example, see Section A2.2.2, Appendix 2).

$$Costs_{i,s} = Costs_{i,R} \frac{Q_{i,s}}{Q_{i,R}}$$

where: $Costs_{i,s}$ = the capital or operating costs for subsystem i of system s due to the electricity generation system; $Costs_{i,R}$ = the capital or operating costs for subsystem i at actual size; $Q_{i,s}$ = the capacity of subsystem i necessary for system s; and $Q_{i,R}$ = the capacity of subsystem i at actual size.

To produce the capital requirements indicator, I, using Equation 3.3.9 (Chapter 3) (for example, see Section A2.2.2.5, Appendix 2), requires equivalent life factors, λ_s (Table 4.3.51). These factors represent the relative life of subsystem s to the life of the entire system. This analysis assumes that for electricity generation systems, the life of the system is that of the generation subsystem. Most economic assessments of electricity generation assume a life of 20 to 30 years (see Section 2.4.3.4, Chapter 2). In this assessment, the system life is 30 years.

To produce the annual repayment, C_R , which represents the annual charge for using I (for example, see Section A2.2.2.5, Appendix 2), requires the real cost of capital, *i* (%), and life of loan, n (years). A common assumption is that n should equal the life of the project, N (years). N includes an allowance for construction time, which is variable depending on the scale of the system. Experience has shown that construction time is longer for coal than natural gas electricity generation plants (see for example DOE (U.S.A.) (1999)). This analysis includes economic impacts from this construction period by introducing interest during construction (IDC). IDC is the interest paid on

Chapter 4: Electricity Generation Systems and Data

Equation 4.3.48

\$\$\$\$ This proportion is an assumption. Actual consumption of natural gas for electricity generation is

^{\$\$\$\$} This proportion is an assumption. Ac often highly variable.

Chapter 4: Electricity Generation Systems and Data

Table 4.3.51: Numbers of subsystems necessary during the operating life of a system (30 years). A value of greater than one implies that that subsystem's life is less than 30 years and a replacement system will be necessary at the end of its life.

Fhel	Subs	ystei	m lives for one	gençi	ration subsystem	life
	Mining	Rei	Transport	Ref.	Transmission	Ref.
	_	A2	1.5 conveyors	A5		
BIC & BICE	2		4 road	A6		
			2 rail	A7		
			2 ocean	A8	1	A12
BrC	1	A3	1	A9		
NG & ING	1 A	Δ <i>Λ</i>	1 pipelines	A10		
		A4	1 ocean	A11		

capital before its use in this period^{*****}. Commonly accepted values for *i* are between 6 and 7 % (Data Ref. A13).

To produce the value added indicator, VA, using Equation 3.3.3 (Chapter 3) (for example, see Section A2.2.2.5, Appendix 2), requires the sales revenue (S) and the costs of materials produced upstream (F) and labour (L). Table 4.3.52 shows factors to estimate S from the outputs of each subsystem. These figures assume export of LNG and BICE to Japan. Hence, S factors for BICE and LNG systems are the average prices paid for fuels and electricity in Japan, not Australia. Factors to produce F are the factors to produce S for the preceding subsystem (i.e. the factor to produce S for the mining subsystem is the factor to produce F for the transport subsystem). Table 4.3.53 shows the L factors for each subsystem, expressed as a percentage of the subsystem's O & M costs.

4.3.6.3.2 Factors to Normalise Economic Indicators to the Australian Region

Australian normalisation factors for the economic indicators have never been reported for this purpose. However, I and VA are published as indicators of Australian economic performance, while AC and CVA may be estimated from other published values. Table 4.3.54 shows normalisation factors for the capital requirements (I or C_R), annualised costs (AC), value added (VA) and capital inclusive value added (CVA) indicators. For an example of normalisation, see Section A2.3, Appendix 2.

Sustainability of Australia's Electricity Generation

Table 4.3.52: Factors to estimate sales revenue (S) values from subsystem outputs, for estimating value added (VA) (\$ (AUD 2000), per t for mining and transport and per MWh for generation and transmission subsystems).

Fuel	Systems	Subsystems							
		Mining	Ref.	Transport	Ref.	Generation	Ref.	Transmission	Ref.
BIC	ST, IGCC	26.85	A14	40.60	A22	39.93	A28	61.18	A30
BICE	ST, IGCC	32.35	A15	59.12	A23	77.00	A29	92.18	A31
E-C	ST	5.31	A16	5.89	A24	20.07	128	61 18	A 30
	IGCC	6.95	A17	7.75	A25	37.75	h20	01.18	7.30
NG	ST, OCGT, CCGT	120.99	A18	165.88	A26	39.93	A28	61.18	A30
	ST	17.39	A19				}		
LNG	OCGT	18.62	A20	342.12	A27	77.00	A29	92.18	A31
	CCGT	16.06	A21					 	

percentage of operating and maintenance costs (O & M).

Evel	Systems				· · 6	ubsystems	•												
		Mining	Ref.	Traisport	Ref.	Generation	Ref.	Transmission	Ref.										
BIC	ST	56.80	122	10.70	437	42.1	A42												
BIC	IGCC	50.80	A32	10.20		48.7	A43												
DICE	ST	10.85	A 2 2	38.20	130	42.1	A42												
DICE	IGCC	49.03	A33	30.29	A30	48.7	A43												
P+C	ST	66 10	131	0.00	130	74.6	A44												
DIC	IGCC	00.10			, , , , , , , , , , , , , , , , , , ,	36,8	A45	10.0	Δ <i>4</i> 0										
	ST		A35			18.3	A46	10.0											
NG	OCGT	25.89		A35 18.50	A35 1	A35 18.50	A35 18.50	A35 18.50	18.50	18.50	18.50	A35 18.50	A35 18.50	A35 18.50	A35 18.50 A	A40	44.4	A47	4 1
	CCGT					33.3	A48												
-	ST				i i	18.3	A46												
LNG	OCGT	28.92	A36	12.12	A41	44.4	A47												
	CCGT		[33.3	A48												

Indicator 👘	Factor (S billion (AUD 2000) per year)	Ref.
f or CR	44.9	B1
AC	1070	B2
VA	169	B3
CVA	124	B4

Chapter 4: Electricity Generation Systems and Data

Table 4.3.53: Labour factors (L) for estimating value added (VA). L expressed as a

Table 4.3.54: Australian normalisation factors for economic impacts. Factors are indicators of the economic impacts due to all sources within Australia.

The interest rate for IDC is the cost of debt (13-14 %, Data Ref. A13).

Chapter 4: Electricity Generation Systems and Data

4.3.6.4 SOCIAL FACTORS

4.3.6.4.1 Factors to Produce System Social Flows

The conversion factors for economic flows (Table 4.3.47 and Table 4.3.50) apply for social flows (for example, see Section A2.2.3.2, Appendix 2).

To produce the indirect employment (El) indicator, using Equation 3.4.2 (Chapter 3) (for example, see Section A2.3.3.1, Appendix 2), requires industry classifications and EM factors (see Table 4.3.55). The El indicator values use Type 2A factors, as they are the most complete measures of indirect employment (see Section 3.4.2, Chapter 3).

Table 4.3.55: Employment multipliers (EM) to estimate the value of the indirect employment (EI) indicator from direct employment (ED) (ABS (2001b)). Gen. and Trans. = Generation and Transmission.

Industry	Fuel	Subsystem		EME	actors	
Classification			<u>1</u> A	IR	2A	2 B
Coal; oil and gas	All	Mining	2.451	3.702	7.46	6.46
Electricity	All	Gen. and Trans.	1.645	2.322	4.275	3.275
Road transport	BIC, BICE	Transport (Road)	1.441	1.745	2.698	1.698
Rail, pipeline,	BIC, BICE	Transport (Rail)	1.302	1.614	2.949	1.949
other transport	NG, LNG	Transport (Pipeline)	1.561	1.979	3.396	2.396
Water transport	LNG, BICE	Transport (Ocean)	1.706	2.239	3.639	2.639

4.3.6.4.2 Factors to Normalise Social Indicators to the Australian Region

Australian regional normalisation factors for social indicators have never been reported for this purpose. However, Australian employment (as an indicator of economic performance), LTI and fatalities (as indicators of workplace safety) have been reported separately. Table 4.3.56 shows normalisation factors for the employment (ED and El), lost time injuries (LTI) and fatalities (Fatal) indicators. For an example of normalisation, see Section A2.3, Appendix 2.

4.3.6.5 SUMMARY OF CONVERSIONS NECESSARY TO PRODUCE INDICATORS

The conversion factors presented allow the presentation of the data collected in Sections 4.3.2 to 4.3.4 as on a system output basis, e.g. per MWh of electricity delivered. Furthermore, this section provides factors necessary to produce sustainability indicators

Sustainability of Australia's Electricity Generation

 Table 4.3.56: Australian normalisation factors for social impacts. Factors are indicators of the social impacts due to all sources within Australia.

Indicator
ED and EI
LTI
Fatal

for each of the electricity generation systems, using the methods presented in Chapter 3.

4.3.7 SUMMARY OF ELECTRICITY GENERATION SYSTEMS AND DATA

The descriptions presented define the electricity generation systems, which consume Australian fossil fuels. The data collected and conversions described enable the production of sustainability indicators for electricity generation systems, which consume Australian fossil fuels. The basis for indicators is 1 MWh of electricity delivered to the electricity distribution network.

4.4 APPLICATION OF UNCERTAINTY AND SENSITIVITY ANALYSES

4.4.1 OVERVIEW OF THE APPLICATION OF UNCERTAINTY AND SENSITIVITY ANALYSES

The production of indicators of uncertainty requires the collection and manipulation of a broad range of data. Much of this data is either an average of data from many sources, or an average over a period (usually 1 year). The range of values combined to obtain these data averages may be considerable. It is possible that one or many of these source values are as probable as the data average to be an 'accurate representation of the 'true' average. Thus, the value of the data average is uncertain. Moreover, data collection relies on a number of assumptions that may not be accurate. Data uncertainty and poor assumptions may influence the accuracy of the collected data and, consequently, the indicators. Section 3.3.3.4 (Chapter 3) discusses methods for assessing the affect of these uncertainties. This section examines how to apply two of these methods.

Chapter 4:Electricity Generation Systems and Data

Factor	Ref.
 8.92 x 10 ⁶	BI
1,54 x 10 ⁵	B2
 372	B3

uncertainty and sensitivity analysis, to this sustainability assessment of electricity generation systems.

Section 4.4.2 discusses the application of uncertainty analysis to estimate uncertainties caused by the data collected in Section 4.3. Section 4.4.3 discusses the application of sensitivity analysis to estimate uncertainties caused by assumptions in Section 4.3.

4.4.2 APPLICATION OF UNCERTAINTY ANALYSIS

Uncertainty analysis involves the estimation of the range of outcomes possible from the LCA based on variability in the data used to construct the study. In general, an uncertainty analysis will determine the variability of the impact assessment scores generated based on the individual variability of input data. Others have acknowledged the wide range of uncertainty in supply chains for electricity generation systems as a difficulty in assessment of their sustainability (IAEA (2000)). Thus, this analysis tests methods developed for use in LCA, all appearing in Section 3.2.3.7 (Chapter 3). Time and resource constraints limit the analysis to the brown coal and black coal steam turbine systems (ST-BrC and ST-BIC). These same constraints required the use of 'rules of thumb' to estimate ranges for some data. The quantitative and qualitative data necessary for the uncertainty assessment appears in Appendix 1.

4.4.3 APPLICATION OF SENSITIVITY ANALYSIS

Sensitivity analysis tests the influence of decisions made in sustainability assessment on the indicator values. Some key decisions in data collection include the

- methods used to allocate material flows in natural gas and black coal mining;
- assumed distances for LNG and export black coal (BICE) transport;
- electrical generation efficiencies of the generation subsystems; and
- chosen system capacity in the economic and social indicator methods.

Section 4.3.6.2.1 details an allocation method for natural gas mining based on relative fuel gas consumption. This same section assumes no allocation necessary for the black coal mining. Commonly, the basis for allocation is the relative economic values of the subsystems' products. Table 4.4.1 shows the product of allocation and conversion factors obtained when using an economic basis for allocation (see Section

Table 4.4.1: Factors for natural gas and black coal mining to convert from subsystem to system basis when using the economic allocation method. Units: subsystem output basis per GWh of electricity delivered.

Fuel	Technoky	y Mining	Ref.
1317 *	ST	4.56 x 10 ⁺	Δ1
1310	IGCC	2.56 x 10 ⁴	A2
1 1 4 1	ST	672 x 10 ⁴	Δ <u>3</u>
1910 1	IGCC	3 77 x 10 ⁻⁴	Δ4
	ST	1.32 × 10 ⁻⁴	45
NG	OCGT	1.52×10^{4}	Δ6
	CCG1	7 77 x 10	Δ7
	ST	8 83 x 10	
LNG	OCGI	1 20 × 10 ⁴	Λ9
	CCGI	615 \$ 10 1	A10

A2.2.1.2, Appendix 2). The sensitivity analysis compares the indicators produced using economic value and the original bases for allocation

The transport distance for LNG and export black coal (BICT) assumes that the destination in Japan. However, the U.S.A. has also been a destination for Australian LNG (13500 km), and Europe is a common destination for BICE (21500 km). The sensitivity analysis compares the indicators produced using these and the original distances

The electrical generation efficiency dictates the amount of fuel necessary for the generation subsystems. The data collection methodology adopted ensures that the data in Section 4.3 refers to generation subsystems with Australian average electrical efficiencies. However, Section 4.3 acknowledges that in practice a range of electrical efficiencies occurs, due to differences in operation, age, and minor improvements (see Table 4.4.2). The sensitivity analysis compares the indicators produced using the minimum, maximum and original electrical efficiencies

The method for producing economic indicators assumes a system capacity of 1000 MW delivered (see Section 4.3.6.3.1). This capacity represents neither a large nor a small capacity system. In Australia, capacities range from, below 100 MW for some NG-OCGT systems, to over 2000 MW for some BIC-ST systems. The sensitivity analysis compares indicators assuming capacities of 500 MW and 2000 MW with the indicators assuming a capacity of 1000 MW.

Chapter 4.1 Jectricity Concration Systems and Data

Table 4.4.2: Minimum, average and maximum electrical generation efficiencies of generation subsystems. Also shown are multipliers (upstream factors) to estimate the additional impacts for minimum and maximum efficiencies (multiply each conversion factor of Table 4.3.47 by the upstream factor).

Fort Technolo		Efficiencies			Lipstream Factor			
		Min.	Ave	Max.	Min.	Ave.	Max	
Diff and Diff.	.51	30	33	37	111	00	0.902	
DR MADE F	IGCC	40	51	52	1 26	1.00	0.997	
lier	51	21	27	28	116	1.00	0.895	
1311	RECC	40	48	48	1 19	1.00	1.00	
	ST	30	35	35	117	E 00	1 (Ю	
NG and LNG	-OCG1	17	26	37	151	1.00	0.694	
	CCG]	44	50	\$7	1 14	1.00	0.881	

4.4.4 SUMMARY OF THE APPLICATION OF UNCERTAINTY AND SENSITIVITY ANALYSES

The assumptions and decision points identified have most influence on the uncertainty of the environmental indicators of sustainability for the electricity generation system Uncertainty analysis and sensitivity analysis allow the estimation of the influence of this uncertainty on the sustainability indicator results. This section establishes several probable choices for each assumption or decision point in the electricity generation systems. This information, will allow the analysis of the uncertainty and sensitivity of the sustainability indicators

4.5 SUMMARY OF ELECTRICITY GENERATION SYSTEMS AND DATA

Each of the electricity generation systems this chapter describes is important in current and future electricity generation using Australian fossil fuels. Material, economic and social flows data for these systems, in conjunction with given conversion factors and the methods Chapter 3 presents, will allow the production of indicators of environmental, economic and social sustainability for the electricity generation systems.

and the second process of the second
This chapter selects the brown coal and black coal steam turbine (BrC-ST and BlC-ST) systems for testing some promising uncertainty techniques. Additionally, it identifies assumptions and decisions having potentially great influence on the accuracy. of the results for a sensitivity analysis.

158

5. RESULTS OF THE SUSTAINABILITY ASSESSMENT

5.1 OVERVIEW OF RESULTS OF THE SUSTAINABILITY ASSESSMENT

The preceding chapters present methods (Chapter 3) and information (Chapter 4) for the production of indicators of sustainability for electricity generation systems, which consume Australian fossil fuels. Additionally, these chapters include methods for estimating the uncertainty of these indicators due to uncertainties in the data and the sensitivity of the indicators to assumptions. Consequently, with this information it is possible to produce indicators of the sustainability for these systems and test their robustness to data uncertainties and assumptions.

Section 5.2, contains environmental indicators for the electricity generation systems, and results of data quality and sensitivity analyses. Similarly, Sections 5.3 and 5.4 contain economic and social indicators for these systems, and results of a sensitivity analysis. Section 5.5 combines the environmental, economic and social indicators using normalisation.

5.2 ENVIRONMENTAL INDICATOR RESULTS

5.2.1 OVERVIEW OF ENVIRONMENTAL INDICATOR RESULTS Environmental indicators are one type of indicator of sustainability. They provide simple measures of the effect of material consumption and emissions on the environment. This section presents environmental indicators of sustainability for each of the electricity generation systems.

Section 5.2.2 presents environmental indicators for each electricity generation system. Each of these systems has four subsystems: fuel mining; fuel transport;

Chapter 5:Results of the Sustainability Assessment

Chapter 5: Results of the Sustainability Assessment

electricity generation and electricity transmission. Section 5.2.2 also presents estimates of each of these subsystems contribution to the system indicators. Sections 5.2.4 and 5.2.5 present the results of the sensitivity and uncertainty analyses, which estimate the affect of data uncertainty and value-based decisions on the indicators.

5.2.2 SYSTEM INDICATOR SCORES

Sections 4.3.2 to 4.3.6 (Chapter 4) contain the data required to produce the environmental indicators. Sections 4.3.2 to 4.3.4 contain this data on a subsystem basis, and Section 4.3.6 contains factors for its conversion to system basis (Chapter 4). Equation 3.2.4 allows production of the indicators from this data (see Section 3.2.3.5, Chapter 3). Equation 3.2.4 requires the CF factors of Section 4.3.6.2.2 (Chapter 4) for each environmental indicator. The environmental indicators produced using this method appear in Table 5.2.1.

5.2.3 OBSERVATIONS FROM THE INDICATORS

5.2.3.1 CLIMATE CHANGE (CC)

Of the systems, ST-BrC have the worst CC indicator scores (1210 kg CO2-eq per MWh delivered), due to their relatively low electricity generation efficiency, derived from the fuel's high moisture content, and the high carbon to hydrogen ratio of the brown coal fuel (see Figure 4.3.4, Chapter 4). Predictably the next two worst systems are also low efficiency, coal fuelled processes, ST-BICE (1110) and ST-BIC (1060). The transport distance of the export system (ST-BICE) is greater than the domestic case, so its indicator score is worse (ST-BIC). The system in fourth position is a natural gas fuelled system, OCGT-LNG, due to its low efficiency. Its indicator score (1050) is similar to that of the ST-BIC system, and worse than the BIC- and IGCC-BICE systems. This evidence supports the claim that CC scores for natural gas systems can be worse than a black coal system (see Section 1.1, Chapter 1). In contrast, the combination of high efficiency and lower carbon to hydrogen ratio of the natural gas based fuels (see Figure 4.3.4, Chapter 4) ensures that the CCGT-NG (458) and CCGT-LNG (566) systems have the best indicator scores. This evidence supports the claim that CC scores for natural gas systems can be better than coal systems (see Section 1.1, Chapter 1).

	Tech.		Indicator											
		ÇÇ.	AD.	EU	PS	III	E TS	,PM	~SW	ND.	RD Á	RDW	IN	EX
	ST	1060	6.68	0.437	0.314	5270	4.9	0.548	87.3	2.37	0.774	6.29	11.3	13.7
ā	IGCC	683	0.171	0.0261	0.0193	74.6	0.289	0.038	63	1.3	0.542	3.71	6.55	8.26
щ	ST	1110	7.34	0.459	0.35	5830	5.92	1.05	310	2.65	5.73	9.49	17.1	20.6
BIC	IGCC	709	0.577	0.0588	0.0457	370	0.842	0.315	187	1.45	3.14	5.48	9.75	12.3
U U	ST	1210	3.13	0.218	0.148	2410	2.62	0.493	21.8	2.68	0.185	9.83	13.9	12.4
Ъ.	IGCC	750	2.41	0.0244	0.105	2430	0.44	0.0218	13.7	0.657	0.244	5.26	7.42	6.85
	ST	724	0,944	0.236	0.158	42.2	2.21	0.035	0.0213	139	5.54	5.58	6.9	11.9
^D Z	OCGT	847	1.47	0.368	0.228	60.9	3.64	0.285	0.0282	0.00	7.59	7.46	9.24	15.9
	CCGT	458	0.415	0.0998	0.0982	33.6	0.965	0.0372	0.0148	25.2	3.98	3.91	4.85	8.39
1	ST	878	1.44	0.333	0.369	177	3.16	0.0714	0.0711	144	11.1	6.45	8.15	13.8
Z	OCGT	1050	2.14	0.499	0.511	241	4.91	0.335	0.0948	5.13	15	8.63	10.9	18.4
	CCGT	566	0.764	0.168	0.246	128	1.62	0.0627	0.0497	27.9	7.84	4.52	5.72	9.7

5.2.3.2 ACIDIFICATION (AD)

Black coal systems have the worst AD indicator scores, ST-BICE (7.34 kg SO2-ca per MWh delivered), and ST-BIC (6.69). Emissions of SO₂ and NO₃ are the major contributors to the AD indicator scores. BIC has higher proportion of sulfur than the other fuels (see Figure 4.3.4, Chapter 4), which during its combustion (in the generation subsystem) becomes SO₂. The score of the ST-BICE system is worse than in many operating systems, which often treat flue gas to reduce SO₂ emissions. Contrariwise, the IGCC-BIC system has the best AD score (0.171). This system removes almost all sulfur from the flue gas (>98 %). Without removing sulfur, its score would be third worst (4.50). Similarly, without sulfur removal, the ST-BICE system is worse than the ST-BIC system. The scores for the brown coal system's are approximately half those of equivalent black coal fuelled systems. This reflects the much lower sulfur content in brown coal (see Figure 4.3.4, Chapter 4). All natural gas fuelled systems have relatively

Sustainability of Australia's Electricity Generation Chapter 5: Results of the Sustainability Assessment

Table 5.2.1: Environmental indicators for electricity generation from Australian fossil fuels (CC = climate change (kg CO_{2-co}), AD = acidification (kg SO_{2-co}), EU = eutrophication (kg PO_4^{3} ..., PS = photochemical smog (kg $C_2H_{4,c}$), HT and ET = human and eco-toxicity (kg 1,4-dichlorobenzene), PM and SW = particulates and solid waste emissions (kg), WD = water consumption (t), RD A and RD W = Australian and world resource depletion (kg Sb_{c}), EN and EX = energy consumption and exergy destruction (GJ), all per MWh of electricity delivered).

low AD scores (up to 2.12), as the sulfur content of raw natural gas is minimal in Australia (see Figure 4.3.4, Chapter 4). AD scores for natural gas systems are mainly due to nitrogen oxides (NO_x), which are less powerful acidifying gases per unit mass than SO₂.

5.2.3.3 EUTROPHICATION (EU)

The OCGT-LNG (0.499 kg PO_{4-eq} per MWh delivered), ST-BICE (0.459), and ST-BIC (0.437) have worst EU scores, due to their relatively high NO_x production. Importantly, the systems whose electricity generation technology has stringent NO_x controls (IGCC and CCGT) have the best scores (i.e. IGCC-BrC (0.0244) and IGCC-BIC (0.0260)). The ST-BrC system has lower NO_x levels than the ST-BIC system, due to its lower flame temperature (see Section 4.3.3.3.1, Chapter 4). Consequently, its EU scores (0.218) is better. NG and LNG systems have similar scores to the coal systems, as, while their fuel contains less nitrogen, they have higher flame temperatures, resulting in similar levels of NO_x emissions.

5.2.3.4 PHOTOCHEMICAL SMOG (PS)

The LNG based systems, OCGT-LNG (0.511 kg C_2H_{4-eq} per MWh delivered) and ST-LNG (0.369), have worst scores, due to substantial NO_x production in the transport subsystem. This subsystem includes LNG production subsystem, which consumes a low hydrocarbon and high nitrogen content fuel gas to provide energy for the intensive liquefaction process. These NO_x emissions from the LNG production subsystem are also important contributors to the poor LNG scores for AD, EU, HT and ET. Fugitive hydrocarbon releases, while included in the assessment, made minimal contribution to this impact.

5.2.3.5 HUMAN (HT) AND ECO-TOXICITY (ET)

The indicator scores for HT and ET show NG and LNG systems to have lowest texicity, while BIC- and ST-BICE have worst toxicity. These scores have limited value as they include only a limited range of substances (H₂S, NO_x, SO₂, N₂O and particulates) (see Section 4.3.5, Chapter 4). Of these, SO₂ has the greatest influence on the indicator

Sustainability of Australia's Electricity Generation

scores. Thus, the BIC- and IGCC-BICE systems, with sulfur removal, have substantially better scores than the BIC- and ST-BICE systems, without sulfur removal.

5.2.3.6 PARTICULATES (PM)

The PM scores of the coal systems are substantially worse than for gas systems, with the greatest impact, ST-BICE (1.05 kg particulates per MWh delivered), corresponding to the system requiring the greatest amount of coal handling operations (4, including all stockpiles and transfers during transport). The second largest score, ST-BIC, entails half as many handling operations for a similar coal requirement. Its score (0.548) is approximately half that of the ST-BICE system. This indicates that limiting transfers will reduce black coal PM scores. The brown coal systems have somewhat lower the particulate emissions from coal handling than do black coal systems, due mainly to the high coal moisture content (see Figure 4.3.4, Chapter 4). However, their PM scores are still high (ST-BrC 0.493), because of emissions from their combustion in the generation subsystem. PM scores for the ST-NG, IGCC-BrC, IGCC-BIC, and the CCGT-NG systems are all minimal (0.022-0.038).

5.2.3.7 SOLID WASTE (SW)

SW indicator scores for the coal systems are far worse than are the scores of the natural gas systems. The major contributors to PM are waste coal (from coal cleaning operations), ash (from coal combustion), and ash/limestone mixtures (from in-bed desulfurisation operations). The greatest scores are for the export black coal systems, ST-BICE (310 kg solid waste per MWh delivered) and IGCC-BICE, as only these systems produce both waste coal and coal-ash mixtures. The domestic black coal systems are next, which produce either ash (ST-BIC 87.3) or ash/limestone mixtures (IGCC-BIC 63.0). The very low ash content of the brown coal (approximately 0.9 % by mass, c.f. black coal 8-25 % by mass, see Figure 4.3.4, Chapter 4) ensures that despite its greater mass consumption, the SW scores (ST-BrC 21.8, IGCC-BrC 13.7) are much lower than equivalent black coal systems. Overburden, the non-coal material removed to mine coal, is not included in SW, as mining generally returns this material continuously or in the rehabilitation phase (see Chapter 4). Inclusion of overburden would accentuate the differences between the coal and gas systems. Moreover, as the

overburden produced in BIC systems is 4-5 times that of BrC systems, inclusion would accentuate the differences between black and brown coal systems.

5.2.3.8 WATER DEPLETION (WD)

WD scores for all NG and LNG systems, except OCGT systems, are substantially greater than BIC and BrC systems. These systems, in all Australian cases, use oncethrough cooling water for condensers in the generation subsystem, and for the LNG case, in the LNG vaporisation step of the transport subsystem. While the treatment of some of this water with chemicals, such as chlorine, does occur, the average chemical composition of the water discharged is unchanged. The main consequence is an increase in temperature of this water by up to 8 °C. The water consumption of BrC and BlC plants assumes the use of circulating cooling water for condensers. These systems use considerably less water, but wastewater from these systems requires treatment to remove contaminants before discharge. Ninch of the water consumed for circulating systems is lost through evaporation in cooling towers.

5.2.3.9 RESOURCE DEPLETION (RD A AND RD W), ENERGY DEPLETION (EN) AND EXERGY DESTRUCTION (EX)

RD A, RD W, EN and EX are all indicators of the impact on sustainability caused by consuming natural resources. The ST-BICE (17100 MJ energy per MWh delivered), ST-BrC (13900), ST-BIC (11300), and OCGT-LNG (10900) systems have worst EN indicator scores. Yet, for the EX indicator, the order is different: ST-BICE (20600 MJ exergy per MWh delivered), OCGT-LNG (18400), OCGT-NG (15900), and ST-LNG (11900). The order changes because the combustion of liquid transport fuels (such as fuel oil and diesel) and natural gas destroys greater exergy than an equivalent amount of the coal fuels. The ST-BICE system uses considerable amounts of diesel in both the mining and transport subsystems. The LNG and NG systems consume natural gas in the mining, transport and generation subsystems. The ST-BICE (9.83 kg Sb..eq per MWh delivered), ST-BICE (9.49), and OCGT-LNG (8.63) systems have the worst RD W indicator scores. In contrast, the NG and LNG systems have the worst RD A indicator scores (i.e. OCGT-LNG 15.0 kg Sb..eq per MWh delivered, ST-LNG 11.1, and CCGT-LNG 7.84).

Sustainability of Australia's Electricity Generation

5.2.3.10 ALL INDICATORS

No single electricity generation system has the worst score for every indicator. However, the CCGT-NG systems have the best scores in many categories. The ST-BrC system has the greatest indicator score for CC, but is not the worst contributor to any other indicator, except RD W. Thus, BrC is not the worst environmental performer. Therefore, these results do not support claims to the contrary, like the quote at the beginning of Chapter 1. The advanced coal generation technology, IGGC, improves significantly the indicator scores of both the BrC and BIC systems from their respective ST systems. However, the AD and SW indicator scores of the IGCC-BIC system are still poor in comparison with systems consuming BrC and NG. The indicators scores for LNG systems are generally better than for those of BICE systems, except in indicators where NO_x emissions are important contributors (i.e. AD, PS, EU).

5,2.3.11 SUBSYSTEM CONTRIBUTIONS

Each of the fossil fuel and technology combinations for electricity generation from Australian fossil fuels contains four subsystems. Figure 5.2.1 shows the contribution of each subsystem to the total environmental indicator magnitude for each combination.

Contributions of the non-generation subsystems are considerable for many of the indicators. The contribution of mining to the indicators of natural gas systems is uniformly greater than the contributions of mining to the indicators of the coal systems. The contributions of the mining and transport subsystems for LNG systems are even greater than the natural gas systems, due to the considerable energy requirements of liquefaction of natural gas, and lengthy ocean transport of LNG. The transport subsystem of IGCC-BICE systems contributes highly in many categories (AD, EU, PS, HT, ET, PM) due mainly to the relatively low generation emissions of these systems. The contributions of the transport subsystems to RD A for the BICE and LNG systems are also considerable.

5.2.4 UNCERTAINTY ANALYSIS

.....

5.2.4.1 OVERVIEW OF UNCERTAINTY ANALYSIS

This section contains a data quality analysis of the ST-BIC and ST-BrC systems. The





presentation is in the same order as the discussions of data quality analysis methods in Section 3.2.3.7 (Chapter 3). Thus, Section 5.2.4.2 includes results using numerical methods, Section 5.2.4.3 qualitative methods, Section 5.2,4.4 combined methods, and Section 5.2.4.5 comparisons.

5.2.4.2 NUMERICAL UNCERTAINTY

Estimating numerical uncertainty uses Monte-Carlo simulations for both black and brown coal cases (see Section 3.2.3.7.2, Chapter 3). Varied in the simulations are all data necessary to produce the CC, AD and RD W indicators, except the CF factors. Each simulation involved 10000 calculations of each of these indicators, using XISim⁴ software. As many data point's uncertainty distributions are unknown, a number of common types are tested: triangular, uniform, normal and log normal (see Figure 3.2.2, Chapter 3). The basis for these distributions is the indicator's maximum, minimum and mean values. The effect of this choice is considerable (see Table 5.2.2). The minimum mean value and standard deviation occur when using normal distributions, and the maximum standard deviations occur when using uniform distributions. In the calculation of the AD potential of the brown coal system the use of triangular and uniform distributions worsens the mean indicator value by over 300 %.

Table 5.2.2: Estimates of the uncer the use of Monte-Carlo simulation (

	Probability	CC (kra	e per-	AD (kg	ob-se Ther	RDW	1 Carter
	Distribution	NIN MAR	b		Nh)	, ner y	(Wh)
3. at 1	. I YPC	Mein	SE SD of	Mean	SU SU	Mean	SU2
-	Average	1040	-	6.77		6.37	•
, B	Triangular	1082	309	7,98	2.90	6.51	1.92
	Uniform	1080	460	9.32	4.53	4.55	2.89
lac	Normal	1038	255	6.88	2.21	6,40	1.53
	Log Normal	1099	334	7.38	2.83	6.79	2.16
	Average	1341	-	3.47	-	10.9	•
Ö.	Triangular	1319	102	12.0	6.1	11.1	0.6
เองท	Uniform	1386	138	19.3	8.2	11.3	0,9
	Normal	1338	76	3.34	4.67	10.9	0.5
`	Log Normal	1346	76	4.92	3.38	10.9	0.5

Some additional simulations determine which data are the greatest contributors to the final uncertainty profile (see Table 5.2.3). These simulations remove the uncertainty

Sustainability of Australia's Electricity Generation Chapter 3:Results of the Sustainability Assessment

tainty	of	some	environmental	indicators	from
SD = st	lan	dard o	leviation).		

Table 5.2.3: Simulations using Monte-Carlo simulation to indicate the influence of individual data uncertainty on indicator uncertainty (Triangular distribution, SD = standard deviation), A from Table 5.2.2.

Parameter held constant	Black	СС (kgcox. s Сояl	, per MWI Browj	S Coal
No Variable Fix A	Mcam -	SD 300	Mean /	SD-SD
Power Station Efficiency	1082	65	1368	102 59
Coal Lost in Transit	1111	277	1382	101
Coal-Bed Methane	1055	304 298	1363 1319	80 102

profile for a chosen data point, and thus hold the chosen data point constant in all simulations. Holding the power station efficiency constant substantially reduces the standard deviations and therefore the variability of the results. It also makes the uncertainty ranges of the black and brown coal systems similar. In the brown coal case, the variation due to uncertainty in the emission factor of CO₂ from coal combustion is also a significant contributor. Other data uncertainties, such as the amount of coal tost as dust in mining and transportation, and coal-bed methane emission factor, are minor contributors to the uncertainty of the results.

5.2.4.3 QUALITATIVE UNCERTAINTY

Qualitative uncertainty measures for the black and brown coal systems use a single set of indicators (Table 3.2.3, Chapter 3) to ensure a fair comparison. These indicators are all data level indicators, while both methods include process and system level indicators in their intended indicator sets.

Table 5.2.4 shows the qualitative uncertainties calculated using the method of Wrisberg (see Section 3.2.3.7.3, Chapter 3). Aggregation is for each subsystem independently as well as for the total system to highlight the areas in which quality is poorest (i.e. has highest score). Reliability and completeness are the major contributors to qualitative error for both black and brown coal cases. Addition of the '%' column shows the relative contributions of each subsystem to the total indicator. Thus, as the transmission data indicators contribute less than 10 % to the totals, increases in the quality of the transmission data will not substantially decrease the totals. Mining and transport contributed higher than expected proportions to the totals.

Table 5.2.4: Qualitative uncertainty scores for the data of the ST-BIC and ST-BIC systems calculated using the Wrisberg method (Min. = Mining, Tsp. = Transport, Gen. = Generation and Trm. = Transmission).

Sub-					Sci	bre				ang sa sang sa sa Ng sa sa sa sa sa Ng sa sa sa sa sa sa
sys	Reliat	ollity	Comple	teness	Temp	oral	Geogra	phical	Techno	ogical
tem !!	Score	%	Score	%	Score	"%a."	Score	₽⁄a	Score	%
Black	Coal		a laga a la Marana Angala. Ng Kanada ang kanalarang ang kanalarang kanalarang kanalarang kanalarang kanalarang Ng Kanada ang kanalarang kanalarang kanalarang kanalarang kanalarang kanalarang kanalarang kanalarang kanalarang		و دو هر اد در و مرکز در در در و				a tan ƙasar ƙasar ƙwallon ƙasar ƙasar ƙasar ƙasar ƙasar ƙasar ƙasar ƙasar ƙasar ƙas	
Min.	2.87	35	3.60	40	1.20	29	1.60	39	1.67	42
Tsp.	3.57	20	4.14	21	2.14	24	1.71	20	1.29	15
Gen.	2,47	38	2.26	32	1.37	41	1.21	38	1.26	40
Trm.	4.00	7	5.00	7	2.00	6	1.00	3	1.00	3
Total	2.86	100	3.16	100	1.47	100	1.42	100	1.40	100
Brown	(Coal)								t e and served a abs il t of the same	
Min.	2.82	32	2.64	26	1.00	19	1.18	25	1.36	31
Tsp.	3.83	23	4.00	21	3.00	32	2.00	23	1.00	12
Gen.	1.64	37	2.23	44	1.09	42	1.14	48	1.18	53
Trm.	4.00	8	5.00	9	2.00	7	1.00	4	1.00	4
Total	2.39	100	2.73	100	1.39	100	1.27	100	1.20	100

Table 5.2.5 shows the total qualitative uncertainties calculated using the method of Rousseaux et al. (see Section 3.2.3.7.3, Chapter 3). When a target quality score of 1 or 2 is set, the acceptability (i.e. the percentage of inventory scores equal to the target score or less) for each indicator is lower for black coal than brown coal. This indicates that the data quality of the brown coal system is better than for the black coal system. Comparing variability scores, it is noticeable that both systems have high temporal scores.

systems calculated using the Rousseaux et al. method.

Indicator	Acceptab	ility at Ob	cctive Qua	lity Score	Variability
		2.00	3	4	
Black Coal					
Reliability	0.0	51.2	76.7	86.0	40.9
Completeness	0.0	48.8	62.8	72.1	54.1
Temporal	79.1	86.0	93.0	95.3	75.9
Geographical	74.4	86.0	97.7	100	44,4
Technological	81.4	81.4	97.7	_100	51.7
Brown Coal					
Reliability	36.6	61.0	73.2	90.2	81.3
Completeness	2.4	68.3	78.0	78.0	58.6
Temporal	85.4	90.2	92.7	92.7	85.9
Geographical	80.5	92.7	100	100	27.7

Table 5.2.5: Qualitative uncertainty scores for the data of the ST-BIC and ST-BIC

Technological 90.2 90.2 100 100 30.2

5.2.4.4 COMBINED UNCERTAINTY

(the fits

 χ^{μ} i

The combined uncertainty measures for the black and brown coal systems use the methods of Kennedy et al. and Meier, and the indicators shown in Table 3.2.3 (Chapter 3). Presented in Table 5.2.6 are CC indicator uncertainties, along with the numerical simulation results reported in Table 5.2.2. For the black coal system, the use of the Kennedy et al. method modified the mean value by a small amount (± 5 %), while increasing the standard deviation (and hence uncertainty) by 40-90 %. The use of the Meier method has little effect on either the mean or the standard deviation. For the brown coal system, both methods left the mean unchanged, while increasing the standard deviation dramatically. In fact, the standard deviations of the Kennedy et al. brown coal systems are very similar to those of the black coal system, although still lower as a percentage of the mean values. Therefore, it is clear that the brown coal system has lower uncertainty than the black coal system.

Table 5.2.6: Uncertainty in the CC indicator obtained using Monte-Carlo simulation, and the methods of Kennedy at al. and Meier (SD = Standard Deviation).

Distribution	Method	C	C (kgcor	ea, per NIW	b)
		Black	Coal	Brown	i Coat
a an		Mean	SD	Mean	SD
	Numerical	1082	309	1319	102
Triangle	Kennedy et al.	1062	545	1355	558
	Meier	1059	331	1367	158
	Numerical	1080	460	1386	138
Uniform	Kennedy et al.	1087	663	1423	550
	Meier	1108	495	1393	230
	Numerical	1038	255	1338	76
Normal	Kennedy et al.	1027	478	1354	551
	Meier	1021	259	1342	135
	Numerical	1099	334	1346	76
Lognormal	Kennedy et al.	1102	572	1331	524
	Meier	1074	332	1353	139

5.2.4.5 COMPARISONS

The differential case (brown coal system minus black coal system) (Table 5.2.7) can determine which system has least environmental impact. While the average values

Sustainability of Australia's Electricity Generation Chapter 5:Results of the Sustainability Assessment

indicate that the black coal system has lower CC and RD W impacts and higher AD impact, the numerical ranges overlap indicating that these outcomes are not certain.

black coal systems obtained using Monte-Carlo simulation (per MWh).

-System	Probability Distribution Type	C (kgr: Mean	C _{05-eq}) SD	A) (kgso Mean	D 244) 25D	RD (kgs Mean	W h-vy.) SD
	Average	301	-	-3.30	-	4.54	*
CE DIC	Triangular	237	321	4.02	6.60	4.55	2,05
51-BIC -	Uniform	306	484	10.0	9.2	6.77	2.8
31-BIC	Normal	300	267	-3.53	5.05	4.52	1.62
L	Log Normal	247	341	-2.46	4.57	4.15	2.21

Figure 5.2.2 shows a normalised difference probability distribution plot for the triangular CC impact (calculated as (brown coal-black coal)/brown coal). This shows that there is a 12 % chance that the brown coal system is at least 10 % better than the black coal system, while there is a (100-41) or 59 % chance that the black coal system is 10% better than the brown coal system.



Figure 5.2.2: Cumulative distribution function of the difference between the CC indicators, obtained using Monte-Carlo simulation, of the brown and black coal systems.

5.2.5 SENSITIVITY ANALYSIS

The sensitivity analysis tests the influence of three decisions on the indicator values: allocation between products of the black coal and natural gas mining subsystems; transport distances (for LNG and BICE systems); and electricity generation efficiency.

Table 5.2.7: Uncertainty in the difference between indicators for the brown and

The results of the sensitivity analysis have the same form as the main results shown in Table 5.2.1*.

The variation of the allocation procedure for black coal mining from unallocated to economic value based allocation result in minor (less than 1 %) variations in environmental indicators. Variation of the natural gas mining allocation procedure also minimally changes the NG system environmental indicators. However, the changes to the LNG system indicators from this variation are more substantial. With economic allocation, the emission related indicators (CC, AD, EU, PS, PM and SW[†]) improve by 10-40 %, and requirement-related indicators (RD A, RD W, EN and EX) improve by 7-8 % over the base values in Section 5.2.2.

The variation of BICE transport subsystem destination from Japan to Europe worsens CC, SW, PM, AD W, EN and EX by less than 4 %, AD, PS and EU of the ST-BICE system by greater amounts, and AD, PS and EU of the IGCC-BICE system and AD A of both BICE systems by large amounts. The variation of LNG transport subsystem destination from Japan to U.S.A. worsens all indicators by less than 10 %, except for AD A, which worsens by 40 %.

Variation of the electrical generation efficiency produces considerable variation, in most cases between 85 and 115 % of the average value (for example, Table 5.2.8).

Table 5.2.8: Changes in environmental indicators when varying electrical generation efficiency as part of a sensitivity analysis (per MWh). Efficiencies appear in Table 4.4.2. EIV is ratio of the average of each indicator's percentage range and the value at the average efficiency.

System		(COrea)	System	CC (kg	CO _{2-ca})
	EIV"	Range		EIV	Range
ST-BrC	90.0-115	1090-1410	ST-NG	100-111	724-736
ST-BIC	90.1-111	956-1180	ST-LNG	100-113	878-917
ST-BICE	89.2-110	987-1220	OCGT-NG	69.4-151	588-1280
IGCC-BrC	100-115	750-886	OCGT-LNG	69.3-151	731-1590
IGCC-BIC	99.7-119	681-857	CCGT-NG	88.0-114	404-523
IGCC-BICE	88.6-110	699-880	CCGT-LNG	88.0-114	499-647

Sustainability of Australia's Electricity Generation

5.2.6 SUMMARY OF ENVIRONMENTAL INDICATOR RESULTS

The environmental indicators presented here show the major affects of electricity generation systems, using Australian fossil fuels on the environment. The subsystem breakdown establishes the proportional influence of each subsystem on the total system indicator. Comparisons of these indicators allow the ranking of each electricity generation system for environmental impact. A discussion of the results of the sensitivity and uncertainty analyses will establish the robustness of these rankings to data variability and assumptions. Chapter 6 discusses the implications of these results on the sustainability of electricity generation systems.

5.3 ECONOMIC INDICATOR RESULTS

5.3.1 OVERVIEW OF ECONOMIC INDICATOR RESULTS

Economic indicators are a second type of indicator of sustainability. They present simple measures of either the consumption or generation of money on the economy. Thus, this section presents economic indicators of sustainability for each of the electricity generation systems.

Section 5.3.2 presents the economic indicators for each electricity generation system. As with the environmental indicators (see Section 5.2), Section 5.3.2 also presents estimates of the contribution of each subsystem to the system indicators. Section 5.3.3 presents the results of the sensitivity analysis, which estimates the affect of value-based decisions on the indicators.

5.3.2 SYSTEM INDICATOR SCORES

Sections 4.3.2 to 4.3.6 (Chapter 4) contain the data required to produce economic indicators. Sections 4.3.2 to 4.3.4 contain this data on a subsystem basis, and Section 4.3.6 contains factors for its conversion to system basis (Chapter 4). System basis capital costs (1) use the λ factors of Section 4.3.6.3 in Equation 3.3.9 (Chapter 3). System annualised costs (AC) use the i and n values of Section 4.3.6.3 in Equation 3.3.8 (Chapter 3) to produce the C_R for Equation 3.3.12 (Chapter 3). The O & M in Equation 3.3.8 is the sum of the system O & M (or O) values. System value added (VA) uses the

Chapter 5:Results of the Sustainability Assessment

The full set of sensitivity results can be examined by private arrangement

^{*} See Table 5.2.1 or the nomenclature for definitions.

S, L and F values of Section 4.3.6.3 in Equation 3.3.3 (Chapter 3). The system capital inclusive value added (CVA) method subtracts the system C_R values of the AC factor from the system VA values. The AC, VA and CVA indicators are indicators of wealth generation, while 1 is an indicator of capital expenditure. The economic indicators produced using this method appear in Table 5.3.1.

Table 5.3.1: Economic indicators for electricity generation from Australian fossil fuels (I = capital cost (S billion (AUD 2000)), I_R , AC, VA and CVA = relative capital cost, annualised cost, value added and capital inclusive value added (S (AUD 2000) per MWh of electricity delivered)).

Fael	Technology			Indicator		
		$\mathbf{I}_{\mathrm{res}}$	J.		• VA.,	(CVA)
	ST	2.29	450	66.9	42.7	9.27
DIC.	IGCC	1.96	271	40.4	50.5	30.4
BICE	'ST	3.17	439	79.3	73.9	41.3
	IGCC	2.95	408	61.2	79.7	49.4
	ST	3.18	441	46.4	57.3	24.6
BR	IGCC	1.81	251	32.3	54.9	36.3
	ST	1.87	258	45.6	41.3	22.2
NG	OCGT	1.54	213	50.4	35,9	20.1
	CCGT	1.31	181	34.7	45.8	32.4
LNG	ST	3.74	518	76.1	59,9	21.5
	OCGT	3,99	553	84.2	56.0	14.9
	CCGT	2.70	374	61.6	64.0	36.2

Each of the fossil fuel and technology combinations for electricity generation from Australian fossil fuels contains four subsystems. Figure 5.3.1 shows the contribution of each subsystem to the total economic indicator magnitude for each combination. The generation subsystems of OCGT systems (II and K) have negative VA, as the sales price is insufficient to repay fuel costs and operating expenses. However, OCGT systems generally operate only when the sales price of electricity is high. The distribution of subsystem contributions for the economic indicators is more even than the environmental indicators (see Figure 5.2.1). Environmental indicators tend to have dominant contributions from either the generation or mining subsystem. The contribution of the transport subsystem to LNG system's I and AC indicators includes liquefaction, ocean transport, vaporisation and pipeline transport.

The VA indicator shows both BrC systems generate greater wealth than do any of the

Sustainability of Australia's Electricity Generation Chapter 5:Results of the Sustainability Assessment



Figure 5.3.1: Subsystem contributions to the economic indicators for electricity generation from Australian fossil fuels (A-L as in Figure 5.2.1). Subsystem contributions for I and I_R are identical. Tabulated subsystem contributions appear in Section A3.2 (Appendix 1).

BIC and NG systems. The AC indicator shows that the ST-BIC system produces least wealth. Furthermore, the AC indicator shows that the CCGT-NG system produces the greatest wealth of the systems currently available, but the IGCC-BrC system offers the potential for even greater wealth generation. The I indicator shows that, of the domestic alternatives, the NG systems have the lowest capital costs, while the ST-BrC systems have the greatest capital costs. Unexpectedly, the IGCC-BrC system has the lowest capital costs of any coal (BrC or BlC) system.

When considering the export systems, the best LNG systems (CCGT-LNG) have lower I, than the best BICE systems (IGCC-BICE). The VA and CVA indicators show IGCC-BICE generating significantly greater wealth than the CCGT-LNG systems. Yet, the AC indicator scores are virtually identical. As the sales of electricity are identical for each system (1 MWh), this indicates that these two systems have equivalent wealth generation.

5.3.3 SENSITIVITY ANALYSIS

The sensitivity analysis tests the influence of three decisions on the indicator values:

Sustainability of Australia's Electricity Generation

Chapter 5:Results of the Systainability Assessment

transport distances (for LNG and BICE systems); electricity generation efficiency; and the assumed system capacity. The results of the sensitivity analysis have the same form as the main results shown in Table $5.3.1^3$.

The variation of BICE transport subsystem destination from Japan to Europe worsens all the economic indicators by a minor amount (less than 8 %). The variation of LNG transport subsystem destination from Japan to U.S.A. worsens I by a small amount (less than 5 %), VA and AC by a larger amount (less than 20 %), and CVA by a substantial amount (27-90 %). Significantly, the CVA value for OCGT-LNG systems is negligible for transport to the U.S.A. (less than \$ 1 per MWh),

Increasing the electrical generation efficiency improves all the economic indicators generally by small amounts (less than 10 %), except for NG- and OCGT-LNG systems where improvements are uniformly greater (26-260 %). The improvements in CVA of ST-BIC and NG- and ST-LNG systems are greater than for other ST systems (18-45 %).

Increasing the assumed system capacity (i.e. size) from 500 to 2000 MW obviously increases (worsens) their capital cost requirements (1) (between 151 % and 226 % of the cost of a 500 MW system). However, per unit of output, I_R (\$ per MWhpa), capital requirements improve (20 % to 40 %). Larger plants often have reduced investment and operating expenses per unit of output (economies of scale). AC, VA and CVA generally improve by 6 to 45 %, but the improvements in CVA are greater.

5.3.4 SUMMARY OF ECONOMIC INDICATOR RESULTS

The economic indicators presented here can show the influence that investment and operation of electricity generation systems have on the economy. The subsystem indicators establish the proportional influence of each subsystem on the total system indicator. Comparisons of these indicators allow the ranking of each electricity generation system for economic impact. A discussion of the results of the sensitivity analysis will establish the robustness of these rankings to data variability and assumptions. Chapter 6 discusses the implications of these results on the sustainability of each electricity generation system.

Sustainability of Australia's Electricity Generation

5.4 SOCIAL INDICATOR RESULTS

5.4.1 OVERVIEW OF SOCIAL INDICATOR RESULTS Social indicators are a third type of indicator of sustainability. These indicators present simple measures of two causes of impacts on social welfare: employment and employee health and safety. Thus, this section presents indicators of social sustainability for each of the electricity generation systems.

Section 5.4.2 presents social indicators for each electricity generation system. As for the environmental indicators (see Section 5.2), Section 5.4.2 also presents estimates of the contribution of each subsystem to the system indicators. Section 5.4.3 presents the results of the sensitivity analysis, which estimates the affect of value-based decisions on the indicators.

5.4.2 SYSTEM INDICATOR SCORES

Sections 4.3.2 to 4.3.6 (Chapter 4) contain the data required to produce the social indicators. Sections 4.3.2 to 4.3.4 contain this data on a subsystem basis, and Section 4.3.6 contains factors for its conversion to system basis (Chapter 4). No further manipulations are necessary for the social indicators. The social indicators produced using this method appear in Table 5.4.1.

Each of the fossil fuel and technology combinations for electricity generation from Australian fossil fuels contains four subsystems. Figure 5.4.1 shows the contribution of each subsystem to the total social indicator magnitude for each combination.

Table 5.4.1 shows that BIC and BICE systems have the greatest direct (ED) and indirect (El) employment. In these systems the mining, transport and generation subsystems each require comparatively greater employment than the corresponding sta_k of other systems. BIC and BICE systems also have the greatest health and safety impacts, as measured using the lost time injuries (LTI) and fatalities indicators (Fatal).

5.4.3 SENSITIVITY ANALYSIS

The sensitivity analysis tests the influence of four decisions on the indicator values: transport distances (for LNG and BICE systems); electricity generation efficiency; the assumed system capacity; and the chosen multipliers for indirect employment. The

Chapter 5:Results of the Sustainability Assessment

[‡] See footnote ^{*}, page 174.

Table 5.4.1: Social indicators for electricity generation from Australian fossil fuels (ED and EI = direct and indirect employment (10⁶ x employees), LTI = lost time injuries (10⁷ x LTI), and Fatal = fatalities (10⁹ x fatalities), all per MWh of electricity delivered).

Fuel	Technology		Indicator				
		ËD	EI	LTI	Fatal		
NC	ST	80.2	449	51.8	10.2		
	IGCC	57.2	304	30.3	5.81		
BICE	ST	133	735	85.4	14.6		
	IGCC	87.4	466	49.2	8.43		
BrC	ST	37	212	4.01	0.785		
	IGCC	28,9	151	3.11	0.785		
	ST	52.1	348	9.78	3.92		
NG	OCGT	61.7	444	13.2	4.81		
L	CCGT	42.2	260	6.93	3.18		
	ST	48.2	311	9,09	2.87		
LNG	OCGT	58.7	394	12.4	3.88		
L <u></u>	CCGT	37.7	234	6.34	2.02		



Figure 5.4.1: Subsystem contributions to the social indicators for electricity generation from Australian fossil fuels (A-L as in Figure 5.2.1). Tabulated subsystem contributions appear in Section A3.2 (Appendix 3).

results of the sensitivity analysis have the same form as the main results shown in Table 5.4.1[§].

The variation of BICE transport subsystem destination from Japan to Europe results in no change in any of the indicators. The variation of LNG transport subsystem destination from Japan to U.S.A. increases ED by 12-14 % and El by 8-9 %, and worsens the LTI and Fatal indicators by approximately 8 % and 23-25 % respectively.

Increasing the electrical generation efficiency decreases EI and ED and improves UTI and Fatal indicator scores. In general, the changes in EI, LTI and Fatal are of the same magnitude as the decreases in ED. Changes are generally between 12 % and 24 %, except for OCGT systems where the changes are nearer 60-80 %.

Increasing the assumed system capacity from 500 to 2000 MW generally decreases El and ED and improves LTI and Fatal indicator scores by less than 10 %. Improvements in Fatal are greater for ST-BrC and -IGCC (75 %) and ST-NG, -OCGT, and -CCGT (47 % to 57 %) systems.

5.4.4 SUMMARY OF SOCIAL INDICATOR RESULTS

The social indicators presented here can show the affect that the operation of electricity generation systems have on the employment and the health and safety of employees. With the toxicity environmental and wealth generation economic indicator, these social indicators allow an understanding of the relative influence of each system on social welfare. Additionally, the subsystem indicators establish the proportional influence of each subsystem on the total system indicator. Comparisons of these indicators allow the ranking of each electricity generation system for social impact. A discussion of the results of the sensitivity analysis will establish the robustness of these rankings to data variability and assumptions. Chapter 6 discusses the implications of these results on the sustainability of each electricity generation system.

5.5 SUSTAINABILITY INDICATOR RESULTS

Production of the final set of sustainability indicators entail the normalisation of the

[§] See footnote *, page 146.

Sustainability of Australia's Electricity Generation Chapter 5:Results of the Sustainability Assessment

sustainability indicators for each fossil fuel and technology combination. The normalised sustainability indicators indicate the relative importance of their magnitude to that impact within the region. Section 3.5.2 (Chapter 3) contains a discussion of normalisation, and Section 4.3.6 (Chapter 4) presents the normalisation factors for the Australian region. Figure 5.5.1 shows the normalised sustainability indicators produced when dividing the sustainability indicators in Table 5.2.1, Table 5.3.1 and Table 5.4.1 by their corresponding Australian normalisation factors. Each score represents the contribution to Australia's yearly impact on sustainability that the delivery of 1 MWh of electricity to consumers would have, if produced using that system. For example, the ST-BrC normalised CC indicator score represents the contribution that 1 MWh of electricity produced in a ST-BrC system would make to the yearly CC score for Australia. HT, ET, PM and WD are not normalised, due to a lack of comprehensive Australian data for the contributing substances of most importance.

Figure 5.5.1 shows that many of the normalised indicator scores are of similar magnitudes. All of the environmental indicators and the VA economic indicator have magnitudes between 10^9 - 10^{10} . The AC, ED, LTI and Fatal indicators each are one magnitude smaller, or 10^{11} . Only for I^{**} is the magnitude greater, indicating that this indicator has great importance.

Of the normalised environmental indicators, CC has the greatest magnitudes of any emission-related indicator, but the AD and EU indicator scores are of similar magnitudes. The resource depletion scores are of higher magnitude when considering the needs of the world (RD W, EN and EX), than when considering Australia alone (RD A). Thus, Australia is relatively rich in resources necessary for its electricity generation systems. The greatest differences for RD W and RD A are in the domestic coal cases (ST-BrC, ST-BlC, IGCC-BrC, and IGCC-BlC), highlighting the abundance of coal resources available for Australia's future development.

Of the economic indicators of wealth generation, VA and CVA are of greater magnitude than AC.

The magnitudes of the LTI and Fatal social indicators are very similar, indicating that electricity generation systems are equally great contributors to both Australian worker injuries and fatalities.



Sustainability of Australia's Electricity Generation

elaitestell toaqmit boeilann

	•					é
	ED E1 LTT Fabi	Social	* IGCC-BrC same value	E ¹ OCGT-LNG	ed using Australian dat	contains a table of the
	10 ¹¹ x 10 ¹⁰ x 10 ¹⁰ x 1	Economic	Importance of greater value questionable	D _r ST-BICE	n fossil fuels normalis	tion A3.3, Appendix 3
	EX 0		ater value implies tter sustainability	CCGT-NC	n from Australia	the scale). Seci
	DA ROW EN	•	ue implies of Gru taurability	BrsT-BIC	ctricity generatio	iere scores exceed
ninamuininasingpitataaninaa saatussaatus	H MS	nvironment	Lower val better sus	(*.01 ×	s for cle	udes wh
	B	نت		re 2.54 x 1 score 1.57	dicator	magnit
. : : : : : : : : : : : : : : : : : : :	0 PS		ator labels:	د او و در دو در من	ıbility iı	ate the
	CC V		Shaded indic	AJST-BrC (4	Sustaina	top indic
0 2 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	•		L	, · -	Figure 5.5.1:	Numbers at

Chapter 5:Results of the Sustainability Assessment

183

indicator

^{**} The normalised 1 indicator is the normalised relative capital cost (I_R) (see Table 5.3.1).

5.6 SUMMARY OF RESULTS OF THE SUSTAINABILITY ASSESSMENT

It is possible to generate the sustainability indicators for electricity generation systems. Examination of these indicators enables the identification of which system has the poorest score for each of the impacts on sustainability known to be important to electricity generation systems. CCGT-NG systems have best scores for almost all of the indicators. Importantly, the environmental indicators show that no one system has worst scores for every indicator. Thus, this evidence does not indicate that consuming BrC for electricity generation is necessarily worse for the environment than NG and BIC. The normalisation procedure allows the estimation of the significance of each indicator to Australia's annual contribution of impacts on sustainability. This procedure highlights capital requirements (I) as being most important, while all of the environmental indicators have similar significance.

The non-generation subsystems are substantial contributors to the magnitudes of the system indicators. These contributions are not uniform for all indicators and systems.

The analysis of data uncertainties indicates that the ST-BrC system indicators are less uncertain than the ST-BIC system indicators. However, the uncertainties in bot! f these system's indicators are very large. The analysis of the sensitivity of the indicators to assumptions indicates, that some indicator scores are highly influenced by: LNG allocation procedure (environmental indicators); LNG transport distance (economic); efficiency (all); and assumed system capacity (economic and social indicators). The sensitivity of the indicators of different systems is not uniform. For example, the ST-BrC system's CVA indicator improves by 6.7 % from minimum to maximum electrical efficiency, while for OCGT-NG systems the improvement is 133.8 %.

Therefore, further discussion is necessary to determine the implications of these results on the sustainability of electricity generation systems and to assess the performance of the developed method for sustainability assessment.

6. IMPLICATIONS THE **OF** SUSTAINABILITY ASSESSMENT 6.1 OVERVIEW THE **IMPLICATIONS** THE **OF** OF SUSTAINABILITY ASSESSMENT

The major outcome of Chapter 5 is the production of indicators of sustainability for electricity generation systems, which consume Australian fossil fuels. These indicators are measures of a number of impacts on the economy, the environment and social welfare. What do these measures imply about the relative impacts on sustainability of each electricity generation system? Can one identify the best or worst performing electricity generation system from indicators of individual impacts on sustainability? Section 6.2 seeks to answer these and related questions.

This thesis develops a method for the sustainability assessment of electricity generation systems. Its basis is the use of individual indicators to measure each type of impact on sustainability and it incorporates many tools designed exclusively for environmental assessments. Section 6.3 discusses the merits of the sustainability assessment method.

Section 2.4.2 (Chapter 2) claims that the measurement of sustainability should allow a more comprehensive basis for the planning of electricity generation networks. Section 6.4 discusses the potential of using indicators produced with the developed method as the basis for assessing changes to electricity generation networks. The sustainability assessment method may be useful in assessing the sustainability of systems other than electricity generation systems. Section 6.4 also discusses the potential for the sustainability assessment method to be a generalised tool for assessing impacts on sustainability.

Chapter 7 contains the formal conclusions for this chapter.

6.2 SUSTAINABILITY **ELECTRICITY INDICATORS O**F **GENERATION**

6.2.1 OVERVIEW OF SUSTAINABILITY INDICATORS OF ELECTRICITY GENERATION

This section discusses the relative sustainability of the electricity generation systems and discusses the implications of using these indicators.

Section 6.2.2 discusses the relative sustainability performance of the electricity generation systems based on the indicator and normalised indicator scores. Section 6.2.3 discusses the implications of the use of the 'life-cycle' perspective on the indicated sustainability performances. Section 6.2.4 discusses the implications of some limitations highlighted during the uncertainty and sensitivity analysis, and the gathering of data for the electricity systems.

6.2.2 SUSTAINABILITY PERFORMANCE

6.2.2.1 OVERVIEW OF SUSTAINABILITY PERFORMANCE

Section 6.2.2.2 discusses the domestic systems, which consume BIC, BrC and NG, while Section 6.2.2.3 discusses the export systems, which consume BICE and LNG.

6.2.2.2 DOMESTIC SYSTEMS

The electricity generation systems consuming brown coal (BrC) have middle to bad environmental indicators, some good and some bad economic indicators and good social indicators. Of the environmental indicators, climate change (CC), world resource depletion (RD W), and energy depletion (EN) have worst scores. Reducing CC is the goal of most ST-BrC system operators (see for example, Loy Yang Power (2000)). Proposed methods include: pre-drying the coal, using advanced electricity generation technology (i.e. IGCC) and changing to another fossil fuel (fuel switching). Drying the coal may reduce water contents from BrC to 20-25 %, thus increasing the energy density of the fuel and electricity generation efficiency (Strauss et al. (2001)), improving CC scores in a ST-BrC and reducing the capital cost of the generation subsystem (see Section 4.3.3.3.1, Chapter 4). However, emerging technologies for

Sustainability of Australia's Electricity Generation Chapter 6(Implications of the Sustainability Assessment

drying coals, such as the MTE, are not yet ready for commercialisation in Australia. There are many proposed advanced electricity generation technologies for BrC, one of which is the assessed IGCC-BrC system. The replacement of existing ST-BrC systems with IGCC-BrC systems results in a 38 % reduction of the CC indicator. Fuel switching for ST-BrC systems is not simple. The design of ST-BrC systems is such that existing generation subsystems cannot replace BrC with BlC or NG without major modifications. The discussion of BrC systems (Section 4.3.3.3.1, Chapter 4) states that the boiler temperatures for BrC are lower than for BIC and NG. Consequently, the tubing within the boiler is unable to survive at the higher temperatures caused when consuming BIC or NG. Thus, changing fuels may require the complete replacement of the boilers, which are expensive items. Potentially, replacing a ST-BrC with:

- a ST-BIC or ST-BICE system reduces CC by 13 % or 9 % respectively;
- BrC system); and
- a CCGT-NG system reduces CC by 62 %.

Figure 5.5.1 (Chapter 5) enables broader sustainability perspective on the benefits of these options for CC reduction. The IGCC-BrC option betters all indicators except value added (VA) and employment (direct (ED) and indirect (ID)). Furthermore, the I value for IGCC-BrC systems, includes new mining, transport and transmission subsystems. A new IGCC-BrC system may use the existing subsystems, from the replaced ST-BrC system, and thus its 1 value will be lessened from \$ 1.81 billion to \$ 1.04 billion (AUD 2000) (I from Table 5.3.1 and Figure 5.3.1, Chapter 5). The BIC options significantly worsen acidification (AD), eutrophication (EU), solid waste generation (SW), annualised costs (AC), VA, lost time injuries (LTI) and fatalities (Fatal) and improve RD W, EN, ED, El and capital requirements (I). The CCGT-NG option significantly worsens Australian resource depletion (RW A) and VA, but improves all other categories.

The resource depletion indicators (RD A, RD W, EN and exergy destruction (EX)) show BrC sys. ms as both worst and best, depending on the indicator. The RD W indicator shows the ST-BrC system as worst, while the similar RD A indicator, shows it to be better than all systems except the more advanced IGCC-BrC. The two indicator normalised scores have different magnitudes (RD W 10⁹, RD A 10¹¹), indicating

• IGCC-BIC and IGCC-BICE system reduces CC by 35-40 % (similar to an IGCC-

Sustainability of Australia's Electricity Generation ____ Chapter 6 Implications of the Sustainability Assessment

different relevancies to Australia. BrC consumption is the major contributor to both indicators scores. This indicates that the consumption of BrC in Australia has less relevance to the sustainability of Australia than the world. The EN indicator shows ST-BrC as the worst system, while the similar EX indicator shows ST-BrC as a midrange system. This indicates that the processes used to generate electricity from BrC consume relatively large amounts of energy resources, but are relatively more efficient at extracting useful energy (exergy) from those resources.

Of the economic and social indicators, I is worse and ED and EI are lower for BrC systems than for other fuels. The IGCC-BrC I indicator score, which includes coal drying, is much lower than the ST-BrC score, and similar to the 1 indicators of IGCC-BIC systems. However, the reliability of this estimate is unknown, as the used IGCC and MTE dryer technologies are in their infancy (see Section 4.3.3.3.1, Chapter 4). The integrated nature of the BrC generation system, with one company generally owning the mining, transport and generation subsystems, aids in limiting costs (as shown by AC). Consequently, fewer employees are necessary. For example, only one senior management team is necessary, rather than three. In addition, similar tasks in the subsystems, such as maintenance, may require only a single maintenance team.

The electricity generation systems consuming black coal (BIC) have middle to bad environmental indicators, middle to bad economic indicators and good and bad social indicators. Of the environmental indicators, CC, AD, EU and SW have worst scores, As with BrC, the goal of most companies operating BIC electricity generation systems is the reduction of CC. The replacement of existing ST systems with IGCC systems, would improve CC (by 36 %), SW (by 28 %) and EU (by ~100 %). This also results in improvements in all economic and social indicators. Changing fuel to NG significantly improves the CC indicator, and the AD, PS, EU and SW indicators. Unlike BrC, it is possible to consume NG fuel in existing ST-BIC plants, with minimal capital expenditure. This change (to ST-NG) would improve the CC indicator by 32 % over ST-BIC systems.

Improving AD requires reducing emissions of NO_x and SO₂. Of these emissions, the control of SO_2 is more common as SO_2 has a greater relative contribution to AD. The introduction of equipment to reduce SO₂ emissions ('desulfurisation') to power stations in each case would substantially improve the AD (by up to 96 %). These systems

commonly consume limestone, producing gypsum. Gypsum is a useful material; extensively used in the manufacture of plasterboard, as a soil conditioner, and elsewhere. However, the lack of the use of desulfurisation indicates that it is not economic to produce gypsum in Australia in this manner. Thus, the likely destination of the generated gypsum would thus be landfill, further increasing the already large solid waste indicators of these systems. Moreover, the desulfurisation process will consume energy, for example in the preparation and pumping of the limestone-water slurry. Therefore, the addition of desulfurisation improves AD, but correspondingly worsens SW and other emission related indicators (i.e. CC, EU and PS). Such a modification will worsen I (for the desulfurisation equipment), AC, VA and CVA (all due to limestone purchase and gypsum disposal costs). It may also improve ED and El, while worsening LTI and Fatal (through increased transport and bulk material handling activity). The use of desulfurisation in Australia is unlikely as BIC industry sources claim: 'SO_x emissions are not an issue in Australia due to a combination of low sulfur coals and low industrial intensity. There is also generally a higher tolerance of Australian flora to SO_x' (BHP Minerals Technology (2001))[†]. Therefore, the addition of desulfurisation without regulations enforcing reductions in SO₂ is unlikely.

The resource depletion indicators (RD A, RD W, EN and EX) show BIC systems as middle range performers. The RD W, RD A, EN and EX indicators show the ST-BIC system as between the performances of the ST-BrC and CCGT-NG systems. As with BrC, the RD W and RD A indicator normalised scores have different magnitudes (10⁹) and 10^{11}), indicating different relevancies to the world and Australia. BIC is the major contributor to RD W, while other fuels (i.e. diesel) contribute significantly to RD A. This indicates that the consumption of BIC in Australia has lesser relevance, than the consumption of other fuels, to the sustainability of Australia than the world.

Of the economic and social indicators, AC, CVA. LTI and Fatal are worse, while ED and EI are better for BIC systems than other fuels. While BIC systems have lower I than ST-BrC systems, the costs of mining and transport of BIC (fuel costs) are significantly higher. The product of the AC indicator scores for ST-BrC and ST-BlC from Table 5.3.1 (Chapter 5) and the fractional contributions of the mining and transport subsystems from Figure 5.3.1 (Chapter 5) provides an estimate of the costs of fuel for the two systems: \$ 16.6 (AUD 2000) per MWh for ST-BrC and \$ 36.7 (AUD

^{*} In this thesis, SO₂ and SO_x are interchangeable.

Sustainability of Australia's Electricity Generation — Chapter 6:Implications of the Sustainability Assessment

^{*} This view is simplistic (see Section 6.3.3).

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

2000) per MWh. Thus, even though the ST-BrC system consumes approximately 3 times as much coal (see Tables 4.4.20 and 4.4.9, Chapter 4), its fuel costs are less than the ST-BIC system. While relativities between prices paid for fuels are not necessarily similar to the relativities between their costs of production, examination of the relative prices paid for BIC and BrC at the generation subsystem (see Table 4.3.52, Chapter 4, column 'Transport'), indicates higher underlying costs for BIC fuel production. Comparatively poor CVA scores also indicate higher fuel costs for BIC (CVA excludes labour costs).

BIC systems have greatest employment, measured using ED and El. The mining, transport and generation subsystems each have comparatively greater employment than the corresponding subsystems for the other fuels (see Figure 5.4.1, Chapter 5). However, BIC systems also have the greatest health and safety impacts, as measured by LTI and Fatal. The greater number of employees (ED) for BIC systems could conceivably cause the high LTI and Fatal scores, as these safety incidents are associated with workers. This supposition will be true if the ratios of LTI and Fatal to ED are similar for each system (see Table 6.2.1). Table 6.2.1 shows that the LTI/ED and Fatal/ED ratios for the BIC systems are much greater than for the other systems. Mining, particularly underground, is the greatest contributor to the safety indicator scores of BIC systems (see Figure 5.4.1, Chapter 5). In BIC mining, approximately four fatalities occur per annum (National Occupational Health & Safety Commission (2002)). For NG and LNG mining less than one fatality occurs per annum (National Occupational Health & Safety Commission (2002)), while since 1993 there have been no BrC mining fatalities (Natural Resources and Environment (2001)).

As with BrC, the use of the advanced technology IGCC-BIC systems, rather than ST-BIC systems, enhances all economic and social indicators (except for ED and EI). The IGCC-BIC systems consume less coal, and thus require smaller coal mining and transport subsystems.

The electricity generation systems consuming natural gas (NG) have good to bad environmental indicators, good to bad economic indicators and middle social indicators. Of the environmental indicators, PS and EU have poor scores. NG systems emit proportionally greater quantities of NO_x than the other fuels. Emissions of NO_x heavily influence the PS and EU indicators. However, the CCGT-NG systems have much lower impacts than the OCGT-NG and ST-NG systems. Consequently, the replacement of

ED from Table 5.4.1, Chapter 5).

Fuèl	., Technolog	y R	ítio
		Delline	Fatal/ED
RIC	ST	0.646	0.127
DIC	IGCC	0.530	0 102
BICE	ST	0.640	0,110
	IGCC	0.563	0.097
BrC	ST	0.108	0.021
	IGCC	0.108	0.027
NG	ST	0.188	0.075
	OCGT	0.213	0.078
	CCGT	0.164	0.075
LNG	ST	0.188	0.059
	OCGT	0.211	0.066
	CCGT	0.168	0.054

current ST and OCGT systems with CCGT systems should be a priority, especially in locations where such plants are in operation over extended periods. Operation of OCGT systems for utility generation has hitherto been for peak power provision, but the use of OCGT systems for private electricity generation is common in Australia.

The resource depletion indicators (RD A, RD W, EN and EX) show NG systems as good and bad performers depending on the employed generation technology. The RD W indicator score for CCGT-NG systems is near best, while the OCGT-NG system's score is worse than ST-BIC systems. However, the RD A indicator scores for NG systems are much greater than the other fuels. Thus, the consumption of NG in Australia has greater resource depletion relevance to Australia than the consumption of BrC or BIC. As with BrC, the RD W and RD A indicator normalised scores have different magnitudes (10⁹ and 10¹¹), indicating different relevancies to Australia and the world. NG is the major contributor to RD W and RD A, indicating that the consumption of NG has lesser relevance to the sustainability of Australia than the world.

Of the economic and social indicators, VA indicators are poorer, while I, CVA and EI are better for NG systems than for other systems. NG systems require less capital (I) than other systems, as their generation subsystems are less complex than the

190

Table 6.2.1: Ratios of LTI to ED and Fatal to ED, to indicate the frequency of safety lacidents per employees for Australian electricity generation systems (LTI, Fatal and

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

corresponding BIC and BrC systems (requiring no fuel storage, fuel treatment or flue gas cleaning) and produce little ash. A further consequence of this reduced complexity and capital investment is reduced construction time. This in turn, reduces the interest accrued during construction and the risk of large time and expenditure overruns. The poor VA scores may indicate higher production costs for NG systems than for the other fuels, as electricity price is a constant. Yet, the AC indicator shows that production costs of NG systems are not highest. This is due to the definition of VA, which states that VA depends on two other factors: sales (S) and labour costs (L) (Equation 3.3.3, Chapter 3). A high L will worsen the VA indicator scores. The ED indicator shows NG systems to have higher employment than BrC systems and lower employment than BIC systems. In contrast, NG systems have good CVA scores due to their low I scores.

Of the domestic systems, consuming BrC, BIC and NG, CCGT-NG systems have best scores for most indicators. Nevertheless, their RD A scores are worse than for the other fuels, indicating greater scarcity within Australia for the combustion of NG over BrC and BIC. Both IGCC-BrC and IGCC-BIC systems have better RD A indicators, but their other indicators are generally much worse than CCGT-NG systems. Of the two IGCC systems, BrC systems have generally better indicator scores, especially economic and safety indicators. Yet, the IGCC-BrC system has a greater CC indicator. IGCC-BrC and IGCC-BIC systems are not yet commercially available (see Sections 4.4.2.3.1.1 and 4.4.3.3.1.1, Chapter 4). These results highlight the importance of accelerating the development of these IGCC systems, and the MTE dewatering technology that allows the IGCC-BrC to function, to fulfil their promised high performance and low cost.

6.2.2.3 EXPORT SYSTEMS

The electricity generation systems consuming export fuels, black coal (BICE) and liquefied natural gas (LNG) systems, have worse environmental indicators than their equivalent domestic (BIC and NG) systems. For the BICE systems the changes are generally minor except for the SW indicator, derived from the need to clean the coal. The coal cleaning process rejects a large amount of the coal mined in the mining stage (up to 30 % of raw coal is lost as solid waste). BICE systems also emit more particulates than BIC systems, as in the transport system a greater number of stockpiling and handling operations occur. In the BICE systems, coal stockpiling may occur at the

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

mine, domestic port, foreign port and power station. In contrast, in the BIC systems, coal stockpiling occurs only twice, at the mine and power station. For the LNG systems, the changes are considerable, due to substantially greater: fuel gas consumption in the liquefaction of NG to LNG; LNG and fuel oil consumption during transport, and NG consumption during subsequent vaporisation of LNG. The scores of the OCGT-LNG system are substantially worse than nll other natural gas and LNG systems, due to the low electricity generation efficiency of these systems.

The resource depletion indicators (RD A, RD W, EN and EX) show similar trends to those of the other environmental indicators. Yet, the RD A score for BICE systems worsens, from the performance of BIC systems, out of proportion to the other indicators. This is due to the consumption of fuel oil in the ocean transport part of the transport subsystem. The reserves necessary to produce fuel oil (i.e. crude oil) in Australia are much smaller than NG, BIC and BrC reserves. Thus, the consumption of fuel oil in Australia has a much higher relevance to sustainability (as characterised by their equivalence factors, CF) than the consumption of NG, BIC or BrC. The same does not occur for RD W, indicating that fuel oil has lesser relevance to sustainability for the world than for Australia.

Of the economic indicators, 1 and AC worsen, but the higher prices paid for BICE, LNG and electricity in the export destination country (Japan, see Table 4.3.52, Chapter 4) ensure they have better VA and CVA scores. Of the social indicators, the BICE systems have better employment (ED and EI) scores and worse safety (LTI and Fatal) scores than the corresponding BIC systems. The cause of these effects is the necessity of mining greater quantities of coal in the mining subsystem, to allow for coal cleaning. Meanwhile, the LNG systems have slightly better social indicators, than NG systems, due to the lower direct employment (ED) necessary for LNG systems.

Of the export systems, consuming BICE or LNG, LNG systems have better scores than ST-BICE systems (except for OCGT-LNG systems). Switching to IGCC-BICE systems would enable better indicator scores than the ST-LNG and OCGT-LNG systems, except for the SW and safety (LTI and Fatal) indicators. However, greater benefit results from switching to CCGT-LNG systems, which have best scores for all indicators except for RD A and the employment indicators (ED and EI). As earlier shown when discussing domestic NG systems, the consumption of NG in Australia has greater relevance to Australia than the consumption of BIC. Consequently, the RD A

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

indicator for CCGT-LNG systems is worse than both the IGCC-BICE and ST-BICE systems.

6.2.2.4 OVERALL

The indicators of sustainability in Chapter 5 indicated no one system has either best or worst environmental indicator scores for every indicator. Comparing the relative contributions to each sustainability indicator shows the relative performance of systems for each indicator (see Sections 6.2.2.2 and 6.2. 3). The use of the normalisation method establishes that most impacts are of similar importance to Australia's yearly impacts on sustainability (see Section 5.5, Chapter 5). The importance of the capital expenditure indicator (I) to Australia appears greater than are any of the other indicators. Thus, electricity generation projects are highly significant investments of capital. The normalised SW indicators of BIC and BICE systems also have similar magnitudes to the normalised I indicator. However, the source of the normalisation value for SW (see Table 4.3.49, Chapter 4) does not detail all inclusions. Thus, it may exclude solid wastes from electricity generation and other similar industries and understate Australian annual solid waste generation. As this normalisation factor is uncertain, the magnitude of the normalised SW indicator is also uncertain. The basis for most recent comparisons of electricity generation has been economic indicators, such as AC, and CC (see Section 1.1, Chapter 1 and Section 2.4, Chapter 2). As the normalised indicators of other impacts have similar magnitudes, the basis for these comparisons should be broader, including all of these indicators.

6.2.3 'LIFE-CYCLE' PERSPECTIVE

Sustainability assessment requires the use of the 'life-cycle' perspective (see Section 2.3.5, Chapter 2). A glance at the subsystem contributions figures of Chapter 5 (Figures 5.2.1, 5.3.1 and 5.4.1) shows that substantial proportions of each of the indicator scores are from subsystems other than the generation subsystem. Some indicators would lose all relevance if measured with a boundary restricted to the generation subsystem alone. For example, the social indicators, LTI and Fatal, have minimal contributions from the generation subsystem. Without the mining subsystem contributions, these indicators for all systems are very similar. Thus, the knowledge that BIC and BICE systems have

Sustainability of Australia's Electricity Generation ___ Chapter 6:Implications of the Sustainability Assessment

considerably greater LTI and Fatal indicator scores would have been lost. Additional examples with important contributions from non-generation subsystems include: for LNG systems, energy c_1 sumption in mining due to the liquefaction of NG and for BICE systems, solid wa c_2 incration in mining due to coal cleaning.

Contributions to impacts on sustainability may be large in real terms even when they have small indicator scores, if the output of the system (i.e. electricity) is large, as is typical for these systems. The product of the indicator score (i.e. kg per MWh) and the output (i.e. MWh per annum) represent the real impact on sustainability (i.e. kg per annum).

Subsystem contributions may indicate where the best opportunities are for improving performance. For example, trying to improve the health and safety (LTI and Fatal) performance of a B₁C or BlCE system by improving performance in the generation subsystem will have little effect (see Figure 5.4.1, Chapter 5). The most effective way to improve system health and safety performance is by improving health and safety performance in the mining subsystem. Accordingly, an assessment of the impacts of sustainability caused by electricity generation systems, determined with these subsystems absent, would be misleading.

The system boundary used is not strictly speaking compatible with the 'life-cycle' perspective. Section 4.2.3 (Chapter 4) highlights exclusions from the analysis, such as construction, decommissioning and secondary processes. The ability to exclude parts of the 'life-cycle' is necessary to ensure both the data collection and sustainability assessment remain manageable (Section 3.2.3.3, Chapter 3).

6.2.4 LIMITATIONS

6.2.4.1 UNCERTAINTY ANALYSIS

Uncertainty analysis is necessary to establish the robustness of the sustainability indicators to uncertainty in data used in their calculation (see Section 3.5, Chapter 3). Yet, there are no established methods for uncertainty analysis of assessments using 'life cycle' boundaries (see Section 3.2.3.7, Chapter 3). Therefore, this assessment investigates some uncertainty appraisal methods for the CC, AD and RD W environmental indicators of the ST-BrC and ST-BlC systems. The results of this investigation are in Section 5.2.3 (Chapter 5).

Sustainability of Australia's Electricity Generation ____ Chapter 6:Implications of the Sustainability Assessment

The first method is a numerical Monte-Carlo analysis (see Section 3.2.3.7.2, Chapter 3). The results indicate that there is large uncertainty in the indicators, as characterised by the standard deviation (SD) (see Table 5.2.2, Chapter 5). Thus, it is possible to obtain a wide range of scores for the three indicators. Moreover, these ranges overlap. Therefore, while the indicators in the first line of the table, calculated using average values of input data, show worse CC and RD W indicators for ST-BrC systems, but a better AD indicator, it is possible for these conditions to reverse. For example, if a ST-BIC system has low electricity generation efficiency, its CC and RD W scores can be worse than for a ST-BrC system having high efficiency. The normalised difference method (see Figure 5.2.2, Chapter 5) shows, in an objective manner, that there is a greater probability that ST-BIC systems have better CC scores than ST-BrC systems. In all likelihood, the use of this method on other indicators will show that the probability that the relative performances of the systems shown in Chapter 5, is logator most, if not all, indicators. In these other cases, the probabilities are likely to be similar (for example, there may be 45 % probability that a ST-BrC is better than a ST-BIC, and a 45 % probability that the ST-BIC is better than the ST-BrC for the same indicator). Thus, this type of uncertainty analysis is useful in proving that the identified differences between the systems are real and likely to occur in practice.

A separate Monte-Carlo analysis of contributions to the uncertainty in the CC indicators establishes that electricity generation efficiency is the largest contributor for both systems (see Table 5.2.3, Chapter 5). Thus, the variation of efficiency provides a fair approximation of the quantitative uncertainty of the systems. The removal of uncertainty due to this efficiency established that the scale of uncertainties due to other data sources for both systems is similar. Thus, the difference in the scale of uncertainty for both systems (BIC 309, BrC 102) is due mainly to the efficiency data.

The greater uncertainty in the BIC efficiency data is due to the wide range of efficiencies in Australian ST-BIC systems. Electricity generation efficiency varies with both plant age and design. Even the efficiency of an individual power station may also fluctuate during operation, with changes in operating conditions. Some important influences on efficiency during operation include variances in fuel quality, ash deposition on boiler tubes, startup and shutdown, and electricity generation at design, high or low levels. For the ST-BrC, the CO₂ emission factor for BrC combustion also is a significant contributor. The moisture content of the coal is the major influence on the

CO₂ emission factor. Three mines produce BrC and the moisture content of BrC in one mine is under 60 %, while in another it can be up to 66 % (by mass) (see Figure 4.3.4, Chapter 4). Such variations in moisture content affect the amount of coal consumed, and thus the consumption and emission of carbon for the same output of electricity.

The qualitative method of Wrisberg (see Table 5.2.4, Chapter 5) shows that reliability and completeness are the major contributors to qualitative error for both systems. The cause of the poor reliability scores is the use of some data without an identified source plant (unverified data) and data based on assumptions. The cause of the poor completeness scores is the lack of data for some facilities within Australia.

The qualitative method of Rousseaux et al. (see Table 5.2.5, Chapter 5) shows that the data quality of the ST-BrC systems is better than for the ST-BlC systems. The ST-BrC system uses mainly data obtained directly from the small number of existing systems, of high reliability. Contrariwise, the ST-BIC system uses mainly reports published by a minority of the many existing systems. Thus, the reliability and completeness scores of the ST-BrC system data are better. Both systems have high variability scores for the temporal indicator indicating a wide spread of data temporal scores. Most of the data sources for these systems are recent (within three years of the year of study) having data indicator scores of 1 or 2 (see Table 3.2.3, Chapter 3). However, it also includes a number of very detailed older sources (10 years from the year of study) with data indicator scores of 5. Similarly, the variability score of the ST-BrC system is high as its data includes a small number of estimates (indicator score 5), while most data are plant measurements (indicator score 1). Thus, this method shows more promise in locating and identifying the cause of deficiencies in data quality than does the method of Wrisberg.

The uncertainty assessment does not include the NG systems. Yet, estimation of the likely outcome is possible, using the knowledge collected from the ST-BrC and ST-BlC assessments. Variation of electrical generation efficiency allows the estimation of numerical uncertainty for NG systems. This variation is part of the sensitivity analysis (see Section 6.2.4.2). Qualitative uncertainty is dependent on the qualities of the data sources. The NG systems have a greater reliance on aggregated sector-wide data, due to a lack of public, site-specific, information sources. This evidence suggests that the qualitative uncertainty of the NG systems could be greater than for both the BrC and BIC systems.

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

6.2.4.2 SENSITIVITY ANALYSIS

Sensitivity analysis is necessary to establish the robustness of the sustainability indicators to decisions and assumptions in their derivation (see Section 3.5, Chapter 3). Therefore, sensitivity analyses have investigated the influence of key decisions and assumptions in the preparation of indicators for each electricity generation system (see Sections 5.2.4, 5.3.3 and 5.4.3, Chapter 5).

Of the environmental indicator sensitivity results, the allocation method and transport distance for LNG systems are the important influences. Using the economic allocation method considerably improves the indicators of LNG systems. However, the economic allocation method predicts a much lower consumption rate of natural gas in the LNG mining subsystem than observed in practice. Thus, the indicators determined using this method are not as representative as those produced using the allocation method based on gas consumption rate. The indicator most affected by varying LNG transport distance is RD A. LNG systems consume NG during transport of full LNG vessels and fuel oil during transport of empty LNG vessels (return journeys after delivery). Section 6.2.2 states that the consumption of NG and fuel oil have greater relative impact on sustainability for Australia than does the consumption of BIC. Of NG and fuel oil, fuel oil has greater relevance to sustainability for Australia. Thus, the RD A indicator is particularly susceptible to transport distance. However, changes to the other indicators are minimal and the magnitude of the normalised RD A indicator is low. Consequently, the robustness of the sustainability performances derived from the environmental indicators in Chapter 5 is high.

Of the economic indicator sensitivity results, the transport distance for LNG systems and assumed system capacity are the important influences. The indicator most affected by varying LNG transport distance is CVA. CVA is the difference between VA and annualised capital (C_R) (see Equation 3.3.14, Chapter 3). When extending LNG transport distance, VA worsens because the operating costs of transport increase. Moreover, it takes greater time to deliver each load of LNG by a transport vessel, requiring more vessels to deliver LNG at the same rate. Thus, greater I is necessary. Both of these changes worsen CVA. The assumed system capacity has important effects on all economic indicators. Each economic indicator includes an estimate of operating or capital cost in its method. All of the operating and capital cost estimations

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

are power law correlations (see Section 3.3.3.3 and 3.3.3.4, Chapter 3) based on either production capacity (Q) or production $(P)^{\ddagger}$. Thus, assumed system capacity should be a dominant influence on the robustness of the economic indicators. However, an analysis of relative economic performance with very large changes in assumed system capacity shows little variation, except in VA and CVA (see Table 6.2.2). The variations in VA and CVA occur because the increase in electricity sales with capacity (and production) is linear, while increases in costs have power law dependence on capacity. As the indicators show only minor sensitivity to wide changes in these important variables, the robustness of the sustainability performances derived from the economic indicators in Chapter 5 is high.

Table 6.2.2: Changes in the rankin generation systems, using the eco changes the assumed system capa improves one place, 0 = no change i

Fuel	Technology		- Ind	icator	÷.,.,
		<u> </u>	AC.	VA	CVA
BIC	ST	0	0	2	-1
	IGCC	1	1	-2	0
BICE	ST	1	0	0	2
	IGCC	0	-1	0	0
BrC	ST	-1	0	-3	1
	IGCC	1	2	-2	-1
NG	ST	-2	-1	-2	-4
	OCGT	0	-2	0	-3
	CCGT	0	0	-2	-4
LNG	ST	0	0	3	3
	OCGT	0	0	5	3
	CCGT	0	1	1	4

Of the social indicator sensitivity results, the transport distance for LNG systems and assumed system capacity have important influences on the Fatal indicator. However, there is a substantial difference between the poor performance of BIC systems, and the better performing NG and BrC systems. Therefore, the relatively small changes to the Fatal indicator, resulting from varying the transport distance, make little difference to

* Production capacity directly influences production rates.

ngs (1 = best, 12 = worst) of different electricity
nomic indicators, when the sensitivity analysis
city from 500 MW to 2000 MW (1 = ranking
in ranking, -2 = ranking worsens by two places).

their relative social performances.

The final sensitivity analysis varies the electrical generation efficiency of the generation subsystem. Section 6.2.4.1 shows that for these systems, varying electrical generation efficiency provides a rough estimate of the uncertainty of the indicators because of numerical uncertainties in the data used to generate them. For all the systems, the resultant indicator variations are generally between 90 and 110 % of the indicator score for the average efficiency, except for OCGT systems. As with uncertainty assessment, the ranges of many indicators overlap. Section 6.2.4.1 discusses overlap between the CC indicators of ST-BIC and ST-BrC systems. The sensitivity analysis shows that similar overlap occurs for the IGCC-BrC and IGCC-BIC systems. These systems are currently in development, indicating that it may be possible to develop an optimised IGCC-BrC system with CC indicators similar to those of an IGCC-BIC system.

The performance of OCGT systems with low efficiencies (< 25 %) is concerning, given the widespread use of such turbines in industry. The impacts of the lowest efficiency OCGT gas systems are worse than the least efficient coal systems. Hence, from an environmental viewpoint the replacement of low efficiency OCGT gas systems should be a priority, where these operate over extended periods. A particular matter of concern is the current trend of recycling old, inefficient turbines for use in OCGT systems for peak power generation, justified by their projected occasional use. The inefficiencies of the turbines themselves compounded with the inherent inefficiencies of the OCGT technology result in very poor environmental performance. Once installed, unforeseen failures of normally operating, higher efficiency plant may lead to more frequent use of the OCGT system than originally envisioned, with serious environmental penalty.

6.2.4.3 ADDITIONAL IMPACTS

The sustainability assessment aims to include impacts of most importance to electricity generation systems (see Section 1.1, Chapter 1). Selection of these impacts has been by critical examination of previous environmental, economic and social assessments of electricity generation (see Section 2.5, Chapter 2). Nevertheless, electricity generation systems contribute to a number of other impacts. Some environmental impacts excluded because of a lack of established or quantitative data include:

- Electromagnetic field (EMF) radiation effects;
- Dioxins in foods;
- Acid mine drainage; and
- Subsidence.

EMF radiation effects on humans, plant and animals are not well understood. Nevertheless, some attribute increases in the incidence of leukaemia (cancers of the blood) in young children to EMF radiation from transmission cables (WHO (2001)). Dioxins are a persistent chemical that has a number of effects on human health, including cancer (WHO (1999)). Many types of industrial processes emit dioxins and electricity generation systems are not the most important emitters. Dioxins concentrate in fatty foods, such as butter, and recent tests have shown that Australian butters have low dioxin concentrations (Kalantzi, et al. (2001)). This indicates that dioxin emissions from Australian electricity generation are of comparatively low importance (perhaps due to a lower concentration of emitters). Acid mine drainage occurs when water dissolves some components in coal, increasing its own acidity and toxicity. Acid mine drainage has particularly been a problem for BIC systems, due to the great amounts of overburden and coal moved and stored during mining. Such problems can be minimised through proper management of water flows at mines. Subsidence is the lowering of surface levels through either underground mining or artesian water removal. Underground mining can lead to rapid subsidence and the loss of property and lives. Artesian water removal leads to slow subsidence, over a wide area. Subsidence is a substantial problem for BrC systems (for example, Gloe (1984)).

The economic indicators exclude the risk of failure. Risk is important in economic assessments, as projects with high risk (of failure) are unlikely to proceed to construction. Important contributors to economic risk are: the magnitude of I; the time for construction; sensitivity to capital overruns (i.e. new technologies and fir t of a kind plant have high sensitivity); and the flexibility to adapt to changes in regulatory or product requirements and to produce wealth if major function fails. NG systems have lower I and construction times than do any of the BrC and BIC systems. Lower I and construction times reduce the risk of cost rises during construction. Sanction estimates of I for new projects characteristically have an accuracy of ± 10 %, so overruns of up to 10 % are not uncommon. An increase of 10 % in I for NG systems is an increase of \$

Sustainability of Australia's Electricity Generation ___ Chapter 6:Implications of the Sustainability Assessment
Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

130 to 190 million, while for a ST-BrC system it is \$ 320 million. The IGCC-BrC and IGCC-BIC systems are both new technologies. Actual I for new technology systems is often greater than predicted, and thus have higher risk. Another aspect of risk is flexibility. As NG is a widely used fuel and chemical feedstock, the NG production and delivery parts of the NG systems may still generate wealth should electricity generation itself become unprofitable. Thus, NG systems have greater flexibility to invest in plant improvements as new technology develops, or in additional plants or other sectors. Investing in additional plants or other sectors spreads risk between the investments (thus, reducing the risk of the entire investment failing to generate wealth), and may enhance wealth generation from the same amount of I. Quantification of this risk is known as risk analysis (Hertz (1964)), and Van Groenendaala and Kleijnen (1997) discuss some methods for risk analysis for large investment projects, including electricity generation and NG transmission subsystems. The measurement of economic risk is beyond the scope of the sustainability method applied in this thesis.

Section 3.4.3 (Chapter 3) identifies that the LTI and Fatal indicators ignore health effects on workers and the nearby public that occur later in life, are long term or 'chronic'. One important effect is the so-called black lung disease, which affects workers exposed to BlC dust (see Page *et al.* (1997)). As the LTI and Fatal indicators of BlC systems are already worse than the other systems, the exclusion of this impact from the indicators would augment this effect. BrC dust shows no such effects on health (Finocchiaro *et al.* (1997)).

Each of the impacts identified here are difficult to measure using indicators. Nevertheless, the method should allow for the inclusion and qualitative description of such impacts in assessments of sustainability performance. None of these impacts should significantly affect the relative sustainability performance of the electricity generation systems as measured by the qualitative indicators reported in Chapter 5.

6.2.4.4 SCOPF OF THE ASSESSMENT

The sustainability assessment aims to compare electricity generation systems consuming Australian fossil fuels (see Section 1.2, Chapter 1). As the indicators describe Australian average electricity generation systems, individual systems are likely to have different sustainability profiles. The use of uncertainty and sensitivity analyses

Sustainability of Australia's Electricity Generation _ Chapter 6:Implications of the Sustainability Assessment

enables an appraisal of the upper and lower limits of indicators for these individual systems. Used within Australia are other types of electricity generation technology, including cogeneration (for example, Morwell, Vic.), fluidised bed (Redbank, NSW), many internal combustion engine systems (private), hydroelectricity and solar and wind electricity generation. These systems, apart from hydroelectricity (8.6 % in 1997/98, Electricity Supply Association of Australia (1999)), produce very little of the total electricity generated annually. Therefore, the domestic systems (BIC, BrC and NG) examined are generally representative of the bulk of Australia's electricity generation systems.

The export systems (BICE and LNG) assume that the foreign generation and transmission subsystems are identical to those within Australia. However, this assumption is not accurate in all cases. For example, BICE consuming electricity generation systems in Japan use flue gas desulfurisation to reduce AD. Some foreign generation systems also treat flue gas for NO_x , thus improving AD, PS, EU, HT and ET impacts. Consequently, the export system indicators are less representative of actual performance than arc the domestic system indicators. The export systems are likely to have improved environmental indicator scores, but worse economic indicator scores (due to the additional I and O & M for the desulfurisation and NO_x flue gas treatment equipment).

The assessment includes within the total system boundary impacts from generating electricity overseas with exported Australian fossil fuels. The assessment compares the impacts on sustainability caused by consuming Australian fossil fuels in different ways, without seeking to apportion responsibility between fuel supplier (Australia) and electricity generator (foreign) for respective impacts.

6.2.4.5 SUBSTITUTABILITY OF ELECTRICITY GENERATION SYSTEMS

This sustainability assessment method compares the performance of average electricity generation systems. Section 6.2.2 discusses options for replacing systems with poor indicator scores with other systems with better indicator scores. This assumes that such system replacement is feasible. Using this assumption, CCGT-NG systems, which have best scores in most categories, would increase their share of production. Yet, the assumption of substitutability may not be valid in many cases, because of a number of constraints. This section discusses these constraints on the substitutability of electricity

generation systems.

There are limitations in both the amount of NG resources available and their delivery. At the consumption rates of 1999/2000, Australia had reserves for 105 years $(AGA (2001))^{\delta}$. In comparison, Australia had reserves of BIC and BrC, at the consumption rates of 1997/98, for 258 and 744 years respectively" (ABA (2001)). Conversion of all electricity generation systems to CCGT-NG would reduce the NG reserve to 55 years^{††}. This estimate excludes recent and planned expansions of LNG and other natural gas intensive exports, the possible use of natural gas as a source of hydrogen, and assumes that electricity demand does not increase, which would further erode this value. Complete depletion of Australia's resources of NG would require importation of LNG (if available) or could demand more difficult and expensive NG extraction and transportation strategies (such as, the pipeline transportation of gas from Western Australia to the eastern states). Both of these would substantially worsen the sustainability performance of electricity generation (see Figure 5.5.1, Chapter 5 for LNG case)^{‡‡}. On a world scale, while BIC and BrC coal reserves are sufficient to provide electricity at any rate of consumption for centuries, NG reserves are much lower, but sufficient at current rates for several decades^{§§} (IAEA (2000)). Thus, worldwide conversion of electricity generation systems to CCGT-NG may cause NG supplies to fail sooner than predicted (Hammond (2000), IAEA (2000), IChemE (2002b)). Even before its eventual depletion, as other nations deplete their domestic supplies of NG and become more dependent on imports, the price of NG will increase, and thus affect the economic impacts on sustainability for NG consuming systems (IChemE (2002b)).

The transport of NG from mining to generation subsystems is by pipelines. Pipeline

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

projects in Australia presently have uncertain paths to gain approval for their construction (Ministerial Council on Energy (2002)). Consequently, there is little certainty in new pipeline projects to increase NG supply to major centres on the east coast of Australia. Therefore, as CCGT-NG plants are major NG consumers and usually close to major centres, there are no large CCGT-NG plants on the east coast, nor are there proposals for new plants. Equation 4.3.44 and Equation 4.3.46 (Chapter 4) indicate that the capital and operating costs of pipeline transport increase with distance, L, for the same NG delivery (Q and P). For example, in the ST-NG 1.47 Mtpa of NG delivery is over 927 km, for capital cost of \$ 234 million (AUD 2000). Doubling this distance, to 1854 km, increases the capital costs to \$ 389 million (AUD 2000). Consequently, the economic indicators of sustainability will worsen, and the average price of natural gas may have to rise to ensure the profitability of natural gas transportation.

If the replacement generation system is to be a coal system fuelled by a different coal (e.g. a BIC system replacing a BrC system) then there are two options. The first is to locate the replacement system near its resource and deliver electricity with new transmission lines. This may result in the need for very long transmission lines (e.g. from BIC mines in NSW to Victoria). Yet, transmission losses, and consequently the environmental effects per unit of delivered electricity, increase substantially with distance. Thus, these transmission losses may negate the improvements obtained by switching fuels. As with NG transport, it may be difficult to obtain approval such a project (Ministerial Council on Energy (2002)). The alternative option is to locate the replacement system at the location of the replaced system. This will extend the transport subsystem. Should that fuel be BIC, a comparison of the scores for BIC and BICE systems (Figure 5.5.1, Chapter 5) indicates that the use of domestic quality, BIC would be preferable over BICE. Critically, should that fuel be BrC, its transport over any great distance is uneconomic. Thus, for BrC only the first option is available. To compare the actual sustainability impacts of these options, modifications to the electricity generation systems presented would be necessary.

Replacement of any system because of its sustainability performance is difficult in privately owned electricity generation networks as there is no incentive for investment to improve performance (IChemE (2002b)). In Australia, as in the U.K. (IChemE (2002b)), the oldest systems have lowest debt, can sell their electricity to the market at

⁵ These ratios are notoriously unreliable, and often drastically underestimate true resource life, as the resource base of the planet is not well defined (Hammond (2000)). For example, the resource life of oil from Bass Strait has increased from 20 to 23 years over the last 5 years (NRE (VIC) (2000)).

^{**} These are underestimates as they include only reserves currently expected to be profitable. Under the same assumptions, NG resources will last only 48 years.

^{††} This value is an estimate based on natural gas consumption and reserves from AGA (2001) (Table

^{2.10),} electricity generation by fuel statistics in Electricity Supply Association of Australia (1999), and system natural gas consumptions from this work (see Section A2.5, Appendix 2).

^{‡‡} This is also true for depletion of BIC and imports of BICE (Figure 5.5.1, Chapter 5).

^{§§} Estimates vary from 40 to 70 years (Hammond (2000)).

Sustainability of Australia's Electricity Generation ____Chapter 6:Implications of the Sustainability Assessment

lowest cost, and thus dominate. These plants generally have worst impacts on environmental sustainability. Thus, for Victoria, NG use was for about 8 % of electricity requirements in 1995, before full privatisation (Victorian Government (2001)). However, in 1999, after privatisation, only 0.2 % of electricity used NG (Victorian Government (2001)). The remainder was BrC use in ST-BrC systems, which had worse environmental impacts, but lower costs. Compounding the effect, the owners of these systems have no incentive to improve performance, as it requires investment capital (for plant improvements or development and proving of new technologies), and would increase their costs (Matthes and Timpe (2000)). For example, Section 6.2.2.2 shows the IGCC-BrC system as a substantial improvement over ST-BrC systems. Yet, substantial capital investment is required to advance the coal drying (MTE) and IGCC technology to commercial scale. The result of these influences is an electricity generation network providing electricity at lower costs to consumers. Under these conditions, new systems will have low profitability and thus there is no incentive for investment in them (IChemE (2002b)). The United Nations (2002) recognise these problems and recommends the removal of these impediments to improved sustainability. The difficulty lies in achieving this in a competitive electricity market, where governments should not apply direct regulatory control (IAEA (2000)). IAEA (2000) suggests: 'price controls, taxes, tradeable emission permits, development and transfer of advanced electricity generation technology, and directed research and development funding schemes' as methods for governments to achieve sustainability goals.

The operating costs exclude governmental taxes, on resources and company profits (see Section 3.3.3.4, Chapter 3). Taxes are uncertain sustainability impacts as they are dependent on many variables, including resource type (different tax rates apply for BrC, BIC and NG) and location (different states apply different tax rates). Moreover, taxes have both a negative sustainability impact on operators, by decreasing profitability, and positive sustainability impact on society, when spent on social welfare services, such as health, education, housing, etc. Including taxes may affect the relative economic sustainability performance of the systems.

6.3 THE SUSTAINABILITY ASSESSMENT METHOD

6.3.1 OVERVIEW OF THE SUSTAINABILITY ASSESSMENT METHOD

This thesis has developed a sustainability assessment method with many new and enhanced features. Its basis is the use of individual indicators to measure each type of impact on sustainability. The method incorporates uncertainty and sensitivity analysis, and normalisation tools designed for use in other assessments. This section discusses the applicability of the use and integration of these methods to assessment of the sustainability of electricity generation systems.

Section 6.3.2 discusses the effectiveness of the indicator method in assessing the sustainability of electricity generation systems. The chosen indicators dictate the effectiveness of an indicator method. In the assessment, more than one indicator represents some of the impacts on sustainability. Section 6.3.3 discusses the indicators chosen for the method. Section 6.3.4 discusses the suitability of the sensitivity and uncertainty methods for the sustainability assessment method. Section 6.3.5 discusses the use of normalisation both as a method for presenting results and for assessing the relative sustainability of electricity generation systems.

6.3.2 THE INDICATOR METHOD

Section 6.2 shows that the developed method measures impacts on sustainability, and compares the relative sustainability of different options and the relative importance of different impacts on regional sustainability. However, the method neglects some impacts known to be important to these systems, but difficult to quantify (Section 6.2.4.3). Modification of the method is necessary to include these impacts in evaluations of sustainability performance.

Section 2.3.2 (Chapter 2) identifies other methods for sustainability assessment. Applications of these other methods for sustainability assessment have been to dissimilar systems, limiting opportunities for direct comparisons. Application of the developed sustainability assessment method to these dissimilar systems may allow verification of this method and realisation of its advantages and disadvantages.

6.3.3 THE INDICATORS

Multiple indicators in Chapter 5 represent some of the impacts identified as important to electricity generation systems. For example, the HT, ET, LTI and Fatal indicators represent different aspects of the 'toxicity' and 'health and safety' impacts: HT on the health of humans (employees and the general population); ET on the health of plants and animals***; and LTI and Fatal on the safety of employees.

The RD W, RD A, EN and EX indicators represent the 'resource depletion' indicator category. This category indicator is important for fossil fuel electricity generation systems as they consume large quantities of fossil fuels. The category indicator attempts to measure the impact on sustainability of consuming materials now, and thus limiting their availability in the future. Thus, it should include: present resource scarcity and the needs of the future for that resource. Yet, estimating the needs of the future is problematic: planners have difficulty estimating needs for even a few decades^{†††}. Sustainability assessment requires estimation of the needs of people many generations into the future (50-100 years). Thus, present methods are measures of resource scarcity. EN is a measure of the energy resources consumed by the process. It is an un-weighted measure, assuming that all fuels are interchangeable (energy is energy no matter the source). However, BrC systems cannot consume low moisture content fuels like NG or BIC (see Section 6.2.2). Similarly, NG systems cannot burn solid fuels like BrC or BIC, and the mining and transport equipment present in all systems cannot suddenly switch their fuel, from say diesel to BrC. Furthermore, it does not account for the relative scarcity of particular resources and excludes non-energy resources. While, in electricity generation systems non-energy resource consumption is minor, other types of systems may consume large quantities of such materials. Thus, EN is an incomplete measure of resource depletion. EX is an advanced energy measure, which estimates the loss of useful energy when consuming resources. Its advantages over EN include the ability to include other resources and differentiate between different types of energy sources. The inclusion of other resources is by estimation of the lost useful energy

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

during their production. Unfortunately, these values change with time, as the technology for the production of resources develops. In spite of these additional benefits, like EN, EX is a measure of resource consumption, not resource scarcity. The RD W and RD A indicators are both measures of resource scarcity. Each indicator applies the same method, but the information used to generate equivalence factors (CF) factors for RD W is global, while for RD A it is Australian (see Equation 3.2.1, Chapter 3). The RD W and RD A indicators for electricity generation systems have substantially different scores (see Figure 5.5.1, Chapter 5). Section 6.2.2 attributes this difference to Australia's high level of resources, in comparison to its consumption rates. Comparing the regional (RD A) and world (RD W) indicators and expressing them as a ratio (Table 6.3.1), allows estimation of the possibility of future dependence on resources from outside the region. This indicates that LNG systems have highest probability of being the first type of system requiring resources from outside Australia, and BrC systems have lowest probability. Thus, both of these indicators have merits that necessitate their inclusion.

and RD W from Figure 5.5.1).

Fuel	Technolog	y Ratio
סומ	ST	0.0147
BIC	IGCC	0.0174
	ST	0.0719
BICE	IGCC	0.0682
	ST	2.24 x 10 ⁻³
ыс	IGCC	2.24 x 10 ⁻³
- · · ·	ST	0.118
NG	OCGT	0.121
	CCGT	0.121
	ST	0.204
LNG	OCGT	0.206
	CCGT	0.206

Section 6.2.2 assumes that if a system has lower RD A scores than RD W scores, its resource consumption is of lower importance to Australia than the world. Yet, this assertion ignores the fact that Australia is dependent on imports of other resources (such

Table 6.3.1: Ratio of RD A to RD W to show the relative probability of Australia requiring to import fossil fuels for each of the electricity generation systems (RD A

Section 4.4.6, Chapter 4 explains the exclusion of HT and ET from the sustainability performance discussion (Section 6.2.2) for these systems.

^{†††} The State Electricity Commission of Victoria, for example, predicted in 1976 that demand in 1992 would reach 52500 GWhpa, with a peak demand of 9600 MW (SECV (1976)). The actual demand in 1992 was 39109 GWhpa, with a peak demand of 6005 MW (SECV (1993)).

Sustainability of Australia's Electricity Generation ____Chapter 6:Implications of the Sustainability Assessment

as crude oil). Australia exports its abundant resources to fund the importation of scarcer resources. Consequently, consumption of any Australian resource is important.

The AC, VA and CVA indicators represent the 'wealth generation' indicator category. In the sustainability performance discussion (Section 6.2.2) I is proven important, but VA excludes I. Thus, while VA indicates that CCGT-NG systems provide less wealth than BrC and BIC systems, both indicators that include I, AC and CVA, indicate BIC systems to provide least wealth. Thus, the CVA indicator is important to ensure inclusion of these capital-related effects. The AC indicator is also an indicator of wealth generation, but it includes I and labour costs. Thus, it is a more complete indicator of wealth generation than either VA or CVA. However, AC is only an indicator of wealth generation because of the assumption that the value of electricity sold from each system is equivalent. When this assumption is inaccurate, then net present value is more appropriate. Nevertheless, each of the included indicators provides unique information about wealth generation performance.

The ED and EI indicators represent the 'employees' indicator category. ED indicates the numbers of employees directly employed by the system. High ED is a positive social impact in one sense, spreading wealth generated widely. Alternatively, high ED may count as a negative economic indicator, reducing the system's wealth generation. El indicates the numbers of employees whose employment occurs because of the system. It thus includes: employees of contractors, employees of companies selling materials and services to the industry, and employment generated through their spending. El is thus a more apt measure of the employment impact of these systems. Nevertheless, producing EI uses industry wide averages, and thus may not accurately estimate the EI that occurs for these systems. For example, the EI factor for the generation subsystems, 'Electricity', does not differentiate between BrC, BIC and NG systems. These systems have vastly different numbers of employees for their output (see Figure 5.5.1, Chapter 5). Nor does the 'Electricity' factor differentiate between the generation or transmission subsystems^{‡‡‡}. Thus, the reported EI scores are less certain than are ED scores. Therefore, when presenting El indicator scores, ED and EI scores should both appear.

6.3.4 UNCERTAINTY AND SENSITIVITY ANALYSIS

Data quality assessment is important for the development and acceptance of LCA as a decision-making tool (see Section 3.2.3.7.1, Chapter 3). It is also important for sustainability assessment, as the method uses LCA to estimate environments, indicator scores. Yet, assessments of data quality are rare, and there is no consensus on the appropriate methods to use. Each of the methods tested indicates greater uncertainty in the black coal system results than in the brown coal system results. The Monte-Carlo method (see Section 3.2.3.7.2, Chapter 3) enables a single uncertainty range to be reported for each environmental impact result, allowing the results to be presented using standard quality assessment tools (i.e. box plots). The use of the normalised difference probability distribution plot, enables quantification of the difference between two systems if their uncertainty ranges overlap (see Figure 5.2.2, Chapter 5). The qualitative methods enable tracing of the source of this difference to the greater usage of data derived from plant measurements for the brown coal case. The two combinatorial methods, Kennedy et al. and Meier, produce very different results (see Table 5.2.6, Chapter 5). There is no evidence that either approach produces results that are more accurate, or more representative of qualitative uncertainty, than is the other. Moreover, as they are reliant on subjective (expert) reasoning, neither method produces a measure of uncertainty that is more relevant than that of the numerical uncertainty method alone. These methods provide no greater information than the quantitative methods, and present far less information than the qualitative and quantitative methods presented separately. Therefore, uncertainty assessments should present results from both numerical and qualitative assessment methods separately.

The use of the Monte-Carlo method shows that different results occur when using different types of uncertainty distributions (see Table 5.2.2, Chapter 5). In the case of the AD indicator for ST-BrC systems, the use of triangular and uniform distributions worsens the mean impact value by over 300 %. The uncertainty distributions of the NO_x and SO₂ emissions, which are the major contributors to AD, are highly skewed, i.e. their mean and minimum values are similar, and distant from the maximum value. This is approximately a log-normal distribution (see Figure 3.2.2, Chapter 3), and neither the uniform or triangular distributions can successfully approximate log-normal distributions. The mean of a uniform distribution must lie halfway between its extents. The mean of a triangular distribution is limited such that its average must lie between 29

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

^{*}** It also includes distribution of electricity.



Figure 6.3.1: The maximum and minimum positions for the mean of triangular distributions. Positions of b and c calculated using laws for similar triangles and equal probabilities (area left and right must equal half).

and 71 % of the distance between its extents (see Figure 6.3.1). The fact that such problems can occur, limits the practicality of standard, arbitrary distributions. When estimates for the distribution of each data point are not available (i.e. insufficient data), then the use of a skewed distribution, such as the triangular or log-normal distributions is recommended. They allow the average data point to be closer to either its minimum or maximum, and require only a data point's average, minimum, and maximum values (see Figure 3.2.2, Chapter 3). The triangular distribution probably overestimates the true uncertainty profile (a skewed normal distribution) and thus provides a conservative estimate.

Of the tested qualitative methods, neither weights quality scores by contributions to the indicator score^{§§§}. Thus, their scores are indicative of the quality of the data collection methodology (i.e. how good relative to theoretical perfection), not of the quality of the indicator scores. Nevertheless, the Rousseaux *et al.* method does allows the location and identification of the causes of deficiencies in data quality. As such, neither method's results indicate the quality of the environmental impact results. The comparison of these qualitative methods may be unfair as both methods recommend the inclusion of other indicators in their intended indicator sets.

6.3.5 NORMALISATION AND PRESENTATION OF INDICATOR RESULTS

The aim of normalisation is to enhance understanding of the relative importance and

Sustainability of Australia's Electricity Generation ___ Chapter 6: Implications of the Sustainability Assessment

magnitude of the indicators (see Sections 3.2.3.5 and 3.5, Chapter 3). Much of the sustainability performance discussion (Section 6.2.2) requires only the pre-normalised indicator scores (of Sections 5.2, 5.3 and 5.4, Chapter 5). For example, pre-normalised scores allow comparisons of the relative performance of systems for particular indicators (i.e. the CC score for ST-BIC systems is better than the CC score for ST-BrC systems). However, pre-normalised scores cannot provide indicators of the relative importance of a system's performances for different indicators. For example, is the CC score for ST-BIC systems better or worse than its AD score? The normalised scores describe importance to the Australian regional sustainability by providing an appreciation of relative performance in different is problematic, as the normalised scores are not strictly indications of the relative severity of the impacts. Thus, a difference is only significant if there are large, order-of-magnitude differences between the indicators (i.e. 10^8 and 10^9).

The validity of the comparison of normalised economic and social indicators with environmental indicators is not established. If this method can compare completely different types of environmental types of impact, such as CC and RD W, then it is equally acceptable to compare different types of sustainability impact. This is the first time to the author's knowledge that normalisation has been used to compare environmental, economic and social impacts.

Another method to try to enhance understanding of the i-dicator results is weighting (see Section 3.2.3.5, Chapter 3). Weighting relies on subjective, value judgements to assign importance to the magnitude of each indicator's score. It is important that the use of value judgement does not undermine the value of the quantitative, scientifically based work up to this point of decision making.

^{§§§} The method of Lindeijer et al. (see Berg et al. (1999)) reportedly does this, however evaluation of this method is difficult as it is only available in Dutch.

6.4 POTENTIAL APPLICATIONS OF THE SUSTAINABILITY ASSESSMENT METHOD

6.4.1 OVERVIEW OF POTENTIAL APPLICATIONS OF THE SUSTAINABILITY ASSESSMENT METHOD

The developed method assesses the impact on sustainability of electricity generation systems. This data may have other important uses. For example, where a company, or network operator, desires to provide a new electricity generation system, in a sustainable manner, the method guides the choice of the most appropriate technology for this goal. Similarly, electricity consumers may desire to select the most sustainable electricity supplier, from a choice of different electricity generation systems. Measurements of impacts on sustainability may also be of use for other systems.

Section 6.4.2 discusses the potential for the use of the method in the planning of electricity supply systems. Section 6.4.3 discusses the potential for the use of the method by electricity consumers. Section 6.4.4 discusses the potential application of, and limitations in, using the method for systems other than electricity generation systems.

6.4.2 ELECTRICITY SUPPLY PLANNING

Traditional approaches to the planning of electricity supply attempt to meet anticipated demands of consumers, reliably and at lowest possible cost (IAEA (2000)). This requires development of an electricity demand forecast and a capital works budget and schedule to meet this forecast. Government owned electricity supply networks would produce these internally. In deregulated networks, governments develop forecasts, but private companies own and develop the electricity generation network. In these networks, governments must ensure that forecast electricity supplies are met, through legislation or other means (IAEA (2000)). Inclusion of environmental concerns in previous planning methods has been by costing measures to achieve regulated limits on emissions (IAEA (2000)). Consequently, optimisation of existing electricity supply networks has been for low electricity costs (i.e. the AC indicator), which is only one type of sustainability impact. Emerging planning methods are using pricing incentives to attract investors and customers to renewables, such as the MRET scheme in

Sustainability of Australia's Electricity Generation Chapter 6.Implications of the Sustainability Assessment

Australia, which have their own problems (for example, see Ministerial Council on Energy (2002)). Using the presented sustainability assessment method, electricity supply planning could integrate all sustainability impacts known to be important for electricity generation systems.

Electricity supply plans generally modify networks in three ways:

- 1. Addition of new electricity generation plant;
- 2. Replacement of existing plant; and
- 3. Acting to reduce demand.

When considering the addition of a new electricity generation plant, the first step is to identify possible options. Sections 4.2.2.3, 4.2.3.3 and 4.2.4.3 (Chapter 4) discuss the most likely options for new power stations within Australia. The use of the sustainability method enables the estimation of the probable sustainability impacts of each choice as sets of indicators (see Figure 5.5.1, Chapter 5). The basis for selecting an option can now be the sustainability performance (see Section 6.2). Alternatively, the planners may require certain performances for particular impacts, but consider some other impacts of little consequence. The assessment may include these biases by weighting the indicator scores appropriately. For example, a planner may consider low climate change emissions and low electricity costs to be of greatest importance, and resource depletion and employment of low importance. In this example, high weightings will apply to the CC and AC indicators, and low weightings to the RD W. RD A and ED indicators. However, conclusions based on these weighted results will be debatable, as these weighting schemes have no scientific basis (see Section 6.3.5).

An alternative basis for these assessments might be the minimisation of network sustainability impacts. The electricity generation network consists of a number of electricity generation plants, and thus electricity generation systems. The sustainability indicator scores for the network will equal the sum of sustainability indicator scores from using this method on each of the systems, weighted by their respective electricity generation. For example, a network consisting of a ST-BIC system (CC score = 1060 kg CO2-eq per MWh) producing 30 GWh of electricity, a ST-BrC system (1210) producing 15 GWh and a CCGT-NG (458) system producing 10 GWh will have a CC score of approximately 991 kg CO2as per MWh (scores from Table 5.2.1, Chapter 5). If the existing network has poor scores for some indicators, the basis for choosing between

Sustainability of Australia's Electricity Generation Chapter 6:Implications of the Sustainability Assessment

options could be to minimise additional, or reduce, contributions to this indicator score.

Periodically network planners replace older electricity generation plants with new plant. In this case, impacts on sustainability usually decrease, as new plants normally have higher efficiencies than those replaced. Yet, this may not be so for all such replacements and for all sustainability impacts. For example, consider the replacement of an average ST-BrC plant with a ST-BlC plant. Section 6.2.2 discusses such a replacement in the context of ameliorating the CC impact, and shows that while this change improves some indicator scores, it worsens many more. The same assessment highlights that replacement with a CCGT-NG improves most indicators. This option may be a better choice on sustainability grounds. An additional benefit of the method is that it allows quantification of the scale of improvements, or deteriorations, in impacts caused by the change. For example, using the previous example, the change in CC score for replacing a ST-BrC system (CC score = $1210 \text{ kg CO}_{2-cq}$ per MWh) with a ST-BIC system (1060) is 150 kg CO_{2-co} per MWh, or 12.4 % of the old CC score.

The last type of change is demand reduction. Reducing demand involves improving the efficiency at which end users consume electricity. Governments operate many schemes to improve the energy efficiency of consumers, and industries often attempt to improve their energy efficiency to reduce electricity costs, albeit often at the expense of increased capital costs. Estimation of the quantitative improvement in sustainability caused by reductions in demand requires an estimate of the sustainability impacts of electricity consumption. The sustainability assessment method can produce such an estimate: the electricity generation network's sustainability indicators. The product of these indicators and the reduction in energy demand is a measure of the improvement in sustainability performance. For example, if demand for electricity from a network containing a ST-BIC, ST-BrC and ST-NG, with a CC score of 991 kg CO_{2m} per MWh (see earlier example), is reduced by 100 MWhpa, then the reduction in CC is 9.91 x 10⁴ kg CO_{2eq} per annum.

It may be of benefit when using this method for electricity supply planning to introduce an indicator of supply reliability. The current indicator is usually reserve (or peak reserve) capacity, and this may serve. However, this measure does not account for the likelihood of an individual system to fail. For example, older plant more often fails than newer plant. Another factor may be the propensity of workers to call for strike action. A complete reliability indicator could include these and other factors in its

measure.

6.4.3 OPTIMUM ELECTRICITY SOURCE PLANNING

Operators of electricity generation networks could use the sustainability method to estimate and trade-off sustainability impacts. Electricity generation networks, like Australia's National Energy Market, source electricity from many different types of plant. The total sustainability of the network will depend on the proportions of electricity sourced from each type of plant****. This method allows quantification of the relative impact on sustainability when deciding between two different types of plants for any amount of electricity. For example, an operator requires 100 MW, which is available from either ST-BrC (CC score = 1210 kg $CO_{2,eg}$ per MWh) or CCGT-NG systems (458) (scores from Table 5.2.1, Chapter 5). Thus, the choice of the CCGT-NG system rather than the ST-BrC alternative negates 7.52 x 10⁴ kg CO_{2-sa} per MWh of CC impact. These quantities could be of use as guides for trading impacts. For example, if electricity from a ST-BrC system is available at a lower cost than electricity from a CCGT-NG system, then the operator can quantitatively compare differences in cost and sustainability. Trading impacts may limit the magnitudes of individual impact magnitudes, by ensuring a wide range impacts, rather than great amounts of a few¹¹¹¹

6.4.4 APPLICABILITY TO NON-ELECTRICITY SYSTEMS

The sustainability method, although developed for electricity generation systems, is flexible, allowing the quantification of impacts from any fossil fuel electricity generation system. The basis for the choice of indicators for this method is that they must measure an impact known to be important for electricity generation systems. Consequently, the indicators in Chapter 5 are applicable only for systems whose important impacts align with those of electricity generation systems. Fortunately, many systems have this characteristic. Examples include non-electricity natural gas delivery. petroleum fuels and products, steel and other metals, chemicals, fertilisers, and plastics

والمرافقة فسيفت بتوهد وودون والتواري والتوارين والمراجع والمراجع والمراجع والمراجع

"" Section 6.4.2 describes the procedure for estimating the sustainability of an electricity peneration

**** IAUA (2000) uses a similar argument for limiting environmental impacts by using a greater diversity

network

of electricity sources. gravity for the second s

Sustainability of Australia's Electricity Generation _____Chapter 6: Implications of the Sustainability Assessment

manufacturers. For some of these systems, additional impacts may occur which the methods indicators do not measure (c.f. Section 6.2.4.3). The sustainability assessment method is flexible and can include additional indicators for those impacts as appropriate.

One simple example of an application using other systems is to include other types of electricity generation systems (oil, nuclear, renewables). Such an analysis would enable the widening of the analyses in Sections 6.2, 6.4.2 and 6.4.3 from the most probable options to all possible options.

Another example is to use the analysis for all types of systems consuming large amounts of one resource. For example, apart from electricity generation, the production of many chemicals, such as hydrogen, methanol, ammonia, urea and other fertilisers. consume large amounts of NG. The analysis would enable the quantification of the relative sustainability impacts of each consumer of the resource. Such information could aid in resource use planning by governments, or companies with interests in multiple consumers of the resource.

7. CONCLUSIONS AND RECOMMENDATIONS

7.1 OVERVIEW OF THE RESEARCH FINDINGS

7.1.1 THE SUSTAINABILITY OF ELECTRICITY GENERATION SYSTEMS

The indicators of sustainability of electricity generation systems indicate no one type of system has worst environmental or sustainability indicator scores for every indicator. Thus brown coal is not a 'dirtier fuel' than black coal and natural gas using a basis of either overall environmental or sustainability performance. The normalised indicators, normalised using newly developed Australian indicator scores, are mostly of similar magnitude. Nevertheless, capital expenditure (1) is clearly of a higher magnitude and thus most important for the Australian region. Consequently, while climate change (CC) deserves inclusion in decision-making processes, so too do the other impacts on sustainability.

The electricity generation systems consuming brown coal (BrC) do not have worst indicator scores in all categories. A major advantage of BrC is its economic performance. BrC systems have low electricity costs and high value added. However, the investment costs of BrC steam turbine systems are very great. The electricity generation systems consuming BrC outperform those consuming black coal (BIC) for the majority of indicators. The acidification (AD), solid waste (SW), and worker injuries (LTI) and fatalities (Fatal) indicator scores for BIC systems are particularly poor. The electricity generation systems consuming natural gas (NG) in combined cycle gas turbine (CCGT-NG) systems are the most sustainable of the considered systems. The performance of open-cycle gas turbine (OCGT-NG) and steam turbine (ST-NG) systems is significantly worse than these CCGT-NG systems. Therefore, the conversion to, or replacement with, CCGT-NG systems of existing systems with these configurations should be a priority.

While, CCGT-NG systems are the best option of the systems considered, the reserves of Australian natural gas are substantially less than the coals. Thus, the consumption of natural gas has greater relative impact, as measured by the newly developed Australian

resource depletion indicator (RD A), than the consumption of the coals. Furthermore, constraints in the addition of new gas transport pipelines limit the rapid expansion of natural gas consumption which conversion or replacement of coal systems to gas would require. The use of advanced electricity generation technology for the coal systems has substantially greater sustainability than the conventional technologies. However, advanced technologies are not commercially available at present for either of the coals.

Foreign electricity generation from natural gas (LNG) and black coal (BICE) exported from Australia is less sustainable than domestic electricity generation, due to the need to prepare and transport the fuels over long distances. Of the options, the CCGT-LNG systems are most sustainable. However, as for domestic NG systems, consuming natural gas resources to produce LNG has relatively greater resource depletion impact, as measured using RDA, than the consumption of black coal.

Uncertainty and sensitivity analyses show that the ranking for the systems, using the sustainability indicators, is robust to data uncertainty and decisions made during the formation of the indicators. Nevertheless, an analysis of data uncertainties indicates that it is possible for the relative climate change (CC) performance of BrC and BlC steam turbine (ST) systems to reverse under specific circumstances. However, the normalised difference uncertainty method shows, in an objective manner, that there is a higher probability of the ST-BIC system having better CC performance than the ST-BrC system, than the reverse situation.

The uncertainty analysis tests several methods for estimating the magnitude of uncertainty in the indicators due to uncertainties in the data used in their production. All methods show that the climate change, acidification and world resource depletion indicators (RD W) of the ST-BIC system have greater uncertainty than the corresponding ST-BrC indicators. The qualitative methods establish that the greater use of data derived from plant measurements for the ST-BrC systems cause this difference. Uncertainty in electricity efficiency has greatest influence on the values of the indicators, and thus a sensitivity analysis using this efficiency can estimate the effect of data uncertainty on electricity generation systems. The sensitivity analysis shows that decisions such as the method to allocate impacts between products, transport destination (for export systems) and assumed system is re (estimated by capacity) have considerable influence on the sustainability indicator sec es.

Sustainability of Australia's Electricity Generation

7.1.2 ABOUT THE SUSTAINABILITY METHOD

The developed method allows production of indicators for all impacts on sustainability identified as important for electricity generation systems consuming Australian fossil fuels. Important features of the method are its ability to provide quantitative indicators, use of a scientific basis for quantification of environmental impacts and accommodation of the 'life-cycle' perspective. Including only the electricity generation subsystem would significantly reduce the importance of some indicators, such as the BIC and BICE systems' worker injuries and fatalities indicators. The method can compare the sustainability performance of systems with different combinations of fuel and electricity generation technology.

The developed method allows estimation of the relative magnitude of each impact on regional sustainability, through normalisation. The application of normalisation to economic and social indicators is uncommon and environmental normalisation factors from authoritative sources for Australia are not available. Thus, the use of normalisation required the estimation of new normalisation factors for Australia using diverse sources of Australian material, economic and social flows data. Normalisation does not involve non-scientific methods (c.f. weighting) to attach 'importances' to each indicator. However, normalisation provides an objective method for assessing the relative magnitude of contributions to different types of sustainability impact.

The EN and EX indicators are not necessary for the quantification of the resource depletion impact on sustainability as they do not measure resource scarcity. RD A, a new Australian based indicator of resource depletion, can show both Australian resource scarcity and, together with RD W, differentiate between Australian and worldwide resource scarcity.

Analysis of the robustness of the indicators to data quality assessment is important for the development and acceptance of LCA as a decision-making tool. Yet, the use of data quality assessment in published LCA studies is rare, and there is no consensus on the methods to use. The Monte-Carlo method produces a single uncertainty range for each environmental indicator. When knowledge of a data source's numerical uncertainty profile is unavailable, the uncertainty analysis findings recommend the use of a skewed distribution (such a the triangular or log-normal distributions). The qualitative methods, such as that of Rousseaux et al., can link differences in the magnitude of uncertainty in the indicators of different systems to the types of data used

Chapter 7: Conclusions and Recommendations

to estimate their indicators. The methods that combine numerical data uncertainty with quantitative uncertainty in a single uncertainty measure are reliant on subjective (expert) reasoning, provide no additional information and present far less information than if the qualitative and quantitative methods are presented separately. Therefore, when an uncertainty analysis is necessary, the best approaches currently available are the Monte-Carlo method (quantitative) and Rousseaux et al. (qualitative) methods. As the methods for data quality assessment are still underdeveloped, uncertainty analysis is recommended only where the differences between the options is debatable and the choices have impacts of considerable magnitude.

7.2 LIMITATIONS OF THE RESEARCH

7.2.1 OF THE SUSTAINABILITY OF ELECTRICITY GENERATION SYSTEMS

The sustainability assessment relies on a number of assumptions, which may influence the magnitudes of the sustainability indicators. Firstly, the system boundary excludes secondary processes (such as construction), as others have reported their contributions as minor, and is thus not strictly compatible with the 'life-cycle' perspective. The data collection excludes some environmental data, as reliable emissions data for certain species is unavailable, necessitating the exclusion of toxicity impacts from the discussion of sustainability performance. Difficulties in quantifying some effects with indicators resulted in the exclusion of some impacts identified as important for electricity generation systems. The assessment assumes that the electricity generation systems consuming the exported Australian fuels (LNG and BICE) have electricity generation and transmission systems characteristic of Australian systems, rather than their own systems. Finally, the scope of the assessment is average Australian electricity generation systems, and thus the sustainability profiles for individual systems or for similar systems in other countries, may have minor differences.

7.2.2 OF THE SUSTAINABILITY METHOD

While the method attempts to select indicators for each impact important to electricity generation systems, some impacts are difficult to quantify with indicators, such as the

Sustainability of Australia's Electricity Generation_

effect of subsidence. The method allows the inclusion of qualitatively measured impacts through discussion, but does not provide a generalised procedure.

The validity of using normalised scores for comparing economic and social impacts with environmental impacts is not established, as no other workers have apparently used this method.

Section 3.2.3 (Chapter 3) identifies shortcomings of the method for producing environmental indicators (LCA). Of those noted, this work considers resource depletion, uncertainty analysis, normalisation and weighting. The presence of such difficulties may limit the accuracy of some of the environmental indicator scores.

7.3 RECOMMENDATIONS FOR FURTHER RESEARCH

The developed sustainability method is able to assess the sustainability of Australian electricity generation systems. Nevertheless, its application to a system, on which others have previously applied another sustainability assessment method, is necessary to verify the stated advantages of the method developed in this thesis.

The basis for the conclusions and recommendations of the uncertainty analysis is the data of only two systems (ST-BrC and ST-BlC) and three indicators (CC, AD and RD W). The conclusions drawn about uncertainty analysis rely on this basis (see Sections 7.1.1 and 7.1.2). Extension of the uncertainty assessment to all systems and indicators will test the validity of these conclusions.

The sustainability assessment method has no defined procedure for the integration of impacts, which are difficult to quantify, into assessments of sustainability performance (see Section 7.2.2). Development of such a procedure will enhance the value of the method and the sustainability performance assessments it produces.

Section 6.4 (Chapter 6) identifies a number of potential applications for the developed method. Tests of the method's ability to apply to electricity generation network supply and source planning can commence with the data contained in this thesis. However, testing of the method for assessing the sustainability of other systems (for example, oil, nuclear and renewable energy systems) will require further, timeconsuming data collection.

APPENDICES

APPENDIX 1: DATA SOURCES FOR THE SUSTAINABILITY ASSESSMENT

A1.1 OVERVIEW OF DATA SOURCES FOR THE SUSTAINABILITY ASSESSMENT Sections A1.3 to A1.5, detail the data sources behind the subsystem data given in Chapter 4. Section A1.3 details the data sources for systems consuming either domestic or export black coal. Section A1.4 details the data sources for systems consuming brown coal. Section A1.5 details the data sources for systems consuming natural gas or LNG. Section A1.7 details the data sources necessary to conversion of this data into sustainability indicators. Section A1.8 details the data sources for the fuel emission factors. Section A1.9 shows the data for the normalisation factors. Section A1.10 provides examples of the aggregation methods used to produce data from multiple sources.

A1.2 EXPLANATION OF TABLE TITLES The tables presented in this appendix enable the duplication of the data shown in Chapter 4. Each table entry has a reference number ('Ref.'), which refers to its position in Chapter 4. The first part of this number is a letter, which indicates to which of the included technologies the data belongs. For example, in the black coal mining subsystem, A indicates the data is for underground mining and B for surface mining. The number following this letter indicates the data's position in the relevant table in Chapter 4. Indication of positional aspects such as system, subsystem, and type of data (material, economic or social), is by identical sub-sectioning in Chapter 4 and this appendix.

Entries in the second column ('Agg.') indicate which aggregation procedure is used to produce the data from the sources shown in the next column ('Sources'). Possible entries are:

- N (no aggregation);
- EN (estimated as negligible);
- YA (yearly average);
- FA (factored average);
- SA (simple average);

Appendix 1:Data Sources

Sustainability of Australia's Electricity Generati

- PW (production-weighted average);
- CW (capacity-weighted average);
- MW (employment or employee hours-weighted average);
- FE (estimate obtained by applying a factor to another data value);
- WE (calculation built in parts using separate sources for each part);
- MC (solution of multiple linear, multivariate equations);
- R (estimate relying on ratios based on old data);
- EM (emission factor based calculation); and
- CP (correlation using power function).

Section A1.8 shows emission factors for the fuels encountered.

Some data sources contain multiple values for the desired data value. Entries in the fourth column ('S. Agg.') indicate which aggregation procedure is used to aggregate these multiple values. Possible entries are:

- N (no aggregation);
- YA (yearly average);
- RA (rolling average); and
- EC (economic correlation).

Section A1.10 provides further details and examples of these aggregation methods.

The final two columns, 'PM Scores' and 'Range', include data necessary to produce an uncertainty assessment on the BIC-ST and BrC-ST system's environmental indicators. The first column ('PM Scores') contains pedigree matrix scores (see Table 3.2.3, Chapter 3), where R, C, T, G and E are the reliability, completeness, temporal correlation, geographical correlation, and technological correlation indicator scores. The final column ('Range') contains the minimum and maximum values found for each data point. Superscript values in this column, and throughout this appendix, are magnitudes (i.e. ⁶ equals x 10^6).

A1.3 BLACK COAL AND EXPORT BLACK COAL

A1.3.1 Mining Subsystem

A1.3.1.1 Material Flows

Ref. Agg. Source(s) S. PM Scores Range Agg. R C T C F Min. Max.

C1	SA	DMR (2000), Alyward and Sutherland (2000), DME	N	2	2	1		1	0	41.9
<u></u>	EA	(1998) and DME (2000)	VA		3	1			112	ንነ
C2	PA -	BIIP (1999a), BIIP (1999b),	1A (1)		4	1	ſ		14.5	£1.
		A BUB (1999C), BUD (1999d)	(U), SA							
		BHP (1996), BHP (1996),	οΛ (S)							
		RHP (1999), BHP (1999),	(3).							
		Pacific Coal (1999) B								
C3	See C	2		4	5	1		3	1.68 4	3.8
C4	See S	ection A1.4,1,1 A4		4	5	1	1	3	912	13
C5	SA	Theodore Coal (1981), Spath	N	3	4	5	1	1	1.2 *	2.4
		<i>et al.</i> (1999), Tasman Asia Pacific (1997)								
C6	N	Spath et al. (1999)	N	2	3	1	3	2	2.74 5	2.7
Ċ7	PW	DMR (2000)	N	2	2	1	1	1	1.99 *	1.9
C8	N	Dharmappa et al. (2000)	N	2	2	1	1	i		
C9	PW	As in C2, Additional: Coal	-	1	2	1	1	1	1.37	3.9
		and Allied (1999) B								
C10	See C	7		2	2	1	1	1	1.99 6	1.9
cn	N	Mass balance		· ·	*	. .	• . 			
C12,	EM	Diesel, Petrol EM	N	3	5	1	2	3	8.00 4	1.0
C14-							:		121	1.6
C19							ļ		2	3
							•		1123	15
			f r -				ļ		437	6.0
									227	3.1
									48	64
C13	EM	Diesel, Petrol EM, and Smith	N	3	5	1	2	3	17.4	90
		and Sloss (1992), NGGIC	1						2.95 *	4.0
	ł	(2000), Williams (1998) M								

-	
fr s	••
	•
_	-

Appendix 1: Data Sources

Sustainability of Australia's Electricity Generation

Ref.	Agg.	Source(s)	S		PM.	Sco	res		Rai	ige
			Agg.	R	Ċ	े।	Ċ		Min.	Max.
XI	SA	Aylward and Sutherland		5	3	1	1	1	1	24
		(2000), DME (2000), DMR								
		(2000), confidential sources								
		and estimates.								
X2	SA	Spath <i>et al.</i> (1999), Szabo		2	3	1	3	1	16.1	2.9.9
		(1978)		<u>-</u>	-			 		17.4
X3	See X	1		5	3	1	1		13.6	17.4
X4	See X	1		5	3		ļ 	1	115	900
X5	PW	Freightcorp (2000), Queensland		 	I				3150	8645
		Rail (2000)				1			16.2	795
λ6 	See X	 		3	3	I 	•		10.5	20.5
N/ No		· I	,		 		 	 	54.7	60.5
A8 	See A	De training Commission I	NÌ		i				1944 1970 - 1970	07.5
X9	we	(1998)	[N							
XI0	N	Productivity Commission	N							
		(1998)								
XH	N	Woodside Energy Ltd. (1998)	N							
Λl	N	Sazbo (1978)	N	2	2	5	3	1	0.044	2.46
Λ2	PW -	Sazbo (1978), Woodside	N	3	3	5	3	1	1.69 5	63.3 5
		Energy Ltd. (1998) RI, Szabo								
	1	(1978) Rd								
A3	WE	OECD (1985), Chadwick et al.	N	5	5	1	1	1	9.97 *	9.98 *
• • •		(1987)					 	•		
A4	See A	\3	• • • • • • • • • • • •	5	5	1	1	1	8.94	1.31
Λ5-	EM	Diesel EM	N	3	5	1	2	3	5.28	1.98 '
AП							ĺ		43.2	1.62
			ļ		Ì				818	3.07
				İ					15.2	500
				1					7.02	2.80
		مىلىرىتى تۈرمىتىنى بىلىرىكى بى بىلىرىكى بىلىرىكى بىل	 	<u> </u>		 			2.90	1,11

AI.3.1.2	Economic Flows
----------	----------------

Ref.	Agg.	Source(s)	S .		PM	Se	ores		R	nge
			Agg.) }	С	'T	G	E	Min.	Max.
A1	СР	Anonymous (2001a), DITC	EC							
		(1980-1986), Errington and								
		Kirby (1993)								
٨2	СР	DOI (U.S.A.) (1993), ACCC	EC							
		(1986), EIA (U.S.A.) (2000),								
		Errington and Kirby (1993)		-		" 				
B1	СР	NREC (1987), BHPBilliton	EC					1.2 7 3. 7 3		
		(2001a), DITC (1980-1986),								
		Macarthur Coal (2001), Mining				*u 	A.,	3 5 7		
		Technology (2002)								
B2	CP	DOI (U.S.A.) (1993), ACCC	EC							
1		(1986), Tasman Asia Pacific						4 -		
		(1997), Simes (2000), EIA								
		(U.S.A.) (2000)						е 7 1 т		
CI	FA	A1 and B1			a sut	х <u>у</u>			· · ·	
C2	FA	A2 and B2			urrini jipenni					i i di second

A = Underground Mining, B = Surface Mining, C = Average Mine.

AI.3.1.3 Social Flows

Ref.	Age.	Source(s)	S.	PM Scores Range						
			Agg.	R	e	I	G	· E	Min.	Max.
CI	PW	Coal Services (2002), NRM	N							
		(Qld.) (2001)	1							
C2	PW	NOHSC (2000)	N						-	
C3	PW 1	NOHSC (2000), MCA (2002)	N	••••••		·				
ί C	`≊ Aver	age Mine.	i	L		L		I	بد مملاً سميما	المرجب ومراجع

A1.3.2 Transport Subsystem

A1.3.2.1 Material Flows

Appendix 1:Data Sources

Sustainability of Australia's Electricity Generation

				<u> </u>					1.55 3	5.80 4
A12	EM	Diesel EM for particulates, see	N	3	5	1	2	3	314	1.18+
		A3 for dust		5	5	1	1	ł	8,94 4	1.31 5
BI	WE	Spath <i>et al.</i> (1999) for coal cleaning and ports.	N							
B2	See 1	315								
B3	WE	See A2	N							
B4	SA	Smit and Nieuwlaar (1994), Simapro 5 (2001)	N							
B5	N	Spath et al. (1999)	N	-						
B6	See B	15		-						
B7	See B	15		-						
B8- B14	See A	5-A11								
B15	WE	See A12, additionally Smit and	N	-		i				
		Nieuwlaar (1994) for ship							· · · · ·	
		transport, and Spath ct al.								
		(1999). Smit and Nieuwlaar (1994) for coal cleaning					3 - N			

A1.3.2.2 Economic Flows

Ref.	Agg.	Source(s)	S. Agg.	R	PM C	l Se T	ores G	E	Ra Min.	nge Max.
Al	СР	Macarthur Coal (2001),	EC							
		Freebairn and Trace (1988),								
		Easton (1988), CFAC (111.)		ļ						
	ļ	(1977)								
A2	СР	CFAC (III.) (1977)	EC							
A3	СР	Lee (1986), CFAC (111.) (1977)	EC							
A4	СР	PWC (Qld.) (1997), Merritt	EC						a, ¹ 2 d	
		(1986). Estonian Railways								
		(2002), AISE (2002), Tradeport								
······································		(2000) for ports. Drewry	-					ga ta di Alita ya Alita ya		

		Shipping Consultants (1995) for ships					
Α5	СР	Szabo (1978), Freebairn and Trace (1988), BIE (1995), Easton (1988)	EC				
A6	CP	Brennan (1990), NEAC (1980), Szabo (1978), Lema (1990), CFAC (III.) (1977)	EC				
Λ7	СР	Lee (1986), Szabo (1978), Lema (1990), CFAC (III.) (1977), EPA (U.S.A.) (1994)	EC				
Λ8	СР	Drewry Shipping Consultants (1995)	EC				

AI.3.	.2.3	Social Flows								
Ref.	Agg.	Source(s)	S. Agg.	PN R	1 Se	ore T	s G	E	Rang	e Max.
Al	WE	BIE (1992) (Road and Rail)	N							
Λ2		NOHSC (2002)								
A3	See /	12		12						
Bl	WE	BIE (1992) (Road and Rail). BTCE (1998) (Ship)	N							
B2	SA	NOHSC (2002), AMSA (2002)	YA							
B3	See I	32	L •-							

A1.3.3 Generation Subsystem

A1.3.3.	I = M	laterial Flows									
Ref.	Agg.	Source(s)			S.	2. J	PM Sc	ores		Ra	nge
					Agg.	12	СТ	G		Min.	Max.
Al	PW	ECNSW	(1991a)	and	YA	2	1 3	1		4.38 -2	8.49 -2
		confidential									

Appendix 1:Data Sources

Appendix 1:Data Sources

A2	PW	DMR (2000), DME (2000)	N	3	2	1	1	l	1.25 %	7.55 5
A3-	PW	Delta Electricity (1999b).	YA	2	2	1	1	1	4.98	19.9
A4		Tarong Energy (2000b)							5.47	21.9
A5	PW	Delta Electricity (1999b),	YA	2	2	l	1	ì	1.83 2	4.00 ²
		ECNSW (1991a), Stanwell								
		(1999), and confidential								
A6-	N	Delta Electricity (1999b),	N	2	2	ī	1	1	32.5	130
A15		ECNSW (1980)							22.5	90.0
									3.75	15.0
									3.00	12.0
									0.10	0.40
									0.15	15.0
									7.50 -2	7.50
									5.00 -2	5.00
									2.00 -2	2.00
									2.00 -2	2.00
A16	N	Delta Electricity (1999b)	YA	2	1	1	1	1	1.70 -1	170
		•		-						
A17	FE	Dragun <i>et al.</i> (1986)	N	3	2	4	1	1	1.39 5	2.60 5
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta	N YA	3	2	4	1	1	1.39 ⁻⁵ 1.25 ⁻⁶	2.60 ⁵ 2.34 ⁶
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC	N YA	3	2	4	1	1	1.39 ⁻⁵	2.60 ⁵ 2.34 ⁶
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP	N YA	3	2	4	1	1	1.39 ⁻⁵	2.60 ⁵ 2.34 ⁶
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong	N YA	3	2	4	1	1	1.39 ⁻³	2.60 ³ 2.34 ⁶
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell	N YA	3	2	4	1	1	1.39 ³	2.60 ⁵ 2.34 ⁶
A17 A18	FE PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999)	N YA	3	2	4	1	1	1.39 ³	2.60 ⁵ 2.34 ⁶
A17 A18 A19	FE PW See A	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999)	N YA	3	2 2 Se	4 1 e A	1	1	1.39 ³ 1.25 ⁶	2.60 ⁵ 2.34 ⁶ 1.085
A17 A18 A19 A20	FE PW See A PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta	N YA YA	3 3	2 2 See 2	4 1 e A 1	1	1	1.39 ⁻⁵ 1.25 ⁻⁶ 1.044 5.54 ⁻⁴	2.60 ⁵ 2.34 ⁶ 1.085 1.34 ⁵
A17 A18 A19 A20	FE PW See A PW	Dragun <i>et al.</i> (1986) Dragun <i>et al.</i> (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong	N YA YA	2	2 2 See 2	4 1 e A	1	1	1.39 ⁻⁵ 1.25 ⁻⁶ 1.044 5.54 ⁻⁴	2.60 ³ 2.34 ⁶ 1.085 1.34 ⁵
A17 A18 A19 A20	FE PW See A PW	Dragun et al. (1986) Dragun et al. (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong Energy (1998), Stanwell	N YA YA	3 3 2	2 2 See 2	4 1	1	1	1.39 ⁻⁵ 1.25 ⁻⁶ 1.044 5.54 ⁻⁴	2.60 ⁵ 2.34 ⁶ 1.085 1.34 ⁵
A17 A18 A19 A20	FE PW See A PW	Dragun et al. (1986) Dragun et al. (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong Energy (1998), Stanwell (1999)	N YA YA	3 3 2	2 2 See 2	4 1 1	1	1	1.39 ³ 1.25 ⁶ 1.044 5.54 ⁴	2.60 ⁵ 2.34 ⁶ 1.085 1.34 ⁵
A17 A18 A19 A20 A21	FE PW See A PW WE	Dragun et al. (1986) Dragun et al. (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong Energy (1998), Stanwell (1999) Day (1986), ECNSW (1982)	N YA YA	2	2 2 See 2	4 1 1 1	1	1	1.39 ⁻⁵ 1.25 ⁻⁶ 1.044 5.54 ⁻⁴ 1.51 ⁻⁶	2.60 ⁵ 2.34 ⁶ 1.085 1.34 ⁵ 7.56 ⁵
A17 A18 A19 A20 A21 A22	FE PW See A PW WE PW	Dragun et al. (1986) Dragun et al. (1986), Delta Electricity (1999b), SPCC (1979), ECNSW (1980), BHP Research (1999), Tarong Energy (2000b), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong Energy (1998), Stanwell (1999) 1 Pacific Power (1999), Delta Electricity (1999b), Tarong Energy (1998), Stanwell (1999) Day (1986), ECNSW (1982) Tarong (2000a), Tarong	N YA YA YA YA	3 3 3 2 2 2	2 2 2 2 2 2 2 2	4 1 1 1 1 1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1	1.39 ⁻⁵ 1.25 ⁻⁶ 1.044 5.54 ⁻⁴ 1.51 ⁻⁶ 1.87 ⁻⁵	2.60 ⁵ 2.34 ⁶ 1.085 1.34 ⁵ 7.56 ⁵ 5.71 ⁵

Sustainability of	'Australia's	Electricity Gene
-------------------	--------------	------------------

A23	MW	Based on Section A1.4.3.1 A26		5	2	1	4	4	2.73 -3	2.73
A24	YA	Pacific Power (1999)	N	3	4	1	1	1	3.15 *	3.15 6
A25-	EM	BIC-ST, Diesel, Petrol, Fuel	N	2	2	1	 l	1	2.73 5	1.65 6
A32		Oil EM							2.77	166
									5.12	30.9
									2.40	14.5
									8.87 ²	5.37 ³
									3.31 ⁻¹	200
									1.12 ³	6.73 ³
									1.32 ²	801
B1	ĊW	DOE (U.S.A.) (1999), DOE	N							
		(U.S.A.) (2000f)								
B2	SA	See B1 and Leonard et al.	N							
		(2001)								
B3 &	WE	Leonard et al. (2001) and	N							
B6		assumptions								
B4-	See A	3-A4								
B5										
B7-		See B1, Smit and Nieuwlaar								
B17		(1994), Sierra Pacific								
		Resources (2001), and See								
		A6-A15								
B18	See A	16								
B19	See B]								
B20	WE	Calculated from emissions	·							
B21	See B	1		1						
B22-	N	DOE (U.S.A.) (2000f)	N							
B23										
B24	WE	DOE (U.S.A.) (1999), Delta	N	ŝ,						
		Electricity (1999b) and A21			D.					
B25	CE	Estimate based on A22	N		1.0	L.C			10.000	

ration_

Appendix 1:Data Sources

Appendix UData Sources

Sustainability of Australia's Electricity Generation

A1.3.	3.3	Social Flows								
Ref.	Agg.	Source(s)	S. 31		PM	See	bres		Ra	nge
			Agg.	R	C	٦,	G	Ľ	Min.	Max.
Al	N	Macquaric Generation (2000)	N							
A2	MW	Tarong Energy (2002),	N							
		Stanwell (2002), Macquarie								
		Generation (2000), CS Energy								
		(2000)								
B1	SA	DOE (U.S.A.) (1999) and	N							
		Section A1.4.3.3 A1								

A1.3.4 Transmission Subsystem

A1.3.	4.1 1	Material Flows								
Ref.	Agg.	Source(s)	S .		PM	Sco	res		Ra	nge
Ń			Agg.	R	С	Т	G	E.	Min.	Max.
A1	MC	Macquarie Generation (2000),	YA	4	5	2	1	1	0.980	0.993
		Pacific Power (2000), Delta				ŀ				
		Electricity (1999a), Tarong								
		Energy (2000a), Stanwell								
		(2000), CSEnergy (2000),				ļ				
		ESAA (1999) for generation,				ĺ				
		TransGrid (2001), Powerlink								
		(2001) for distances, ESAA								
		(1999) for interstate.								

A1.3.	4.2 1	Economic Fl	ows									
Ref.	Agg.	Source(s)			S.		PM	Sci	ires		Ra	nge
					Agg.	R	С	Т	G	E	Min.	Max.
Al	СР	Hirst and	Kirby (200)1), GPU	EC							
		Powernet	(1987),	Hydro								
		(Canada)	(2001),	Kinhill								
		Stearns (19	87), Treasu	ıry (Qld.)								
		(2001)										

B26	MW	DOE (U.S.A.) (2000f) and Section A1.4.3,1 A26	N	
B27	WE	Nautilus Institute (1999) and A24	N	
B28- B35	EM	BIC-IGCC, Diesel, Petrol, NG-CCGT, Coke Breeze EM	N	

A1.3.3.2 Economic Flows

Rela	Agg.	Source(s)	S. PM Scores						Ra	nge
			Agg.	R	Ċ	٦Ľ (G	E	Min.	Max.
Al	СР	NREC (1987), ERDC (1992),	EC							
		NEA (1986), EIA (1996-7),								
		DOE (U.S.A.) (1999), GTZ				n in Mysin Mese				
		(1996), DoE (U.K.) (1988),								
		IEA (1992), Stanwell (2002),								
		CSEnergy (2001), Tarong								
		Energy (1999), Intergen (1999),				n agus Sinn Sinn				
		Gallagher and Pierce (1986),						en en Lighter		
		IEA (1993), Olds (1973)								
A2	СР	NREC (1987), ERDC (1992),	EC				с 1. С 1.			
		NEA (1994), DOE (U.S.A.)								
		(1999), ECNSW (1991b), GTZ								
		(1996), NEA (1986), NEA								
		(1992), IEA (1992), Sayers and								
		Shields (2001), Simes (2000)								
BI	СР	DoE (U.K.) (1988), DOE	EC							
		(U.S.A.) (1999), DOE (U.S.A.)								
		(2000a-f), IEA (1992), Power								
		Technology (2002), IEA (1993)								
B2	СР	ERDC (1992), DOE (U.S.A.)	EC							
		(1999), DOE (U.S.) (2000a-f),								
		1EA (1992), Maude (1993)	. <u></u>							

Appendix LiData Sources

A2	CP	Freeman et al. (2002),	EC	
		Powerlink (2002), Electranet		
		(2002a), Electranet (2002b),	1.00	
		UMS Group (2000), Simpson	r L	
		(2001)		

A1.3.4.3 Social Flows

Ref.	Agg.	Source(s)	S.	PM Scores Range
			Agg.	R C T G E Min. Max.
Al	CW	ElectraNet (2003), SPIPowernet (2003)	N	
A2	MW	Electranet (2003), Transgrid (2003), Fong (2002)	YA	
A3	N	Fong (2002)	YA	

A1.4 BROWN COAL

A1.4.1 Mining Subsystem

A1.4.1.1 Material Flows

Ref.	Agg.	Source(s)	S. (PM	Sce	ires		Ra	nge
			Agg.	R	C	Т	G	E	Min.	Max.
Al	PW	Confidential (Multiple)	N	4	2	1	1	i	3.61	5.00
A2	See A	7	L = ·	2	2	1	1	1	2.44 8	6.25 8
A3-	PW	Confidential (Multiple)	N	4	2	1	1	1	1.20 *	2.30 5
A4									838	1.31 ³
A5	Estim	ate from A8		2	2	1	1	I	85	8.53 ³
A6	PW	Confidential Source	N	3	3	1	1	1	4.26 7	1.70 8
A7	PW	Yallourn Energy (1999), Hazelwood Power (1999).	YA	2	2	1	1	l	2.44 8	6.25 *
		Loy Yang Power (1999)								
A8	SA	Hazelwood (1999)	YA	2	2	1	1	1	88	8.81 3
A9	PW	Hazelwood Power (1999) and Confidential (Multiple)	N	2	2	1	1	1	1.64 *	5.68 *
A10-	EM	Diesel, Petrol EM	N	3	5	1	2	3	3.87 4	7.24 5

Sustainability of Australia's Electricity Generation

A17	1.16	314
	5.87	3.35 ³
	0.11	62
	54.7	3.12 4
r :	21.3	1.21 4
	11.0	6.32 ³
	2.37	1.31 ³

A1.4.1.2 Economic Flows

Ref.	Agg.	Source(s)	S.		PM	Sec	ores		R	inge
			Agg.	R	С	1	G	E	Min.	Max.
A1	СР	VBCC (1982), DITC (1986),	EC						C ANG	
		SECV (1989)								
A2	СР	VBCC (1982), VAO (1993),	EC	4						
	ļ 1	Ernst and Whinney (1982)								

A1.4.1.3Social FlowsRef. Agg.Source(s)A1PWConfidential (Multiple)

A1.4.2 Transport Subsystem

A1.4.2.1 Material Flows

Ref.	Agg.	Source(s)	S. Agg.	R	PM See C T	ores G E	Ra Min.	nge Max.
AI	WE	See A1.3.2.1 A1 for consumption rate, Confidential for distance	N	4	3 5	3 1	0.161	2.57
A2- A4	WE	Based on assumptions	N	5	5 1	1 1	1.00 ⁹ 495 0.50	1.00 ⁹ 4.95 ⁴ 5.00 ³

Δp	pendix	1:Data	Sources
_			

	, S. Agg.	r R	PM Sco C – T	G E	Ra Min.	nge Max.
)	N					

Appendix EData Sources

Sustainability of Australia's Electricity Generation

AL.4.2.2 ECONOMIC FIOWS	A1.4.2.2	Economic Flows
-------------------------	----------	----------------

Ref.	Agg.	Source(s)	S. Agg.		PM	Sec	res	й., с.,		Range	
				n na seanna an seanna an seanna an seanna an seanna an seanna an seanna an seanna an seanna an seanna an seann Seanna an seanna an s	3 .	n na		- 6	N.11.	1998) 1997 - N. 1997	
				ст і. 5-1 с			7			11 . - 1944 Geografie	
Al	CP	VBCC (1982)	EC								
A2	See A	l	L								

A1.4.2.3 Social Flows

No estimates of social flows for brown coal transport.

A1.4.3 Generation Subsystem

A1,4.3,1 Material Flows

Ref.	.Agg.	Source(s)	S .		PM	Sc	ores		Range			
			Agg.	R	C		G	E	Min.	Max.		
Al	WE	Confidential (Multiple)	YA		2	ì	1	1	5.37 -2	0.11		
A2	PW	Yallourn Energy (1999), Hazelwood Power (1999), Loy Yang Power (1999)	YA	1	2	l	1	1	1.25 6	1.62 *		
A3-	R	SECV (1993) and	N	2	2	3	1	1	0	641		
A5		Confidential (Multiple)							0	1.36 ³		
									0	32.9		
A6-	PW	Confidential (Multiple)	YA	1	1	1	1	1	0	0.90		
A8				! :					4.63	7.06		
				 : :					0	10.2		
A9-	N	Confidential	YA	1	2	1	1	1	266	544		
A15,									26.5	54.3		
A17									44.2	90.4		
	ļ								11.0	22.6		
			- - - 						0.47	0.97		
									7.90 ⁻²	0.16		
									0.55	1.13		
				1					0.51	1.04		
Al6	PW	Confidential (Multiple)	YA	1	2	1	l	1	9.3	26.9		
A18	Based	l on A23	•	1	2	1	1	1	2.53	5.17		

A19	PW	Yallourn Energy (1999),	YΛ	1	2	1	1	1	L.90 ^{°6}	3.17
		Hazelwood Power (1999),								
		Loy Yang Power (1999),								
		Confidential								
A20	PW	Confidential (Multiple)	YA	1	2	1	1	1	7.66 4	1.17
Λ21	Basis	I GWh + Al	L	1	2	1	1	1	1.05	1.1
Λ22	PW	Yallourn Energy (1999),	YA	4	2	l	1	1	1.44	2.80
		Hazelwood Power (1999),								
		Loy Yang Power (1999)								
A23	PW	Hazelwood Power (1999),	ŶΛ	1	2	1	1	l	2.89	5.9
		Loy Yang Power (1999)								
A24	N	Confidential	ΥΛ	1	2	1	1	I	9.11 5	1,90
A25,	PW	Yallourn Energy (1999),	YA	11	2	Ī	1	1	7.67 5	1.79
A27		Loy Yang Power (1999)		1	1	1	1	1	1.10 5	1.6
A26	N	Confidential	ΥΛ	1	2	1	1	1	0.00273	0.2
A28-	EM	BrC-ST, Fuel Oil,	N	2	2	1	2	1	9.83	1.3
A35		Briquettes, NG-ST, Diesel,		5	2	1	1	1	0.652	60
		Petrol, LPG EM		3	5	1	2	3	2.06	20
				ຼີ 2	2	1	1	1	1.61	16
				3	5	1	2	3	153	1.5
					ŀ			:	21.3	2.0
			•	-				i	182	1.8
				:			:		44.6	4.6
B1	WE	Based on Section A1.3.3.1	N							
		B1 plus penalty for MTE				• . • .				
		drier.								
B2	WE	Assumption backed by simulation	N							
B3	WE	Based on Section A1.3.3.1	N			.				
		B3	•							
B4-	MW	Based on A6-A8	N	• • • • •	<u> </u>					
B5										
B6-	R	Based on Section A1.3.3.1	N	-	+					
-			. = .		13	£ -:	<u>}.</u>	1. dit	1	1

238

239

Appendix 1:Data Sources

Sustainability of Australia's Electricity Generation Appendix EData Sources

Sustainability of Australia's Electricity General	ion
---	-----

B15		B8-B17, some ratios based on differences in fuel qualities (i.e. S contents).							
B16	N	Based on Section A1.3.3.1 B18	N						
B17	C	Calculated from other water flows.	N						
B18	R	Based on Section A1.3.3.1 B19 and simulation	N						
B20	C	Based on coal composition.	N			 			
B19	Basis	1 GWh + Bl							
B22	See B	16		-				 	
B23	R	Based on A24	N						
1324	WE	Based on requirements and MTE drier wastewater (Confidential)	N						
B25	MW	A26	N	-		 	·	 	
B26	PW	Based on Section A1.3.3.1 B27	N				 - -		
B27- B34	EM	BrC-IGCC, NG-CCGT, Diesel, Petrol EM	N	-			. :		

A1.4.3.2 Economic Flows

Ref.	Agg.	Source(s)	S .		PM	Se	ores		Ŕa	nge
			Agg.	R	C	T	G	E	Min.	Max.
A1	СР	NREC (1987), ERDC (1992),	EC							
		SECV (1989)								
A2	СР	Ernst and Whinney (1982),	EC							
	 	ERDC (1992), Loy Yang								
		Power (1999), SKM (1997)								
BI	СР	Section A1.3.3.2 B1,	EC							
	ł	Confidential for MTE costs						2014 1947 - 1947 1947 - 1947		
B2	R	Based on Section A1.3.3.2 B2	EC							

A1.4.	.3.3	Social Flows								
Ref.	Agg.	Source(s)	S. Agg.	R	PM C	Se T	ores G	E	Ra Min.	nge Max.
ΛI	PW	Confidential (Multiple)	YA							
٨2	MW	Confidential (Multiple)	YA							
<u>B1</u>	CW	DOE (U.S.A.) (1999) and DOE (U.S.A.) (2000f)	ΥΛ							
B2	Base	i on A2	-I							

AI.J.J. Transmission Subsyste

A1.4	.4 Iri	ansmission 2	subsystem									
A1.4	4.1	Material Flo	ws									
Ref.	Agg.	Source(s)			S. Agg.	R	РМ С	Sce T	ores G	E	Ra Min.	oge Max.
AI	MC	Yallourn Hazelwood Yang Po generation interstate.	Energy I Power (19 ower (19 , ESAA (1	(1999), 999), Loy 99) for 1999) for	ΥΛ	4	5	2		1	0.961	0.993

A1.4.4.2 Economic Flows See Section A1.3.4.2.

A1.4.4.3 Social Flows See Section A1.3.4.3.

A1.5 NATURAL GAS AND LIQUEFIED NATURAL GAS

A1.5.1 Mining Subsystem

A1.5.1.1 Material Flows Ref. Agg. Source(s)

A1, YA Basis + APPEA (199

	S. Agg.	R	PM C	See T	ores G	E	Ra Min.	nge Max.
9)	N							

Sustainability of Australia's Electricity Generation Appendix 1:Data Sources

Δ.5		Τ		Sec.	1	-ixalisis	90.090	198823	eren and a second	-
A.)	<u> </u>									
A2-	YA	APPEA (1999)	N							
A4	ļ									
A6	N	MMS (1998)	N						101100	
A7	PW	EPS (Canada) (1989),	N							
		Canada-Newfoundland								
		Offshore Petroleum Board								
		(2001)								
A9	PW	FPA (11SA) (1999) DTI	N	-8888. 72927					273853234 2928532483	2893335 2963-99
		(11 K) (2000) Confidential								
A 10	San A									
A10-	See A	10								
A11		······································			欬					
A12,	N	EPS (Canada) (1989)	N							
A14										
A13	PW	UKOOA (1999), DTI (2000)	YA							
A16	PW	Scott et al. (1997), Dedman	N		SU.					
		(2000), Santos (1999b),								
		Sampson (2000)								
A17-	EM	NG-F, Diesel, Petrol,	N		aras Xoz					
A19		Aviation Fuel, Crude Oil								
		EM [*] for combustion. APPEA								
		(1999) for fugitive								
A 20	EM	NG E Discel Bytrol								
AD4		Autotion Fuel Out Off	IN .							
A24		Aviation ruel, Crude Oil								
• 		EM								
A25	YA	APPEA (1999)	YA							
				بنا يمنين سميه				سهيد ا		

A1.5.1.2 Economic Flows

R ef.	Agg.	Source	e(9)			S. Agg.	PN R C	1 Score T (s F E	Ra Min,	nge Max.
Al	СР	ACIL	(1998),	OMV	(2001),	EC					

^{*} Special EM for NG mining (see Section A1.8)

	and	Santos (1999a), Woodside							na kongel angles Sangan antara c
	FA	Offshore Petroleum (1992),							
		Woodside (2000), BHPBilliton				80.2 ()			
		(2001b), Anonymous (2002c),							
		APPEA (2001a), Mobil							
		Australia (1999), OTD (N.T.)							
		(2001) for offshore production,						i de la comercia de l	
		FA for onshore production							
A2	СР	ExxonMobil (2002),	EC						
		BHPBilliton (2001b), Santos			40 S S S8 3				
		(1999a), Woodside (2000),							
		UKOOA (2001) for production							
B1	СР	YPC (2001), Hydrocarbon	EC						
		Processing (2001,1999), DoE							
		(U.K.) (1980), Special							
		Correspondent (1977), Mallet							
		(1987), Anonymous (2002c,							
		1997), Santos (1999a) for		4.163 78)					
		processing							
B2	СР	Zauner (2000), DoE (U.K.)	EC						
		(1980), AGA (2001) for	1						
		processing		38			 		

A1.5.1.3 Social Flows

Ref.	Agg.	Source(s)	S. Agg.	R	PM C	i Sc T	ores . G	E	Ra Min.	nge Max.
A1	PW	Woodside (2000) offshore, Santos (1999b) onshore	YA							
A2- A3	MW	NOHSC (2002), APPEA (2001b)	N							
B1	PW	Woodside (2000)	YA	14.ea 224						
B2- B3	See A	2-A3	·							

242

Sustainability of Australia's Electricity Generation Appendix EData Sources

B15		Oil (Water) Diesel, Petrol,		
		LPG EM for combustion,		
	,	Snam (1999) and AGA (1997)		
	 	for lugilive	-	
B13-	EM	Natural Gas (Water, OCGT),	Ν	
B21		Fuel Oil (Water) Diesel, Petrol,		
		LPG EM		

A1.5.2.2 Economic Flows

Ref.	Agg.	Source(s)	S.	PM Scores					Range		
			Agg.	R	Ċ	Т	G	E	Min.	Max.	
Al	СР	Origin Energy (2002),	EC								
		McDonald (2001), Duke						898) 2008/-			
		Energy (2000a, 2000b), Epic									
		Energy (2001), YPC (2001),									
		Anonymous (1998a, 1998b),									
		Klaassen <i>et al.</i> (1999)				88.29 17 - 20 1					
A2	WE	HEE (2002b), Cedigaz (2000),	EC								
	and	OTD (N.T.) (2001), YPC									
	СР	(2001), BG Group (2002),									
		Hydrocarbon Processing									
		(2001, 1999), TotalFinaElf	Į								
	1	(2000) for LNG plants.									
		Flower (1998), TotalFinaElf									
		(2000), Klaassen, et al. (1999)									
		for ocean transport				KS.	K.		e and and and and and and and and and and		
		Anonymous (2002b), Cedigaz									
		(2000), Bradner (2002),									
		Hydrocarbon Processing					K.				
		(2001), Rigzone (2003),						ŀ.			
		TotalFinaElf (2000) for			Ľ						
		vapourisation plant.									
		Supporting information:]					

A1.5.2 Transport Subsystem

A1.5.2.1 Material Flows

Ref.	Agg.	Source(s)	S.		PM	Se	oreș		Ra	nge
			Agg.	R	Ċ	Т	Ģ	E	Min.	Max.
A1-	YA	AGA (1997)	N							
A4										
A17	See A	1-A4								
A18	EM	NG-OCGT, Diesel, Petrol,	N						•	
		LPG EM for combustion,								
]		AGA (1997) for fugitive and					X			
		other								
A19-	EM	NG-OCGT, Diesel, Petrol,	N							
A20		LPG EM, AGA (1997) for								
		ıugitive								
A21-	EM	NG-OCGT, Diesel, Petrol,	N							
A25		LPG EM								
A26	YA	AGA (1997)	N							
B1	PW	Tamura et al. (2001), Snam	N							
		(2001), Snam (1999) for								
		vaporisation, A1 for pieline								
B2	N	Tamura et al. (2001)	N							
B3-	See A	2-A4								
B5										
B6	N	Snam (1999)	N							
B10	WE	Tamura et al. (2001) for ocean	N							
		transport, Snam (2001), Snam								
		(1999), Tamura <i>et al.</i> (2001)								
		for vaporisation, A2 for								
		pipeline								
B11	N	Snam (1999)	N							
B12	See B	6	 _							
B13-	EM	NG-OCGT and Water, Fuel	N							
L	L	L	<u> </u>		1		<u> 1997</u>	<u>1. 199</u> 7		<u>a sta 2280</u>

244

Appendix 1:Data Sources

Sustainability of Australia's Electricity Generation

		Programs (EPA) (2001)							
A3-	MW	Based on Section A1.4.3.1	N						
A5		A6-A8							
A6-	PW	SECV (1973) and Sections	N						
A12		A1.3.3.1 and A1.4.3.1							
A13	Sec A	21	L						<u> </u>
A14	PW	SECV (1973), Generation	N	N. K			383		
		Victoria (1997), Optima							334 (J) 334 (J) 32 (S) 40
		Energy (2000)							
A15	PW	SECV (1973), Generation	N			200 200		<u>.</u>	
		Victoria (1997)							
A20-	N	SECV (1973)	N						
A21				9.4.5 3					
A22	See S	ection A1.3.3.1						N. Salak	
A23	N	Generation Victoria (1997)	N						
		and mass balance							
A24	MW	Based on Section A1.4.3.1	N						
		A26							
A25-	EM	NG-ST, Diesel, Petrol, LPG	N		æ.				
A32		EM							
BI	PW	ECNSW (1991b), NERC	Ϋ́Α						
		(2000)							
B2	PW	Alstom (2001), SECV (1993)	N				l.		24.5
B3	YA	SECV (1993)	N				İ.	222	
B4-	See A	3-A5	L. <u>.</u>						
B6									
B7-	PW	Based on A6-A12 with	N						
B8		reduced water requirements			İ.				
B9	N	SPIA (2001)	N						
B10	PW	Pavri and Moore (2001),	N						
		Davis and Black (2000),							
		NYPA (2001)							
B12	See B	9	L,	t		han		1	

ŗ <u></u>		TotalFinaElf (2000), EncrSca					
		Transport (2002), Golar LNG					
		(2001), OTA (U.S.A.) (1980)					
BI	СР	Systek (2002), ADB (2002),	EC				
		HEE (2002b), NiSource					
		(2001), GPU GasNet (2002),					
		Freeman et al. (2002)					
B2	СР	HEE (2002a), BG Group	EC		S. (.)		
		(2003), EIA (2002, 1997),			NA NA		
		OTA (U.S.A.) (1980), Enoch					
		and Frey (2002), Martin				a an an an an an an an an an an an an an	
		(2001)					

A1.5.2.3 Social Flows

Ref.	Agg.	Source(s)	S.		PM	Se	ores		Ra	nge
			Agg.	R	C	1	G	E	Mio.	Max.
A1	PW	Epic Energy (2003)	N							
A2-	SA	DOL (U.S.A.) (1996), DFFP	YA							
A3		(U.S.A.) (1997)								
B1	WE	Golar LNG Ltd. (2002) for	N							
		administration, Eddy (1998)							nia († 59. Artista	
		for crewing								
B2-	YA	AMSA (2002)	N							
B3										

A1.5.3 Generation Subsystem

A1.5.3.1 Material Flows

Ref.	Agg.	Source(s)	S. Agg.	PM Scores Range R. C T G E Min. Max.
Al	PW	NERC (2000), Confidential	N	
A2	PW	ETSA Corporation (1997), Office of Atmospheric	YA	

246

Appendix 1:Data Sources

B13	N	NYPA (1990)	N						and a start of the	
B14	MW	Based on Section A1.4.3.1	N							
		A26			R.			Star Star		
B15-	ĒM	NG-OCGT, Fuel Oil-OCGT,	N							
B22		Diesel, Petrol, LPG EM	-							
CI	PW	NYPA (2001), NERC (2000)	YA							
C2	SA	NYPA (2001), Alstom	N							
		(2001), DIT (2000)								
C3	PW	See B3	N				400		1.11.1.1.1.1. 1.1.1.1.1.1.1.1.1.1.1.1.1	
C4-	See A	3-A5	L							
C6										
C7-	WE	Based on A6-A12 with	N							
C13	-	modified water requirements								
C14	WE	Based on A13 and B9	N							
C15	WE	Clark Public Utilities (2001),	Ň						NA (AL) Na (AL)	
		BHP Minerals Technology								
		(2001), NYPA (2001)					2002			
C16	N	NYPA (2001)	N							
C18	SA	NYPA (2001), SECV (1973)	N							
C19	See C	14	L							
C20	See S	ection A1.3.3.1								
C21	WE	Based on other flows	N							
C22	MW	Based on Section A1.4.3.1	N					ðs. Seð		
		A26								
C23-	EM	NG-CCGT, Fuel Oil-CCGT,	N							
C30		Diesel, Petrol, LPG EM								
	·	L		K. 2. 02.4	1	Land	2000(00-0)	L 200		CA 7 August Conference Conference Conference Conference Conference Conference Conference Conference Conference C

A1.5.3.2 Economic Flows

Ref.	Agg.	Source(s) S. PM Scores Range Agg. R C T G E Min, Max.
Al	CP	GTZ (1996), Johnson (1987), EC
		Olds (1973), NEA (1998,

[_ ····	1992), SECV (1987)							NA: N	
A2	СР	GTZ (1996), NEA (1992,	EC							
		1998), Freehill, Hollingdale								
		and Page (1999), SECV (1987)						S.A.		
61	СР	ERDC (1992), GTZ (1996),	EC	188) X23) 1						
		BGAG (2002), Johnson (1987),								
		Anonymous (2002a), SECV								
		(1987), Driver and Ritchey								
		(1998)								
B2	СP	ERDC (1992), GTZ (1996),	EC							
		SECV (1987), Driver and							Contrast.	
		Ritchey (1998)								
CI	СР	DOE (U.S.A.) (1999), National	EC						(
		Power (2000), ERDC (1992),								
		GTZ (1996), Treasury (Qld.)								
		(2000), ECNSW (1991b),								
		SCPPA (2002), SECV (1987),					N.			7
		Driver and Ritchey (1998)								
C2	CP	ERDC (1992), DOE (U.S.A.)	EC							
		(1999), ECNSW (1991),								
	ļ	SCPPA (2002), IEA (1992),								en neue
		GTZ (1996), NEA (1998),								
		SECV (1987), Driver and								
		Ritchey (1998)								
		,		10.000 0.000	1	 A 20025 	- ·····C :20	en 1996 - Si Xe	March 1997 Control of	

A1.5.3.3 Social Flows

Ref.	Agg.	Source(s)	S.		PN	l Sc	ores		Ra	nge
			Agg.	R	C	T	G	Ľ	Min:	Max.
A1, B1	MW	Ecogen Energy (1999)	N							
Cl	R	DOE (U.S.A.) (1999) and	N							
		Section A1.3.3.3 A1								
A2,	EN	Ecogen Energy (1999)	N							
B2, C2			 				Ľ			

248

Appendix 1:Data Sources

Appendix 1:Data Sources

Sustainability of Australia's Electricity General

3.2.4

A1.6.2 E	conom	ic Factors							
Ref.	Agg.	Source(s)	S.		PM	Sco	res	Ra	nge
		DMD (2000) Alwayad	Agg.	K				Min:	Max.
A1	FW	and Sutherland (2000)	14						
		DMF (1998) and DMF							
		(2000)			ilia k L'Ang				
Λ2	N	Freebairn and Trace	N						
		(1998)							
A3, A9,	N	SECV (1989)	N				2000 1941		
A12									
Λ4	N	Assumption (economics	N						
		based on entire field							
		estimate, no Australian							
	1	field has stopped							
		production)					· · · ·		
A5	N	CFAC (III.) (1977)	N						
A6	N	Spath <i>et al.</i> (1999)	N						
Λ7	N	Freebairn and Trace	N 						
· · · · · · · · · · · · · · · · · · ·		(1998)							
A8	N	Chrzanowski (1983)	N	-	ļ				
A10	N	Pipelines are long-life	N						
	N	Flower (1998)							
AH A12	N	Picharde (1998) DOF							
M12		(11 S.A.) (1999) SKM							
		(1997)							
A14-	YA	Simes (2000)	EC						
A15,			1						
A22							Ś		

A1.5.4 Transmission Subsystem

A1.5.4.1 Material Flows

Ref.	Agg, Source(s) S. Agg.	PM Scores Range	
		R C T G E Min. Mr	ix .
Al	See Section A1.3.4.1 A1		

A1.5.4.2 Economic Flows See Section A1.3.4.2.

A1.5.4.3 Social Flows See Section A1.3.4.3.

A1.6 CONVERSIONS NECESSARY 'O PRODUCE INDICATORS FROM COLLECTED DATA

A1.6.1 Environmental Factors

Ref.	Agg.	Source(s)	S. •		PM	Sci	ores		Ra	nge
)		Agg.	R	C	Т	G	E	Min.	Max,
Al	MC	AGA (2000a) for gas	Ν		•					
-		consumption, APPEA (1998)								
		for production								
BI	Eq.	ABS (1998c)	YA							
	3.2.1									
C1-	Eq.	ABS (2000), ABS (1998b),	YA							
C4	3.2.4	ABS (1998c)								
C5	Eq.	ABS (2000), NGGIC (2000)	YA							
	3.2.4								:	
C6	Eq.	NGGIC (2000), NPI (2001)	YA							
	3.2.4									
C7	Eq.	NGGIC (2000)	YA							
	3.2.4									
C8	Eq.	NPI (2001), NGGIC (2000)	YA							
	3.2.4				1					
C9	Eq.	ABS (1998a)	YA							

uion		 Appendix 1:Data Sources								
	r						·····			

Appendix 1:Data Sources

Sustainability of	Australia's	Electricity	Genera
-------------------	-------------	-------------	--------

A46	N	Frechill, Hollingdale and Page (1999)	N	
Λ47	N	ERDC (1992)	N	
A48	N	IEA (1992)	N	
A49	N	National Grid (2001)	N	
Bl	N	ABS (2003b)	YA	
B2	N	ABS (2003c)	YA	
B3	N	ABS (2003a)	YA	
B4	N	ABS (2003a), ABS (2003b)	YA	

A1.6.3 Social Factors

Ref.	Agg.	Source(s)	S. Agg.	PM R.C	Scores T G	Ra E Min.	nge Max.
Bl	N	ABS (2003d)	RA				
B2-B3	N	NOHSC (2002)	YA				

A1.7 APPLICATION OF UNCERTAINTY AND SENSITIVITY ANALYSES

Ref.	Agg.	Source(s)		S.	PM D C	Scores	Range
				. (*66)			
BI-B4	MC	Simes (2000)		YA			statist (same) 1922 - Statist
B5-	MC	APPEA (1999),	ABS	YA			
B10		(2001)					

A1.8 EMISSION FACTORS

Fuel	Туре	Substance	Source(s)	Value (kg per kg)
BIC	ST	CO ₂ , CH ₄ , VOC,	NGGIC (1998a)	2.19, 2.16 ⁻⁵ , 4.08 ⁻⁵ ,
		N ₂ O, NO _x ,		1.92 ⁻⁵ , 7.11 ⁻³ ,
		CO, SO ₂		2.64 -4, 8.88 -3

A16-	N	O & M of proceeding	EC		<u>.</u>			
A17,		subsystem.						
A19-								
A21,								
A24-								
A25			- -					
A18,	YA	Zauner (2000)	EC					
A26,								
A28,								
A30								
A23,	YA	Institute of Energy	EC				an ang ang ang ang ang ang ang ang ang a	
A27		Economics (Japan) (2002)						
A29,	YA	IEA (1999)	EC					
A31								
A32	N	ACCC (1986)	N					
A33	SA	ACCC (1986), Errington	i N					
		and Kirby (1993)						
A34,	N	Ernst and Whinney (1982)	N					
A39,								
A44								
A35-	SA	EIA (2002), Brennan et al.	N					
A36		(1980)						
A37	N	Freebairn and Trace	N					
		(1988)						
A38	N	BTCE (1998)	N			502		
A40	N	DoE (U.K.) (1980)	N					
A41	SA	DoE (U.K.) (1980),	N					
		Bourgeois (1990), OTA						
		(1980)						
A42,	N	DOE (U.S.A.) (1999)	N					
A43								
A45	N	DOE (DOE) (U.S.A.)	N					
		(2000f)						
		L		[3030]	98-18 1	<u></u>	<u> 7500</u> 0	

Estimation of most gaseous and some particulate (non-dust) emissions was by emission factors (EM). Section A1.10.3 contains an example of emission factor use.

Appendix 1:Data Sources

Sustainability of Australia's Electricity Generation

BIC	ST	PM	Pacific Power (2000),	1.06 -3
			Delta Electricity (1999b),	
			Tarong Energy (2000a),	
			NPI (1999)	
	IGCC	CO ₂ , CH ₄ , NO _x ,	DOE (U.S.A.) (1999),	2.43, 1.16 -4, 5.13 -4,
		CO, SO2, PM	Leonard <i>et al.</i> (2001)	5.83 ⁻⁴ , 1.98 ⁻⁴ , 2.33 ⁻⁵
		VOC	Smit and Nieuwlaar	7.58 ⁻⁷ , 3.48 ⁻⁵
		N ₂ O	(1994), DOE (U.S.A.)	
			(1997), Sierra Pacific	
			Resources (2001), IEA	
			(1993), IEA (1992),	
			Keeler (2001)	
BrC	ST	CO ₂ , CH ₄ , VOC,	NGGIC (1998a)	0.824, 4.32-6, 1.47-5,
		N ₂ O, NO _x ,		1.21 -5, 1.14 -3,
		CO, SO ₂		1.50 ⁻⁴ , 1.29 ⁻³
		PM	Yallourn (1999),	3.36 -4
			Hazelwood Power (1999),	
			Loy Yang (2000)	
ĺ	IGCC	CO ₂ , CH ₄ , VOC,	Based on BIC-IGCC and	0.950, 4.18 ⁻⁴ , 2.73 ⁻⁶ ,
		N ₂ O, NO _x ,	BrC-ST	1.30 -5, 1.92 -4,
		CO, O ₂ , PM		2.70 ⁻⁴ , 2.55 ⁻³ , 2.61 ⁻⁵

NG	ST	CO ₂ , CH ₄ , VOC,	NGGIC (1998a)	2.62, 5.12 -6, 3.29 -5,
		N ₂ O, NO _x ,		5.12 ⁻⁶ , 5.95 ⁻³ ,
		CO, SO ₂		8.20 -4, 1.18 -7
		РМ	NPI (1999)	6.65 ⁻³
	OCG	CO ₂ , CH ₄ , VOC,	NGGIC (1998a)	2.63, 3.48 -4, 8.06 -3,
	Т	N ₂ O, NO _x ,		5.12 ⁻⁶ , 8.38 ⁻³ ,
		CO, SO ₂		1.94 ⁻³ , 1.18 ⁻⁷
		РМ	NPI (1999)	9.21 -4
	CCG	CO ₂ , CH ₄ , VOC,	Argonne National	2.71, 1.13 ⁻⁴ , 6.34 ⁻⁵ ,
	Т	N ₂ O, NO _x ,	Laboratory (1996)	4.98 ⁻⁵ , 3.54 ⁻³ ,
		CO, SO ₂ , PM		1.67 ⁻⁴ , 1.40 ⁻⁵ , 1.63 ⁻⁴

Fuel	st	CO ₂ , CH ₄ , VOC,	NGGIC (1998a)	2.98, 4.29 *, 3.43 *,
Oil		N ₂ O		2.57 -5
		NO _x , CO,	NPI (1999)	8.40 ⁻³ , 6.30 ⁻⁴ ,
		SO ₂ , PM		5.95 -2, 3.28 -4
	Ship	CO ₂ , CH ₄ , VOC,	Based on Fuel Oil-ST and	2.98, 2.51 -3, 2.23 -2,
	(W)	N2O, NO _s ,	BIC-ST	2.05 -4, 5.49 -2,
		CO, SO ₂ , PM		7.44 ⁻⁴ , 5.95 ⁻² , 3.28 ⁻⁴
Fuel C	Dil-IGCC	(see Fuel Oil-ST)		<u> </u>
Fuel (Dil-OCG	T (see Fuel Oil-ST)		
Fuel C	Dil-CCG	T (see Fuel Oil-ST)	_ <u>, </u>	
Coke	Breeze (s	see BIC-ST)		
Bri-	ST	CO ₂ , CH., VOC,	Based on BrC-ST,	1.71, 8.98 -6, 3.05 -3,
que-		N ₂ O, NO _x ,	NGGIC (1998a) and	2.51 -5, 2.37 -3,
ttes		CO, SO ₂ , PM	DISR (1999)	3.11 ⁻⁴ , 2.69 ⁻³ , 6.99 ⁻⁴
Die-	Auto-	CO ₂ , CH ₄ , VOC,	NGGIC (1998a), NGGIC	3.13, 2.56 -4, 4.85 -3,
sel	moti-	N ₂ O, NO _x ,	(1998c)	8.98 -5, 4.51 -2,
	ve	CO, SO ₂		1.75 -2, 9.16 -3
		PM	NGGIC (1998b)	1.86 -3
	Ship	CO ₂ , CH ₄ , VOC,	APPEA (1999)	3.2, 2.70 - , 2.40 - 3
	(W)	N ₂ O, NO _x , CO		2.20 ⁻⁴ , 5.90 ⁻² , 8.00 ⁻³
		SO2, PM	See Diesel Automotive	9.16 ⁻³ , 1.86 ⁻³
Pet-	Auto-	CO ₂ , CH ₄ , VOC,	NGGIC (1998a), NGGIC	3.06, 1.21 -2, 9.28 -4,
rol	moti-	N₂O, NO _x ,	(1998c)	4.64 -5, 8.82 -3,
	ve	CO, SO ₂		5.10 -3, 6.96 -4
		PM	NGGIC (1998b)	2.01 -3
LP	Auto-	CO ₂ , CH ₄ , VOC,	NGGIC (1998a), NGGIC	2.95, 1.09 -3, 2.03 -2,
G	moti-	N ₂ O, NO _x ,	(1998c)	4.96 ⁻⁵ , 2.17 ⁻² ,
	ve	CO, SO ₂		0.271, 3.97 -4
		PM	Wang and Huang (1999)	3.28 -3

Natural Gas mining subsystem emission factors (APPEA (1999) for CO₂ to CO, SO₂ and PM based on NG-ST, Fuel Oil-ST (for crude oil), diesel (automotive), and petrol (automotive) for aviation fuel).

254

Appendix 1:Data Sources

Appendix 1:Data Sources

Subst-	NG-F	Crude	Diesel	Diesel	NG	NG	Aviation
ance		Oil	(turbines)	(engines)	(turbines)	(engines)	Fuel
CO ₂	2.75	3.2	3.2	3.2	2.75	2.75	3.2
CH ₄	3.50-2	3.30-4	8.00-5	1.40 -4	4.20 -4	2.80-2	8.70.3
VOC	1.50-2	2.97-3	7.00-4	1.90-*	5.10*5	3.00-3	8.00-4
N ₂ O	8.10-3	2.20-4	2.20-4	2.20-4	2.20-4	2.20-4	2.20
NO _x	1.50-3	3.70 -3	9.40-3	7.00 -2	6.70-3	7.60-2	1.25-2
<u>CO</u>	8.70 ⁻³	1.80 -2	2.10-3	1.90*2	2.70 -3	9.60-3	5.20-3
SO ₂	1.18-7	5.95-2	9.16-3	9.16	1.18-7	1.18-7	6.96 -4
РМ	9.21 -	3.28-4	1.86-3	1.86 ⁻³	9.21	9.21	2.01 .3

A1.9 AUSTRALIAN NORMALISATION DATA FOR THE ENVIRONMENTAL INDICATORS This section shows the base data used to generate the Australian normalisation factors, shown in Table 4.3.49 (Chapter 4), for the environmental indicators.

Table of the data used to produce normalisation factors for the CC, EU, AD, PS and SW indicators (see Section 4.3.6.2 for sources).

Substance	Mt'per annum	Substance	Mt per annum
CO2	312.1	PFCs	0.0002
СН4	5.5	SF6	0
N2O	0.089	SO2	1.8
NOx	2.6	Ammonia	0.016
со	18.6	Solid Waste	2.12205×10^4
NMVOC	1.9		

Table of the data used to produce normalisation factors for the RD W and RD A indicators (see Section 4.3.6.2 for sources).

Substance	Production (MIt)	Reserve	Substance	Production (Mt)	Reserve
antimony (Sb)	1.3	89.9	uranium (U)	5	622
cadmium (Cd)	1.9	132	zinc (Zn)	1.1	3.99 x 10 ⁴
cobalt (Co)	0.9	414	bauxite	43.1	3.02 x 10 ⁶
copper (Cu)	0.5	2.37 x 10 ⁴	iron (ore)	147	1.78 x 10 ⁷
gold (Au)	289	4,45	manganese (ore)	2.1	1.18 x 10 ⁵
lead (Pb)	513	1.87 x 10 ⁴	Black Coal	2.52 x 10 ⁵	1.72 x 10 ⁸
lithium (Li)	3	166	Brown Coal	5.30×10^4	1.44 x 10 ⁸

Sustainability of Australia's Electricity Generation

Substance	Production (Mt)	Reserve (Mt)	Substance .	Production (Mt)	Reserve (Mt)
nickel (Ni)	113	6370	Natural Gas	2.36 x 10 ⁴	2.55 x 10 ⁶
silver (Ag)	1	43.3	Oil	2.07 x 10 ⁴	4.86 x 10 ⁵
tantalum (Ta)	0.3	8.1	Condensate	2850	1.97 x 10 ⁵
tin (Sn)	12,2	120	LPG	2590	1.99 x 10 ⁵
tungsten (W); wolfraam	0.1	0.9			

A1.10 AGGREGATION PROCEDURES

A1.10.1 Overview of Aggregation Procedures In this section are details of the methods used to produce aggregated data from multiple sources. Sections A1.3 to A1.8 provide details on the data sources and the methods applied in each case.

A1.10.2 Averages (YA, FA, SA, RA, PW, CW and MW)

A1.10.2.1 SA (simple average), YA (yearly average) and RA (rolling average) A simple we is an average of multiple values. A yearly average is an average of values from multiple years. For example: NG consumption in Generation subsystem (Data Ref. A2).

Vear	kg per GWhpa
1994-95	196044
1995-96	199759

Thus:

$$V_{YA} = \sum_{y=1}^{2} V_y / 2 = \frac{196044 + 1997}{2}$$

A rolling average (RA) is a yearly average over consecutive years (i.e. 1995-1997 is a 3-year rolling average).

A1.10.2.2 FA (factored average) A factored average is an aggregate produced using averages (V_1 and V_2), where the aggregate is not a simple sum ($V_A \neq V_1 + V_2$). In this case, factors (F₁ and F₂) enable summation. For example: Diesel consumption in BIC mining subsystem (Data Ref. C3). Underground and surface mining consume different amounts of diesel. The proportion of average BIC produced in underground and surface mines respectively is not 50 %.

256

Appendix 1: Data Sources

¹⁵⁹ = 197901.5 kg per GWhpa

Appendix 1:Data Sources

Mine Type	Diesel (kg per Mtpa)	% of coal
Surface	21730	24.7
Underground	1625023	75.3

Thus:

 $V_A = F_1 V_1 + F_2 V_2 = (21730 x 24.7) + (1625023 x 75.3) = 1646752 \text{ kg per Mtpa}$

A1.10.2.3 PW (production), CW (capacity) and MW (employment or employee hoursweighted averages)

Production-weighted averages (V_{PW}) assume that systems that produce (P) more of the desired product are more accurate estimates (V_i) than those that produce less. For example: BIC, ST generation subsystem, coal consumption.

Power Station	Coal (kg per GWhpa)	Generation (GWhpa)
Bayswater	483589	15364
Eraring	755259	7400

Thus:

$$V_{PW} = \sum_{i=1}^{T} (V_i, P_i) / \sum_{i=1}^{T} P_i = \frac{(V_1, P_1 + V_2, P_2)}{(P_1 + P_2)} = \frac{(483589 \times 15364 + 755259 \times 7400)}{(15364 + 7400)}$$

= 571902 kg per GWhpa

Similarly, capacity-weighted averages (V_{CW}) assume that systems that have greater capacity (C) to produce the desired product are more accurate estimates (V_i) than those that have less capacity.

$$\mathbf{V}_{CW} = \sum_{i=1}^{T} \left(\mathbf{V}_{i} \cdot \mathbf{C}_{i} \right) / \sum_{i=1}^{T} \mathbf{C}_{i}$$

Employment-weighted averages (V_{EW}) assume that systems that have greater employment (E) are more accurate estimates (V_i) than those that have less employment, for social flows.

$$V_{EW} = \sum_{i=1}^{T} \left(V_i \cdot E_i \right) / \sum_{i=1}^{T} E_i$$

For other flows, an employment-weighted average (V_{EW}) is a value produced by weighting another systems value (V_{OS}) by the relative employment in each system (E_{EW} and Eos). For example: BIC, IGCC generation subsystem, sewerage emissions.

Fuel	TechnologyEn	ployment (per GWh)Sewerage (kg per GW)
BrC	ST	0.0231	0.0273
BIC	IGCC	0.0281	VBIC-IGCC

Thus,

A1.10.3 Emission factors (EM)

Emission factors (EM) assume consistent fuel compositions. The estimation of the emission of each substance (E_i) uses the consumption rate of a fuel (m_F) and the appropriate emission factor for the fuel and substance (EM_{Ei}) . For example: BIC mining subsystem, CO₂ and methane emissions. Diesel and petrol rates are from Section 4.3.2.1.2. The EM are from Section A1.7.

Substance	Rate (kg per Mtpa)	EMco2 (kg per kg	EM _{CH4})(kg per kg)
Diesel	1.50 x 10 ⁶	3.13	2.56 x 10 ⁻⁴
Petrol	1140	3.06	1.21 x 10 ⁻²

Thus:

 $E_{CO2} = m_D x EM_{D,CO2} + m_P x EM_{P,CO2} = 1.50 x 10^6 x 3.13 + 1140 x 3.06$ $= 4.70 \times 10^{6} \text{ kg per Mtpa}$ $E_{CH4} = m_D x EM_{D,CH4} + m_P x EM_{P,CH4} = 1.50 \times 10^6 x 2.56 \times 10^{-2} + 1140 x 1.21 \times 10^{-2}$ $= 3.84 \times 10^4$ kg per Mtpa

A1.10.4 Economic Correlations (CP and EC)

A1.10.4.1 CP (correlation using power function) subsystem, capital cost (Data Ref. A1).

 $1 = 13.6(Q)^{0.663}$

Other details: standard error $(s^2) = 327$; correlation coefficient (R) = 0.85; number of data points = 61 (3 excluded outliers); and references = 15. These other details are

Sustainability of Australia's Electricity Generation

Appendix 1:Data Sources

 $V_{BIC-IGCC} = V_{BrC-ST} (E_{BIC-IGCC} / E_{BrC-ST}) = 0.0273 (0.0281 / 0.0231) = 0.0218 \text{ kg per GWh}$

Economic source data aggregation uses a power-law correlation (see Equations 3.3.6 and 3.3.10). This correlation uses a computer program (CurveExpert 1.3°), which also provides estimates of error and other details. For example: BIC, ST generation

Equation 4.3.20

for: Q = 50 to 2800 MW and I =\$ 168 to 2410 million (AUD 2000).

available on request.

A1.10,4.2 EC (economic correlation)

Economic correlation converts economic values (EV) of non-basis currency and age into the appropriate basis (AUD 2000). Two steps are necessary. The first step is to convert the original source currency into AUD using market currency conversion factors (MCF). MCF are available from many sources, including http://pacific.commerce.ubc.ca/xr/ and www.fxtop.com. The second step uses historic consumer price indexes (CPI) to change the age of the estimate. Short-term CPI are available from ABS (2002a), and long term from ABS (2002b).

$$EV_{(AUD 2000)} = EV_{(X Y)}.MCF_{\left(\frac{AUD}{Y}, Y\right)}.\frac{CPI_{2000}}{CPI_{Y}}$$

where: $EV_{(AUD 2000)}$ is the converted estimate; $EV_{(X|Y)}$ is the original estimate in currency X and with basis (or published) year Y; $MCF_{(AUD/X,Y)}$ is the market currency factor for converting X currency to AUD currency in year Y; and CPI₂₀₀₀ and CPI_y are the consumer price indexes for Australia in the years Y and 2000.

For example: BIC, ST generation subsystem, capital cost. DOE (U.S.A.) (1999) estimate for ST system (397.5 MW) is \$ 446859 (USD 1998) (January). $MCF_{(AUD/USD,1998)} = 1.523$ AUD per USD in 1998. CPI_{2000} and $CPI_{1998} = 131.3$ and 120.3 (ABS (2002a)).

Thus:

$$EV_{(AUD 2000)} = EV_{(USD 1998)} \cdot MCF_{(AUD (USD 1998))} \cdot \frac{CPI_{2000}}{CPI_{1998}} = (446859)(1.523)(\frac{131.3}{120.3})$$

= \$742796 (AUD 2000)

A1.10.5 Other Types (FE, WE, MC and R)

A1.10.5.1 FE (estimate obtained by applying a factor to another data value) Only one FE estimate appears in the data set, high quality water consumption (m_{HOW}) in the BIC generation subsystem. This value is the product of the low quality water consumption ($m_{LOW} = 1.78 \times 10^6$) and a factor (f = 10 % of $m_{LOW+HOW}$).

 $m_{HOW} = m_{LOW} \cdot f/(1-f) = (1.78 \times 10^6)(0.1/(1-0.1)) = 1.98 \times 10^5 \text{ kg per GWh}$

V_i values are averages. For example: see Section A1.10.2.2.

$$\mathsf{V}_{\mathsf{PW}} = \sum_{i=1}^{\mathsf{T}} \big(\mathsf{V}_i . \mathsf{F}_i \big)$$

A1.10.5.3 MC (solution of multiple linear, multivariate equations) Some reported data (V) is for multiple products (i.e. A + B + C). Thus: $V = V_A + V_B$ V_{c} . It is difficult to establish the contribution of one particular product (i.e. V_{A}) from this reported data. However, if a number of different sources (1, 2 and 3) report V, then:

$$V_{1} = V_{A,1} + V_{B,1} + V_{C,1}$$
$$V_{2} = V_{A,2} + V_{B,2} + V_{C,2}$$
$$V_{3} = V_{A,3} + V_{B,3} + V_{C,3}$$

If the proportional contributions of each product can be assumed identical for each source (i.e. the same amount of V is caused for the same amount of A produced), then: $P_{B}V_{1} + P_{C}V_{1}$ $P_{B}V_{2} + P_{C}V_{2}$

$$V_{1} = V_{A,1} + V_{B,1} + V_{C,1} = F_{A} \cdot V_{1} + V_{C,1}$$
$$V_{2} = V_{A,2} + V_{B,2} + V_{C,2} = P_{A} \cdot V_{2} + V_{C,3}$$
$$V_{3} = V_{A,3} + V_{B,3} + V_{C,3} = P_{A} \cdot V_{3} + V_{C,3}$$

Solving this set of equations uses matrix methods:

$$\begin{bmatrix} V_{1} \\ V_{2} \\ V_{3} \end{bmatrix} = \begin{bmatrix} P_{A} \cdot V_{1} + P_{B} \cdot V_{1} + P_{C} \cdot V_{1} \\ P_{A} \cdot V_{2} + P_{B} \cdot V_{2} + P_{C} \cdot V_{2} \\ P_{A} \cdot V_{3} + P_{B} \cdot V_{3} + P_{C} \cdot V_{3} \end{bmatrix} =$$

or
$$\begin{bmatrix} P_{A} \\ P_{B} \\ P_{C} \end{bmatrix} = \begin{bmatrix} V_{1} & V_{1} & V_{1} \\ V_{2} & V_{2} & V_{2} \\ V_{3} & V_{3} & V_{3} \end{bmatrix}^{-1} \begin{bmatrix} V_{1} \\ V_{2} \\ V_{2} \\ V_{3} \end{bmatrix}$$

For example: Environmental conversion factors, NG mining subsystem, fuel gas consumption (Data Ref. A1).

A1.10.5.2 WE (calculation built in parts using separate sources for each part) A worked estimate (V_{WE}) uses multiple sources to determine parts of the estimate (V_i), and then sums them using factors (F_i). FA is a special case of WE, in which the various

 $P_{B}V_{3} + P_{C}V_{3}$

$$\begin{bmatrix} V_1 & V_1 & V_1 \\ V_2 & V_2 & V_2 \\ V_3 & V_3 & V_3 \end{bmatrix} \begin{bmatrix} P_A \\ P_B \\ P_C \end{bmatrix}$$

Appendix 1:Data Sources

State			Productio	on (kt)		Fuel Gas
	NG	I.PG	ENG	Condensate	Crude	(kt)
Victoria	3846	1159	0	576	10622	664
Queensland	994	125	0	142	626	70
WA	4284	528	7397	4704	9031	1477
SA	3722	411	0	380	569	215
NT	355	0	0	0	961	4
Australia	13201	2223	7397	5802	21809	2431

Thus:

664		$[3846.P_{NG} + 1159.P_{LPG} + 576.P_{Cond.} + 10622.P_{Oil}]$
70		$994.P_{NG} + 125.P_{LPG} + 142.P_{Locul.} + 626.P_{Oul}$
1477		$4284.P_{NG} + 528.P_{LPG} + 7397.P_{LNG} + 4704.P_{Cond.} + 9031.P_{Oil}$
215	=	$3722.P_{NG} + 411.P_{LPG} + 380.P_{Cond.} + 569.P_{Oil}$
4		$355.P_{NG} + 961.P_{Oil}$
2431		$[13201.P_{NG} + 2223.P_{LPG} + 7397.P_{LNG} + 5802.P_{Cond.} + 21809.P_{Oil}]$

OF

664		3846.P _{NG} + 12358.P _{L.iquid}
70		994.P _{NG} + 892.P _{Liquid}
1477		$4284.P_{NG} + 7397.P_{LNG} + 14263.P_{L.iquid}$
215	=	3722.P _{NG} + 1360.P _{Liquid}
4		355.P _{NG} + 961.P _{Liquid}
2431		$[13201.P_{NG} + 7397.P_{LNG} + 21809.P_{Liquid}]$

Using a number of sets of three equations from the second form (i.e. below) enables the estimation of the fuel gas consumption for each type of fuel: $P_{NG} = 0.0410$ kg fuel gas per kg natural gas, $P_{LNG} = 0.106$ kg fuel gas per kg LNG, and $P_{Liouid} = 0.0363$ kg fuel gas per kg liquids (LPG, crude oil and condensate).

Example set:

[70]		994.P _{NG} + 892.P _{Liquid}
1477	=	4284.P _{NG} + 7397.P _{LNG} + 14263.P _{Liquid}
215		3722.P _{NG} + 1360.P _{Line1}

or

P _{NG}		70	994	0	892	-1	0.0488
P_{LNG}	=	1477	4284	7397	14263	=	0.124
P _{Liquid}		215	3722	0	1360		0.0245

Thus $P_{NG} = 0.0488$ kg fuel gas per kg natural gas, $P_{LNG} = 0.124$ kg fuel gas per kg

Sustainability of Australia's Electricity Generation

LNG, and $P_{Liquid} = 0.0245$ kg fuel gas per kg liquids (LPG, crude oil and condensate).

A1.10.5.4 R (estimate relying on ratios based on old data); modified using the ratio of the old and new requirement (F_N/F_0) . $m_N = m_O(F_N/F_O)$

A1.11 SUMMARY OF DATA SOURCES FOR THE SUSTAINABILITY ASSESSMENT The data collected allow the production of sustainability indicators for electricity generation systems, which consume Australian fossil fuels. These tables allow the duplication of the data shown in Chapter 4.

Ratio based estimates (V_R) use ratios from one source, with data from another, to produce an estimate. There is one such case: BrC, ST generation, auxiliary fuel consumption. The auxiliary fuel requirements were similar for each of three included power stations, in both an old public source ($F_0 = 0.11$ GWh per MW capacity) and a new confidential source (F_N) . The auxiliary fuel types and their respective material requirements were in the old source, but not the new. The old fuel mix was thus

Appendix 2: Examples of Calculation Steps

APPENDIX 2: EXAMPLES OF SUSTAINABILITY INDICATOR PRODUCTION STEPS

A2.1 OVERVIEW OF EXAMPLES OF SUSTAINABILITY INDICATOR CALCULATION STEPS This appendix contains details and examples of the calculation methods for the sustainability indicator methods. Section A2.2 details the methods used to convert the subsystem basis data of Chapter 4 into the system basis. Section A2.3 details the process of normalisation.

A2.2 BASIS CONVERSION AND CONVERSION FACTORS

A2.2.1 Material Flows

A2.2.1.1 Conversion Data

Section 4.3.6.2.1 describes the method for producing conversion data to change the basis of data from subsystem to system. For example: BIC, ST system.

The method requires the material flow data of the basis material for each subsystem (see Table data from Section 4.3.2). The rates 'To' refer to the flow of the reference substance required by the subsystem named in the 'Subs_-tem' column (mBS, Subsystem, To). The rates 'from' refer to the flow of the reference substance from the subsystem named in the 'Subsystem' column (m_{BS,Subsystem,From}). The qualifier 'to be' indicates that the basis is an input material flow.

Subsystem	Basis Substance		To	tates,	From
Mining	Black Coal	:		1 x 10 ⁹	kg per Mtpa mined
Transport	Black Coal	1 x 10 ⁹	kg per Mtpa to be	9.98 x 10 ⁸	kg per Mtpa to be
Generation	Electricity	4.49 x 10 ⁵	transported kg per GWh generated	1	transported GWh per GWh generated
Transmission	Electricity	1	GWh per GWh to be delivered	0.986	GWh per GWh to be delivered
System	Electricity			1	GWh per GWh delivered

Step 1: Convert the transmission (Trm) basis into the system (Sys) basis.

Transmission data has 'to be' in its units, and thus has an input basis. Conversion to output basis involves division of the input by the system output.

$$F_{BS-S,Trm} = \frac{m_{BS,Trm,To}}{m_{BS,Trm,From}} = \frac{1}{0.986} = 1.01 \text{ GWh to be delivered per GWh delivered}$$

Sustainability of Australia's Electricity Generation

This F_{BS-S,Trm} is the factor used to convert all transmission material flows into the system basis.

Step 2: Convert the generation (Gen) basis into the system basis. Generation data is already in output basis, thus no conversion to output basis (see Step 1 as an example) is necessary.

$$F_{\text{BS-S,Gen}} = \frac{1}{m_{\text{BS,Gen,From}}} \left(\frac{m_{\text{BS,Trm,To}}}{m_{\text{BS,Trm,From}}} \right) = \frac{1}{m_{\text{BS,Gen,From}}} \left(F_{\text{BS-S,Trm}} \right) = 1 \times 1.01$$

=1.01GWh generated per GWh delivered

As in Step 1, this F_{BS-S,Gen} factor is used to convert all generation material flows into the system basis.

Step 3: Convert the transport (Tsp) basis into the system basis. Transport data is in input basis, and thus conversion to output basis is necessary.

$$F_{\text{BS-S,Tsp}} = \left(\frac{m_{\text{BS,Tsp,To}}}{m_{\text{BS,Tsp,From}}}\right) \frac{1}{m_{\text{BS,Gen,From}}} \left(\frac{m_{\text{BS,Trm,To}}}{m_{\text{BS,Trm,From}}}\right) = \left(\frac{1 \times 10^9}{9.98 \times 10^8}\right) \times 1 \times 1.01$$
$$= 1.02 \text{ GWh generated per GWh delivered}$$

The basis of the transport subsystem is Mt to be transported (see table above), not GWh generated. Conversion uses the rate of coal required per GWh of electricity generated.

$$F_{BS-S,Tsp} = F_{BS-S,Tsp}.m_{BS,Gen,Required} = \frac{4.59 \times 10^5 \text{ kg/GWh}}{10^9 \text{ kg/Mt}} \times 1.02 \text{ GWh/GWh}$$

basis.

Step 4: Convert the mining (Min) basis into the system basis. Mining data is already in output basis, thus no conversion to output basis is necessary.



Appendix 2: Examples of Calculation Steps

 $= 4.56 \times 10^{-4}$ Mt to be transported per GWh delivered

This FBS-S, Tsp factor is used to convert all transport material flows into the system

$$\frac{1}{m_{\text{BS,Gen,From}}} \left(\frac{1}{m_{\text{BS,Gen,From}}} \right) \left(\frac{m_{\text{BS,Gen,Required}}}{10^9} \right) \left(\frac{m_{\text{BS,Trm,To}}}{m_{\text{BS,Trm,From}}} \right)$$

Appendix 2: Examples of Calculation Steps

$$= \left(\frac{1}{m_{BS,Min,From}}\right) F_{BS-S,Gen}$$

 $=4.56 \times 10^{-4}$ Mt to be transported per GWh delivered

This F_{BS-S,Min} factor is used to convert all generation material flows into the system basis (except for the NG mining subsystem (see Section A2.2.1.2)).

A2.2.1.2 Producing System Material Flow Data **Producing allocation factors:**

For fuel gas allocation of NG mining subsystem:

Section A1.9.5.3 (Appendix 1) contains the derivation of the relative fuel-gas consumption factors for natural gas and LNG (P_{NG} and P_{LNG}). Producing the allocation factors, shown in Section 4.3.5.2, for the NG mining stage from these factors requires two steps. Step 1 converts the P_{NG} and P_{LNG} values into percentages :

%
$$P_{NG} = \left(\frac{P_{NG}}{m_{Fuel Gas,Min}}\right) m_{NG,Min} = \left(\frac{0.0410}{2408}\right) 13201 = 22.5\%$$

where : $m_{FuelGas,Aust,Min} = P_{NG} \cdot m_{NG,Aust,Min} + P_{LNG} \cdot m_{LNG,Aust,Min} + P_{Liquids,Aust,Min}$

$$= 0.0410 x 13201 + 0.106 x 7397 + 0.0363 x 29834$$
$$= 2408$$

Step 2 uses this percentage to produce the allocation for the yearly averaged natural gas and petroleum productions $(15164 \text{ kt} \text{ and } 51143 \text{ kt})^{\dagger}$ to produce the allocation factor $(\mathbf{F}_{\mathbf{A}})$:

$$F_{A,Min,NG} = \left(\frac{\% P_{NG}}{m_{NG,Aust.Min}}\right) m_{Petroleum,Aust.Min} = \left(\frac{22.5\%}{15164}\right) 51143$$

 $= 0.758 \frac{\text{material flows per kg natural gas}}{\text{material flows per kg petroleum}}$

The same process applies for LNG systems.

* P_i from Section A1.9.5.3, Appendix 1 and m_{i.Aust.Min} are yearly averages from APPEA (2000).

Sustainability of Australia's Electricity Generation

This factor is then multiplied with the $F_{B-S,Min}$ factor produced in Section A2.2.1.1 to generate an allocated conversion factor, FB-S,Min,A.

For economic allocation of NG and BIC mining subsystems: NG mining: Relative economics of products from ABS (2001), relative production from APPEA (1999).

Product 👋	S/kg	_m;	S per kg	$\sim P_i$.
	product	(Mtpa)	petroleum	<u>(%)</u>
Crude Oil	0.160	21.8	0.0691	45.7
Condensate	0.0947	5.8	0.0101	7.21
Natural Gas	0.132	13.2	0.0346	22.8
LPG	0.224	2.22	0.00985	6.52
LNG	0.182	7.4	0.0267	17.7
Total		50.4	0.151	100

For example (NG):



allocation method.

Fuel	Description .	Value	, Units
	% of coal for domestic consumption	43	% by mass
BIC	Deletine Cent	75	% BIC value of BICE value
	Relative Cost	37.3	% value of BIC of 1 t of All Coal
	% of coal for export	57	% by mass
BICE		100	% BICE value of BICE value
	Relative Cost	62.7	% cost of BICE of 1 t of All Coal
All	Bul due Gent	90.2	% All Coal value of BICE value
Coal	Relative Cost	100	% cost of All Coal of 1 t of All Coal

266

Appendix 2: Examples of Calculation Steps

$$\int_{NG} \frac{m_{NG,Aust.Min}}{m_{Petroleum,Aust.Min}} = (0.132) \frac{13.2}{50.4} = 0.0346$$

Production of allocation factors from this P_{NG} factor is identical to Step 2 of the fuel gas

BIC mining: Relative economics of BIC to BICE from Simes (2000).

⁺ m_{NG, Aust,Min} and m_{Petroleum, Aust,Min} are yearly averages from APPEA (2000).

Appendix 2: Examples of Calculation Steps

(% cost of BIC compared to BICE)(BIC % by mass) % cost of BIC for 1 t of All Coal = -7(% cost of BIC compared to BICE)(BIC % by mass) + (% cost of BICE compared to BICE)(BICE % by mass)

The allocation factors ($F_{BIC,Min}$ and $F_{BICE,Min}$) are obtained as:

$$F_{A,BIC,Min} = \frac{\% \cot of BIC \text{ for 1 t of All Coal}}{BIC \% \text{ by mass}} = \frac{37.3}{43} = 0.859$$
$$F_{A,BIC,Min} = \frac{\% \cot of BIC \text{ for 1 t of All Coal}}{BIC \% \text{ by mass}} = \frac{62.7}{57} = 1.11$$

Producing system basis material flow data:

Example: Diesel consumption in BlC-ST system

Subsystem		Rates
Mining	1.5×10^6	kg per Mtpa mined
Transport	7.36 x 10 ⁵	kg per Mtpa to be transported
Generation	9.1	GWh per GWh generated
Transmission	0	GWh per GWh to be delivered

Thus, using the conversion factors (F_{BS-S,i}) from Section A2.2.1.1:

$$m_{\text{Diesel,S}} = m_{\text{Diesel,Min}} x F_{\text{BS-S,Min}} + m_{\text{Dieset,Tsp}} x F_{\text{BS-S,Tsp}} + m_{\text{Diesel,Gen}} x F_{\text{BS-S,Gen}} (+ m_{\text{Diesel,Trm}} x F_{\text{BS-S,Trm}}) = (1.5 \times 10^6 x 4.56 \times 10^{-4}) + (7.36 \times 10^5 x 4.56 \times 10^{-4}) + (9.1 \times 1.01) (+ 0 \times 1.01)$$

=1028 kg per GWh delivered

Similarly for electricity consumption in BIC-ST system:

Såbsystem	* * . 		Rates
	Consumed	Emitted	$\langle Units \rangle \sim \langle v \rangle$
Mining	19.7	0	GWh per Mtpa mined
Transport	0.508	0	kg per Mtpa to be transported
Generation	5.67	1.06	GWh per GWh generated
Transmission	1	0.986	GWh per GWh to be delivered
System	1.08	2.08	GWh per GWh delivered

If an allocation procedure was used for the subsystem, such as for the natural gas mining system, the allocated conversion factor (F_{BS-S,i,A}) replaces the conversion factor (F_{BS-S,i}) in the above procedure.

Removing electricity from the material flow data:

Using the technique established in Section A2.2.1.2, 'Producing system basis material

Sustainability of Australia's Electricity Generation

flow data', allows the generation of a table of diesel and electricity consumption and emission data in the system basis (kg per GWh delivered):

Consumption	Mining	Transport	Subsystem Generation	Transmission	System
Electricity	8.98 x 10 ³	2.3×10^4	5.75 x 10 ⁻²	1.01	1.07
Diesel	686	332	10.1	0	1028
Emission	Mining	Transport	Generation	Transmission	System
Electricity	0	0	1.07	1	2.07
Diesel	0	0	0	0	0

Step 1: Assume all of generation electricity consumption is internal (only startup power would be sourced from the grid). Remove transmission systems electricity carriage (1 MW enters and leaves). Thus, the system consumes 1028 kg of diesel and 2.28×10^{-2} GW of electricity to deliver 1 GWh of electricity.

Consumption			Subsystem		
	Mining.	Transport	Generation	Transmission	System
Electricity	8.98 x 10 ⁻³	2.3 x 10 ⁻⁴	0	1.39 x 10 ⁻²	2.28 x 10 ⁻²
Diesel	686	332	10.1	0	1028
Emission	Mining	Transport	Generation	Transmission	System
Electricity	0	0	0	1	1
Diesel	0	0	0	0	0

electricity. Continuing this substitution for all subsystems gives:

Consumption	Mining	Trànsport	Subsystem Generation	Transmission	System
Electricity	2.06×10^{-4}	5.31 x 10 ⁻⁶	0	3.13 x 10 ⁻⁴	5.24 x 10 ⁻⁴
Diesel	696	332	10.1	14.1	1052
Emission	Mining	Transport	Generation	Transmission	System
Electricity	0	0	0	1	1
Diesel	0	0	0	0	0

1011	•		
	1	or	t

Step 2: Replace electricity consumption with an equivalent amount of consumption and emissions. For the mining subsystem 8.98 x 10^3 GWh of electricity is required. 1 GWh of electricity consumes 1028 kg of diesel and 2.28 x 10^{-2} GW of electricity. Thus, 8.98 x 10⁻³ GWh of electricity consumes 9.23 kg of diesel and 2.06 x 10⁻⁴ GWh of

Appendix 2: Examples of Calculation Steps

Step 3: Repeat Step 2 by iteration or matrix inversion methods until the electricity consumptions are zero.

Consumption			Subsystem		
	Mining	Transport	Generation	Transmission	System
Electricity	0	0	0	Ō	0
Diesel	696	332	10.1	14.6	1053
Emission	Mining	Transport	Generation	Transmission	System
Electricity	0	0	0	1	1
Diesel	0	0	0	0	0

The process is identical for all other systems, and materials.

A2.2.1.3 Producing Environmental Indicators

Producing environmental indicators uses Equation 3.2.4. For example: World resource depletion (RD W) for BIC, ST subsystem. This table shows the converted material flows data for black coal and diesel.

Consumption (kg per GWh)	Mining	Transport	Subsystem	Transmission	System
Black Coal	4.61 x 10 ⁵	108	0	6490	4.67 x 10 ⁵
Diesel	696	332	10.1	14.6	1053

The characterisation factors, CF_{RD W,c}, for black coal (0.0134 kg_{Sb-cq} per kg black coal) and diesel (0.0201 kg_{Sb-eq} per kg diesel, assumes equivalent to oil) are from Guinée et al. (2001). Using Equation 3.2.4 gives:

 $ICS_{RD W} = RD W = m_{RD W, 2lack Coal} x CF_{RD W, Black Coal} + m_{Diesel} x CF_{RD W, Diesel}$

 $= 4.67 \times 10^{5} \times 0.0134 + 1053 \times 0.0201$

= 6280 kg_{Sb-ca} per GWh delivered

A2.2.2 Economic Flows

A2.2.2.1 Producing subsystem 1 and O&M costs

I and O&M costs are correlations (see Section A1.9.4.1, Appendix 1, for information on producing correlated costs) based on an estimated capacity (Q) for I or production (P) for O&M. For example: BIC generation subsystem, I costs. The capital cost function for this subsystem is Equation 4.3.20. The assumed capacity for the system is 1000

Sustainability of Australia's Electricity Generation

MW, which means the required capacity for the generation subsystem (Q_{BIC.Gen}) is 1014 MW (see Section A2.2.2.5). Using Equation 4.3.20: $I_{BIC.Gen} = 12.5(Q_{BIC.Gen})^{0.663} = 12.5(1014)^{0.663} = $1230 \text{ million} (AUD 2000)$

A2.2.2.2 Producing subsystem I and O&M costs using Equation 4.3.48 Some subsystem costs assume a larger capacity than that required, i.e. BIC mining, and NG mining and transport, as present systems also supply other markets (BIC for domestic and export use, NG for electricity and domestic and industrial heat and energy). Equation 4.3.48 allows this to be included. For example: BICE mining subsystem, I costs. The capital cost function for this subsystem is Equation 4.3.3. The assumed capacity for the system is 1000 MW, which means the required capacity for the mining subsystem (Q_{BIC,Min}) is 2.32 Mtpa (see Section A2.2.2.5). Using Equation 4.3.3:

$$I_{BIC,Min} = 130(Q_{BIC,Min})^{0.690} + 90.0$$

mine costs:

$$(I_{BIC,Min})_{DM} = 130(Q_{BIC,Min,DM})^{0.690} + 90.0(Q_{BIC,Min,DM})^{0.616} = 130(2.01)^{0.690} + 90.0(2.01)^{0.616}$$
$$= $349 \text{ million} (AUD 2000)$$

2.67 of these mines $((I_{BIC,Min})_{DMT})$ cost \$ 931 million (AUD 2000). Equation 4.3.48 estimates the proportion due to domestic consumption:

$$I_{BIC,Min} = (I_{BIC,Min})_{DMT} \frac{Q}{(Q)_{DMT}} = 931 \frac{2.32}{5.35} = \$404 \text{ million} (AUD 2000)$$

This value assumes, for simplicity of explanation, that each mine is of the same size. In the actual calculation, 2 mines are assumed to provide 62 % of coal by conveyor, 0.3 mines provide 16 % by road, and 0.37 mines provide 22 % by rail. However, the same method applies. For BICE, 0.3 mines provide 11 % by road, while 2.37 mines provide 89 % by rail.

Appendix 2:Examples of Calculation Steps

A survey of power stations and mines shows that each power station has an average of 2.67 distinct mines. Of these mines, 43.4 % of coal was for domestic consumption. Assuming each of these mines is of equal size, then the total amount of coal produced for 2.32 Mtpa of domestic coal is 2.32/0.434 = 5.35 Mtpa. 2.67 distinct mines (DM) produce this 5.35 Mtpa, and thus each mine produces 5.35/2.67 = 2.01 Mtpa. Thus each

A2.2.2.3 Producing a subsystem VA indicator

Producing subsystem VA indicators requires sales (S), upstream purchases (F), labour costs (L) and operating and maintenance (O&M) values. S, F and L values are in Section 4.3.5.3.1, while O&M costs come from various correlations in Section 4.3 (see 'Producing a subsystem I/O&M cost' for information on producing O&M values from these correlations). Equation 3.3.3 gives VA as:

$$VA = \sum_{l=1}^{L} \sum_{s=1}^{S} \{S_{s,l} - [(O \& M_{s,l} + F_{s,l}) - L_{s,l}]\}$$

For a single subsystem, s, and a single year, l, then Equation 3.3.3 becomes: $VA_{r} = S_{r} - [(O \& M_{r} + F_{r}) - L_{r}]$

For example: BIC generation subsystem. In this example, $S_{EIC,Gen} =$ \$ 39.9 (AUD 2000) per MWh, $F_{BIC,Gen}$ (previous subsystem's, S or $S_{BIC,Ten}$) = \$ 40.6 (AUD 2000) per t. $L_{BIC,Gen} = 42.1 \%$ of O&M, and O&M_{BIC,Gen} = \$ 9.41 (AUD 2000) per GWh.

Step 1: F_{BIC,Gen} is converted to the current basis by multiplying with the amount of BIC consumed to generate 1 MWh of electricity from the generation stage. This requires multiplication by the transport subsystem conversion factor ($F_{BS-S,Tsp} = 4.56 \text{ x}$ 10^{-4}) and division by the generation factor (F_{BS-S,Tsp} = 1.01) (see 'Producing system I, NPC, VA and CVA indicators'). Thus, $F_{BIC,Gen} =$ \$ 18.3 (AUD 2000) per MWh.

Step 2:

$$VA_{BIC,Gen} = S_{BIC,Gen} - [(O \& M_{BIC,Gen} + F_{BIC,Gen}) - L_{BIC,Gen}]$$

= $S_{BIC,Gen} - [(O \& M_{BIC,Gen} + F_{BIC,Gen}) - L_{BIC,Gen} (O \& M_{BIC,Gen})]$
= $39.9 - [(9.41 + 18.3) - 0.421(9.41)]$
= \$16.5 (AUD 2000) per MWh

A2.2.2.4 Producing subsystem C_R, NPC and CVA indicators

Producing subsystem C_{R} indicators requires capital costs (I), cost of capital (i) and life of loan (n) values. i and n values are in Section 4.3.6.3.1, while I values come from various correlations in Section 4.3 (see 'Producing a subsystem I/O&M cost' for information on producing I values from these correlations). Equation 3.3.8 gives C_R as:

$$C_{\rm R} = \frac{1.i}{\left[1 - \frac{1}{\left(1 + i\right)^{\rm n}}\right]}$$

'Producing a subsystem I/O&M cost'). Equation 3.3.12 gives NPC as:

$$NPC = O \& M + C_R$$

subsystem VA indicator'). Equation 3.3.14 gives CVA as:

$$CVA = VA - C_R$$

For example: BIC generation subsystem. In this example, I =\$ 1230 million (AUD) 2000) (see 'Producing a subsystem I/O&M cost'), O&M = \$ 9.41 million (AUD 2000), VA = 16.5 (AUD 2000) per MWh (see 'Producing a subsystem VA indicator'), i = 6.2% and n = 30 years.

$$C_{\rm R} = \frac{\text{L}i}{\left[1 - \frac{1}{\left(1 + i\right)^{\rm n}}\right]} = \frac{(1230)(0.062)}{\left[1 - \frac{1}{\left(1 + 0.062\right)^{30}}\right]} = \$91.3 \text{ million (AUD 2000) per annum}$$

Dividing by the system yearly production at 1014 MW ($P_{BIC} = 5089$ GWhpa) gives $C_{R} =$ \$ 17.9 (AUD 2000) per MWh. Thus: NPC = $O \& M + C_p = 9.41 + 17.9 = $27.3 (AUD 2000) per MWh$ $CVA = VA - C_p = 16.5 - 17.9 = \$ - 1.44 (AUD 2000) \text{ per MWh}$

A2.2.2.5 Producing system I, NPC, VA and CVA indicators The method is similar to that of material flows (see Section A2.2.1.2, 'Producing system basis meterial flow data'). However, unlike the material flows, the assumed capacity and production undergo conversion, not the subsystem indicators produced. This is because for economic data, the assumed capacity or production is the independent parameter in their correlations. The system indicators are thus the sum of the subsystem value produced using the converted capacities and productions. For example: BIC-ST

Producing subsystem NPC indicators requires annualised capital repayment (C_R) and operating costs (O&M) values. For C_R values see above and for O&M values (see

Producing subsystem CVA indicators requires annualised capital repayment (C_R) and value added (VA) values. For C_R values see above and for VA values (see 'Producing a
Appendix 2: Examples of Calculation Steps

system capacities and productions. Mass flows BIC conversion factors (see Section A2.2.1.1):

Subsystem	Conversion Factors
Mining	4.56 x 10 ⁻⁴
Transport	4.56 x 10 ⁻⁴
Generation	1.01
Transmission	1.01

Assumed system capacity (Q_{BIC.S}) is 1000 MW. Assuming an average production time (load factor) of 58.1 % (for Australian BIC-ST systems), then:

 $P_{BIC,S} = Q_{BIC,S}.LF_{BIC}.t(hours per year) = \frac{1000}{1000}(0.581)(24 \times 365) = 5089 \,GWhpa$

The capacity (O) or production (P) are multiplied by the conversions:

For example:

 $Q_{BIC,Min} = Q_{BIC,S} \cdot F_{A,BIC,Min} = 1000 (4.56 \times 10^{-1}) = 2.32 \text{ Mtpa}$

Subsystem	🧈 🗘 🔪	D
Mining	2.32 Mtpa	2.32 Mtpa
Transport	2.32 Mtpa	2.32 Mtpa
Generation	1014 MW	5159 GWhpa
Transmission	1014 MW	5159 GWhpa
System	1000 MW	5089 GWhpa

A2.2.3 Social Flows

A2.2.3.1 Producing a EI from a ED

Indirect employment (EI) production uses direct employment (ED) and multipliers (EM). Equation 3.4.2 gives EI as:

ST CT

$$EI = \sum_{s=1}^{\infty} \sum_{c=1}^{\infty} EM_c \ x \ ED_{s,c}$$

For a single subsystem, s, then:

$$EI_s = \sum_{c=t}^{C_T} EM_c \ x \ ED_{s,c}$$

Most subsystems have only one type of industry classification, except transport,

Sustainability of Australia's Electricity Generation

where road, rail and pipeline, and ocean transport have different classifications. For example: BIC, ST generation. In this subsystem, ED is 2.71×10^{-5} employees per MWhpa, and its industry classification is electricity. The EM for electricity is 4.275 indirect employees per direct employee. Thus:

$$EI_{BIC,Gen} = \sum_{c=1}^{1} EM_c \ x \ ED_{BIC,Gen,c} =$$

= 1.16×10^{-4} indirect employees per MWh

A2.2.3.2 Producing system ED, EI, LTI and Fatal indicators system basis material flow data').

A2.3 NORMALISATION

$$ICS_{RD W}^{Australia,N} = \frac{ICS_{RD W}}{ICS_{RD W}^{Australia}} = \frac{6.2}{4.80}$$

A2.4 NG RESOURCE LIFE UNDER ALL CCGT-NG SCENARIO Electricity production information for Australia is available from ESAA (1999) in Table 2.4 and Chart 2.5. The table below shows the generation information (GWhpa) obtained from these charts. The NG figure for ST assumes that the contribution of other petroleum fuels to OCGT and CCGT generation is minimal.

	ST	OCGT	CCGT.	Total
NG	7870	1540	931	1.03 x 10
BIC	9.91 x 10 ⁴			9.91 x 10
BrC	5.40×10^4			5.40 x 10
Hydro				1.52 x 10
All	1.61 x 10 ⁵	1540	931	1.63 x 10

This NG electricity generation consumes 98.782 PJpa of natural gas (ESAA (1999),

 $= 4.275 x 2.71 x 10^{-5}$

The method is the same as that of material flows (see Section A2.2.1.2, 'Producing

Normalisation is the division of an indicator score (ICS_a) by a regional indicator score (ICS_{*}^{R}) (Equation 3.2.5). For example: BIC, ST system, RD W score. ICS_{RD} w for the BIC, ST system 6.29 kg Sb.ca. per MWh of electricity delivered (see Table 5.2.1). $ICS_{RD W}^{Australia}$ for the Australian region is 4.80 x 10⁹ kg Sb_{.eq} per annum. Thus:

 $\frac{6.29}{80 \times 10^9} = 1.31 \times 10^{-9} \frac{\text{kg Sb.}_{eq.} \text{ per MWh delivered}}{\text{kg Sb.}_{eq.} \text{ per MWh delivered}}$

Table 2.5). Australian natural gas consumption in 2000 was 1218.7 PJpa (AGA (2001)). Thus, non-electricity requirements are 1120 PJpa.

To provide all, non-hydro electricity, CCGT-NG plants must provide 1.48 x 10⁵ GWhpa. The natural gas requirements for each GWh of electricity are 1.59×10^5 kg (Table 4.3.41, Chapter 4). Thus, the CCGT-NG systems would require 2.35 x 10^{10} kg of NG. Using a natural gas energy density of 39 MJ per m³ (IOR Energy Pty Ltd (2000)) and mass density of 0.762 (NPI (1999)) then, 2.35 x 10^{10} kg of NG is 1205 PJpa. Thus, the total NG requirements would be the non-electricity requirements (1120) PJpa) plus the CCGT-NG requirements (1205 PJpa), or 2325 PJpa. The total reserve in 2000 was 127928 PJ (AGA (2001)). Thus, the NG reserve life under this condition is 55 years (127928/2325 years).

A2.5 SUMMARY OF EXAMPLES OF SUSTAINABILITY INDICATOR CALCULATION STEPS These examples allow the duplication of the results shown in Chapter 5.

Sustainability of Australia's Electricity Generation Appendix 3: Contributions and Normalised Indicators

APPENDIX 3: CONTRIBUTIONS AND NORMALISED SUSTAINABILITY INDICATORS

A3.1 OVERVIEW OF CONTRIBUTIONS AND NORMALISED SUSTAINABILITY INDICATORS Section A3.2 contains the actual contributions of subsystems to the total indicators, shown in Chapter 5 as Figures 5.2.1, 5.3.1 and 5.4.1. Section A3.3 contains the normalised sustainability indicators, shown in Chapter 5 as Figure 5.5.1.

A3.2 SUBSYSTEM INDICATOR CONTRIBUTION TABLES

energy consumption and exergy destruction (GJ)).

-	a a	Sub.		4. <u>1</u> . 6. 57		Indicator			
June (1 00		CC	AD	EU	PS	НЛ	ET	PM
		Min	45.4	0.0221	7.06×10^{-3}	0.015	552	0.0820	6.20×10^{-3}
		т.	43.4	0.0001	7.90 X 10	0.015	55.5	0.0629	0.20 X 10
	H	1sp.	1.5	0.0127	2.05×10^{-1}	1.31×10^{-5}	5.12	0.0613	0.0511
	S	Gen.	999	6.5	0.421	0.294	5130	4.69	0.483
U.		Tm.	14.7	0.0928	6.07 x 10 ⁻³	4.36 x 10 ⁻³	73.1	0.068	7.61 x 10 ⁻³
B		Min.	26	0.0212	3.97 x 10 ⁻³	0.0107	6.27	0.0386	1.49 x 10 ⁻³
	Ю	Tsp.	0.971	6.29 x 10 ⁻³	1.10 x 10 ⁻³	9.59 x 10 ⁻⁴	2.2	0.0339	0,0287
	õ	Gen.	647	0.141	0.0206	7.40 x 10 ⁻³	65.1	0.212	7.28 x 10 ⁻³
		Trn.	9.48	2.37 x 10 ⁻³	3.62 x 10 ⁻⁴	2.68 x 10 ⁻⁴	1.04	4.01 x 10 ⁻³	5.28 x 10 ⁻⁴
		Min.	67.5	0.131	0.012	0.0226	88.8	0.136	0.0157
	ĥ	Tsp.	29.3	0.612	0.019	0.0293	528	1.02	0.535
	Ś	Gen.	999	6.5	0.421	0.294	5130	4.69	0.483
B		Trn.	15.4	0.102	6.37 x 10 ⁻³	4.86 x 10 ⁻³	80.9	0.0822	0.0145
BIC		Min.	37.3	0.0307	5.35 x 10 ⁻³	0.0141	10.8	0.0537	3.98 x 10 ⁻³
	2	Tsp.	15.5	0.397	0.0321	0.0235	289	0.564	0.299
	١ <u>ठ</u>	Gen.	647	0.141	0.0206	7.40 x 10 ⁻³	65.1	0.212	7.28 x 10 ⁻³
		Tm.	9.84	8.01 x 10 ⁻³	8.17 x 10 ⁻⁴	6.34 x 10 ⁻⁴	5.13	0.0117	4.37 x 10 ⁻³
Ų	L	Min.	7.66	0.0221	1.88 x 10 ⁻³	1.26 x 10 ⁻³	15.7	0.0213	3.16 x 10 ⁻³
Ē	Ś	Tsp.	1.11	2.86 x 10 ⁻³	2.00 x 10 ⁻⁴	1.36 x 10 ⁻⁴	2.21	2.46 x 10 ⁻³	5.22 x 10 ⁻⁴

Table of environmental indicator subsystem contributions (Units (all per MWh of electricity delivered): $CC = climate change (kg CO_{2-eq}); AD = acidification (kg SO_2-$ _{ca}); EU = eutrophication (kg PO₄³⁻, _{ca}); PS = photochemical smog (kg C₂H_{4-e}); HT and ET = human and eco-toxicity (kg 1,4-dichlorobenzene); PM and SW = particulates and solid waste emissions (kg); WD = water consumption (t); RD A and RD W = Australian and world resource depletion (kg Sb.c); and EN and EX =

277

Sustainability of Australia's Electricity Generation ____ Appendix 3:Contributions and Normalised Indicators

		Sub.				Indicator			
Fue	Teel		CC	AD	, EU	PS	TH	ET	₽Й
	H	Gen.	1180	3.04	0.211	0.143	2340	2.54	0.479
	N	Trn.	26.5	0.0683	4.76 x 10 ^{.3}	3.23 x 10 ⁻³	52.6	0.0572	0.0108
ပု		Min.	4.89	0.0166	1.95 x 10 ⁻³	4.25 x 10 ⁻³	9.59	0.0195	7.48 x 10 ⁻⁴
B	B	Tsp.	0.649	1.16 x 10 ⁻³	1.35 x 10 ⁻⁵	3.14 x 10 ⁻⁴	1.16	3.16 x 10 ⁻⁴	1.18 x 10 ⁻⁴
	ğ	Gen.	728	2.34	0.0219	0.0977	2370	0.411	0.0204
	Ì	Trn.	16.4	0.0525	5.31 x 10 ⁻⁴	2.28 x 10 ⁻³	53.1	9.60 x 10 ⁻³	4.75 x 10 ⁻⁴
		Min.	67.4	0.21	0.0447	0.106	40.2	0.433	0.0166
	ST	Tsp.	8.31	4.21 x 10 ⁻⁴	8.35 x 10 ⁻⁵	7.49 x 10 ⁻³	0.107	2.43 x 10 ⁻³	2.00 x 10 ⁻³
		Gen.	647	0.733	0.191	0.0445	1.82	1.77	0.0164
		Trn.	0.816	1.06 x 10 ⁻³	2.66 x 10 ⁻⁴	1.78 x 10 ⁻⁴	0.0476	2.49 x 10 ⁻³	3.95 x 10 ⁻⁵
		Min.	90.1	0.28	0.0597	0.141	53.7	0.578	0.0222
0	5	Tsp.	11.1	6.56 x 10 ⁻⁴	1.35 x 10 ⁻⁴	0.01	0.147	3.51 x 10 ⁻³	2.72 x 10 ⁻³
z	8	Gen.	745	1.19	0.308	0.0766	6.93	3.06	0.26
		Trn.	0.955	1.66 x 10 ⁻³	4.15 x 10 ⁻⁴	2.57 x 10 ⁻⁴	0.0686	4.11 x 10 ⁻³	3.22 x 10 ⁻⁴
		Min.	47.2	0.147	0.0313	0.0739	28.2	0.303	0.0116
	5	Tsp.	5.8	2.46 x 10 ⁻⁴	4.60 x 10 ⁻⁵	5.24 x 10 ⁻³	0.0745	1.59 x 10 ⁻³	1.40 x 10 ⁻³
	2	Gen.	405	0.267	0.0683	0.0189	5.28	0.659	0.0242
L		Tm.	0.517	4.68 x 10 ⁻⁴	1.12×10^{-4}	1.11 x 10 ⁻⁴	0.0379	1.09 x 10 ⁻³	4.20 x 10 ⁻⁵
		Min.	198	0.616	0.131	0.31	118	1.27	0.0487
	F	Tsp.	32.5	0.0945	0.0105	0.0146	56.9	0.108	6.26 x 10 ⁻³
1	S	Gen.	647	0.733	0.191	0.0445	1.82	1.77	0.0164
		Tm.	0.99	1.63×10^{-3}	3.75×10^{-4}	4.16 x 10 ⁻⁴	0.2	3.56×10^{-3}	8.05 x 10 ⁻⁵
		Min.	265	0.823	0.175	0.414	158	1.7	0.0651
9	5	Tsp.	44	0.129	0.0146	0.02	76.2	0.15	9.22 x 10 ⁻³
1	8	Gen.	745	1.19	0.308	0.0766	6.93	3.06	0.26
		Tm.	1.19	2.42×10^{-3}	5.62 x 10 ⁻⁴	5.76 x 10 ⁻⁴	0.272	5.54 x 10 ⁻³	3.77 x 10 ⁻⁴
		Min.	139	0.432	0.092	0.217	82.7	0.891	0.0341
	5	Tsp.	22.2	0.065	7.10×10^{-3}	0.01	39.8	0.0731	4.37 x 10 ⁻³
		Gen.	405	0.267	0.0683	0.0189	5.28	0.659	0.0242
		Tm.	0.638	8.62 x 10 ⁻⁴	1.89 x 10 ⁻⁴	2.78 x 10 ⁻⁴	0.144	1.83 x 10 ⁻³	7.07 x 10 ⁻⁵

Table of environmental indicator subsystem contributions (continued).

		Sub.			Indie	ator		
	21		SW	WD	RD A	.RD W	EN	EX
		Min.	0.784	0.33	0.405	6.19	11.1	13.3
ပ	E-	Tsp.	0.0707	5.49 x 10 ⁻⁴	0.181	8.13 x 10 ⁻³	0.0177	0.0183
<u>ه</u>	s.	Gen.	85.3	2	0.178	6.58 x 10 ⁻³	0.0141	0.121
		Trn.	1.21	0.0328	0.0107	0.0873	0.157	0.19

Fuel	Tech	Sub.	SW	WD	Indica RD A	itor RD W	EN	EX
		Min.	0.318	0.18	0.421	3.66	6.45	7.88
ပ	B	Tsp.	0.0368	1.75 x 10 ⁻⁴	0.102	4.25 x 10 ⁻³	9.40 x 10 ⁻³	9.63 x 10 ⁻³
۳	ğ	Gen.	61.8	1.1	0.0122	4.52 x 10 ⁻⁴	1.03 x 10 ⁻³	0.249
		Trn.	0.874	0.018	7.53 x 10 ⁻³	0.0515	0.0909	0.115
		Min.	4.1	0.49	0,662	9.15	16.4	19.7
	F	Tsp.	216	0.121	4.81	0.2	0.424	0.452
	<u>م</u>	Gen.	85.3	2	0.178	6.58 x 10 ⁻³	0.0141	0.121
円		Trn.	4.3	0.0368	0.0795	0.132	0.237	0.286
ΞĮ		Min.	1.39	0.266	0.547	5.31	9.4	11.6
	Ю	Tsp.	121	0.0661	2.54	0.101	0.216	0.263
	<u>1</u> 0	Gen.	61.8	1.1	0.0122	4.52 x 10 ⁻⁴	1.03 x 10 ⁻³	0.337
		Tm.	2.6	0.0202	0.0436	0.0761	0.135	0.171
		Min.	0.132	5.89 x 10 ⁻⁴	0.0766	9.6	13.6	12
	Ľ	Tsp.	0.027	2.17 x 10 ⁻³	1.69 x 10 ⁻⁴	8.99 x 10 ⁻³	0.0127	0.0113
	S	Gen.	21.2	2.62	0.104	3.89 x 10 ⁻³	8.30 x 10 ⁻³	0.0587
ပါ		Trn.	0.476	0.0517	4.03 x 10 ⁻³	0.214	0.304	0.27
ā		Min.	0.0434	7.37 x 10 ⁻⁴	0.235	5.15	7.25	6.62
	g	Tsp.	0.0102	2.43 x 10 ⁻⁴	3.29 x 10 ⁻⁴	2.52 x 10 ⁻³	3.56 x 10 ⁻³	3.28 x 10 ⁻³
	00	Gen.	13.3	0.646	3.39 x 10 ⁻³	1.25 x 10 ⁻⁴	2.89 x 10 ⁻⁴	0.0781
		Trn.	0.298	0.0105	5.33 x 10 ⁻³	0.115	0.162	0.149
		Min.	0.0211	5.06 x 10 ⁻⁴	5.53	5.57	6.89	11.8
	<u>د _</u>	Tsp.	2.79 x 10-6	0.0184	6.73 x 10 ⁻³	9.59 x 10 ⁻⁴	1.42 x 10 ⁻³	2.08 x 10 ⁻³
	S	Gen.	2.25 x 10 ⁻⁴	139	5.85 x 10 ⁻³	2.69 x 10 ⁻⁴	5.81 x 10 ⁻⁴	0.037
		Tm.	2.38 x 10 ⁻⁵	0.157	6.25 x 10 ⁻³	6.29 x 10 ⁻³	7.78 x 10 ⁻³	0.0134
		Min.	0.0282	6.76 x 10 ⁻⁴	7.38	7.44	9.21	15.8
r٦.	E	Tsp.	4.99 x 10-6	1.66 x 10-7	9.36 x 10 ⁻³	1.61 x 10 ⁻³	2.31 x 10 ⁻³	3.48 x 10 ⁻³
ž	Х	Gen.	0	2.63 x 10 ⁻⁴	0.184	6.82 x 10 ⁻³	0.0146	0.0156
		Tm.	3.18 x 10 ⁻⁵	1.06 x 10-6	8.55 x 10 ⁻³	8.41 x 10 ⁻³	0.0104	0.0179
		Min.	0.0148	3.55 x 10 ⁻⁴	3.87	3.9	4.83	8.3
	5	Tsp.	1.37 x 10-6	2.33 x 10 ⁻³	4.57 x 10 ⁻³	5.18 x 10 ⁻⁴	8.03 x 10 ⁻⁴	1.13 x 10 ⁻³
	١ <u>ठ</u>	Gen.	1.74 x 10 ⁻⁵	25.20	0.0958	3.55 x 10 ⁻³	7.59 x 10 ⁻³	0.0717
	Ŭ	Tm.	1.67 x 10 ⁻⁵	0.0284	4.48 x 10 ⁻³	4.41 x 10 ⁻³	5.46 x 10 ⁻³	9.45 x 10 ⁻³
		Min.	0.0619	1.49 x 10 ⁻³	6.93	6.28	7.8	13.4
	,	Tsp.	8.86 x 10 ⁻³	4.17	4.11	0.167	0.342	0.376
0	ST	Gen.	2.25 x 10 ⁻⁴	139	5.85 x 10 ⁻³	2.69 x 10 ⁻⁴	5.81 x 10 ⁻⁴	0.037
Ē		Trn.	7.99 x 10 ⁻⁵	0.162	0.0125	7.27 x 10 ⁻³	9.19 x 10 ⁻³	0.0155
		Min.	0.0828	1.99 x 10 ⁻³	9.27	8.39	10.4	17.8
	5	Tsp.	0.0119	5.12	5.51	0.23	0.466	0.517

278

Sustainability of Australia's Electricity Generation Appendix 3:Contributions and Normalised Indicators

Sustainability of Australia's Electricity Generation Appendix 3:Contributions and Normalised Indicators

uel	ech	Sub.	SW	wo	Indica	tor A P	0 W F	N FN
ł	Ų	Gen.	0	2.63 x 10 ⁻⁴	0.184	6.82 x 10 ⁻³	0.0146	0.0156
	0	Trn.	1.07 x 10 ⁻⁴	5.78 x 10 ⁻³	0.0169	9.74 x 10 ⁻³	0.0123	0.0208
g		Min.	0.0434	1.04 x 10 ⁻³	4.86	4.4	5.47	9.36
FN	5	Tsp.	6.17 x 10 ⁻³	2.72	2.88	0.113	0.235	0.256
	Р <u></u>	Gen.	1.74 x 10 ⁻⁵	25.2	0.0958	3.55 x 10 ⁻³	7.59 x 10 ⁻³	0.0717
		Trn.	5.60 x 10 ⁻⁵	0.0315	8.84 x 10 ⁻³	5.10 x 10 ⁻³	6.45 x 10 ⁻³	0.0109

Table of economic indicator subsystem contributions (I = capital cost (\$ billion (AUD 2000)), AC, VA and CVA = annualised cost, value added and capital inclusive value added (\$ (AUD 2000) per MWh of electricity delivered)).

Ē	ch.	58		. Indi	cator	
<u> </u>	Ţ.	1	I.	AÇ	VA -	CVA .
		Min.	0.664	24.8	5.71	-3.97
<u>ں</u>	Ē	Tsp.	0.260	11.9	-0.327	-4.12
B	S	Gen.	1.23	27.3	16.5	-1.44
		Tm.	0.137	2.82	20.8	18.8
		Min.	0.571	14.4	3.18	-2.68
U U	2	Tsp.	0.217	7.18	-0.522	-2.75
BI	ğ	Gen.	1.03	16.7	26.9	16.4
		Tm.	0.137	2.1	20.9	19.5
	ST	Min.	1.16	31.5	11.3	-0.634
		Tsp.	0.642	25.0	0.842	-5.75
		Gen.	1.23	20.7	47.0	34.3
Щ		Tm.	0.137	2.1	14.8	13.4
BI	2	Min.	0.794	19.3	6.25	-1.9
		Tsp.	0.417	17.3	-1.15	-5.44
	ğ	Gen.	1.60	22.5	59.8	43.4
		Trn.	0.137	2.1	14.8	13.4
		Min.	0.678	14.5	4.99	-1.97
		Tsp.	0.122	2.09	0	-1.25
	Ś	Gen.	2.17	26.9	31.2	8.95
ပ္	_	Tm.	0.214	2.88	21.1	18.9
B		Min.	0.476	9.99	3.37	-1.51
	Ŋ	Tsp.	0.0878	1.49	0	-0.902
	ğ	Gen.	1.04	18.0	30.5	19.8
		Tm.	0.214	2.88	21. i	18.9
0	E	Min.	0.580	28.6	7.8	1.89
Z	<u>م</u>	Tsp.	0.234	3.21	8.47	6.06

Sustainability of Australia's Electricity Generation ____ Appendix 3:Contributions and Normalised Indicators

ંગ	, i	SS	(e)	Indi	cator	
. E	Fe		1	AC	VA .	<u> </u>
	- F =4	Gen.	1.04	13.6	3.87	-6.81
	s,	Trn.	0.0113	0.273	21.1	21.0
		Min.	0.737	38.8	10.3	2.75
	G	Tsp.	0.294	4.00	11.6	8.61
0) Ö	Gen.	0.500	7.31	-7.16	-12.3
Ž	•	Trn.	0.0113	0.273	21.1	21.0
		Min.	0.437	20.0	5.66	1.17
	CCGT	Tsp.	0.179	2.49	5.83	3.99
		Gen.	0.680	11.9	13.2	6.19
j		Tm.	0.0113	0.273	21.1	21.0
		Min.	0.647	10.2	1.04	-5.61
	ST	Tsp.	2.04	52.0	39.6	18.6
		Gen.	1.04	13.6	4.29	-6.39
		Tm.	0.0113	0.273	15.1	14.9
		Min.	0.817	13.6	1.51	-6.89
<u>o</u>	5	Tsp.	2.67	63.0	59.4	32.0
E	Ň.	Gen.	0.500	7.31	-20	-25.1
		Trn.	0.0113	0.273	15.1	14.9
		Min.	0.492	7.35	0.666	-4.39
	GT	Tsp.	1.52	42.0	23.5	7.90
	- Š	Gen.	0.680	11.9	24.7	17.8
		Tm.	0.0113	0.273	15.1	14.9

Table of social indicator subsystem contributions(ED and EI = direct and indirect employment (10^6 x employees), LTI = lost time injuries (10^7 x LTI), and Fatal = fatalities (10⁹ x fatalities), all per MWh of electricity delivered).

cl	äh.	SS .		Indicator				
Ē	Tei		ED *	EI	LTI	Fatal		
		Min.	38.8	289	44.2	7.02		
	ц	Tsp.	12.4	35.8	4.77	2.42		
(ઝ	Gen.	27.1	116	2.72	0		
0		Trn.	1.86	7.97	0.128	0.72		
B		Min.	21.8	163	24.8	3.94		
	8	Tsp.	6.98	20.1	2.68	1.36		
	ğ	Gen.	27.1	116	2.72	0		
ł		Tm.	1.31	5.61	0.0901	0.507		
E E		Min.	57.1	426	65	10.3		
BK	Ň	Tsp.	47.9	188	17.5	3.78		

280

281

Sustainability of Australia's Electricity Generation ____ Appendix 3: Contributions and Normalised Indicators

el		SS			Indicator	
Ě.	Ic		ED	El	LII	Fatal
	T	Gen.	27.1	116	2.72	0
	S	Trn.	1.31	5.61	0.0901	0.507
U E E E		Min.	32.1	239	36.5	5.81
Ē	IGCC B	Tsp.	26.9	105	9.84	2.12
		Gen.	27.1	116	2.72	0
		Tm.	1.31	5.61	0.0901	0.507
		Min.	16.8	125	0	0
		Tsp.	0	0	0	0
	N N	Gen.	18.2	77.8	3.87	0
U V		Trn.	2.03	8.68	0.139	0.785
Ē		Min.	8.67	64.7	0	0
	8	Tsp.	0	0	0	0
	ğ	Gen.	18.2	77,8	2.97	0
		Trn.	2.03	8.68	0.139	0.785
		Min.	38	283	9.38	2.45
]		Tsp.	6.12	30,6	0.38	1.44
	N N	Gen.	7.89	33.7	0.0108	0
		Trn.	0.109	0.467	7.50 x 10 ⁻³	0.0422
Ì		Min.	51.7	386	12.8	3.33
U	GT	Tsp.	6.12	41.6	0.38	1.44
Z	8	Gen.	3.81	16.3	5.22 x 10 ⁻³	0
		Tm.	0.109	0.467	7.50 x 10 ⁻³	0.0422
		Min.	26.4	197	6.53	1.7
	G I	Tsp.	6.12	21.3	0.38	1.44
	S	Gen.	9.56	40.9	0.0131	0
		T m .	0.109	0.467	7.50×10^{-3}	0.0422
		Min.	34.1	254	8.42	2.2
	Ē	Tsp.	6.15	22.5	0.649	0.629
	s	Gen.	7.89	33.7	0.0108	0
		Tm.	0.109	0.467	7.50×10^{-3}	0.0422
	_	Min.	46.4	346	11.5	2.99
Ŭ	5	Tsp.	8.34	30.7	0.881	0.847
ב	8	Gen.	3.81	16.3	5.22 x 10 ⁻³	0
		Trn.	0.109	0.467	7.50×10^{-3}	0.0422
		Min.	23.7	177	5.86	1.53
	61	Tsp.	4.31	15.7	0.454	0.445
	8	Gen.	9.56	40.9	0.0131	0
		Tm.	0.109	0.467	7.50 x 10 ⁻³	0.0422

A3.3 NORMALISED INDICATORS

Fuel	Tech				Indica	ntor				
		CC	AD	FU	PS 👘	SW .	RD A	RÐ Ŵ	EN	FΧ
Unito	f Impact	kg CO ₂₋₅₄	kg SO _{2 eq}	kg PO ₄	$\log C_2 H_{\rm trained}$	kg	🔄 kg Sh	-64	. (Ð
Magn	itude				10	,			42.	
DIO	ST	2.22	1.93	1.27	0.318	4.12	0.0192	1.31	2.34	1.56
RIC	IGCC	1.43	0.0493	0.0758	0.0195	2.97	0.0134	0.773	1.36	0.941
	ST	2.33	2.12	1.33	0.355	14.6	0.142	1.98	3.54	2.35
BICE	IGCC	1.49	0.167	0.171	0.0463	8.82	0.0778	1.14	2.02	1.41
	ST	2.54	0.904	0.635	0.150	1.03	4.58 x 10 ⁻³	2.05	2.89	1.41
BrC	IGCC	1.57	0.694	0.0709	0.106	0.644	6.05 x 10 ⁻³	1.10	1.54	0.78
	ST	1,52	0.272	0.686	0.160	1.01 x 10 ⁻³	0.137	1.16	1.43	1.30
NG	OCGT	1,77	0.424	1.07	0.231	1.33 x 10 ⁻³	0.188	1.55	1.92	1.81
	ссбт	0.959	0.120	0.290	0.0994	6.98 x 10 ⁻⁴	0.0986	0.814	1.00	0.955
LNG	ST	1.84	0.417	0.968	0.374	3.35 x 10 ⁻³	0.274	1.34	1.69	1.57
	OCGT	2.21	0.618	1.45	0.517	4.47 x 10 ⁻³	0.371	1.80	2.26	2.10
	СССТ	1.19	0.221	0,488	0.249	2.34 x 10 ⁻³	0,194	0,941	1.19	1.10

Table of normalised economic and social indicators. Units all (unit of impact per MWh of electricity delivered) per (unit of impact in Australia per annum)).

Ruel	Fuel Technology Indicator								
8 N.2		Ī	ÂC	NA .	CVA	ED	EI	LII	Fatal
Unito	Impact			S		Emplo	yees .	LTI	fatalities
Magni	tude	10-8	10-11	1	0 ¹⁰			10 "	
DIC	ST	10.0	6.26	2.52	0.745	0.899	5.03	3.35	2.73
ыс	IGCC	6.03	3.78	2.98	2.44	0.641	3.41	1.96	1.56
	ST	9.78	7.42	4.36	3.32	1.50	8.24	5.52	3.93
BICE	IGCC	9.10	5.73	4.70	3.97	0.979	5.23	3.18	2.27
D -C	ST	9.81	4.34	3.38	1.98	0.415	2.37	0.259	0.211
BIC	IGCC	5.59	3.02	3.24	2.91	0.324	1.69	0.201	0.211
	ST	5.75	4.27	2.44	1.78	0.584	3.9	0.633	1.05
NG	OCGT	4.75	4.72	2.12	1.61	0.692	4.98	0.852	1.29
	CCGT	4.03	3.24	2.7	2.60	0.474	2.91	0.449	0.855
	ST	11.5	7.12	3.54	1.73	0.541	3.49	0.588	0.77
LNG	OCGT	12.3	7.87	3.30	1.20	0.658	4.41	0.800	1.04
•	CCGT	8.33	5.76	3.78	2.91	0.423	2.63	0.41	0.542

Table of normalised environmental indicators. Units all (unit of impact per MWh of electricity delivered) per (unit of impact in Australia per annum)).

283

Nomenclature

en en centre de la contract de la contract

Sustainability of Australia's Electricity Generation

NOM	ENCLATURE
Nomen	clature
Symbo	Explanation
3	Equivalent life of subsystem

λς	Equivalent life of subsystem s (system life, n, divided by life of subsystem, s)
۵ ۵	Exponent for power law estimates of O&M
W	Exponent for power law estimates of I
Å	Pre-factor for power law estimates of I
AC	Annualised Costs (economic indicator), usually includes O&M, C _P , and F
AD	Acidification (environmental indicator)
ADP	Abiotic Depletion Potential: measure of abiotic depletion.
AP	Acidification Potential: measurement of acidification.
Ass.	Assumed
AUD	Currency of Australia (Australian Dollars).
Aust.	Australia
В	Pre-factor for power law estimates of O&M
BIC	Black coal (domestic quality black coal transported wholly within Australia)
BICE	Export quality black coal (transported to Japan)
BrC	Brown coal
$C_{x,v}$	ADP parameter: consumption of substance x in region y for 1 year
C	EVA method: cost of capital
C_2H_{4-eq}	Ethylene equivalents; units of photochemical smog formation
CA	EVA method: capital expenditure in calculation year
CC	Climate Change (environmental indicator)
CCGT	Combined-Cycle Gas Turbine (Combustion Technology)*
$CF_{x,y}$	Factor representing the relative impact of substance x to the reference substance
	for LCA impact category y
CH_4	Methane
CML	Centre of Environmental Studies, Leiden (Netherlands)
CO	Carbon monoxide
CO ₂	Carbon Dioxide
CO _{2-cq}	Carbon Dioxide equivalents; units of climate change.
Conv	Conveyor
C_{R}	NPV Method: annual capital investment recovery cost
CVA	Capital value added (economic indicator)
Data	Data reference number for Appendix 3
Ref.	
ED	Employment (social indicator method)
EDP	Energy or exergy depletion potential
EI	Indirect employment (social indicator method)
EIA	Environmental Impact Assessment
EM	Employment method: employment multiplier derived from input-output tables
EN	Energy depletion (environmental indicator method)
EP	Eutrophication Potential; measure of eutrophication.
El	Eco-toxicity (environmental indicator)
EU	Eutrophication (environmental indicator)
	<u></u>

^{*} See 'Glossary' for further descriptions.

Symbol	Explanation
EVA	Economic Value Added (
EX	Exergy depletion (environ
F	Economic Methods: Cost
- Fatal	Fatalities indicator
GRP	Currency of Great Britain
Gen	Electricity generation sub
	10 ⁹ Joule: ISO units of en
CIVIL CIVIL	10 ⁹ Watter units of energy
	To waits, units of energy
Gwnpa	Gwn per annun
GWP	Global warming Potentia
H ₂ S	Hydrogen sulphide
HHV	High Heating Value
HT	Human toxicity (environr
i	Interest rate or amortisati
1	Uncertainty assessment p
I	Investment or capital cost
	Capital requirements (ecc
ICS	Characterised score for a
IGCC	Integrated Coal Gasificat
lp	Relative capital cost (eco
IRR	Internal Rate of Return (e
ISO	International Organisation
100 ka	ISO standard unit of mas
ng km	10^3 metres
ҚШ 1.+	10^{3} matrix topped
KL T	VA Mathadi apata of lah
L I	VA Method: costs of faor
L	Distance traversed by tra
LCA	Lite Cycle Analysis or Li
LCI	Life Cycle Inventory
LCIA	Life Cycle Inventory Ana
LHV	Low Heating Value
LNG	Liquefied Natural Gas
LPG	Liquefied Petroleum Gas
lti	Lost time injuries; injuri
	for more than one day.
ltif	Lost time injury frequence
m,	Mass rate (kg per unit) of
MES	VA Method: cost of mate
Min	Fuel mining subsystem
Mt	10 ⁶ metric tonnes
Mina	Mt per annum
MW	10 ⁶ Watts: units of canac
MU/h	10 ⁶ Watt-hours' units of
14T 14 11	Life of system (years)
11 N1	Life of project (years)
NO	Dinitrogen evide (e eres
N ₂ U	Dinitrogen oxide (a greet
NG	Natural gas transported b
NMVOC	Volatile organic compou
NO ₂	Nitrogen dioxide

Nomenclature

(economic indicator method) nmental indicator method) ts for the main fuel consumed

(British Pounds). bsystem nergy y for electricity

al; measure of climate change.

mental indicator) ion factor (%). parameter for Kennedy et al. (1996) method st (economic indicator method) onomic indicator) an LCA impact category tion Combined-Cycle (Combustion Technology)* onomic indicator) economic indicator method) on for Standardisation SS

our (i.e. salaries and wages) ansport or transmission subsystem. ife Cycle Assessment

alysis or Life Cycle Inventory Assessment

ies for which cause an employee to be absent from work

icy; LTI per 10⁶ WH of substance or flow x erial, energy and services.

city for electricity energy for electricity

enhouse gas) by pipeline wholly within Australia ınd

Nomenclature

Sustainability of Australia's Electricity Generation

Symbol.	Explanation
NOPAT	EVA method: net operating profit before tax
NOx	Nitrogen Oxides
NPC	Net Present Cost (economic indicator method)
NPV	Net Present Value (economic indicator method)
NSW	Australian state of New South Wales
0	Operating costs
O&M	Operating and maintenance costs ($O\&M = O - (Capital + F)$)
OCGT	Open-Cycle Gas Turbine (Combustion Technology)* or Gas Turbine
OEI	Overall environmental index (weighted sum of environmental indicators)
Р	NPV Methods: actual production of desired product(s)
	VA Method: purchased materials, energy and service.
PO _{4-eq}	Phosphate equivalents; units of eutrophication.
POCP	Photo-Oxidant Creation Potential; measure of photochemical smog formation
PM	Particulates (environmental indicator method)
PS	Photochemical smog (environmental indicator)
Q	Capacity of subsystem, i.e. maximum production of desired product(s).
Qld.	Australian state of Queensland.
R	VA Method: cost of raw material purchases.
R _{x,v}	ADP parameter: resource of substance x in region y
Raill,	Refers to the domestic and foreign rail transport legs in the BICE transport
Rail2	subsystem.
RD A	Resource depletion, Australian reference region (environmental indicator method)
RD A	Resource depletion, world reference region (environmental indicator method)
Ref.	Data reference number for Appendix 3
S	VA Methods: sales revenue
Sb.eq	Antimony equivalents; units of abiotic depletion.
SD	Standard deviation
SG	Specific gravity: ratio of substance density to water (liquid) or air (gas)
SO ₂	Sulphur dioxide
SO _{2-eq}	Sulphur dioxide equivalents; units of acidification potential.
ST	Steam turbine
SW	Solid waste (environmental indicator method)
Sys.	System
t	ISO metric tonnes
Trm.	Electricity transmission subsystem
Tsp.	Fuel transport subsystem
U.K.	United Kingdom or Great Britain
USD	Currency of the U.S.A. (U.S. Dollars)
VA	Value added (economic indicator method)
Vic.	Australian state of Victoria.
VOC	See NMVOC
WD	Water consumption (environmental indicator method)
WF	Weighting factors to produce an OEI
WH	work hours; hours of employee work time.
WP	LTIF method: worker productivity (hours worked per annum per employee)
x	Uncertainty assessment parameter for Kennedy et al. (1996) method

Systems	Explanation
BIC-ST	Fuel: BIC (domestic
	Technology: ST (ste
BIC-IGCC	Fuel: BIC
	Technology: IGCC (
BICE-ST	Fuel: BICE (export of
	Technology: ST
BICE-IGCC	Fuel: BICE
	Technology: IGCC
BrC-ST	Fuel: BrC (brown co
	Technology: ST
BrC-IGCC	Fuel: BrC
	Technology:IGCC
NG-ST	Fuel: NG (natural ga
	Technology: ST
NG-OCGT	Fuel: NG
	Technology: OCGT
NG-CCGT	Fuel: NG
	Technology: CCGT
LNG-ST	Fuel: LNG (liquefie
	Technology: ST
LNG-OCGT	Fuel: LNG
	Technology: OCGT
LNG-CCGT	Fuel: LNG

1	t i C	m
		-

Nomenclature

Table of Systems (technology descriptions in Nomenclature and Glossary).

e quality black coal transported wholly within Australia) eam turbine)

(integrated gasification with combined cycle gas turbine) quality black coal transported to Japan)

oal transported wholly within Australia)

as transported by pipeline wholly within Australia)

(open cycle gas turbine)

(combined cycle gas turbine) ed natural gas transported to Japan)

Technology: CCGT

287

Glossary

GLOSSARY

- Acidification an LCA impact category and indicator of sustainability (AD) associated with the fall in pH due to the rise in acidifying gases in the atmosphere. Major acidifying gases are NO_x and SO_x, which when reacted with water become HNO₂, HNO₃, and H₂SO₄. These acids mix with falling rain, and among other things cause the discoloration and dissolution of marble structures and a fall in the pH (alkalinity) of lakes.
- allocation the act of proportionally distributing the responsibility for resource consumption, emissions and waste streams from processes in the life cycle. Is used when more than one (useful) product is generated from a subsystem.
- Annualised Costs an economic indicator of sustainability (AC) representing wealth generation including both capital and operating costs. The capital costs are annualised, which is a common method of depreciation when estimating costs for electricity generation systems. Annualisation modifies the discounted annual capital contributions so that they have a constant value (see Figure 3.3.1, Chapter 3)
- artesian water water which exists below the surface in water bearing rocks, sands and gravels. An aquifer is a discrete source layer of artesian water.
- ash non-combustible fraction of coal, which is removed as a solid. Fly-ash is the fraction of ash which is too small to settle in the boiler and is collected by a particulate removal device, i.e. electrostatic precipitator or baghouse. Bottom ash is the fraction of ash that settles in the boiler. Removal is by mechanical, pneumatic (air) or hydraulic (water) means.
- black coal coal which has a higher energy content that 17 MJ per kg. Includes coal of the anthracite, bituminous, and sub-bituminous classifications. This coal has an export value, and contributes significantly to Australian exports.
- brown coal coal which has an energy content lower than 17 MJ per kg. Includes coals of the lignite, and some sub-bituminous (C) classifications. Australian brown coals are characterised by their very high moisture content (50-66 % of their mass), and low energy contents (8-10 MJ per kg), which limits their export value.
- capital costs that fraction of costs which can be considered as once off, and occurs before the operation of the facility, i.e. construction costs. It does not include costs for the maintenance of capital after the facility has begun operating. These and other costs accrued continuously through the life of the plant are considered operating

costs.

and annualised capital costs (C_R) (see annualised costs). CO₂. These conversion factors are equivalency factors (CF). inventory data belongs. to the rise in global temperature caused by greenhouse gases. making coke from black coal. allows such systems to produce efficiencies of over 50 %. continuously and dump directly onto a conveyor belt system. excludes humans (see Human Toxicity).

Capital Inclusive Value Added - an economic indicator of sustainability (CVA) representing wealth generation. CVA is the difference between value added (VA)

characterisation - the process of quantifying the environmental impact. Uses factors representing the relative magnitude of the impact caused by one substance to a reference substance, i.e. methane (CH₄) is 11 times more effective than carbon dioxide (CO₂) in Climate Change, thus 1 tonne CH₄ is characterised as 11 tonnes of

classification - the act of identifying the appropriate impact category to which

Climate Change – an impact category and indicator of sustainability (CC) associated with the output (or emission) of greenhouse gases, for example, CO₂, CH₄, CFC's, etc. Climate Change is, under this definition, the change in climate due

coke breeze - fuel consisting of fines (small particles) generated during the process of

Combined Cycle Gas Turbine (CCGT) – a generation technology used for gaseous fuels (such as natural gas, processing and refinery waste gases, and manufactured gases (e.g. gasified coal or biomass)) or light fuel oils. The key features of a CCGT are the gas turbine, heat recovery steam generator (HRSG), and the steam turbine (Figure 4.2.8, Chapter 4). It is the combined use of a gas and steam turbine, which

criteria - (in sustainability assessment) a target score for an indicator. The setting of criterion is a value choice, and thus criteria have no scientific basis.

draglines – mining machinery used in black coal mining. It is a crane like machine with a large shovel rather than a hook. This design gives a large operational range.

dredger - mining machinery used in brown coal mining. There are two designs in operation in Australia, the bucket-wheel and the ladder types. Both operate

Eco Toxicity - an impact category and indicator of sustainability (ET) associated with the emission of substances that are harmful to animals and plants. This category

electricity generation efficiency – a measure of the effectiveness of energy conversion from fuel energy to electricity in the generation subsystem. It is the ratio of the energy exported as electricity over the energy entering as fuel. This thesis uses a higher heating value (HHV) basis for all energy efficiencies.

<u>Glossary</u>

- eqivalency factor (CF) the value used to convert one substances emission value (say kg substance) to that of the impact category indicator (say kg indicator).
- Eutrophication an impact category and indicator of sustainability (EU) associated with the rise in the concentration of nutrients in the environment. Compounds high in nitrogen (N) or phosphorus (P), such as those found in fertilisers, are the primary cause. Consequences of eutrophication are increased algae and other weed growth, which, in waterways, causes a drop in water oxygen content and thus a drop in the numbers of aquatic animals. Also known as 'nutrification'.
- Extraction of Biotic Resources an impact category associated with the consumption of animal and plant resources. It considers resources regenerated through both direct human activity (i.e. farming) and natural processes (i.e. hunting, fishing, forestry, etc.).
- fuel costs costs accrued through the regular purchase of fuels throughout the life of the plant are called **fuel costs**.
- fugitive emissions emissions of a substance which is not intentional or accidental, i.e. petroleum or natural gas from a leaking pipeline.
- gasification the conversion of coal to a low-energy content gas through partial oxidation, usually in the presence of steam (i.e. $coal + oxygen + steam \Rightarrow carbon$ monoxide + hydrogen + methane + carbon dioxide).
- greenhouse gas any gas which remains in the atmosphere for a sufficient period to absorb energy radiating from the surface of the earth, and thus increase the temperature of the atmosphere (global warming or climate change).
- Human Toxicity an impact category and indicator of sustainability (HT) associated with the emission of substances that are harmful to humans.
- **impact category** a class representing environmental impacts of concern into which LCI results may be assigned (ISO term).
- indicator (of sustainability) a measure representing an environmental, economic or social impact of concern.

integrated gasification combined cycle (IGCC) - a generation technology which

Sustainability of Australia's Electricity Generation

converts a solid fuel to gas (gasification), cleans the product gas, and provides it to a combined cyclc system (Figure 4.2.4, Chapter 4). Gasification in an IGCC is usually in a fluidised bed gasifier. There are many IGCC designs. inter-generational equity – a theoretical situation in which all persons, no matter their location in time, have equal opportunity to live with a high level of well-being, and to access financial and natural resources. The context is usually equal to or greater, rather than equal to (as the strict definition would suggest). intra-generational equity - a theoretical situation in which all persons, no matter their present location, have equal opportunity to live with a high level of well-being, and to access financial and natural resources. In the current situation persons in industrialised countries have greater access than persons in developing countries, as they have generally have greater financial power.

- contained a similar plant, involves no change in land use.
- ultimate disposal.
- (LNG) is common for ocean transport.
- present in a gas stream.

Glossary

Land Use - an impact category associated with the change in how the effected land is used. For example, the building of a plant in a virgin forest involves a change in land use. However, the building of a plant on an industrial site, which previously

Life Cycle Analysis (LCA) – Life Cycle Assessment (or LCA) is a recently developed method of environmental analysis for quantifying the environmental effect of any product, process or service over its entire life cycle, from raw material acquisition to

natural gas - petroleum gas, high in methane (up to 96 %). Production is often coincidental with crude oil and other petroleum products. Liquefaction of natural gas

NMVOC – Non-methane volatile organic compounds. Category of emissions to air, including all organic compounds, except methane, that are likely to evaporate, or are

normalisation - the conversion of impact category indicator and indicator of sustainability values from their absolute value to a dimensionless number representative of its relative contribution to a given region's impacts. This involves dividing the absolute value by the total impact generated by the region.

Open-Cycle Gas Turbine - a generation technology used for gaseous fuels (such as natural gas) and light fuel oils. The gas turbine system consists of a compressor, combustion chamber, turbine, generator and associated auxiliary equipment (Figure

Glossary

Sustainability of Australia's Electricity Generation

4.2.7, Chapter 4).

- operating costs that fraction of costs which is accrued continuously through the life of the facility. It also includes the cost maintaining capital items during operation, O&M is a limited set of these costs, excluding loan and interest repayments and costs of products produced in earlier subsystems (i.e. the black coal used in the generation system). See also capital costs.
- overburden mining term for undesirable material which lies above a body of a particular desirable ore body. Usually used when the ore body is horizontal or near horizontal to the surface.
- **Photo-oxidant Formation** an impact category and indicator of sustainability (PS) associated with the formation of a brown cloud of pollutants, which usually occurs in urban areas. Major precursors are NOx, VOC's, and particulates, which are released through incomplete combustion in cars, and other combustion systems, and fugitive emission sources.
- renewables energy sources that are regenerated naturally, or can be regenerated through the action of man, within the short time spans required for significant consumption. These include sunlight, wind, water (river and sea/wave), and biomass.
- Resource Depletion an impact category and indicater of sustainability (RD) associated with the use of mineral and energy resources. It considers the reduced resources caused through mining, and oil and gas production. It does not consider the energy use and emissions caused through these activities. Also known as 'Extraction of Abiotic Resources'.
- sphere an category of sustainability. Refers to the environment, economy or social classifications in sustainability.
- spoil piles the material produced from a coal mine that is too high in non-coal material to be utilised. It is generally stockpiled near the mine site and is prone to natural decomposition and spontaneous combustion.
- Steam Turbine (ST) a generation technology used for solid fuels, where the solid fuel is broken or pulverised into small particles and carried with the combustion air to the burner mouth. Most commonly the system is used for coal, however any solid fuel can be used, i.e. bagasse, municipal waste, etc. The system consists of many sections, including fuel preparation, boiler, steam turbine, condenser, fans, flue gas

cleanup, stack, and ash handling plant (see Figure 4.2.3, Chapter 4). This system can also consume gas (i.e. natural gas) and liquid (i.e. oils) fuels. Stratospheric Ozone Depletion – an impact category associated with the decrease in ozone (O₃) in the Ozone Layer. The Ozone Layer provides the planet shelter from harmful ultra-violet (UV) radiation, which are produced by the sun. These UV rays have been proven as the major cause of skin cancers in humans and animals. subsystem – a subsystem is a well defined processing step, such as mining, transport and generation in coal generation systems, or production, processing, transmission, and generation in natural gas generation systems. Subsystems link together without crossover.

- development have identical meanings in line with common usage. welfare.
- cycle of the product under study. not undergo any further processing. materials, energy and services for a given system.

Glossary

Sustainable Development – the act of improving the harmful effects of present products, processes, and services, on the environment, economy, and social welfare of the world. It is also, the act of progressing, through advancement, towards the ideal state of sustainability. In this thesis, sustainability and sustainable

Sustainability - theoretical end state in which further sustainable development is no longer necessary (i.e. harmful effects of humans balanced by positive natural and human responses). The ideal state, meeting all needs of the present population, without reducing the probability of future generation's having similar or better

system - in the context of a Life Cycle Assessment (LCA) or Sustainability Assessment the system is the group of subsystems which adequately describe the life

system boundary - describes the boundary between what is considered to be a part of the system and the general environment. All substances crossing this boundary should be raw materials, which have yet to be processed, or emissions, which will

Value Added - an economic indicator of sustainability (VA) representing wealth generation. VA is the difference between the sales revenue and the purchases of

washery rejects - coal containing material which, after the washing process (where high ash and low ash coal are separated), contains too high an ash content for sale.

water (high quality) and water (low quality) - high and low quality water correspond

293

Glossary

to municipal drinking water ('tap water') and surface waters from rivers, lakes, reservoirs, estuaries and seas.

- Water Consumption an impact category and indicator of sustainability (WD) associated with the use of water resources and the increased competition, between other users (human or otherwise), this may cause. This includes the use of surface, sea, ground and artesian water.
- weighting an optional step in the LCIA which involves the combination of individual impact category values into a single index.

REFERENCES

MAIN TEXT

Bureau of Statistics (ABS), 4604.0, Tables 1.18 and 2.10. Australian Bureau of Statistics, Table 3T. Statistics, Canberra, Cat. No. 4608.0.40.001. Australian Bureau of Statistics, pp. 9-18. Association (ACA), March, p. 20. environment.php?tab=environment, last accessed 14/7/2003. Australia). Gas Association (AGA), AGA Research Paper No. 12, May. Association, Greenhouse Challenge program, November. Australian Greenhouse Gas Office (AGO), September, pp. 19. Organisation, Department of Industry, Science & Economics. 23.

4/12/2000.

ABS (2001a), Energy and Greenhouse Gas Emissions Accounts, Australia, Australian

ABS (2001b), 5209.0 Australian national accounts: input-output tables, 1996-97,

ABS (1998), Mineral Account, Volume Change Tables, Australia, Australian Bureau of

ABS (1994), 5237.0 Australian national accounts input-output multipliers, 1989-90,

ACA (1999), Who's Who in Australian Cogeneration, Australian Cogeneration

AGA (2002), 'Making Electricity Generation Cleaner', Hacts and Figures -Environment, Australian Gas Association, http://www.gas.asn.au/factsandfigures/

AGA (2001), Gas Statistics Australia 2001, Australian Gas Association (Canberra,

AGA (2000), Assessment of Greenhouse Gas Emissions from Natural Gas, Australian

AGA (1997), Greenhouse Challenge Facilitative Agreement, Australian Gas

AGO (1999), National Greenhouse Gas Inventory 1997 with Methodology Supplements,

AGSO (1998), Oil and Gas Resources of Australia 1998, Australian Geological Survey

Anderson, K. L. (2001), 'Reconciling the electricity industry with sustainable development: backcasting - a strategic alternative', Futures, Volume 33, pp. 607-

APPEA (1999), APPEA Greenhouse Challenge Document, Australian Petroleum Production and Exploration Association (APPEA), www.appea.com.au, last accessed

- Aspinall, T. O., Isles, P. T., Davidson, C. W., Davis, A. L., Hagan, T. N., McManus, D.A. and Smith, G. H. (1993), 'Strip Mining', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12, Australasian Institute of Mining and Metallurgy, pp. 203-224.
- Audus, H. (1996), 'IEA Greenhouse Gas R & D Programme: Full Fuel Cycle Studies', Energy Conversion and Management, Volume 37, Number 6-8, pp. 837-42.
- Ayres, R.U. (1996), 'Statistical measures of unsustainability', *Ecological Economics*, Volume 16, Number 4, pp. 239–255.
- Azapagic, A. and Perdan, S. (2000), 'Indicators of Sustainable Development For Industry: A General Framework', Trans. IChemE, Volume 78, Part B, July 2000, pp 243-261.
- Barbier, E.B. (1987), 'The concept of sustainable economic development', Environmental Conservation, Volume 14, Number 2, pp. 101–110.
- Baumann, A. E. and Hill, R. (1992), 'Quantification of External Costs of Energy Technologies Applied to Photovoltaic Life-Cycle Analysis', in OECD/IEA. Expert Workshop on Life-Cycle Analysis of Energy Systems, Methods and Experience, *Proceedings*, Organisation for Economic Co-operation and Development (OECD) and the International Energy Agency (IEA), Paris, France, 21st-22nd May 1992, pp. 143-50.
- Bell, S. and Morse, S. (2001), 'Breaking through the Glass Ceiling: who really cares about sustainability indicators?', Local Environment, Volume 6, Number 3, pp. 291-309.
- Berg, N. W. van den, Huppes, G., Lindeijer, E. W., Ven, B. L. van der and Wrisberg, M. N. (1999), 'A Framework for Quality Assessment', in Berg, N. W. van den, Huppes, G., Lindeijer, E. W., Ven, B. L. van der and Wrisberg, M. N., Quality Assessment for LCA, Centre for Environmental Science (CML), Leiden, CML Report 152, http://www.leidenuniv.nl/interfac/cml/ssp/publications/quality.pdf, pp. 2-3, 21, 30.
- BHP Minerals Technology (2001), ACARP-Coal in a Sustainable Society, Australian Coal Association Research Program, ACARP Project C9058, BHP Research, http://www.sustainabletechnology.com.au/, last accessed 26/5/2001.
- BHP Research (1999), LCA of Steel and Electricity Production, ACARP Project C8049, Australian Coal Association Research Program, BHP Research, December, p. 3

frames.html, last accessed on 18/9/2000. Australasian Institute of Mining and Metallurgy, pp. 174-85. sustainability, National Research Council (U.S.A.). BP (2000), Statistical Review of World Energy, BP, June 2000. Longman Cheshire (Melbourne, Aust.), Chapters 4 and 5. Volume 44, Number 3, pp. 409-27. Butterworth Hienemann. Number 3, pg. 315. Monash University, February, pp 173, etc.

International Thomson Publishing, pp. 71-123. Chadwick, M. J., Highton, N. H. and Lindman, N. (eds.) (1987), Environmental Impacts

References

- (Main Report) and case studies, http://www.sustainabletechnology.com.au/
- Blackham, M. (1993), 'Choice Between Underground and Open Cut', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12,
- Board on Sustainable Development (1999), Our Common Journey a transition toward
- Boustead, I. (1995), 'Life-cycle assessment: an overview', Energy Word, pp. 71-94.
- Brennan, D.J. (1990), Process Industry Economics: An Australian Perspective,
- Briassoulis, H. (2001), 'Sustainable Development and its Indicators: Through a (Planner's) Glass Darkly', Journal of Environmental Planning and Management,
- Brigham, E.F. and Ehrhardt, M. C. (2002), Financial Management: Theory and Practice, 10th ed., Harcourt College Publishers (U.S.A.), pp. 528-30.
- Brockway, D. J. and Higgins, R. S. (1991), 'Brown Coal Sampling, Analysis and Composition', Chapter 5. in Durie, R. A. (ed.), The Science of Victorian Brown Coal,
- Buhrs, T. and Aplin, G. (1999), 'Pathways towards sustainability: The Australian Approach', Journal of Environmental Planning and Management, Volume 42,
- Burgess, A.A. (1999), An Environmental and Cost Assessment of Gas Oil
 - Desulphurisation, M. Eng. Sc. Thesis, Department of Chemical Engineering,
- Burgess, A.A. and Brennan, D.J. (2001), 'Application of life cycle assessment to chemical processes', Chemical Engineering Science, Volume 56, pp. 2589-2604.
- Cardwell, H., Feldman, D. and Kahn, J. (1995), 'Measuring Sustainability in a Humid Region', Proceedings of the 22nd Annual Conference on Integrated Water Resource Planning for the 21st Century, Cambridge (MA, USA), May 7-11, 1995, pp. 169-72.
- Carlson, K. E. (1996), 'Fossil Fuels', in Black & Veatch, Power Plant Engineering,

References

Sustainability of Australia's Electricity Generation

of Coal Mining and Utilisation, Beijer Institute, United Nations Environment Programme, USSR Academy of Sciences, and the Swedish National Energy Administration, Pergamon Press, pp. 30-31.

- Clark, J.J., Hindelang, T.J. and Pritchard, R.E. (1989), Capital Budgeting: Planning and Control of Capital Expenditures, 3rd, ed, Prentice-Hall (Englewood Cliffs, USA),
- Clift, R. (1998), 'Life Cycle Assessment How to Slim Down Your Ecoprofile', IChemE Environmental Protection Bulletin, pp. 9-13.
- Consoli, F., Allen, D., Boustead, I., Fava, J., Franklin, W., Jensen, A. A., de Oude, N., Parrish, R., Perriman, R., Postlethwaite, D., Quay, B., Séguin, J. and Vigon, B. (1993), Guidelines for Life-Cycle Assessment: A "Code of Practice", 1st ed., SETAC.
- Coulon, R., Camobreco, V., Teulon, H. and Besnainou, J. (1997), 'Data Quality and Uncertainty in LCI', In J. LCA, Volume 2, Number 3, pp. 178-82.
- Cowell, S. J., Wehrmeyer, W., Argust, P. W., Graham, J. and Robertson, S. (1999), 'Sustainability and the primary extraction industries: theories and practice', *Resources Policy*, Volume 25, Number 4, pp. 277–286.
- CRIEPI (1994), Energy Technology Life Cycle Analysis that Takes CO2 Emission Reduction into Consideration, the Central Research Institute of the Electric Power Industry (CRIEPI), http://criepi.denken.or.jp/RD/nenpo/1995E/seika95kankvo23E .html, last accessed on 15/9/2000.
- Curran, M., A. (1993), 'Broad-based Environmental Life Cycle Assessment', Environmental Science and Technology, Volume 27, Number 3, pp. 431-6.
- de Simone, L. and Popoff, F. (1997), Eco-effciency: The Business Link to Sustainable Development, MIT, Massachusetts.
- Delta Electricity (1999), Delta Electricity Environment Report 1996-1999, Delta Electricity.
- Department of Mines and Energy (1998), 47th Oueensland Coal Industry Review 1997-1998, Department of Mines and Energy (Queensland), pp. 8-9, 16-7, 48-51, 58-9.
- DETR (U.K.) (2001), Achieving a better quality of life, Review of progress towards sustainable development, Government annual report 2000, Department of the Environment, Transport and the Regions (DETR (U.K.)), January, http://www.sustainable-development.gov.uk/ann_rep/susdevel.pdf.
- Division for Sustainable Development (2003), Indicators of Sustainable Development: Guidelines and Methodologies, United Nations Division for Sustainable

Development, Table 4, http://www.un.org/esa/sustdev/natlinfo/indicators/isdms2001/ isd-ms2001 isd.htm, last accessed 8/5/2003. DoE (U.K.) (1998), UK Greenhouse Gas Emission Inventory 1995, Global Atmosphere Division, Department of the Environment (DoE (U.K.)), 19th February, http://www.aeat.co.uk/netcen/airgual/emissions/ipccwww.html, link lost. DoE (U.K.) (1997), Indicators of Sustainable Development for the United Kingdom, Report of the Department of the Environment, Transport & the Regions (U.K.). DOE (U.S.A.) (2000), Transport Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy (DOE (U.S.A.)), PED-IGCC-98-006, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/trnspt3x .pdf,

0400.

last accessed 13/3/2002.

- Metallurgy, pp. 602-11.
- http://pss100.psi.ch/~shirsch/gabe/lca.html, last accessed 15/9/2000.
- 3.
- Volume 37, pp. 5–22.
- DP2/55, November 1997.
- Electricity Supply Association of Australia (ESAA), pp 46.

References

DOE (U.S.A.) (1999), Market-Based Advanced Coal Power Systems Final Report, U.S. Department of Energy (DOE (U.S.A.)), Office of Fossil Energy, May, DOE/FE-

Doherty, J. K. (1993), 'Coal Mine Development, Organisation, Administration and Surface Arrangements', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12, Australasian Institute of Mining and

Dones, R. and Gantner, U. (1996), Life Cycle Assessment (LCA), GaBE Project: Comprehensive Assessment of Energy Systems, the Paul Scherrer Institut (PSI),

ECNSW (1980), Mount Piper Power Station Environmental Impact Assessment, Electricity Commission of New South Wales (ECNSW), August, Volumes 1, 2 and

Ekins, P. and Simon, S. (2001), 'Estimating sustainability gaps: methods and preliminary applications for the UK and the Netherlands', Ecological Economics,

Ekvali, T., Rydberg, T., Hedenberg, Ö., Jacobson, B. B., Pajula, T. and Wessman, H. (1997), Guidelines on Life Cycle Impact Assessment of Pulp and Paper, Nordpap

Electricity Supply Association of Australia (1999), Electricity Australia 1999,

Energy Information Agency (U.S.A.) (1995), Electricity Generation and Environmental

References

Externalities: Case Studies, Office of Coal, Nuclear, Electric and Alternate Fuels, Energy Information Agency (U.S.A.), September, DOE/EIA-0598.

- Energy Research and Development Corporation (1992), Advanced Generation Options for the Australian Electricity Supply Industry and their Impact on Greenhouse Emissions, Phase 2, Energy Research and Development Corporation and Electricity Supply Association of Australia, ERDC 130, January.
- Environment Australia (2000), Commonwealth Environmental Impact Assessment, Australian EIA Network, Environment Australia, http://www.environment.gov.au/ epg/eianet/eia/EPIPbrochure.html, last accessed 5/7/2000.
- Environment Australia (1996), Environment Protection (Impact of Proposals) Act 1974 - Administrative Procedures, As at 1 January 1996, Australian EIA Network, Environment Australia, http://www.environment.gov.au/epg/eianet/legislation/legis com adproc.html.
- EPA (U.S.A.) (2000), Inventory of U.S. Greenhouse Gas Emissions and Sinks: 1990 -1998, Office of Atmospheric Programs, United States Environmental Protection Agency (EPA (U.S.A.)), EPA 236-R-00-001, April, pp. ES-4, ES-9.
- EPA (U.S.A.) (1999), EPA Office of Compliance Sector Notebook Project: Profile of the Oil and Gas Extraction Industry, Office of Compliance, Environmental Protection Agency (U.S.A.), October, pp. 16-27,
- EPS (Canada) (1989), Environmental Code of Practice for Treatment and Disposal of Waste Discharges from Offshore Oil and Gas Operations, Environmental Protection Service, Environment Canada, EPS 1/PN/2.
- Evans, D. G. (1995), The Emission of Greenhouse Gases in the Supply of Electricity to Victoria, A Comparison of Low-Rank Coal with Other Fuels, Cooperative Research Centre: New Technologies for Power Generation from Low-Rank Coal, Report No. 95008, September.
- Eyre, N. and Holland, M. (1992), 'External Costs of Electricity Generation Using Coal', in OECD/IEA, Expert Workshop on Life-Cycle Analysis of Energy Systems, Methods and Experience, Proceedings, Organisation for Economic Co-operation and Development (OECD) and the International Energy Agency (IEA), Paris, France, 21st-22nd May 1992, pp. 124-34.
- Fava, J. A. and Page, A. (1992), 'Application of Product Life-Cycle Assessment to Product Stewardship and Pollution Prevention Programs', Water Science and

- Technology, Volume 26, Numbers 1-2, pp. 275-87. USA.
- journals/lca/letters/finnveden.htm
 - Impact Assessment, February 1996.
- Occupational and Environmental Medicine, Volume 54, pp. 642-5. Kluwer Academic.
- Volume 19, Number 2, pp. 95-106. and Unwin (London, U.K.).

- B, May, pp. 105-19.

Sustainability of Australia's Electricity Generation

Fava, J., Consoli, F., Denison, K., Dickson, K., Mohin, T. and Vigon, B. (1991), A conceptual framework for life-cycle impact assessment, SETAC, Pensacola, Florida,

Finnveden, G. and Lindfors, L.-G. (1998), 'Data Quality of Life Cycle Inventory Data -Rules of Thumb', Global LCA Village, Letters to the Editor, http://www.ecomed.de/

Finnveden, G., ed. (1996), 'Part III: Resources', in Towards a Methodology for Life Cycle Impact Assessment, Udo de Haes, H. A., ed., Europe Workgroup on Life Cycle

Finocchiaro, C., Lark, A., Keating, M., Ugoni, A. and Abramson, M. (1997), 'Does occupational exposure to brown coal dust cause a decline in lung function?',

Funtowicz, S. O. and Ravetz, J. R. (1990), Uncertainty and quality in science for policy,

George, C. (2001), 'Sustainability appraisal for sustainable development: integrating everything from jobs to climate change', Impact Assessment and Project Appraisal,

Gilchrist, R. R. (1971), Managing for Profit, The Added Value Concept, George Allen

Global Reporting Initiative (2002), Sustainability Reporting Guidelines, Global Reporting Initiative, <u>www.globalreporting.org</u>, last accessed 2/1/2003.

Gloe, C.S. (1984), 'Case History No. 9.1. Latrobe Valley, Victoria, Australia', Section 9.1 of Poland, J.F. (ed.), Guidebook to studies of land subsidence due to groundwater withdrawal, United Nations Educational, Scientific and Cultural Organization (UNESCO), http://wwwrcamnl.wr.usgs.gov/rgws/Unesco/, last accessed 22/7/2003.

Golonka, K.A. and Brennan, D.J. (1997), 'Costs and Environmental Impacts in Pollutant Treatment: A Case Study of Sulphur Dioxide Emissions from Metallurgical Smelters', Trans. IChemE, Volume 75, Part B, November, pp. 232-44.

Golonka, K.A. and Brennan, D.J. (1996), 'Application of Life Cycle Assessment to Process Selection for Pollutant Treatment: A Case Study of Sulphur Dioxide Emissions from Australian Metallurgical Smelters', Trans. IChemE, Volume 74, Part

- Goodland, R. and Ledec, G. (1986), Neoclassical Economics and Principles of Sustainable Development, Office of Environmental and Scientific Affairs, The World Bank, Washington, USA.
- Gorokhov, V., Manfredo, L., Ratafia-Brown, J., Ramezan, M. and Stiegel, G. J. (2002), Life-Cycle Assessment of Advanced Power Generation Systems, Technology (Cognizant Communications Corporation, U.S.A.), Volume 8, pp. 217-228.
- Graedel, T. E. (2002), 'Green Chemistry and Sustainable Development', in Clark, J. and Macquarrie, D. (eds.), Handbook of Green Chemistry and Technology, Blackwell Science, Chapter 4, pp. 56-61.
- Guinée, J. (1993), Data for the Normalization Step within Life Cycle Assessment of *Products*, Centre of Environmental Studies, Leiden, December,
- Guinée, J. B. and Heijungs, R. (1995), 'A Proposal for the Definition of Resource Equivalency Factors for use in Product Life-Cycle Assessment', Environ. Toxicol. Chem., Volume 14, pp. 917-925.
- Guinée, J. B., Gorree, M., Heijungs, R., Huppes, G., Kleijn, R., Wegener Sleeswijk, A., Udo de Haes, H. A., de Bruijn, J. A. and van Duin, R. (2001), Life Cycle Assessment, An Operational Guide to the ISO Standards, 1st ed., Centre of Environmental Science, Leiden, Three volumes, I Guidelines, II Operational Models and Data, III Scientific Backgrounds.
- Guy, G. B. and Kibert, C. J. (1998), 'Developing indicators of sustainability: US experience', Building Research and Information, Volume 26, Number 1, pp. 39-45.
- Hammond, G. P. (2000), 'Energy, Environment and Sustainable Development: A UK Perspective', Trans. IChemE, Volume 78, Part B, July, pp. 304-23.
- Harsch, M., Schuckert, M., Eyerer, P. & Saur, K. (1996), Life-Cycle Assessment, Advanced Materials & Processes, Number 6, pp. 43-6.
- Hazelwood Power (1999), Annual Environmental Report 1st July 1998 30 June 1999, Hazelwood Power.
- Heaney, J. P. (1995), 'Sustainable Stormwater Management', Proceedings of the 22nd Annual Conference on Integrated Water Resource Planning for the 21st Century, Cambridge (MA, USA), May 7-11, 1995, pp. 165-8.
- Hedley, T. P. and McDonald, A. (1993), 'Longwall and Shortwall', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12, Australasian Institute of Mining and Metallurgy, pp. 359-90.

- 95-106.
- Economics, John Wiley and Sons (London, UK), pg. 59.
- Australasian Institute of Mining and Metallurgy, pp. 572-88.
- 88.
- 32.
- (U.K.), Chapter 1, pp. 25-6.
- Science, Leiden University, February, pp. 29-49.
- - No.
 - accessed 28/1/2003.

Heijungs, R. (1995), 'Harmonization of Methods for Impact Assessment', Environmental Science and Pollution Research, Volume 2, Number 4, pp. 217-24.

Heijungs, R., Guinée, J., Huppes, G., Lankreijer, R. M., Udo de Haes, H. A., Wegener Sleeswijk, A., Ansems, A. M. M., Eggels, P. G., van Duin, R. and de Goede, H. P. (1992), Environmental Life Cycle Assessment of Products: (in 2 volumes) 1. Guide and 2. Backgrounds, Centre of Environmental Science (CML), Leiden.

Hertz, D. B. (1964), 'Risk analysis in capital investment' Harvard Business Review, pp.

Holland, F. A., Watson, F. A. and Wilkinson, J. K. (1974), Introduction to Process

Holmes, D. S. (1993), 'Brown Coal in the Latrobe Valley', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12,

Horrocks, K. R. S. and Gadsden, W. R. (1993), 'Benification, Handling and Utilisation of Coal', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12, Australasian Institute of Mining and Metallurgy, pp. 572-

Huijbregts, M. A. J. (1998), 'A General Framework for the Analysis of Uncertainty and Variability in Life Cycle Assessment', Int. J. LCA, Volume 3, Number 5, pp. 273-80. Huijbregts, M. A. J., Norris, G. A., Bretz, R., Ciroth, A., Maurice, B., Bahr, B. von, Weidema, B. P. and Beaufort, A. S. H. de (2001), 'Framework for Modelling Data Uncertainty in Life Cycle Inventories', Int. J. LCA, Volume 6, Number 3, pp. 127-

Hunt, G. F. (1991), 'The Steam Turbine', in Davies, E. J., Modern Power Station Practice, Volume C Turbines, Generators and Associated Plant, Pergamon Press

Huppes, G. (1996), LCA Yesterday, Today and Tomorrow, Centre of Environmental

IAEA (2000), Enhanced Electricity System Analysis for Decision Makir. A Reference Book, International Atomic Energy Agency (IAEA), DECADES Project Document 4, http://www-pub.iaea.org/MTCD/publications/PDF/decades-4.pdf, last

IChemE (2002a), The Sustainability Metrics, Institution of Chemical Engineers

(IChemE), http://www.icheme.org/sustainability/index.htm, last accessed 13/1/2002.

- IChemE (2002b), Energy at the Crossroads, Institution of Chemical Engineers (IChemE), a response to the PIU Energy Review and the joint departmental energy policy consultation published in May 2002, http://www.icheme.org/pr and media/ Government relations/energyresponse.pdf, last accessed 13/1/2003.
- IEA (2001), World Energy Outlook, Assessing Today's Supplies to Fuel Tomorrow's Growth, International Energy Agency (IEA).
- IEA (1993), Advanced Power Generation a comparative study of design options for coal, International Energy Agency (IEA), Organisation for Economic Co-operation and Development, IEA Report IEACR/55, Mar.
- IEAGHG (2000), What is the IEA Greenhouse Gas R&D Programme?, International Energy Agency Greenhouse Gas Research and Development Programme (IEAGHG), International Energy Agency, http://www.jeagreen.org.uk/doc1a.htm, last accessed 15/9/2000.
- International Institute for Sustainable Development (2000) What are the general selection criteria for indicators?, International Institute for Sustainable Development (IISD), http://iisd.ca/measure/faqs.htm, last accessed 20/1/2003.
- International Sustainability Indicators Network (2003), 'Communities Working on Sustainability Indicators', Resources, International Sustainability Indicators Network, http://www.sustainabilityindicators.org/resources/WhoWorkingOnIndicato rs.html, last accessed 15/7/2003.
- ISO 14040:1998, Environmental management Life cycle assessment Principles and framework, as AS/NZS 14040, Standards Australia.
- ISO 14041:1999, Environmental management Life cycle assessment Goal and scope definition and inventory analysis, as AS/NZS 14041, Standards Australia,
- ISO 14041:1999, Environmental management Life cycle assessment Goal and scope definition and inventory analysis, as AS/NZS 14041, Standards Australia.
- ISO 14042:2000, Environmental management Life cycle assessment -- Life cycle impact assessment, as AS/NZS 14042, Standards Australia.
- ISO 14043:2000, Environmental management Life cycle assessment Life cycle interpretation, as AS/NZS 14043, Standards Australia.
- ISO/TR 14047, Environmental management Life cycle assessment Examples of application of ISO 14042, International Organisation for Standardisation,

Sustainability of Australia's Electricity Generation International Organisation for Standardisation. http://www.jcb.org.au/statistics.htm, last accessed 6/10/2000. 6. pp. 1013-1018.

- Chemical Engineering, Melbourne, Australia, 23rd-27th September.
 - 4, pp. 199-207.
- of Cleaner Production, Volume 1, Number 3-4, pp. 143-9.
- 141-57, 168-9, 180-6.
- Bioenergy, Volume 18, pp. 341-68.

References

ISO/TR 14049:2000, Environmental management – Life cycle assessment – Examples of application of ISO 14041 to goal and scope definition and inventory analysis,

ISO/TS 14048:2002, Environmental management – Life cycle assessment – Data documentation format. International Organisation for Standardisation.

Joint Coal Board (2000), Joint Coal Board Statistics, Joint Coal Board (NSW),

Kalantzi, O.I., Alcock, R.E., Johnston, P.A., Santillo, D., Stringer, R.L., Thomas, G.O. and Jones, K.C. (2001), 'The Global Distribution of PCBs and Organochlorine Pesticides in Butter', Environmental Science and Technology, Volume 35, Number

Katz, D. L. and Lee, R. L. (1990), Natural Gas Engineering, Production and Storage, McGraw-Hill (International), pp. 297-300.

Kealy, T., Favas, G., Tiu, C. and Chaffee, A.L. (2001), 'Lignite Upgrading by Mechanical-Thermal Expression (MTE)', Proceedings 6th World Congress of

Kennedy, D.J., Montgomery, D.C. and Quay, B.H. (1996), 'Data Quality - Stochastic Environmental Life Cycle Assessment Modelling', Int. J. LCA, Volume 1, Number

Keoleian, G. A. (1993), 'The Application of Life Cycle Assessment to Design', Journal

Kinhill Sterns (1987), Assessment of Environmental Impact of Electricity Transmission Line from Mount Piper to Marulan, Electricity Commission of N.S.W., February, pp.

Kniel, G. E., Delmarco, K. and Petrie, J. G. (1996), 'Life Cycle Assessment Applied to Process Design: Environmental and Economic Analysis and Optimisation of a Nitric Acid Plant'. Environmental Progress, Volume 14, Number 4, pp. 221-8.

Krotscheck, C., Koénig, F. and Obernberger, I. (2000), 'Ecological assessment of integrated bioenergy systems using the Sustainable Process Index', Biomass and

Kukla, G. T., Buffier, M. J., Balks, R. J. and Farnell, G. K. (1993), 'Open Pit Operation', in Hargraves, A. J. and Martin, C. H. (eds.), Australasian Coal Mining Practice, Monograph 12, Australasian Institute of Mining and Metallurgy, pp. 225-

References

37.

- Lavely, L. L. and Ferguson, A. W. (1996), 'Power Plant Emissions Control', in Black & Veatch, Power Plant Engineering, International Thomson Publishing, pp. 418-463.
- Leca, A., Adler, S. and Marin, O. R. (2002), 'The Electricity Capacities' Life Cycle Assessment for a Sustainable Development', International Conference on EcoBalance and Life Cycle Assessment in India, Mumbai, India, February 13-15.
- Lee, J. J., O'Callaghan, P. and Allen, D. (1995), 'Critical Review of Life Cycle Analysis and Assessment Techniques and their Application to Commercial Activities', Resources, Conservation and Recycling, Volume 13, pp. 37-56.
- Lindfors, L.-G., Christansen, K., Hoffman, L., Virtanen, Y., Juntilla, V., Hanssen, O.-J., Rønning, A., Ekvall, T. and Finnveden, G. (1995), Nordic Guidelines on Life-Cycle Assessment, Nord 1995:20, p 22.
- Lowe, A. (1998), Comparing CO_2 Emissions for Power Generation in NSW, The Australian Coal Review, April, pp. 37-42.
- Lowe, A. (2000), Pacific Power (New South Wales), Personal Communication, 11th October.
- Loy Yang Power (2000), Community Report 2000, Loy Yang Power, http://www. lovvangpower.com.au/site/env2000-rep-05.html, last accessed 18/7/2003.
- Loy Yang Power (1999), Environmental Report, Loy Yang Power, http://www. lovvangpower.com.au/envreport1999.htm.
- Lundie, S., Dimova, C., Bligh, M. and Huijbrechts, M. (2000), Australian Research into Life Cycle Impact Assessment, Paper presented at the 2nd National Conference on LCA, Melbourne, Victoria, 23-24 February,
- MacDonald, D, Donner, J. and A., Nikiforuk (1996), Full Fuel Cycle Emission Analysis for Electric Power Generation Options and Its Application in a Market-Based Economy (Alberta Department of Energy), Presented at the Third International Conference on Carbon Dioxide Removal, September 9-11, 1996, Boston, MA, U.S.A., http://www.resdev.gov.ab.ca/enviro/climate/ffc.htm, last accessed 15/9/2000.
- Mann, M. K., Spath, P. L. and Craig, K. R. (1996), 'Economic and Life Cycle Assessment of an Integrated Biomass Gasification Combined Cycle System', Proceedings of the 31st Intersociety Energy Conversion Engineering Conference, Volume 3, August, pp. 135-41.

Matson, R. J. and Carasso, M. (1999), 'Sustainability, Energy Technology, and Ethics',

Sustainability of Australia's Electricity Generation

Renewable Energy, Volume 16, pp. 1200-3. Matthes, F. C. and Timpe, C. (2000), Sustainability and the Future of European Electricity Policy, Heinrich Boell Foundation (Germany). Maurice, B., Frischknecht, R., Coelho-Schwirtz, V. and Hungerbühler, K. (2000), 'Uncertainty analysis in life cycle inventory. Application to the production of electricity with French coal power plants', Journal of Cleaner Production, Volume 8, pp. 95-108. May J.R. and Brennan D.J. (2003a), 'Application of Data Quality Assessment Methods

- Assessment, Volume 8, Number 4, pp. 215-225. Volume 81, Number 5, September.
- Australia, July 17-19 2002.
- Melbourne, Australia, 23rd-27th September.
- Australia), Draft, January, pg 22.
- - Aust.), pp. 49-55.
- Research, Volume 102, pp. 279-94.

References

to an LCA of Electricity Generation', International Journal of Life Cycle

May, J.R. and Brennan, D.J. (2003b), 'Life Cycle Assessment of Australian Fossil Energy Options', Transactions of The Institution of Chemical Engineers, Part B,

May, J.R. and Brennan, D.J. (2002), 'Application of Data Quality Assessment Methodologies to a Life Cycle Assessment of Electricity Generation', Proceedings 3rd Australian Conference on Life Cycle Assessment, Gold Coast, Queensland,

May, J.R. and Brennan, D.J. (2001), 'A Life Cycle Assessment Study of Electricity Production in Australia'. Proceedings 6th World Congress of Chemical Engineering,

McIntosh, M. (2003), Evaluation of the Cost and Performance of Advanced Brown Coal Technologies Expected to be Available by 2010 - Comparison with Alternative Fossil Fuels, Cooperative Research Centre for Clean Power from Lignite (Victoria,

McOuaid, J. (2000), 'The Application of Risk Control Concepts and Experience to Sustainable Development', Trans. IChemE, Volume 78, Part B, July 2000, pp 262-9. Meier, M. A. (1997), Eco-efficiency Evaluation of Waste Gas Purification Systems in the Chemical Industry, LCA Documents, Volume 2, Ecomed (Germany).

Middleton, K. A. (1977), The Economics of Capital Expenditure, Butterworths (Sydney,

Miettinen, P. and Hämäläinen, R. 1. (1997), 'How to Benefit from Decision Analysis in Environmental Life Cycle Assessment (LCA)', European Journal of Operational

- Minerals Council of Australia (2002), Fatalities in the Australian Minerals Industry, 1990-1991 to 1999-2000: A Retrospective Analysis, Minerals Council of Australia, December, http://www.minerals.org.au/downloads/pdf/fatality_retro_dec02.pdf, last accessed 17/3/2003.
- Ministerial Council on Energy (2002), Towards a Truly National and Efficient Energy Market, Ministerial Council on Energy, Commonwealth of Australia, http://www.en ergymarketreview.org/FinalReport20December2002.pdf, last accessed 24/7/2003.
- MMS (1998), Gulf of Mexico OCS Oil and Gas Lease Sales 171, 174, 177, and 180, Western Planning Area, Final Environmental Impact Statement, Gulf of Mexico OCS Region, Minerals Management Service (U.S.A.), MMS 98-0008, pp. IV-13 to IV-18.
- Mohin, T. J. (1994), 'Life Cycle Assessment', in The Environmental Management Handbook, Taylor, B., Pollach, S., Hutchinsin, C. and Tapper, R., eds., Pittman, G. B., Chapter 18.
- Müller-Wenk, R. (1998), Depletion of Abiotic Resources Weighted on Base of 'Virtual' Impacts of Lower Grade Deposits Used in Future, IWÖ - Diskussionsbeitrag Nr. 57, March, electronic resource, http://www.iwoe.unisg.ch/files/db57.pdf, Last accessed 7/4/2000.
- Nakč enovč, N., Grübler, A. and McDonald, A. (eds.) (1998), Global Energy Perspectives, International Institute for Applied Systems Analysis, University Press, Cambridge.
- Nash, J. & Stoughton, M. D. (1994), 'Learning to Live with Life Cycle Assessment', Environmental Science and Technology, Volume 28, Number 5, pp. 236A-237A.
- National Occupational Health & Safety Commission (2002), NOHSC Online Statistics Interactive (NOSI), National Occupational Health & Safety Commission, http://nohsc.info.au.com/, last accessed 20/3/2003.
- Natural Resources and Environment (2001), Statistical Review 2000-2001, Department of Natural Resources and Environment (Victoria), http://www.nre.vic.gov.au/ CA25675500149D09/BCView/926CE032BE2F4885CA256B91001B4C62?OpenDo cument, last accessed 20/3/2003.
- Natural Resources and Environment Committee (Victoria) (1987), Inquiry into electricity supply and demand beyond the mid-1990's: Preliminary report and Draft Recommendations, Natural Resources and Environment Committee, Parliament of

Sustainability of Australia's Electricity Generation

Victoria, December, NGGIC (1999), National Greenhouse Gas Inventory 1997 with Methodology Supplements, National Greenhouse Gas Inventory Committee, Australian Greenhouse Office. NGGIC (1998a), ENERGY: Workbook for Fuel Combustion Activities (Stationary Sources), Australian Greenhouse Office, pp. 87-96. NGGIC (1998b), ENERGY: Workbook for Fugitive Fuel Emissions (Fuel Production, Transmission, Storage and Distribution), National Greenhouse Gas Inventory Committee, Australian Greenhouse Office, pp. 9-25. NPI (1999a), Emission Estimation Technique Manual for Mining, National Pollutant Inventory (NPI) (Australia), March, pp. 44. Nuclear Energy Agency (1998), Projected Costs of Generating Electricity: Update 1998, Organisation of Economic Cooperation and Development. OECD (1985), Environmental Effects of Electricity Generation, Organisation for Economic Co-operation and Development (OECD), pp. 9. Office of Technology Assessment (1994), Studies of the Environmental Costs of Electricity, Office of Technology Assessment (USA), September, OTA-BP-ETI-134. Ogawa, Y, and Yoon, S. Y. (1998), 'A Comparative Analysis of Greenhouse Effects of Fossil Fuels Examined from the Global Viewpoint from Mining to Combustion', Energy in Japan, Number 153, September, pp.10-27. Overcash, M. (2001), Sustainability - Concepts For Research In Life Cycle And Chemical Engineering, Paper Presented at the 6th World Congress of Chemical Engineering, Melbourne, Australia, 23rd-27th September. Page, S.J., Organiscak, J.A. and Lichtman, K. (1997), 'The cost of respirable coal mine dust: an analysis based on new black lung claims', Applied Occupational and Environmental Hygiene, Volume 12, Number 12, pp. 832-9. Pezzoli, K. (1997), 'Sustainable Development: A Transdisciplinary Overview of the Literature', Journal of Environmental Planning and Management, Volume 40, Number 5, pp. 549-74. Bibliography contained in: Pezzoli, K. (1997), 'Sustainable Development Literature: A Transdisciplinary Bibliography', Journal of Environmental Planning and Management, Volume 40, Number 5, pp. 575-601. Riadhi-Belkaoui, A. (1999), Value Added Reporting and Research, State of the Art, Ouorum (Westport, U.S.A), Chapter 1.

- Richards, A. (1998), 'Connecting Performance and Competitiveness with Finance: A Study of the Chemical Industry', Chapter 14 in Arora, A., Landau, R. and Rosenberg, N., Chemicals and Long-Term Economic Growth, Insights from the Chemical Industry, John Wiley and Sons. (New York, U.S.A.).
- Rogner, H. H. and Khan, A. (1998), 'Comparing Energy Options: Progress Report on the Inter-Agency DECADES Project', IAEA Bulletin, Quarterly Journal of the International Atomic Energy Agency, pp. 2-6.
- Ross, S., Evans, D. and Webber, M. (2002), 'How LCA Studies Deal with Uncertainty', Int. J. LCA, Volume 7, Number 1, pp. 47-52.
- Rousseaux, P., Labouze, E., Suh, Y.-J., Blanc, I., Gaveglia, V., and Navarro, A. (2001), 'An overall Assessment of Life Cycle Inventory Quality: Application to the Production of Polyethylene Bottles', Int. J. LCA, Volume 6, Number 5, pp. 299-306.
- Royal Dutch/Shell (2001), Energy Needs, Choices and Possibilities Scenarios to 2050, Royal Dutch/Shell, http://www2.shell.com/home/media-en/downloads/scenarios.pdf, last accessed 13/1/2003.
- Sadar, M. H. (1996), Environmental Impact Assessment, 2nd ed. Carleton University Press, pgs. vii, 2-3, 6-7.
- Schaap, H. (2000), 'Electricity Supply and Sustainability: Can Renewables make a Difference?', Australian Academy of Technological Services and Engineering Symposium, November 2000.
- SECV (1993), Annual Report year ending 30th June 1992, State Electricity Commission of Victoria (SECV), pg. 142.
- SECV (1992), The SECV and the Greenhouse Effect, State Electricity Commission of Victoria, Discussion Paper Number 2 - April 1992.
- SECV (1976), Report on Proposed Extension to the State Generating System, Loy Yang Project, State Electricity Commission of Victoria (SECV), pg. 2.
- SECV (1973), Newport Power Station, Environmental Report, State Electricity Commission of Victoria (SECV), March,
- SETAC (1998), Evolution and Development of the Conceptual Framework and Methodology of Life Cycle Impact Assessment, Addendum to Life-Cycle Impact Assessment: The State-of-the-Art, 2nd ed., edited by Barnthouse, Fava, Humphreys, Hunt, Laibson, Noesen, Norris, Owens, Todd, Vigon, Weitz, and Young, Report of the SETAC LCA impact assessment workgroup, Pensacola, FL, SETAC, January,

Sustainability of Australia's Electricity Generation

- http://www.setac.org/files/addendum.pdf. http://www.pre.nl/simapro/default.htm, last accessed 16/7/2003. International Thomson Publishing, pp. 659-688. Press, pp. 616-8.
- Butterworth-Heinemann Ltd (Oxford), pp. 105-111. 1232-8.
- Cleaner Production, Volume 5, Number 4, pp. 255-62.
- (Toronto, Canada), pp. 20-1.
- VGB/EPRI Conference, Wiesbaden, Germany, May.
- February, p. 32.
- Materials and Processes, Volume 147, Number 2, pp. 37-40.

References

Simapro (2001), Simapro 5.1, PRé Consultants, Amersfoort, The Netherlands,

Simapro (1999), Simapro 4, PRé Consultants, Amersfoort, The Netherlands.

Smith, J. M. (1996), 'Gas Turbines', in Black & Veatch, Power Plant Engineering,

Sørensen, B. (1992), 'What is Life-Cycle Analysis?', in OECD/IEA, Expert Workshop on Life-Cycle Analysis of Energy Systems, Methods and Experience, Proceedings, Organisation for Economic Co-operation and Development (OECD) and the International Energy Agency (IEA), Paris, France, 21st-22nd May 1992, pp. 21-53.

Sørensen, B. (1979), 'Energy Analysis', in Sørensen, B., Renewable Energy, Academic

Spath, P. L., Mann, M. K. and Kerr, D. R. (1999), Life Cycle Assessment of Coal-fired Power Production, NREL (National Renewable Energy Laboratory) (U.S.A.), NREL/TP-570-25119, available from http://www.doe.gov/bridge/home.html.

Speight, J. G. (1993), Gas Processing: Environmental Aspects and Methods,

Starr, C. (1969), 'Social Benefit versus Technological Risk', Science, Volume 165, pp.

Steen, B. (1997), 'On Uncertainty and Sensitivity of LCA-based Priority Setting', J. of

Stern, J. M., Shiely, J. S. and Ross, I. (2001), The EVA Challenge, John Wiley and Sons

Strauss, K., Bergins, C. and Böhlmann, M. (2001), 'Beneficial effects of the integration of a MTE plant into a brown coal fired power station', Lignites and Low Rank coal: Operational and Environmental Issues in a Competitive Climate, Proceedings

Stromburg, L., Haglind, I., Jacobson, B., Ekvall, T., Eriksson, E., Kärnä, A. and Pajula, T. (1997), Guidelines on Life Cycle Analysis of Pulp and Paper, NordPap DP2/30,

Sullivan, J. L. and Young, S. B. (1995), 'Life Cycle Analysis/Assessment', Advanced

Tillman, A. M., Ekvall, T., Baumann, H. and Rydberg, T. (1994), Choice of System

<u>References</u>

Boundaries in Life Cycle Assessment, Journal of Cleaner Production, Volume 2, Number 1, pp. 21-9.

- Treloar, G. (2000), 'Life Cycle Assessment of Construction', in Pathways to Eco-Efficiency, Paper presented at the 2nd National Conference on LCA, Melbourne, Victoria, 23-24 February,
- Udo de Haes, H. A. (2000), Status, Developments, and Future Challenges for Life Cycle Impact Assessment, Paper presented at the 2nd National Conference on LCA, Melbourne, Victoria, 23-24 February,
- Udo de Haes, H. A., ed. (1996), Towards a Methodology for Life Cycle Impact Assessment, Europe Workgroup on Life Cycle Impact Assessment, Part I, Chapter 6, February.
- Udo de Haes, H. A. (1993), 'Applications of Life Cycle Assessment: Expectations, Drawbacks and Perspectives', Journal of Cleaner Production, Volume 1, Number 3-4, pp. 131-7.
- Udo de Haes, H. A., Jolliet, O., Finnveden, G., Hauschild, M., Krewitt, W. and Müller-Wenk, R. (1999), 'Best Available Practice Regarding Impact Categories and Category Indicators in Life Cycle Assessment', International Journal of Life Cycle Assessment, Volume 4, Number 2, pp. 66-74 and Volume 4, Number 3, pp. 167-174.
- UKOOA (1999), 1999 Environment Report, United Kingdom Offshore Operators Association, http://www.ukooa.co.uk/, last accessed 6/11/2001.
- Umberto (2003), Umberto 4.1, ifu, Hamburg (Germany), http://www.umberto.de/ english/index.htm, last accessed 16/7/2003.
- UNCED (1992), Agenda 21, United Nations Conference on Environment and Development (UNCED), Rio de Janeiro, June, http://www.un.org/esa/sustdey/ agenda 21text.htm, last accessed 9/10/2001.
- UNEP (1996), Life Cycle Assessment: What it is and How to do it, United Nations Environment Programme (Paris, France), pp. 14, 37, 52-67, 73.
- United Nations (2002), Report of the World Summit on Sustainable Development, Johannesburg, South Africa, 26 August-4 September 2002, http://www.johannesburg summit.org/, last accessed 13/1/2003.
- Upham, P. (2000), 'An assessment of The Natural Step theory of sustainability', Journal of Cleaner Production, Volume 8, Number 6, pp. 445-54.

Van de Vate, J. F. (1997), 'Comparisons of energy sources in terms of their full energy

Sustainability of Australia's Electricity Generation References chain emission factors of greenhouse gases', Energy Policy, Volume 25, Number 1, pp. 1-6. Van Groenendaala, W.J.H. and Kleijnen, J.P.C. (1997), 'On the assessment of economic risk: factorial design versus Monte Carlo methods', Reliability Engineering &

System Safety, Volume 57, Number 1, July, pp. 91-102. Vattenfall (1999), Vattenfalls' Life Cycle Studies of Electricity, Vattenfall (Sweden), p. 4.

- 49.
- Development, Kluwer Academic Publishers, pp. 1-6.
- 3, pp. 135-40.
- the RIIA), p. 267.
- (Abingdon, UK).
- Development (WCED), Oxford University Press, 1987.
- February.
- Cycle Inventory', Int. J. LCA, Volume 3, Number 5, pp. 259-65.
- Production, Volume 4, Numbers 3-4, pp. 167-74.

Veleva, V. and Ellenbecker, M. (2001), 'Indicators of sustainable production: framework and methodology', Journal of Cleaner Production, Volume 9, pp. 519-

Verbuggen, H. and Kuik, O. (1991), 'Indicators of Sustainable Development: an overview', in Kuik, O. and Verbuggen, H., In Search of Indicators of Sustainable

Vigon, B. W. and Jensen, A. A. (1995), 'Life Cycle Assessment: Data Quality and Databases Practitioner Survey', Journal of Cleaner Production, Volume 3, Number

Wallace, D. (1995), Strategies for Industry and Environmental Policy, Earthscan (for

Watkiss, P. and Forster, D. (2000), A Study for British Energy, AEA Technology

WCED (1987), Our Common Future, World Commission on Environment and

Weidema, B. P. (2000), LCA Developments for Promoting Sustainability, Paper presented at the 2nd National Conference on LCA, Melbourne, Victoria, 23-24

Weidema, B. P. (1998), 'Multi-User Test of the Data Quality Matrix for Product Life

Weidema, B. P. and Wesnæs, M. S. (1996), 'Data quality management for life cycle inventories - an example of using data quality indicators', Journal of Cleaner

WHO (2001), 'Electromagnetic Fields and Public Health: Extremely Low Frequency Fields and Cancer', Fact Sheets, World Health Organisation, Fact Sheet WHO/263, October, http://www.who.int/inf-fs/en/fact263.html, last accessed 22/7/2003.

References

- WHO (1999), 'Dioxins and Their Effects on Human Health', Fact Sheets, World Health Organisation, Fact Sheet WHO/225, June, <u>http://www.who.int/inf-fs/en/fact225.html</u>, last accessed 22/7/2003.
- Wood, E.G. (1978), Added Value the Key to Prosperity, Business Books (London, UK).
- Woodruff, E. B., Lammers, H. B. and Lammers, T. F. (1998), Steam Plant Operation, 7th ed., McGraw-Hill, p. 414-5.
- Woodside Energy Ltd. (1998a), Lifecycle Analysis of Emissions Associated with Power Generation Using Various Fuel Sources, synopsis of two previous CSIRO reports, July.
- Woodside Energy Ltd. (1998b), Additional North West Shelf Gas Venture Liquefied Natural Gas (LNG Facilities) Project, Public Environmental Review/Report, November.
- Woodward-Clyde (1999), Environmental Effects Statement-Maryvale Coal Field Development, Prepared for Yallourn Energy Pty Ltd, January, Section 7.
- World Wide Fund for Nature (2002), *Living Planet Report 2002*, World Wide Fund for Nature (Cambridge, UK).
- Wrisberg, M. N. (1997), 'A semi-quantitative approach for assessing data quality in LCA', Proceedings 7th Annual Meeting of SETAC-Europe, Amsterdam, April 6-10, 1997.

Yallourn Energy (1999), Environmental Report, Yallourn Energy.

Yasukawa, S., Tadokoro, Y. and Kajiyama, T. (1992), 'Life Cycle CO₂ Emission from Nuclear Power Reactor and Fuel Cycle System', in OECD/IEA, *Expert Workshop on Life-Cycle Analysis of Energy Systems, Methods and Experience, Proceedings*, Organisation for Economic Co-operation and Development (OECD) and the International Energy Agency (IEA), Paris, France, 21st-22nd May 1992, pp. 151-60.

APPENDICES

- ABS (2003a), 5606.0 Australian National Accounts: National Income, Expenditure and Product, Australian Bureau of Statistics, Table 1.
- ABS (2003b), 5625.0 Private New Capital Expenditure and Expected Expenditure, Australia, Australian Bureau of Statistics, Table 1A.

ABS (2003c), 5676.0 - Business Indicators, Australia, Australian Bureau of Statistics,

Tables 1, 6, 9, 11 and 13. ABS (2003d), 6203.0 - 6202.0 Labour Force, Australia, Australian Bureau of Statistics, Table 1. ABS (2002a), 6401.0 Consumer Price Index, March Quarter 2002, Australian Bureau of Statistics. ABS (2002b), Australia Now-A Statistical Profile, Prices, CPI - Long-term price series, Australian Bureau of Statistics. ABS (2001), Energy and Greenhouse Gas Emissions Accounts, Australia, Australian Bureau of Statistics. ABS (2000), Energy and Greenhouse Gas Emissions Accounts for Australia, 1992-3 to 1997-8, Australian Bureau of Statistics, Canberra, Cat. No. 4604.0, ABS (1998a), Waste Management Industry 1996-7, Australian Bureau of Statistics. Canberra, Cat. No. 8698.0. ABS (1998b), Mineral Account, Australia, Australian Bureau of Statistics, Canberra, Cat. No. 4608.0. ABS (1998c), Mineral Account, Volume Change Tables, Australia, Australian Bureau of Statistics, Canberra, Cat. No. 4608.0.40.001. ACCC (1986), The Australian Coal Industry Cost Study 1985, Australian Coal Consultative Council, National Research Group, Report by Working Party No. 7, March. ACIL (1998), Socio-Economic Effects of the Minerva Gas Field Development on the Port Campbell Region of Victoria, December, in BHP (1999), Social, Economic and Statutory Planning, Minerva Gas Field Development, Environmental Impact Statement, Environmental Effects Statement (Volume 4), Technical Reports, pg. 8. ADB (2002), Loan 1357-INO, Gas Transmission And Distribution Project, Request By The Borrower For Change In Project Scope, Asian Development Bank, April, http://www.adb.org/Documents/Others/Change_Proj_Scope/chapters.pdf, last accessed 5/8/2002. AGA (2001), Gas Statistics Australia 2001, Australian Gas Association (Canberra, Australia). AGA (2000a), Gas Statistics Australia 2000, Australian Gas Association (Canberra, Australia). AGA (2000b), Personal Communication, George Webb, Research Manager, Australian

References

Sustainability of Australia's Electricity Generation

Gas Association, 6/7/2000.

- AGA (1997), Greenhouse Challenge Facilitative Agreement, Australian Gas Association, Greenhouse Challenge program, November.
- AISE (2002), New Coal Terminal Planned for Rotterdam, Steel News Headlines 10/3/02, Association of Iron and Steel Engineers, http://www.steelnews.com/ world/oct02/world_oct08.htm, last accessed 17/2/2003.
- (2001),Alstom Alstom Power: open cvcle. Alstom Power. http://www.power.alstom.com/servlet/ContentServer?pagename=OpenMarket/Xceler ate/View&inifile=futuretense.ini&c=Page&cid=978978472538. last accessed 30/11/2001.
- AMSA (2002), OH&S (MI) Act Incident Reports for Period 1997/98-2001/02, Australian Maritime Safety Authority, http://www.amsa.gov.au/SP/OHS /OHSincrep.pdf, last accessed 20/3/2003.
- Anonymous (2002a), Progress Energy completes power plant in Desoto County, Fla., Power Engineering, June 3rd.
- Anonymous (2002b), Shell Will Develop LNG Terminal in California, Atlantic Oil and Gas Works Online, 28/3/2002, http://www.oilworks.com/New/ 1032502.html, last accessed 23/9/2002.
- Anonymous (2002c), Clough Awarded Sawan Gas Project, Rigzone Industry News, http://www.rigzone.com/news/default.asp, last accessed 12/7/2002.
- Anonymous (2001a), Peabody breaks ground on new mine, Coal Age, Volume 106, Number 5, May, pg. 22.
- Anonymous (1998a), AGL, Petronas Named Preferred Developer of PNG to Old Gas *Pipeline*, The Australian Pipeliner, July, pp. 20-1.
- Anonymous (1998b), Stepping on the Gas in Queensland, The Australian Pipeliner, July, pp. 22-3.
- APPEA (2001a), Exploration Statistics 1983-2001, APPEA, http://www.appea.com .au/exploration and production/pdfs/EXPLTAB.pdf, last accessed 30/11/2001.
- APPEA (2001b), Health, Safety, Environment and Social Performance Report 2001, Australian Petroleum Production and Exploration Association Limited, http://www.appea.com.au, last accessed 12/3/2003.
- APPEA (1999), APPEA Greenhouse Challenge Document, Australian Petroleum Production and Exploration Association, www.appea.com.au, last accessed

4/12/2000.

APPEA (1998), Production Statistics Calendar Year-1997, Australian Petroleum Production and Exploration Association, http://www.appea.com.au/industrystats/ indstats pdfs/97YearlyProduction.pdf, last accessed 16/2/2001. Aylward, A. and Sutherland G. (eds.) (2000), Australian Coal Year Book, Australia's Mining Series, pp. 16-97. BG Group (2002), City Presentation, BG Group, November, http://www.investis .com/bg/presentations/downloads/2002-11-14b.pdf, last accessed 17/2/2003. BHP (1999a), Appin Collierv Environmental Report 1999, BHP Coal. BHP (1999b), Cordeaux Colliery Environmental Report 1999, BHP Coal. BHP (1999c), Tower Colliery Environmental Report 1999, BHP Coal. BHP (1999d), West Cliff Collierv Environmental Report 1999, BHP Coal. BHP (1999e), Gregory/Crinum Mine Environmental Report 1999, BHP Coal. BHP (1999f), Goonyella Riverside Mine Environmental Report 1999, BHP Coal. BHP (1999g), Norwich Park Mine Environmental Report 1999, BHP Coal. BHP (1999h), Peak Downs Mine Environmental Report 1999, BHP Coal. BHP (1999i), South Walker Mine Er ironmental Report 1999, BHP Coal. BHP (1998), Blackwater Mine Environmental Report 1998, BHP Coal. BHP Minerals Technology (2001), ACARP-Coal in a Sustainable Society, Australian Coal Association Research Program, ACARP Project C9058, BHP Research, http://www.sustainabletechnology.com.au/, last accessed 26/5/2001. BHP Research (1999), LCA of Steel and Electricity Production, ACARP Project C8049, Australian Coal Association Research Program, BHP Research, December, http://www.sustainabletechnology.com.au/frames.html, last accessed on 18/9/2000. BHPBilliton (2001a), A New Day Begins, BHP Billiton Annual Report, BHP Billiton PLC, http://www.bhpbilliton.com/bb/investorCentre/annualReports.jsp, last accessed 29/7/2002. BHPBilliton (2001b), Petroleum Business Briefing, BHPBilliton Limited, November 2001, Sydney, http://www.bhpbilliton.com/bbContentRepository/Presentations/ PetroleumBusinessBriefing-Part1.pdf, last accessed 18/2/2003. BIE (1995), Rail Freight 1995, International Benchmarking, Bureau of Industry Economics, Report 95/22, December.

316

References

BIE (1992), International Performance Indicators - Road Freight, Bureau of Industry

Economics, Australian Government Publishing Service (Canberra).

- Bourgeois, M. G. (1990). How can LNG transportation cost be reduced using membrane carriers, EUROGAS '90, proceedings from the European Applied Research Conference on Natural Gas, Trondheim, Norway, May 28-30, 1990, pp. 667-692.
- Brennan, D. J. (1990), Process Industry Economics, An Australian Perspective, Longman Cheshire, pg.127.
- Brennan, D. J., Anderson, W. J. and Salusinszky, A. L. (1980), Cooper Basin Liquids -A Market at Sydney, Eighth Australian Chemical Engineering Conference, Melbourne, Australia, 24th-27th August.
- BTCE (1998), Characteristics and Cost Structure of the Bulk Shipping Fleet, Bureau of Transport and Communications Economics (Canberra, Aust.), Occasional Paper 93, Table 6.1.
- Canada-Newfoundland Offshore Petroleum Board (2001), Hibernia Field Total Production History, Resource Information, Canada-Newfoundland Offshore Petroleum Board, last accessed October 30, 2001, http://www.cnopb .nfnet.com/resource/hproduct.htm.
- Cedigaz (2000), Cedigaz News Report-March 2001, Cedigaz (France), http://consortium.ifp.fr/CE/CE400GK2.html, last accessed 7/8/2002.
- CFAC (III.) (1977), Comparative Transportation Costs: An Economic and Engineering Analysis of Truck, Belt, Rail, Barge and Coal Slurry and Pneumatic Pipelines, Volume 1. Summary and Conclusions, Centre for Advanced Computing (Illinois University), Prepared for Bureau of Mines (U.S.A.), U.S. Department of Commerce, PB-274 379, August 1977.
- Chadwick, M. J., Highton, N. H. and Lindman, N. (eds.) (1987), Environmental Impacts of Coal Mining and Utilisation, Beijer Institute, United Nations Environment Programme, USSR Academy of Sciences, and the Swedish National Energy Administration, Pergamon Press, pp. 30-31.
- Clark Public Utilities (2001), River Road Technical Facts, Clark Public Utilities, http://www.clarkpublicutilities.com/rrgp_facts.html, last accessed 6/12/2001.
- Coal Services (2002), Summary of New South Wales Black Coal Statistics, Coal Services Pty Ltd., http://www.icb.org.au/documents/stats/statistics.pdf, last accessed [1/3/2003.

Sustainability of Australia's Electricity Generation

CSEnergy (2000), Annual Report 2000, CSEnergy Ltd. CSEnergy (2001), 'Callide C puts Queensland first', Media Release, CSEnergy Ltd. Davis, L. B. and Black, S. H. (2000), Dry Low NOx Combustion Systems for GE Heavy-Duty Gas Turbines, GE Power Systems, GER-3568G. Day, D.G. (1986), Water and Coal, industry, environment and institutions in the Hunter Valley NSW, Centre for Resource and Environmental Studies (CRES), Australian National University, CRES Monograph 14, pp. 30-31. Dedman, R. (2000), Water Production, Personal Communication, Senior Production Geologist of the Western Australian Department of Mines and Energy. Delta Electricity (1999a), Delta Electricity, Submission to New South Wales Treasury. Delta Electricity (1999b), Delta Electricity Environment Report 1996-1999, Delta Electricity.

- ossm0008.txt, last accessed 17/3/2003. January.
- Lowpress pipe/alpc_CHAPTER3.pdf, last accessed 17/3/2003.
- Number 2, pp. 23-34.
- accessed 5/12/2001.
- Industry, Technology, and Commerce, June and December. and Energy (Queensland), pp. 8-9, 16-7, 48-51, 58-9.

DOL (U.S.A.) (1996), Survey of Occupational Injuries and Illnesses, U.S. Department of Labor, Bureau of Labor Statistics, April, http://www.bls.gov/iif/oshwc/osh/os/

DMR (NSW) (2000), 2000 New South Wales Coal Industry Profile, Incorporating Joint Coal Board Statistical Supplement, NSW Department of Mineral Resources (DMR),

DFFP (U.S.A.) (1997), Background Pipeline Risk Data, Chapter 3 in An Assessment of Low-Pressure Crude Oil Pipelines and Gathering Lines, Office of the State Fire Marshal, California Department of Forestry and Fire Protection, Report to the California State Legislature, April, http://osfm.fire.ca.gov/pdf/pipeline/

Dharmappa, H. B., Wingrove, K., Sivakumar, M. and Singh, R. (2000), Wastewater and stormwater minimisation in a coal mine, Journal of Cleaner Production, Volume 8,

DIT (2000), Pelican Point ... Powering Ahead, Directions for South Australia, Media Release, 26/12/2000, http://www.directions-sa.net/STORIES /pelican.htm, last

DITC (1980-1986), Major Manufacturing and Mining Investments, Department of

DME (1998), 47th Queensland Coal Industry Review 1997-1998, Department of Mines

- DME (2000), 49th Oueensland Coal Industry Review 1999-2000, Department of Mines and Energy (Queensland), Tables 4 and 23.
- DoE (U.K.) (1988), Prospects for the use of Advanced Coal Based Power Generation Plant in the United Kingdom, Department of Energy (U.K.), Energy Paper Number 56, July.
- DoE (U.K.) (1980), A North Sea gas gathering system, Department of Energy (U.K.), Energy Paper Number 44.
- DOE (U.S.A.) (2000a), Destec Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-003, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/destx3x_pdf, last accessed 13/3/2002.
- DOE (U.S.A.) (2000b), Shell Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-002, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/shell3x_pdi, last accessed 13/3/2002.
- DOE (U.S.A.) (2000c), Texaco Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-001, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/texac3y_.pdf, last_accessed 13/3/2002).
- DOE (U.S.A.) (2000d), KRW Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-005, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/krw3x_20.pdf, last accessed 13/3/2002.
- DOE (U.S.A.) (2000e), British Gas/Lurgi Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-004, revised June 2000, http://www.netl.doe.gov/coalpower/gasification/system/bgl3x_20.pdf, last accessed 13/3/2002.
- DOE (U.S.A.) (2000f), Transport Gasifier IGCC Base Cases, Process Engineering Division, NETL, Department of Energy, PED-IGCC-98-006, revised June 2000, http://www.netl.doe.goy/coalpower/gasification/system/trnspt3x .pdf, last accessed 13/3/2002.
- DOE (U.S.A.) (1999), Market-Based Advanced Coal Power Systems Final Report, U.S. Department of Energy, Office of Fossil Energy, May, DOE/FE-0400.

Sustainability of Australia's Electricity Generation

DOI (U.S.A.) (1993), A Cost Comparison of Selected Coal Mines from Australia. Canada, Colombia, South Africa, and the United States, United States Department of the Interior, Bureau of Mines, Special Publication, August. Dragun, A. K., Gleeson, V. and Musgrave, W. F. (1986), The economics of water use in the Hunter Region, Centre for Resource and Environmental Studies (CRES), Australian National University, CRES Monograph 13, pp. 11-7. Drewry Shipping Consultants (1995), Coal shipping and costs: sea transport logistics, traffic patterns, trade prospects, infrastructure constraints and freight costs, Drewry Shipping Consultants (London, U.K.), Chapter 5. Driver, R. A. and Ritchey, I. (1998), Issues Arising from the Developing Role of Gaseous Fuels, 17th World Energy Council Congress, Houston, USA - 13-18 http://www.worldenergy.org/wec-geis/publications/default/ September 1998. tech papers/17th congress/2 2 15.asp, last accessed 5/8/2002. DTI (UK.) (2000), Digest of United Kingdom Energy Statistics 2000, Department of Industry and Trade, Appendix B Energy and the Environment (http://www.dti.gov.uk/epa/annexb.pdf), and Appendix C UK oil and gas resources (http://www.dti.gov.uk/epa/ annexc.pdf), last accessed 9/11/2001. Duke Energy (2000a), Duke Energy International Completes \$297 Million Australian Release. August 16, from http://www.duke-Pipeline, News energy.com/decorp/content/newscenter/newsreleases/2000/aug/2000081601.html, last accessed 21/02/2002. Duke Energy (2000b). Duke Energy International announces customers for Tasmanian Gas Pipeline, News Release, November 8, from http://www.dukeenergy.com/decom/content/newscenter/newsreleases/2000/nov/2000110801.htmi, last accessed 21/02/2002. Easton E. (1988), Rail Freight Charges and Costs, in Ramsden, H. (ed.) (1988), Rail Freight Rates and the Minerals Sector, Proceedings of a conference on 20 October 1988 in Sydney, Centre of Policy Studies, Monash University. ECNSW (1991b), New and Alternative Technologies for Electricity Generation, the Electricity Commission of New South Wales, July. ECNSW (1991a), Annual Report 1991, Electricity Commission of New South Wales. ECNSW (1982), Annual Report 1982, Electricity Commission of New South Wales, pg. 61.

- ECNSW (1980), Mount Piper Power Station Environmental Impact Assessment, the Electricity Commission of New South Wales, August, Volume 1, pp. 23, Figure 9.
- Ecogen Energy (1999), Human Resources, Ecogen Energy (Victoria), http://www.ecogenenergy.com.au/body_jhuman.html, last accessed 5/9/2000.
- Eddy, E. G. (1998), The LNG Transportation Link, Chapter 5 in Greenwald, G. B. (ed.) (1998), Liquefied Natural Gas: Developing and Financing International Energy Projects, Kluwer Law International (London, UK).
- EIA (U.S.A.) (2002), Oil and Gas Lease Equipment and Operating Costs 1986 Through 2000, Energy Information Administration, Department of Energy (U.S.A.), January, http://ftp.eia.doe.gov/pub/oil gas/natural gas/data publications/

cost indices equipment production/current/c i.html, last accessed 30/3/2002.

- EIA (U.S.A.) (2000), Coal Industry Annual 2000, Energy Information Administration, Department of Energy (U.S.A.), DOE/EIA-0584(2000), ftp://ftp.eia.doe.gov/pub/ pdf/coal.nuclear/05842000.pdf, last accessed 31/7/2002.
- EIA (1997a), Financial Statistics of Major U.S. Investor-Owned Electric Utilities 1996, Energy Information Administration, Office of Coal, Nuclear, Electric and Alternate Fuels. U.S. Department of Energy, http://www.eia.doe.gov/cneaf/ electricity/invest/invest.pdf, last accessed 8/2/2002.
- EIA (1997b), Issues in Midterm Analysis and Forecasting 1997, Energy Information Administration, Department of Energy (U.S.A.), DOE/EIA-0607(97), http://tonto.eia.doe.gov/FTPROOT/forecasting/060797.pdf, last accessed 18/2/2003.
- EIA (1996), Financial Statistics of Major U.S. Investor-Owned Electric Utilities 1995, Energy Information Administration, Office of Coal, Nuclear, Electric and Alternate Fuels, U.S. Department of Energy, December, DOE/EIA-0437(95)/1, http://tonto.eia.doe.gov/FTPROOT/electricity/0437951.pdf, last accessed 8/2/2002.
- ElectraNet (SA) (2003), Company Profile, and Workplace Safety, ElectraNet Pty Limited, http://www.electranet.com.au/company/home.html, last accessed 17/3/2003.
- ElectraNet (SA) (2002a), SA Transmission Revenue Cap Public Forum, ElectraNet, 4 October, http://www.accc.gov.au/elecaric/regulation/sa_tn/pdf/ElectraNet.pdf, last accessed 19/2/2003.
- ElectraNet (SA) (2002b), A Transmission Network to Power South Australia, ElectraNet, November.

EnerSea Transport (2002), Voltrans™ Advantage: Compelling Economics, EnerSea

Sustainability of Australia's Electricity Generation

Transport, http://www.enerseatransport.com/volands.htm, last accessed 29/1/2003. Enoch, M. and Frey, M. (2002), Pipeline Projects Helpful to Public Gas, Presented at the APGA Gas Supply Conference, Fort Walton Beach, FL May 7, 2002. EPA (U.S.A.) (1999), EPA Office of Compliance Sector Notebook Project: Profile of the Oil and Gas Extraction Industry, Office of Compliance, Environmental Protection Agency (U.S.), October, pp. 16-27, 52-65. EPA (U.S.A.) (1994), ARCS Remediation Guidance Document, EPA 905-B94-003, Chicago, Ill, Great Lakes National Program Office, http://www.epa.gov/glnpo/ arcs/EPA-905-B94-003/EPA-905-B94-003-toc.html, last accessed 8/7/2001. Epic Energy (2003), About Epic Energy, Epic Energy (Australia), http://www.epic.com.au, last accessed 17/3/2003. Epic Energy (2001), Epic Energy Pipeline Delivers Timor Sea Gas, Media Release, 9th May, from http://www.epic.com.au/contentpages/news 09 May.html, last accessed 22/02/2002. EPS (Canada) (1989), Environmental Code of Practice for Treatment and Disposal of Waste Discharges from Offshore Oil and Gas Operations, Environmental Protection Service, Environment Canada, EPS 1/PN/2. ERDC (1992), Advanced Generation Options for the Australian Electricity Supply Industry and their Impact on Greenhouse Emissions, Phase 2, Energy Research and Development Corporation and Electricity Supply Association of Australia, ERDC 130, January, Appendix 1. Ernst and Whinney (1982), Committee of Review into Loy Yang Costs, Report on O&M Costs, Overheads and Capital Costs, 9th December. Errington, D. and Kirby, M. (1993), Australian Coal Industry Study, McIntosh and Company, Australian Research, May. ESAA (1999), Electricity Australia 1999, Electricity Supply Association of Australia (ESAA), pp. 46. Estonian Railways (2002), Coal terminal to bring 5 mln tons of coal to Estonia's Muuga Port, Estonian Railways News, 1st October, <u>http://www.evr.ee/pages.php/</u> 020101.395, last accessed 17/2/2003. ETSA Corporation (1997), Annual Report-1997, ETSA Corporation, Adelaide, pp. 73-74. ExxonMobil (2002), Annual Report 2001, ExxonMobil Corporation, pp. 56-63,

References

http://www2.exxonmobil.com/corporate/newsroom/publications/c_fo_01/pdfs/upstre am fin.pdf, last accessed 18/2/2003.

- Flower, A. R. (1998), 'LNG Project Feasibility', Chapter 3 in Greenwald, G. B. (ed.), Liquefied Natural Gas Developing and Financing International Energy Projects. Kluwer Law International (London, UK).
- Fong, C. (Technical Director) (2002), Annual Report of the Technical Regulator 2001/2002, Technical Director (of the Electricity Act 1996), Government of South Australia, http://www.energysafety.sa.gov.au/images/pdfs/ ElecFinalb.pdf, last accessed 17/3/2003.
- Freebairn, J. and Trace, K. (1988), Principles and Practice of Rail Freight Pricing for Bulk Minerals, Centre of Policy Studies, Monash University.
- Freehill, Hollingdale and Page (1999), Ecogen Energy Sale Agreement, The State of Victoria, Generation Victoria, AES Transpower Holding Pty Ltd, and the AES Freehill. Hollingdale (Melbourne), Corporation, and Page http://www.vgpb.vic.gov.au/major/12/Generation Victoria Ecogen Ecogen Energy Sale Agreement.pdf, last accessed 3/8/2002.
- Freeman, D. J., Findlay, D. A., Bamboat, M., Davison, J. and Forbes, I. (2002), Cost and Performance Of CO₂ and Energy Transmission, Paper presented at the 6th International Conference on Greenhouse Gas Control Technologies, 1st to 4th October 2002, http://www.rite.or.jp/GHGT6/pdf/H4-6.pdf, last accessed 19/2/2003.
- FreightCorp (2000), Services-Resources, FreightCorp web site, http://www. freightcorp.com.au/resources.htm.
- Gallagher, D. R. and Pierce, J. E. (1986), The marginal cost of coal-fired thermal electricity generation in New South Wales 1966/67 to 1981/82, Centre for Applied Economic Research, University of New South Wales, Working Paper No. 93. November. Presented at the 15th Conference of Economists, 25th-29th August 1986. Monash University, Melbourne: From University of Melbourne.
- Generation Victoria (1997), Newport Power Station, Environmental Performance And (Australia). http://www.ecogen-Compliance, Generation Victoria energy.com.au/environ.htm, last accessed 14/2/2001.
- Golar LNG Ltd. (2002), Annual Report 2002, Golar LNG Ltd., http://reports.huginonline.com/883491/111017.pdf, last accessed 14/3/2003. GPU GasNet (2002), Tariff Calculation Examples 2002, GPU GasNet (Victoria).

Sustainability of Australia's Electricity Generation **Options** for South accessed 5/4/2002.

- Hazelwood Power.
- http://www.energyfacts.com/LNG.htm, last accessed 5/8/2002.
- issues/comp_reg/transmission_hirst.pdf, last accessed 4/4/2002.
- accessed 5/4/2002.

- Agency (Paris), pg. 47.
- Development, IEA Report IEACR/55, Mar.

References

GPU Powernet (1999), GPU Powernet Submission to NEMMCO on the Interconnection Options Working Group Paper - Technical Issues and Costs of Interconnection GPU -5th Australia, Powernet. May, http://www.nemmco.com.au/future/interconnectors/sani/pl_ir957v001.pdf. last

GTZ (1996), Environmental Manual for Power Development, (computer program), Deutsche Gesellschaft für Techniche Zusammenarbeit, WorldBank.

Hazelwood Power (1999), Annual Environmental Report 1st July 1998 - 30 June 1999,

HEE (2002a), LNG Value Chain Model, Houston Energy Economics,

HEE (2002b), North American LNG Workshop, Houston Energy Group, LLC, Brown, Williams, Moorhead & Quinn, and Global Pipeline Consulting, March 20, http://www.energyfacts.com/LNG.PPT, last accessed 7/8/2002.

Hirst, E. and Kirby, B. (2001), Transmission Planning for a Restructuring U.S. Electricity Industry, Edison Electric Institute, June, http://www.eei.org/

Hydro (Canada) (2001), Safety a priority during construction of phase three of Avalon Peninsula transmission line upgrade, Newfoundland and Labrador Hydro Group (Canada), 16th February, http://www.nlh.nf.ca/2001NewsRel.htm #Safety, last

Hydrocarbon Processing (2001), Hydrocarbon Processing HPI Construction Boxscore, Hydrocarbon Processing, Volume 80, Number 10, October, Section 2, pp.120-156.

Hydrocarbon Processing (1999), Hydrocarbon Processing's HPI Construction Boxscore, Hydrocarbon Processing, Volume 78, Number 10, October, pp. 3-52.

IEA (1999), Energy Policies of IEA Countries 1999 Review, International Energy

IEA (1993), Advanced Power Generation - a comparative study of design options for coal, International Energy Agency, Organisation for Economic Co-operation and

IEA (1992), Greenhouse Gas Emissions from Power Stations, International Energy Agency Greenhouse Gas Research and Development Program, Organisation for Economic Cooperation and Development (OECD), http://www.ieagreen.org.uk/

References

Sustainability of Australia's Electricity Generation

sr1p.htm, last accessed 7/3/2002.

- Institute of Energy Economics (Japan) (2003), Average Import Price of Energy (2003/03/10 Update), Institute of Energy Economics, http://www.ieej.or.jp/ cgibin/trend-list.pl?Lt792400+WtE+key, last accessed 7/3/2003.
- (1999), Intergen 'Millmerran', Asia-Pacific Projects, Intergen, http://www.intergen.com/project.html, last accessed 29/4/2002.
- IOR Energy Pty Ltd (2000), Engineering Conversion Factors: Fuel Energy Density, http://www.ior.com.au/ecflist.html.
- Johnson, J. (1987), The Newport Power Station: A History of Conflict, The Proceedings of the Queen's Birthday Weekend Conference of the H.R Nicholls Society at 6th-8th Mooloolaba June. 1987. http://www.hrnicholls.com.au/ nicholls/nichvol3/vol36the.htm - Vol 3-6: The Newport Power Station: A History of Conflict – Jack Johnson OBE, last accessed 22/4/2002.
- Kinhill Sterns (1987), Assessment of Environmental Impact of Electricity Transmission Line from Mount Piper to Marulan, Electricity Commission of N.S.W., February, pp. 141-57, 168-9, 180-6.
- Klaassen, G., Grübler, A. and Schrattenholzer, L. (1999), Towards New Energy Infrastructures in Eurasia: a background paper, International Institute for Applied Systems Analysis, Interim Report IR-99-17, http://www.iiasa.ac.at/ Publications/Documents/IR-99-017.pdf, last accessed 4/4/2002.
- Lee, H. M. (1986), Future Economics of Coal Transport, Australian Mining, August, pp. 24-5.
- Lema, J. E. (1990), Attributes of Selected Transportation Modes, 1990 International Joint Power Conference, American Society of Mechanical Engineers, Fuel and Combustion Technology (FACT) Division, Volume 8, pp. 65-71.
- Leonard, R., Rogers, L., Vimalchand, P., Liu, G., Smith, P. V. and Longanbach, J. (2001), Development Status of the Transport Gasifier at the PSDF, Paper presented at Gasification Technologies 2001, San Francisco, California, October 7-10, 2001, http://www.gasification.org/98GTC/GTC01057.pdf, last accessed 10/5/2002.
- Loy Yang Power (1999), Environmental Report, Loy Yang Power, http://www.lovvangpower.com.au/envreport1999.htm.

Macarthur Coal (2001), Monto Coal Project, Macarthur Coal, Press Release, 19th July. Macquarie Generation (2000), Annual Report 2000, Macquarie Generation, pg. 12.

- Gas Journal, January 19.
- 18/2/2003.
- 17/3/2003.
- 17/2/2003.
- accessed 11/4/2002.
- 18.
- www.nationalgrid.com/uk/, last accessed 10/2/2002. National Power (2000) Annual Report 2000, National Power, pg. 2.

References

Mallett, M. W. (1987), Largest U.S. gas processing plant begins operations, Oil and

Martin, P. (2002), Compressed Natural Gas: Innovative approaches to transocean gas transportation, Advantica Technologies Ltd., presentation to SONG, 14th May 2002, http://song.icheme.org/meetings/archive/SONG CNG May 2002.pdf, last accessed

Maude, C. (1993), Advanced power generation – a comparative study of design options for coal, IEA Coal Research, International Energy Agency, IEA CR/55, March. MCA (2002), Fatalities in the Australian Minerals Industry, 1990-1991 to 1999-2000: A Retrospective Analysis, Minerals Council of Australia, December, http://www.minerals.org.au/downloads/pdf/fatality retro_dec02.pdf, last_accessed

McDonald, J. K. (2001), Onshore Pipelines for Onshore Pipelines for Timor Sea Gas, SEAAOC Conference - Darwin SEAAOC Conference - Darwin, 19 June 2001, from http://wyw.pipelinetrust.com.au/2/2-1-2.php, last accessed 21/02/2002.

Merritt, R. D. (1986), The Seward Coal Terminal (1985), Alaska Division of Mining and Geological and Geophysical Surveys (U.S.A.), Public-data File 86-95, December, http://dggs.dnr.state.ak.us/scan2/pdf86/text/PDF86-95.PDF, last accessed

Mining Technology (2002), Mount Owen Coal Mine, Australia, Industry Projects, Mining Technology, <u>http://www.mining-technology.com/projects/owen/</u>, last

MMS (1998), Gulf of Mexico OCS Oil and Gas Lease Sales 171, 174, 177, and 180, Western Planning Area, Final Environmental Impact Statement, Gulf of Mexico OCS Region, Minerals Management Service (U.S.), MMS 98-0008, pp. IV-13 to IV-

Mobil Australia (1999), MEPA - Wandoo Oil Field, Mobil Australia, http://www.mobil.com.au/company/teexpl.htm, last accessed 9/9/2001 (link lost).

National Grid (U.K.) (2001), Regulatory Accounts, National Grid (U.K.),

Nautilus Institute (1999), IGCC in China, A Background Paper for the ESENA Workshop: Innovative Financing for Clean Coal in China: A GEF Technology Risk

References

Guarantee?, Berkeley, California, 27-28 February 1999, http://www .nautilus.org/papers/energy/NIIGCCESENAY3.pdf, last accessed 12/03/2002.

- NEA (1998). Projected Costs of Generating Electricity: Update 1998, Nuclear Energy Agency, Organisation of Economic Cooperation and Development.
- NEA (1994), Power Generation Choices: Costs, Risks and Externalities, Proceedings of an International Symposium, Washington, USA, 23-4 September 1993, Nuclear Energy Agency, Organisation of Economic Cooperation and Development.
- NEA (1992), Projected Costs of Generating Electricity: Update 1992, Nuclear Energy Agency, Organisation of Economic Cooperation and Development.
- NEA (1986), Projected Costs of Generating Electricity from nuclear and coal-fired power stations for commissioning in 1995, A report by an Expert-Group, Nuclear Energy Agency, Organisation of Economic Cooperation and Development.
- NEAC (1980), Strategies for Greater Utilisation of Australian Coal, National Energy Advisory Committee, NEAC Report No. 10, May.
- NERC (2000), Electronic GADS Publications for Windows, 1995-1999, Generating Reliability Report, North American Electric Reliability Council (NERC), December, ftp://www.nerc. com/pub/sys/all_updl/gads/gar/gar1999.exe, last accessed 6/12/2001.
- NGGIC (2000), National Greenhouse Gas Inventory 1998 with Methodology Supplements, National Greenhouse Gas Inventory Committee, Australian Greenhouse Office, July.
- NiSource (2001), Income Statement Data, NiSource, http://www.nisource.com/ press/forward-looking-statements.pdf, last accessed 7/8/2002.
- NOHSC (2002), NOHSC Online Statistics Interactive (NOSI), National Occupational Health & Safety Commission (2002), http://nohsc.info.au.com/, last accessed 20/3/2003.
- NPI (2001), National Pollutant Inventory 2000, National Pollutant Inventory (Australia), http://www.npi.ea.gov.au/, last accessed 30/3/2003.
- NPI (1999), Emission Estimation Technique Manual for Fossil Fuel Electric Power Generation, National Pollutant Inventory (Australia), March, pp. 16-39, 68-9.
- NREC (1987), Inquiry into electricity supply and demand beyond the mid-1990's: Information Update No. 2, Natural Resources and Environment Committee, Parliament of Victoria, April.
- NRM (Qld.) (2001), Queensland Coal Facts 2000/2001, Department of Natural

Sustainability of Australia's Electricity Generation

Resources and Mines (Qld.), http://www.nrm.qld.gov.au/mines/gtherm/coal facts.html, last accessed 11/3/2003. NYPA (2001), New York Power Authority - Combined Cycle Project, Article X Application, New York Power Authority, Case 99-F-1627, Chapter 3.0, http://www.nypa.gov/ccf/NYPA%20Article%20X/ ccftoc.htm. last accessed 30/11/2001.

- Co-operation and Development, pp 26-7, 37-8. Atmospheric Programs (USA), March. Engineering, Volume 77, Number 1, pp. 61-5. Energy, http://www.optima.com.au, last accessed 10/5/2000.
- 9/9/2001 (link lost).
- au/enprod/eraring2/index.htm, last accessed 22/11/2000.

References

OECD (1985), Coal, Environmental Issues and Remedies, Organisation for Economic

Office of Atmospheric Programs (EPA) (2001). EGrid2000. Emissions and Generation Resource Integrated Database, Environmental Protection Agency Office of

Olds, F. C. (ed.) (1973), Capital Cost Calculations for Future Power Plants, Power

OMV (2001), Main Report, Patricia Baleen Gas Field Development, Environmental Effects Statement, Public Environmental Report, March, pg. 2-6, Chapter 6.

Optima Energy (2000), Interesting Facts and Figures, General Information, Optima

Origin Energy (2001), Sea Gas Consults On Pipeline Project, South East Australia (SEA) Gas, Origin Energy, Media Release, October 26, 2001, from http://www.originenc-gv.com.au/news/news_detail.php?pageid=82&newsid=137&P HPSESSID=7998ed5a44a2da9c0d607f627e15f265, last accessed 21/02/2002.

OTA (U.S.A.) (1980), Alternative Energy Futures: Part 1-The Future of Liquefied Natural Gas Imports, Office of Technology Assessment, U.S. Department of Energy, March 1980, NTIS order #PB80-173552, http://www.wws.princeton .edu/cgibin/byteserv.prl/~ota/disk3/1980/8002/8002.PDF. last accessed 18/2/2003.

OTD (N.T.) (2001), Laminaria/Corallina, Northern Territory Oil and Gas Fact Sheet, Office of Territory Development (N.T.), Northern Territory Government, http://www.otd.nt.gov.au/dcm/otd/otd/Fact_Sheets/laminaria.html, last_accessed

Pacific Coal (1999), Annual Environmental Report 1999, Pacific Power, http://www.pp.nsw.gov.au/enviro/annual_report.html, last accessed 13/02/2000.

Pacific Power (2000), Eraring Power Station, Pacific Power, http://www.pp.nsw.gov.

- Pavri, R. and Moore, G. D. (2001), Gas Turbine Emissions and Control, GE Energy Services, GE Power Systems, GER-4211.
- Power Technology (2002), Willem-Alexander Igcc Power Plant, Netherlands, Australia, http://www.power-Industry Technology, Projects. Power technology.com/projects/willem/, last accessed 11/4/2002.
- Powerlink (2001), Grid Map, Powerlink Queensland, http://www.powerlink.com.au/ system/indexmap.htm, last accessed 11/9/2001.
- Powerlink (Old.) (2002), Annual Report, Powerlink.
- Productivity Commission (1998), The Australian Black Coal Industry, Inquiry Report, Volume 1, Report No. 1 July 1998.
- PWC (Old.) (1997), The Expansion of the Dalrymple Bay Coal Terminal, Legislative May 1997. No. 36, of Queensland, Report Assembly http://www.parliament.gld.gov.au/comdocs/public works/pwc36.pdf, last accessed 17/2/2003.
- Oueensland Rail (2000), Personal Communication, Siva Gnanananthan, 23/11/2000.
- Richards, A. (1998), 'Connecting Performance and Competitiveness with Finance: A Study of the Chemical Industry', Chapter 14 in Arora, A., Landau, R. and Rosenberg, N., Chemicals and Long-Term Economic Growth, Insights from the Chemical Industry, John Wiley and Sons. (New York, U.S.A.).
- Rigzone (2003). Enel to become BG Group's partner in Brindisi LNG, Rigzone Oil and Gas Industry News, Sunday, February 16, 2003, http://www.rigzone.com/ news/article.asp?a_id=5645, last accessed 17/2/2003.
- Sampson, L. (2000), Personal Communication, Department of Water Resources (South Australia), 8th December.
- Santos (1999a), Ballera South-West Queensland Gas Project, Santos Technical Services Drafting Office.
- Santos (1999b), Annual Report 1999, Santos Limited, www.santos.com.au, last accessed 30/3/2002.
- Sayers, C. and Shields, D. (2001), Electricity Prices and Cost Factors, Productivity Commission Staff Research Paper, AusInfo, Canberra.
- Scott, S.G., Kozak, A. and Sochocki, V. (1997), Petroleum Production and Reserves -Annual Report 1996/97, Queensland Minerals and Energy Review Series, Department of Mines and Energy Queensland.

- 5/8/2002.
- Commission of Victoria, pg. 142. Commission of Victoria, March. 14/1/2002.
- power gas.html, last accessed 11/07/2002. Prepared for Powerlink (Qld.) by PB Associates, April. Report No. 98007 (Internal). Utrecht University, Report No. 94021, November,

330

Sustainability of Australia's Electricity Generation

References

SCPPA (2002), Magnolia Power Project, Southern California Public Power Authority, http://www.scppa.org/ Magnolia/magnoliaprojectoverview.htm, last accessed

SECV (1993), Annual Report year ending 30th J.ne 1992, State Electricity

SECV (1989), Loy Yang Power Project, The June 1989 Progress Report, State Electricity Commission of Victoria, Design and Construction Group, September,

SECV (1987), Electricity Supply and Demand Beyond the Mid 1990s, Part IIa: Supply Options, State Electricity Commission of Victoria submission to the Parliamentary Natural Resources and Environment Committee Inquiry, March.

SECV (1973), Newport Power Station, Environmental Report, State Electricity

Sierra Pacific Resources (2001), Final Technical Report To The Department Of Energy, Pinon Pine IGCC Project, DOE Award Number (DE-FC21-92MC29309), Reporting Period August 1, 1992 To January 1, 2001, January, http://www.lanl .gov/projects/cctc/resources/pdfs/pinon/PinonFinalReport022201.pdf, last accessed

Simes, B. (2000), The Outlook for Coal Supply & Coal Prices, Barlow Jonker Pty Limited, Presented at 5th Annual Queensland Power & Gas Briefing, Brisbane, QLD, 27-28 November, http://www.coalportal.com/general/articles/ bj paper gld

Simpson, B. (2001), Powerlink Operating and Maintenance Expenditure Review,

SKM (1997), Comparative Cost of Power Study APFBC and IGCC, Sinclair Kight Merz, CRC for New Technologies for Power Generation from Low-Rank Coal.

Smit, R. and Nieuwlaar, E. (1994), Life Cvcle Assessment of Integrated Coal Gasification Combined Cycle, Department of Science Technology and Society,

Smith, I. M. and Sloss, L. L. (1992), Methane Emissions from Coal, International Energy Agency (IEA) Coal Research, IEAPER/04, November.

Snam (2001), Panigaglia, a centre of excellence in terms of the environment and safety, Infogas (Snam, an Eni Group company), Volume X, Number 1, January/February.

- Snam (1999), Health, Safety and the Environment-1999 Report, Snam (an Eni Group company), pp. 23.
- Spath, P. L., Mann, M. K. and Kerr, D. R. (1999), Life Cycle Assessment of Coal-fired Power Production, NREL (National Renewable Energy Laboratory), NREL/TP-570-25119, see electronic copy (Coal/General/NREL).
- SPCC (1979), Proposed Bayswater Power Station: Environmental Impact Assessment, the State Pollution Control Commission of New South Wales, Public Release Document, pg. 2.
- Special Correspondent (1977), Bass Strait Gas, PACE: Process and Chemical Engineering, February, 1977, pp. 31-2.
- SPIA (2001), Works Approval Application for SPIA Gas-Fired Peaking Power Station -Laverton North, SPI Auswire Holdings Pty Ltd, 17 October.
- SPIPowernet (2003), About Us, SPIPowernet, http://www.spipowernet.com.au/, last accessed 17/3/2003.

Stanwell (2002), Annual Report 2002, Stanwell Corporation Limited.

Stanwell (2000), Annual Report 2000, Stanwell Corporation Limited.

- Stanwell (1999), Next Generation Power 1998/99 Environment, Health and Safety Performance Report, Stanwell Corporation Limited, pg. 22.
- Systek (2002), GasMod Gas Pipeline Hydraulics Demo Version 4.80.500, Computer Program, Systek, http://www.systek.org, last accessed 9/8/2002.
- Szabo, M. E. (1978), Environmental Assessment of Coal Transport, U.S. Environmental Protection Agency, Industrial Environmental Research Laboratory, EPA-600/7-78-081, May.
- Tamura, I., Tanaka, T., Kagajo, T., Kuwabara, S., Yoshioka, T., Nagata, T., Kurahashi, K. and Ishitani, H. (2001), Life cycle CO2 analysis of LNG and city gas, Applied Energy, Volume 68, Number 3, pp. 301-319.
- Tarong Energy (2002), Annual Report 2002, Tarong Energy, http://www. tarongenergy.com.au/content/annual_report.htm, Last accessed 27/2/2003.
- Tarong Energy (2000a), Annual Report 1999-2000, Tarong Energy, http://www. larongenergy.com.au/content/annual_report.htm, Last accessed 27/2/2001.
- Tarong Energy (2000b), Environmental Performance Review 1999/2000, Tarong Energy, <u>http://www.tarongenergy.com.au/content/environmental_review.htm</u>, Last accessed 27/2/2001.

Sustainability of Australia's Electricity Generation

- 27/2/2001.
- Project, Theodore Coal Pty. Ltd. accessed 20/2/2003.
- 17/2/2003.
- transgrid.com.au/health.htm. Queensland Treasury. budget2000/index.htm, last accessed 5/4/2002. Association index.htm, last accessed 20/9/2002. Association, http://www.ukooa.co.uk/, last accessed 6/11/2001.

References

Tarong Energy (1999), 'Contractor announced for Tarong North Project', News, Tarong Energy, 1/12, http://www.tarongenergy.com.au/, last accessed 27/5/2002.

Tarong Energy (1998), Environmental Performance Review 1998, Tarong Energy, http://www.tarongenergy.com.au/content/environmental_review.htm, last_accessed

Tasman Asia Pacific (1997), The Scope for Productivity Improvements in Australia's Open Cut Black Coal Industry, Tasman Asia Pacific Pty Ltd., October.

Theodore Coal Pty. Ltd. (1981), Summary Technical Report on the Theodore Coal

TotalFinaElf (2000), TOTAL and LNG, TotalFinaElf, http://www.totalfinaelf.com/cgibin/frames.pl?lang=us&grp=de&sgrp=gn&file=index.htm&lev=2, 22/3/2000, last

Tradeport (2000), Deep Water Coal Terminal Proj. Tradeport Trade Directory (U. S. Department of Commerce - National Trade Data Bank, November 3, 2000). http://www.tradeport.org/ts/countries/colombia/mrr/mark0080.html, last accessed

TransGrid (2003), TransGrid Health and Safety Policy, TransGrid, http://www.

TransGrid (2001), About Us, TransGrid, http://www.transgrid.com.au/about_us.htm.

Treasury (Queensland) (2000), State Budget 2000-2001, Chapter 5, Capital Statement, http://www.treasurv.qld.gov.au/budget/

UKOOA (2001), The 2001 Economic Report, United Kingdom Offshore Operators (UKOOA), <u>http://www.oilandgas.org.uk/issues/economic/econ01/</u>

UKOOA (1999), 1999 Environment Report, United Kingdom Offshore Operators

UMS Group (2000), Operating Expenditure Benchmarking Study, UMS Group Australia for the Office of the Regulator-General, Victoria, Final Report, May 24,

VAO (1993), Open Cut Production in the Latrobe Valley, Victorian Auditors Office, http://home.vicnet.net.au/~vicaud1/sr24/ ags2400.htm, last accessed (6/6/2002).

VBCC (1982), Victorian Brown Coal Resource Development Study, prepared by Kinivil Pty Ltd for the Victorian Brown Coal Council, Volume 4 (Coal Winning).

Williams, D. (1998), Greenhouse Gas Emissions from Australian Coal Mining, The Australian Coal Review (CSIRO Energy Technology), October.

Woodside (2000), Annual Report 2000, Woodside.

- Woodside Energy Ltd. (1998b), Lifecycle Analysis of Emissions Associated with Power Generation Using Various Fuel Sources, synopsis of two previous CSIRO reports, July.
- Woodside Offshore Petroleum (1992), Facts About Australia's North West Shelf Gas Project, Published by Woodside Offshore Petroleum on behalf of the Participants in the North West Shelf Gas Project, Publication Number 291, May.

Yallourn Energy (1999), Environmental Report, Yallourn Energy.

YPC (2001), Responses to questions of February 14, 2001 concerning an all-Alaskan LNG gas pipeline route to Valdez, Yukon Pacific Corp., Our Gas, Our Future, Alaska's Gas Watchdog, <u>http://www.alaskagaspipeline.org/documents/</u> <u>Responses%20to%20Representative%20Guess2.doc</u>, last accessed 25/2/2002.



ERRATA and ADDENDUM (CONTINUED)

pg. 64, Equation 3.2.2: 'Exergy Destruction' replaces 'Exergy Depletion' and 'Energy Depletion' replaces 'Energy Content'. Add footnote: 'For more details on the exergy destruction method see Ayres, R. U., Martinás, K. and Ayres, L. W. (1996), Eco-thermodynamics. Exergy and life cycle analysis, Working Paper (96/04/EPS), INSEAD, Fontainbleau, France.

pg. 65, Equation 3.2.3 note: '...system (e.g. kg...' replaces '...system (kg...' for both occasions and '...substance c (e.g. kg...' replaces '...substance c kg...'

pg. 73, para. 2: "...where one stream contributes twice as much as the other to environmental impact." replaces 'and contributions of 33% and 67% of the environmental impact result respectively.'

pg. 83, para. 1: '...have identical revenue.' replaces '...have identical positive cash flow.'

pg. 83, footnote: 'i = real interest rate' replaces 'i = real inflation rate'

pg. 87, para. 1: 'For example, one project (A) ...' replaces 'For example, a one project (A) ...' and '..., such as interest and loan payments.' replaces ' ..., such as interest and load payments.'

pg. 94, para. 1: '... without contributing to the LTI indicator.' replaces '... without contributing the LTI indicator'. The following replaces the paragraph beginning 'Of the indicator methods, ...':

'Of the indicator methods, only the LCA method (Section 3.2.3) was originally developed to have boundaries with a 'life-cycle' perspective. This section examines whether the operations that allow the LCA method to combine data from subsystems of a 'life-cycle' perspective system to obtain and analyse system scores may also apply to the economic and social indicators of sustainability.

pg. 99, para. 2: '... before data collection begins.' replaces -'before data collection.'

pg. 100, para. 1: 'estimate the effect' replaces 'estimate the affect' and para. 3 '... and assisting duplication of the results of the study...' replaces '... and assisting duplication the results of the study...'. Para. 5: The sentences beginning from 'Consequently, the...' are replaced by: 'Consequently, the accuracy of data from Australian electricity generation systems is greater than from foreign and different systems. Data from these later systems is only used when data from Australian electricity generation systems is lacking. Average values are used whenever multiple sources of data are available. ...'

pg. 102, caption: '(dashed boundary = all impacts, dotted boundary = economic impacts only).' replaces '(grey = all impacts, spotted = economic impacts only).'

pg. 105, figure: replacement of figure with Figure B.

pg. 107, para. 1: After '(Data Ref. C1).' is added 'The composition of black coal is given in Figure 4.3.4.', para. 2: after 'during normal operation.' is added 'Often mined-out areas are rehabilitated during the mining of new areas.' and para. 4: after '...not include overburden.' is added 'Overburden is only a solid waste if it accumulates rather than being re-used as fill.'

pg. 108, Table 4.3.3 caption: Added 'The equations refer to a power law equation using the A and ψ parameters (see Equation 3.3.6, Chapter 3).'

pg. 108, Table 4.3.4 caption: Added 'The equations refer to a power law equation using the B and ω parameters (see Equation 3.3.10, Chapter 3).'

pg. 123, figure caption: Added 'Scaled interval is 0 to 60 m.'

pg. 156, para. 3: 'all methods shown in Section 3.2.3.7 (Chapter 3).' replaces 'methods developed for use in LCA, all appearing in Section 3.2.3.7 (Chapter 3).', 'and the climate change, acidification and world resource depletion indicators.' is added to sentence beginning 'Time and resource constraints limit...' and '(Finnveden and Lindfors (1998))' is added after "rules of thumb".

pg. 157, para. 2: '... assumes that the destination is Japan.' replaces '... assumes that the destination in Japan.'

pg. 169, para. 2: '(see Section 4.4.2, Chapter 4)' is added after 'the CC, AD and RD W indicators' pg. 175, para. 1: 'show the major effects' replaces 'show the major affects' pg. 182, para. 2: '10⁻¹⁰-10⁻⁹' and '10⁻¹¹' replace '10¹⁰-10⁹' and '10¹¹' . I

22 (443) p Technology

Coal-ST, post-1970 Gas and Coal co-firing ST

251.12







.÷.,

 Q^{*}

25.